Infrastructure Funding Models Study

Prepared for: Parramatta City Council Date: May 2016 Project No 10379



Infrastructure Funding Models Study

May 2016

Prepared for

Parramatta City Council

By



GLN Planning Pty Ltd ABN 39 585 269 237

Level 10, 66 King Street Sydney NSW 2000 GPO Box 5013 Sydney NSW 2001

P [02] 9249 4100

E info@glnplanning.com.au F [02] 9249 4111 W glnplanning.com.au



Infrastructure Funding Models Study Parramatta City Council

GLN Planning Pty Ltd ABN 39 585 269 237

Level 10, 66 King Street Sydney NSW 2000 GPO Box 5013 Sydney NSW 2001

Е

W

P [02] 9249 4100 F [02] 9249 4111 info@glnplanning.com.au glnplanning.com.au



 Date of Final Issue:
 10 May 2016

 File Path:
 R:\Projects\10379 Parramatta Infrastructure Funding Models Study\Report\10379_Parra

 IFMS_Draft.docx
 IFMS_Draft.docx

 Project Manager:
 Greg New

 Client:
 Parramatta City Council

 Project Number:
 10379

The purpose for which this report may be used and relied upon is limited for that which it was commissioned. Copyright in the whole and every part of this document belongs to GLN Planning and may not be used, sold, transferred, copied or reproduced in whole or in part in any manner or form or in or on any media to any person without the prior written consent of GLN Planning.

Document history and status

Version	Issued To	Qty	Date	Prepared by:	Reviewed by:
Final	D Hewetson	1-e	27/11/15	MC, GN, PL	GN
VS Update Dra	ft D Hewetson	1-e	14/03/16	MC	GN
Final Draft	D Hewetson	1-e	08/04/16	MC	GN
Final	D Hewetson	1-e	10/05/16	MC	GN



Table of Contents

Exe	cutive Summary	iv
1.	Introduction	1
1.1	Background	1
1.2	Project objectives	1
1.3	Structure of report	2
2.	Development profile and infrastructure demands	3
2.1	CBD development framework	3
	2.1.1 Background	3
	2.1.2 Parramatta CBD Strategy	4
2.2	Current and anticipated development	6
2.3	Anticipated infrastructure demands	7
3.	Conventional development contributions mechanisms	9
3.1	Introduction	9
3.2	Comparison of available mechanisms	9
	3.2.1 Consent conditions requiring works	9
	3.2.2 Consent conditions requiring money and / or land	10
	3.2.3 Voluntary planning agreements	13
3.3	Estimated income from current mechanisms	14
3.4	Summary of opportunities and constraints of the conventional contributions mechanisms	16
4.	Conventional contributions development feasibility analysis	17
4.1	Introduction	17
4.2	Land acquisition and viability measures	17
4.3	Impact of an increase in contributions	18
4.4	Development viability modelling	19
4.5	Commercial development	21
4.6	Potential equity issues for contributions on non residential development	22
4.7	Summary	23
5.	Value sharing and development viability impacts	24
5.1	Introduction	24
5.2	What is value sharing?	24
5.3	Types of value sharing schemes	24
	5.3.1 Local examples	25
	5.3.2 Tax increment financing and growth area bonds	26
5.4	Opportunities for State government-led schemes	27
5.5	Why should Council extract some of the value uplift that results from planning decisions?	28
5.6	Review of value sharing schemes	29
	5.6.1 How do the schemes work?	29
5.7	Calculation of value sharing contribution	32



5.8	Potential value sharing schemes	33
5.9	Impact on development feasibility	35
	5.9.1 Assumptions 5.9.2 Results	35 35
5.10	Recommended floor space value and value sharing rates	37
5.11	Implementation issues	38
	5.11.1 Risks of value uplift not materialising 5.11.2 Probity	38 39
Glos	ssary of terms and abbreviations	45

References

Tables

Table 1	Development potential under the CBD Planning Strategy	7
Table 2	Section 94 contributions and section 94A levies comparison	10
Table 3	Summary of estimated income from different combinations of available mechanisms	15
Table 4	Development feasibility benchmarks	17
Table 5	Development viability testing results	20
Table 6	Comparison of key centre contribution rates for commercial development	21
Table 7	Comparison of Centre market rents	22
Table 8	Local value sharing schemes	29
Table 9	Summary of how value is captured in the schemes	30
Table 10	Summary of \$/m ² of GFA rates paid during 2014 and 2015	33
Table 11	Summary of value sharing regimes at \$805/m ² of additional GFA	35

Figures

Parramatta CBD Strategy area	5
Commercial Core zoned area	6
Growth area bonds	27
Draft CBD Planning Proposal typical FSR outcomes	34
	Commercial Core zoned area Growth area bonds



46

Appendices

Appendix A	Estimated income analysis (conventional schemes)
Appendix B	Standard contributions approach development feasibility analysis
Appendix C	Review of value sharing schemes summary
Appendix D	Value sharing development feasibility analysis and assumptions



Executive Summary

The Parramatta CBD is expected to receive a significant amount of development in the future. This will likely include developments accommodating thousands of new residents and workers. These developments will generate increased infrastructure demands.

The costs of the new and enlarged facilities to meet these demands will be met from a variety of funding sources. The contributions developers pay to the local council will be a primary source.

This report examines the opportunities for Parramatta City Council to extract contributions from the future CBD developments. It examines the conventional contributions mechanisms that are available as well as the opportunities for implementing alternative or innovative tools such as value sharing.

A 3% section 94A levy currently applies to development on land throughout the CBD area. The section 94A mechanism has historically been preferred because of its simplicity and low risk of legal challenge.

An analysis of projected income from a mix of conventional section 94 / section 94A approaches showed that:

- Continuation of the current 3% section 94A levy would yield the lowest income of all the approaches (\$165 million)
- Combinations of section 94 contributions and section 94A levies on different developments would yield higher levies (between \$193 million and \$232 million)
- A 4.5% section 94A levy on all new development would yield the highest income of all the options (\$247 million).

The impacts of the various options on the viability of mixed use development were examined. If Council was to implement a section 94A levy of 4.5% or a new section 94 contributions plan, some Parramatta CBD projects would move from being viable to being on the cusp of achieving the necessary development benchmarks to make them worthwhile to proceed.

Council has estimated that the future CBD infrastructure upgrades will cost at least \$835 million.

Application of a section 94A levy to future development, even at the 4.5% level will only yield a fraction of the income needed to meet this cost. As a result, GLN Planning investigated alternative or non-traditional contributions schemes that are operating in other development areas in Sydney. We tested the potential for so-called 'value sharing' schemes to provide additional funds to meet the infrastructure costs, and also tested the impacts of a 'Phase 1 / Phase 2' value sharing scheme on the viability of hypothetical CBD developments.

Our investigation of these alternative schemes concluded that:

- Significant additional development contributions to fund infrastructure could be received by Council through implementing a value sharing scheme.
- Phase 1 value sharing may negatively impact on viability of some projects as it introduces an unanticipated development cost. However whether the impact results in viability benchmarks being unable to be achieved will depend on the level of sharing and the time since the site was purchased. Our research indicates that a 10-20% level of Phase 1 value sharing could be tolerated in the current market.
- The larger developments that is, developments above 12:1 would remain viable with Phase 2 value sharing at any level. It would be unreasonable however to expect developers to tolerate Phase 2 value sharing to exceed 50%.



Optimum strategy using currently available contributions mechanisms

The optimum contributions strategy for the anticipated future development in the Parramatta CBD, using the current tools, is outlined below:

- (a) Continue the current practice of requiring developers to carry out works on and near their development sites where it is reasonable to do so.
- (b) Continue the current practice of consent authorities imposing a section 94A levy on all eligible development, but seek the support of the Minister for Planning to allow the maximum levy to be increased from 3% to 4.5%.
- (c) Continue to negotiate VPAs with developers to extract additional public benefits associated with new developments.

If on the other hand the Council was unsuccessful in obtaining the Minister's support for a maximum 4.5% levy, and Council still wanted to achieve a higher financial return than the current 3% levy, it could pursue a hybrid contributions strategy that allowed section 94 contributions to be imposed on mixed use development, and a section 94A levy on wholly non residential (i.e. commercial and retail) development.

This approach however would be more resource intensive and may down the track potentially expose Council to legal challenge on the reasonableness of its contributions in individual development approvals. These risks however can be managed.

We therefore recommend that Council prepare a draft section 94A contributions plan that authorises a levy of up to 4.5% on future CBD development and seek the Minister for Planning's endorsement to amend the EP&A Regulation to allow the levy.

Council could then choose to impose a lesser levy on straight commercial office development in order to retain the Parramatta's competiveness in the Sydney suburban office market.

In the event that Council is unsuccessful in gaining the Minister's approval for an increased levy, we recommend that Council pursue a mixed or hybrid contribution strategy of:

- (a) Section 94 contributions applying to all residential and mixed use development
- (b) A 3% section 94A levy applying to all development that does not include a residential component.

Strategy for implementing Council's value sharing scheme

Council is considering implementing a 2-phase value sharing scheme on developments in the Parramatta CBD area in order to help fund the significant CBD infrastructure upgrades.

If Council was to proceed with its value sharing scheme, based on our investigations of similar schemes and our assessment of viability impacts, we recommend that Council incorporate the following elements in the scheme:

- (a) Retain the 3% section 94A levy in preference to seeking the Minister's approval for a higher (4.5%) levy.
- (b) Develop, in consultation with the Department of Planning and Environment, a planning scheme that provides a choice for the developer between pursuing an 'as-of-right' development under current general planning controls and infrastructure contributions; or



pursuing a development under different planning controls that also have value sharing contributions attached to them.

- (c) Prepare a comprehensive infrastructure plan containing details of the different facilities and amenities that will be delivered using the proceeds from value sharing.
- (d) Prepare guidelines that contain details of the value sharing scheme and that show how developers can participate in the scheme.
- (e) Adopt a floor space value in the range of \$700 to \$750 per square metre of GFA for the purpose of assessing value to be shared.
- (f) Consider carefully the rate of Phase 1 value sharing that is implemented. Our investigations of hypothetical developments based on recent average sales data show that larger developments could absorb up to a 20% Phase 1 value sharing contribution and probably meet viability benchmarks. Sites recently purchased for smaller developments (ie. development that achieves 6.9:1 FSR including incentives) could potentially absorb a 10% Phase 1 value share. Developments on sites that have been held for several years would likely be able to absorb 20% or more Phase 1 value share regardless of the size of the development.
- (g) The Phase 2 value sharing rate be set at no more than 50%.
- (h) The arrangement for the developer to provide public benefits is achieved and formalised through the negotiation of a VPA between the developer and the Council.
- (i) Value sharing contributions that are paid by developers be held in a dedicated account that has accountability and reporting protocols that at least reflect the accounting requirements for section 94 and section 94A monies.



1. Introduction

1.1 Background

The Parramatta Central Business District (**CBD**) is located centrally in the demographic heart of the Sydney Metropolitan Area and performs a key economic, social and cultural role. The Parramatta CBD is of metropolitan significance as a regional employment centre, and it will continue to increase in importance as Western Sydney's population continues to grow.

A planning strategy adopted in 2015 by the Parramatta City Council (**Council**) sets targets for 27,000 additional jobs and 7,500 additional dwellings in the CBD area by 2036.

Significant growth in jobs and dwellings is expected to place further demands on Council to provide new infrastructure or augment existing infrastructure, such as open space, community facilities, and local road improvements.

Council has engaged GLN Planning to prepare an infrastructure funding model study to determine the most appropriate mechanism to fund and / or deliver new infrastructure to meet the demands of anticipated growth in the Parramatta CBD.

This study:

- Recommends a strategy to fund local infrastructure that utilises the suite of funding mechanisms currently available under existing legislation.
- Explores the opportunities for alternative funding mechanisms which may be appropriate to apply to future development in and around the Parramatta CBD.
- Is to inform the future direction of infrastructure funding for the Parramatta CBD Planning Framework Review.

1.2 Project objectives

The purposes of this project are as follows:

- Review the currently available contributions mechanisms in terms of potential income, impacts on development feasibility, and risks for implementation.
- Recommend a fair, appropriate and workable development contributions system to apply to development in the Parramatta CBD area.
- Investigate alternative and innovative infrastructure funding mechanisms and models, including specifically schemes that capture some of the uplift in value of development sites as a result of the additional floor space rights.
- Identify any legislative or other regulatory changes needed to implement the preferred infrastructure funding strategy.



1.3 Structure of report

This report is arranged as follows:

Part 1 (this part) provides an outline of the objectives and background of this project.

Part 2 discusses the planning and future development of the Parramatta CBD, including the infrastructure anticipated to be required to support the development.

Part 3 discusses the infrastructure contributions and delivery mechanisms currently available under the Environmental Planning and Assessment Act 1979 (**EP&A Act**) that could be applied to the Parramatta CBD developments, and a comparison of the projected income each of these regimes would generate based on the growth targets set for the Parramatta CBD.

Part 4 examines the development viability impacts associated with implementing the different contributions mechanisms currently available under the EP&A Act.

Part 5 discusses a commentary on alternative funding models for the delivery of infrastructure in the Parramatta CBD, specifically 'value sharing' schemes, including the associated advantages and disadvantages of these schemes. This part also tests the viability impacts of a value sharing scheme that has been proposed by Council.

Part 6 provides a conclusion to the study as well as a series of recommendations for Council's consideration to optimise the collection of contributions on development within the CBD whilst maintaining feasible development outcomes.



2. Development profile and infrastructure demands

Substantial new development is expected to occur in the Parramatta CBD in the future.

Planning policies are being prepared to cater for significant population growth requiring some 1.4 million additional square metres of gross floor area to be constructed in the area by 2036. Council's proposed planning controls facilitate additional floor space of over 2 million square metres on the assumption not all sites will turn over for development.

Council is currently investigating the infrastructure that will be need to support this level of growth.

Council's approach is that the future development should meet most or all of the cost of the infrastructure upgrades through development contributions.

The formulation of an effective infrastructure funding and development contributions strategy is inextricably linked to the type, amount, and rate of development expected to occur in an area. This part of the report therefore describes the anticipated development for the Parramatta CBD.

2.1 CBD development framework

2.1.1 Background

In recent years Council has undertaken studies of future development opportunities in the Parramatta CBD.¹

The objective of these studies was to review the current planning framework and identify opportunities, constraints and market conditions impacting on development; and from this develop draft planning controls and recommendations to ensure Parramatta will fulfil its role as Sydney's western CBD.

These studies and previous resolutions of the Council formed the basis of the Parramatta CBD Planning Strategy. This strategy was adopted by Council in April 2015.

Key actions in the strategy will inform a Planning Proposal currently being prepared by Council to amend the planning controls for the CBD include the following:

- 1. Expansion of the CBD boundaries
- 2. Increase in floor space ratios (FSRs) to predominantly 10:1 and 6:1 across the CBD
- 3. Implementation of density bonuses above these FSRs for developments that achieve design excellence and high environmental performance
- 4. Removal of any height controls, except in some key areas
- 5. Investigation of potential sun access controls to key public spaces
- 6. Expansion of the 'Commercial Core' around Parramatta station
- 7. Setting an employment growth target of 27,000 additional jobs and residential growth target of 7,500 additional dwellings by 2036 for the CBD
- 8. Investigation of infrastructure needs, including funding mechanisms
- 9. Promotion of tower slenderness and design excellence.

¹ Draft Parramatta City Centre Planning Framework Study (2014) and the Draft Parramatta Auto Alley Planning Framework Study (2014)



A number of technical studies have been undertaken to give effect to these key actions, including this infrastructure funding models study.

In order to address point 8 above, the economic analysis undertaken by SGS Economics and Planning proposed a 'value capture' mechanism be applied to land or development which benefits from any uplift as a consequence of modified planning controls (e.g. increases in FSR). This study considers the ability to apply such a mechanism under current NSW legislation including relevant NSW examples, and whether legislation changes may be required to enable value sharing to be established as a viable infrastructure funding mechanism.

At the time that it adopted the CBD Strategy, Council decided that it would seek the Minister for Planning's approval to impose a higher section 94A levy of 4.5% on all development throughout the CBD as the primary means of funding CBD infrastructure upgrades. Also at that time Council considered that value sharing could be a funding mechanism that could apply to only those sites where extra development rights were permitted beyond those allowed under the CBD Strategy.

Since that time, information has been prepared showing that the cost of CBD infrastructure upgrades is likely to exceed \$800 million, and that a conventional section 94A levy is unlikely to generate sufficient income to cover that cost. This study therefore examines the potential development viability impacts of value sharing on developments complying with the densities allowed under the CBD Strategy.

2.1.2 Parramatta CBD Strategy

The CBD Strategy applies to the area shown in **Figure 1** over page.

A single Planning Proposal to give effect to the Parramatta CBD Planning Strategy is due to be completed by the end of 2015 and placed on exhibition by mid 2016.

The Planning Proposal will among other things include provisions that increase the maximum FSR and height controls applying to residential development. The proposed FSRs will be generally 10:1 throughout the CBD area, with some exceptions. There is proposed to be no FSR limit on non residential or office development.

The Planning Proposal will continue to prohibit residential development in the B3 Commercial Core.

The location of the B3 Commercial Core zoned land is shown in Figure 2.





Figure 1 Parramatta CBD Strategy area





Source: Draft Parramatta CBD Planning Proposal, 2016

Figure 2 Commercial Core zoned area

2.2 Current and anticipated development

The CBD area currently has the following development:

- Approximately 4,769 residential dwellings
- Approximately 1,188,312 square metres of commercial, retail and other non-residential development.

The CBD Strategy identified the following targets to 2036:

- 27,000 additional jobs
- 7,500 additional dwellings.

The following floor space, in absolute terms, would need to be developed to achieve these targets:



- 648,000 square metres of commercial floor space²
- 750,000 square metres of residential floor space.³

The controls proposed in the Parramatta CBD Strategy provide ample capacity for these targets to be met.

Table 1 shows the development capacity that will be allowed under the draft CBD Planning Proposal.

Table 1 Development potential under the CBD Planning Strategy

	Additional floor space yield (square metres)	Estimated potential additional job / dwelling yield
No residential allowed in Commercial Core		
Commercial	1,753,980	48,721 jobs
Residential	2,996,315	19,976 dwellings
TOTAL	4,750,295	

The CBD Strategy provides the capacity to achieve the planning targets for dwellings and jobs.

Council has also received numerous Planning Proposals that seek floor space potential above that allowed for in the CBD Strategy.

2.3 Anticipated infrastructure demands

Council is investigating the infrastructure needs that will be generated by the development expected to occur under the CBD Planning Strategy. This work is underway at the time of writing this report but initial estimates have indicated the value of infrastructure works to be in the order of \$835 million.

In discussions with Council officers, it is possible that the future infrastructure requirements for the new CBD development may comprise similar projects to those contained in the CBD Civic Improvement Plan that underpins the current Council's section 94A contributions plan. This plan applies to all CBD development with a development cost above \$250,000.

The Civic Improvement Plan includes the current types of projects:

- Aquatic centre and stadium minor works
- Car park refurbishments and lighting
- Cycleways and other cycling facilities
- Heritage Centre refurbishments
- Heritage conservation works
- Library resources, refurbishments and fittings

³ Assuming an average of 100m² per dwelling.



² Assuming an average density of 24m² GFA per worker.

- Local area traffic management schemes
- Parramatta Discovery Centre relocation
- Parramatta River foreshore improvements
- Pedestrian bridges
- Pedestrian safety facilities
- Playgrounds
- Public art
- Riverside Theatres refurbishments
- Roads and traffic upgrades
- Shared zones
- Streetscape works such as street trees, lighting and street furniture.



3. Conventional development contributions mechanisms

3.1 Introduction

Development contributions are requirements imposed on, or arrangements negotiated with, developers of land to provide land, works, and / or money for infrastructure that is generated by or that is related to new development.

Development contributions are:

- required to address the additional demands on infrastructure caused by new development;
- are usually imposed as conditions of development consent to mitigate the impacts of development on public infrastructure; and
- are a tool to address the provision of infrastructure where developers of land will not or cannot provide that infrastructure.

There are several available mechanisms available to councils to require developers to address the impacts on local infrastructure caused by their developments.

The mechanisms include contributions of land or money or works required in conditions of development consent, or in negotiated agreements with developers.

3.2 Comparison of available mechanisms

3.2.1 Consent conditions requiring works

These are conditions may be imposed by a consent authority on any development, and are authorised under section 80A(1)(f) of the EP&A Act.

A consent authority can require a developer to carry out works, either / both within and outside of the development site, relating to any matter referred to in section 79C (1) applicable to the development.

They are conditions for works to be carried out. Conditions requiring contributions of money cannot be authorised by section 80A(1)(f). Only conditions authorised by section 94 or section 94A can require contributions of money by a developer.

Section 80A conditions are not strictly development contributions, but they do work in conjunction with development contributions. For instance, section 80A conditions may include works that are already listed in a contributions plan, but if the works are also listed in a section 94 contribution plan the section 94 contribution for that development must be adjusted to reflect the works that are required in kind by the section 80A condition. No such adjustment is required where the development is the subject of a section 94A levy.

Generally speaking, section 80A conditions can only require works or activities to be undertaken that are *entirely required* by the development to mitigate that development's environmental impacts. It is usually therefore only works that are immediately proximate to the development site that can be the subject of a section 80A condition.

Typical uses of section 80A conditions include requiring developers to carry out the following:

- new and upgraded road access to the development site
- perimeter footpaths fronting the development site



- drainage (inter-allotment, reticulation and trunk types)
- creation of easements on a development site
- traffic management devices to enable access to a development
- provision of support services (for example for seniors housing developments)
- environmental monitoring activities
- making satisfactory arrangements with utility / energy infrastructure providers (including, for example, rights of way).

The works can be of a public nature (i.e. be provided on public land) and could also be transferred to a council or other public authority ownership following completion. They could also be works on private land and which will remain on private land.

Councils should use section 80A conditions for the delivery of development-generated local infrastructure wherever it is reasonable and practical to do so, in preference to delivering that same infrastructure via section 94 or section 94A contributions. This is because it shifts the responsibility of provision (and therefore risk of any cost overruns) entirely from the council to the developer.

3.2.2 Consent conditions requiring money and / or land

While section 80A conditions of consent can require works from a developer, sections 94 and 94A conditions of consent can require the payment of money and / or the dedication of land free of cost for the provision of local public amenities and services, provided these are set out in an operational Contributions Plan. Voluntary planning agreements under section 93F of the EP&A Act can require any combination of works, land and money but these are voluntary arrangements. Planning agreements are discussed in section 3.2.3.

Section 94 contributions can be monetary or land contributions required by councils from developers for the provision of public amenities and services. Section 94A levies are monetary contributions only.

Table 2 below compares attributes of the two alternative types of development contributions that may be imposed as conditions of consent by consent authorities.

Table 2 Section 94 contributions and section 94A levies comparison

Section 94 contributions	Section 94A levies	
Nexus necessary – developers pay their fair share of the cost of public work.	No nexus necessary – developer pays a tax	
A section 94 contributions plan defines the consequences of development in terms of its generation of needs for specific local infrastructure, and then authorises the development to make land or works contributions consistent with the development's share of demand for that infrastructure.	Section 94A levies are simply a flat rate tax on development. Council must spend the tax revenue on the delivery of local infrastructure identified in the contributions plan that authorised the levy.	



Development only pays a reasonable contribution towards the cost of infrastructure in any particular case	Levy does not necessarily have to relate to the specific demands generated by the development
Despite the provisions of a contributions plan, a council must still consider the circumstances of each development to determine whether it is reasonable to require the contribution in that case, and whether the contribution authorised by the contributions plan is reasonable in that case.	A contributions plan authorising a section 94A levy must include a statement about the relationship between expected development and infrastructure demand. But that relationship or connection does not need to be reflected in the plan's works schedule.
However, a Council can only impose a contribution condition in accordance with a Contributions Plan. This is because a condition may be disallowed or amended by the Land and Environment Court on appeal if it is considered to be unreasonable in the particular circumstances of the case (s94B(3)).	Provided a plan, duly prepared in accordance with the EP&A Act and EP&A Regulation, authorises the levying of development then that is all that is required. No assessment is required of whether it is reasonable to require the levy in the particular circumstances of the development.
	The reasonableness of a condition cannot be disallowed or amended by the Land and Environment Court on appeal (s94B(4)).
Works schedule in a contributions plan determined on basis of demand	Any works schedule acceptable
To satisfy the nexus argument, the Section 94 contributions plan works schedule must be linked to expected development. It must therefore be derived from needs analyses and studies that determine the nature and scale of local infrastructure that will be required to meet future development.	Preparation of a section 94A contributions plan works schedule is more straightforward. It may, for example, relate to a council's prevailing capital works priorities. Levies received may be directed toward funding part or all of the cost of individual items in that schedule. There does not need to be a connection between the contribution rate and works items, but facilities still need to be provided in a 'reasonable time'.
More complex calculation	Simpler calculation
To fulfil the nexus requirement the calculation of a section 94 contribution can be complex. As indicated above, the appropriateness and reasonableness of the contribution must be considered. This involves, in the case of each application, for example, determination of any credits for previous demand and the use of indexed rates.	Calculation is more straightforward. The levy is calculated as a percentage of the cost of the development, and this relies on an accurate development cost report. Clause 25J of the EP&A Regulation provides particulars on the matters that must (and those that must not) be included in the calculation of the cost of development for the purposes of calculating the levy.
Determination of existing demand is necessary	No determination of existing demand
S94 contributions must be reasonable and relate only to the net increase in demand for infrastructure occasioned by the particular development. To establish the net increase in demand, the consent authority must account for any existing demand that exists on the development site.	As there is no need to establish a nexus between the development and the infrastructure the subject of the levy, there is no need for an assessment of existing demand. The flat rate tax levy, if authorised by the contributions plan, can apply each time the same parcel of land is developed.



Intensification of use relevant	Intensification of use relevant
As the nexus must be established in order to calculate the net increase in demand for infrastructure, the level of intensification of a use is a necessary component of this assessment.	Despite the fact that nexus need not be considered, for applications relating to refitting or refurbishing premises, Council must establish whether or not there is proposed to be an enlargement, expansion or intensification of the use. Clause 25J of the EP&A Regulation provides that 'where the development involves an enlargement, expansion or intensification of a current use of land', the costs of fittings and furnishings should be included in the cost of work.
Settlement of contributions liability can be made through money, land or works	Settlement of contribution through money only
The contributions that are imposed are payment of money, or the dedication of land, or both. The liability may be settled through these methods, or it can be settled by the developer providing works-in-kind or material public benefits.	A s94A condition can only be for a monetary contribution and can only be settled by the payment of that contribution, unless a Planning Agreement is entered into.
Residential development contributions are capped	Levy rates must not exceed prescribed maximums
In the case of land in the City of Parramatta LGA, residential section 94 contributions are capped at \$20,000 per dwelling, in accordance with the Direction made by the Minister for Planning in August 2012. No cap applies to non-residential development.	Under the EP&A Regulation, levies for all types of development in the Parramatta CBD are capped at 3 percent of the cost of development. The maximum levy that can be imposed on development elsewhere in the Parramatta LGA outside of the CBD area is 1 percent.
Contributions may only be applied to 'essential works' in certain circumstances	No 'essential works' limits on types of works that may be the subject of levy funding
 Where a council has prepared a contributions plan that authorises a section 94 contribution above the cap: the contributions plan will need to be reviewed by IPART; and the contributions that are collected can only be directed towards items on the 'essential works list'. 	The 'essential works' limitation does not apply. In accordance with s94A(3) money that is collected must "be applied toward the provision, extension or augmentation of public amenities or public services (or towards the recouping the cost of their provision, extension or augmentation)."

Councils can apply a 'mix and match' approach in applying the conventional contributions types. They can adopt contributions schemes that authorise section 94 contributions to apply certain types or areas of development and section 94A levies to apply to other types or areas of development.

However, either a section 94 contribution or a section 94A levy may apply to a single development, but not both. Potential issues may arise where a mixed residential and non-residential development is proposed, and the council had adopted a contributions plan that authorised section 94 and section 94A contributions on different developments. In these cases,



the contributions plan would need to be clear about which of the contribution types applied to the development.

This is particularly relevant for the Parramatta CBD area where much of the future development is likely to have a mixed-use multi-storey character containing both residential and non-residential components. For example, if the development needed to address some of its car parking demand by making a cash contribution, in lieu of providing physical spaces, then the full costs of this are most often met by the consent authority imposing a section 94 contribution not a section 94A levy. If Council however preferred a section 94A levy regime for the CBD that reflects the current practice, then individual development parking shortfalls can be managed through other means (for example through a voluntary planning agreement).

3.2.3 Voluntary planning agreements

The preceding discussion relates to powers of consent authorities in requiring developers to do certain things at the point of development consent. Section 94 contributions and 94A levies comprise the mandatory contributions system.

Voluntary planning agreements (**VPA**s) are the discretionary contributions system in NSW. These are agreements that can be negotiated between developers and planning authorities. They cannot be imposed on a developer.

A VPA is a voluntary arrangement between a developer of land and one or more councils and / or other planning authorities whereby the developer is required to:

- dedicate land free of cost, or
- pay a monetary contribution, or
- provide any other material public benefit, or
- provide any combination of the above,

to be used for or applied towards the provision of a public purpose.

The VPA can be offered by a person lodging a DA or a section 96 modification to a DA, or by a person lodging a Planning Proposal.

Contributions can include dedication of land, payment of money, or the carrying out of works or any combination of these. The VPA between a council and a developer can, if the parties agree, authorise that section 94 contributions or section 94A levies not apply to development on the land covered by the agreement.

Other key features of VPAs include:

- Development consent can't be refused on the grounds that a developer refuses to enter into a VPA.
- May be registered on the title to land if the parties and all persons who have an estate or interest in the land agree to its registration.
- It is not necessary to establish a nexus between the development and the public purposes included in a VPA.

One of the main advantages of planning agreements is that they can enable flexible arrangements for the provision of local infrastructure in connection with new developments. With flexibility comes a responsibility to act fairly and consistently. Councils have a duty to act with probity and transparency in planning agreement negotiations and in the interests of the wider community.



Important principles underlying a council's use of VPAs are as follows:

- Planning decisions should not be bought or sold through planning agreements. The development should be acceptable on its planning merits and not rely on the developer's offer of public purposes as compensation for inferior urban outcomes.
- Planning agreements must be voluntarily entered into. The council's officers or representatives should not create an impression that a favourable planning or development decision is tied to successful negotiation of a planning agreement.
- The council should not allow the interests of individuals or interest groups to outweigh the wider public interest when considering a proposed planning agreement.
- The council should not seek benefits under planning agreements that are unrelated to particular development, nor should the council give undue weight to a proposed planning agreement when considering a Planning Proposal or a DA.
- The council should not improperly rely on its statutory position in order to extract unreasonable public benefits from developers.

Tensions regularly arise between Council's role as a planning / consent authority and as a beneficiary of infrastructure provided by developers. There needs to be discipline in managing these conflicts. Parramatta City Council manages negotiation of VPAs through its 2008 Planning Agreements Policy.

VPAs are now widely used in the NSW planning system to achieve many different public purposes associated with new development. They are most useful in the development process where both the planning authority and the developer see value in entering into an agreement that clarifies and improves on mandatory contributions requirements.

VPAs are an underpinning feature of value sharing schemes that are currently operating in certain development areas in Sydney (refer to discussion in section 5.6). If Council was to embark on its own value sharing scheme then it would likely also need to rely on VPAs to achieve the scheme objectives.

3.3 Estimated income from current mechanisms

We tested the potential income that could be generated by applying the different mandatory contributions mechanisms to the anticipated future Parramatta CBD developments, i.e. section 94 contributions and section 94A levies.

The following contributions strategies were tested:

- 1. Impose section 94A levies on development at the current rate of 3% of cost of development (this is the base case)
- 2. Impose section 94A levies on development at the higher rate of 4.5% of cost of development
- 3. Impose section 94 contributions on development
- 4. Impose section 94 contributions on all residential and the non-residential components of mixed use development, and impose section 94A levies on development comprising of only non-residential development at the current rate of 3% of cost of development
- 5. Impose section 94 contributions on all residential development and the non-residential component of mixed use development, and impose section 94A levies on straight non-residential development at the higher rate of 4.5% of cost of development



Options 1, 3 and 4 can be implemented by Council without Ministerial approval. Options 2 and 5 require the Minister's approval for a higher section 94A levy.

The income analysis was based on extracting contributions from the anticipated floor space that is required to meet the CBD Strategy housing targets. Section 94 contributions would be based on the net increase in floor space attributable to those targets; while section 94A levies would be based on both replacement and additional floor space.

All section 94 contributions attributable to residential development in the analysis are assumed to be \$20,000 per dwelling which is consistent with current State government policy. In the event that the section 94 cap was increased or removed entirely, the income from a section 94 approach would be greater.

The detailed results of this analysis and the assumptions underpinning the projected income are contained in **Appendix A**.

The summary results are shown in **Table 3**. The income results are in 2015 dollars and do not account for escalation of development costs.

Table 3 Summary of estimated income from different combinations of available mechanisms

Opt	tion	Projected income (\$million)	Approval of Minister required?
1	3% section 94A levy (base case)	\$165.2	No
2	4.5% section 94A levy	\$247.7	Yes
3	Section 94 contribution	\$193.2	No
4	 Section 94 contribution on all residential Section 94 contribution on non residential component of mixed use development 3% section 94A levy on straight non residential development 	\$206.6	No
5	 Section 94 contribution on all residential Section 94 contribution on non residential component of mixed use development 4.5% section 94A levy on straight non residential development 	\$232.4	Yes

Continuation of the current section 94A levy scheme yields the lowest income. However, if the Council was permitted to impose a higher 4.5% levy, this would yield the highest income.

A section 94 scheme, whether operating alone or in conjunction with section 94A levies (options 3, 4 and 5), would yield a higher income than the current section 94A scheme. In the case of option 4, this is projected to yield some \$41 million more contributions than the base case. However, the preparation and administration of the section 94 contributions plan would require more Council resources than a flat rate levy. There is also the possibility of unreasonableness appeals being brought against Council for DAs approved under any section 94 scheme.



Option 5 could be rejected on the grounds that it requires the Minister's approval and a greater income would likely be received from just imposing a universal 4.5% section 94A scheme (option 2).

A value sharing scheme based on the increased land and development values as a result of the change in controls proposed in the draft CBD Planning Proposal (discussed in Part 5) presents Council with an opportunity to raise significant *additional* revenue from development in the CBD area.

3.4 Summary of opportunities and constraints of the conventional contributions mechanisms

A 3% section 94A levy currently applies to development on land throughout the CBD area.

In order to meet the extra infrastructure demands that will be generated by the development to be allowed under the CBD Strategy, Council has resolved to investigate increasing the section 94A levy to 4.5%.

Councils and other consent authorities can mandatorily require either section 94 contributions or section 94A levies from development to meet the cost of new infrastructure that is required for new development.

The section 94A mechanism is preferred because of its simplicity and low risk of challenge.

An analysis of projected income from a mix of section 94 / section 94A schemes showed that:

- The current 3% section 94A levy would yield the lowest income
- Combinations of section 94 contributions and section 94A levies on different developments would yield higher levies
- A 4.5% section 94A levy would yield the highest income of all the options.

VPAs are a discretionary mechanism for councils and developers to negotiate the provision of contributions and can be used on an opportunity basis to achieve better and / or more flexible contributions outcomes.

Council currently pursues an 'open for business' philosophy, and does not wish to impose a development contributions regime that discouraged investors in major development projects in the CBD. The impacts of pursuing different types and mixes of contributions mechanisms on the feasibility of development is examined in the following two parts of this report.



4. Conventional contributions development feasibility analysis

4.1 Introduction

At present, Council applies a section 94A levy of 3% on development costs to all development in the Parramatta CBD exceeding \$250,000.

This part of the report examines a range of alternatives to this business-as-usual approach in terms of the impacts on development viability.

This part discusses the following:

- key viability benchmarks used by the development industry
- approach taken to model viability of development
- impacts that various contribution frameworks would have on development within the CBD.

4.2 Land acquisition and viability measures

The cost to purchase a development site is already reasonably established in most areas based on subtracting from the net sale price of the apartments in a building, the known costs of development (including land, construction and development contributions, marketing, statutory fees and professional fees/costs) based on the existing planning controls.

The development feasibility will cashflow these costs and revenues with the aim of achieving a target Internal Rate of Return (**IRR**) of 20% and Profit Margin of around 20%.

IRR and Profit Margin are key benchmarks used by developers and their financiers to determine the viability of a development project (refer **Table 4**). Both IRR and Profit Margin are used to determine the amount that can be paid for the land to meet the benchmark returns for the development to proceed.

A development feasibility analysis will cashflow these costs and revenues with the aim of achieving a target IRR of at least 20% and Profit Margin of around 20%⁴.

Internal Rate of Return	IRR is effectively the interest rate at which the net present value of costs (negative cash flows) of the investment equals the net present value of the benefits (positive cash flows) of the investment. IRR is calculated before interest (i.e. interest is typically not included in the calculation) and accordingly the longer the period for completion of a project before revenue offset costs, the more the IRR will be eroded.
Profit Margin	Profit Margin is calculated by dividing profit (i.e. forecast net revenue minus the total forecast development costs) by total development costs. The return on costs is typically expressed as an amount and a percentage and would be used by all developers and financiers to measure the investment and return of a particular project.

Table 4 Development feasibility benchmarks

⁴ These targets are often referred to as a hurdle rate. These targets may be higher depending on various factors such as experience of developer, market conditions, risk of project etc. and are typically viewed by financial institutions when considering to finance proposed developments.



Both IRR and Profit Margin are benchmarks used to determine the amount that can be paid for the land to meet the benchmark returns for the development to proceed. Within any area developers will also measure the comparative value of an investment by calculating the 'site value' and comparing that to recent acquisitions. Site value is used as a ready reckoner and simply divides the number of dwellings achievable on the site by the proposed purchase price of the land.

The site value is relevant where the proposal incorporates residential dwellings. The planning controls in the Parramatta CBD require the inclusion of retail / commercial floor space in any mixed-use development. However, this retail / commercial floor space is often provided to the minimum required by the controls⁵ in order to maximize the floor space for residential apartments. A developer may simply 'net off' the projected revenue against the costs for construction of this floor space and car parking (i.e. costs will equal revenue), particularly if there is no precommitment by an anchor tenant or other user group to occupy constructed tenancies. This means that financial return of a project is almost exclusively determined based on residential yield.

4.3 Impact of an increase in contributions

As previously mentioned, a developer determines the price they can afford to pay to acquire a site based on the net revenue generated from development of the site less development and statutory costs and development margin (that represents the risk and reward of doing the development). The price a developer offers has to be enough to entice a land owner to sell.

Development contributions are typically treated by developers as a statutory cost that is factored into the development feasibility when considering the acquisition of land. They are one component that informs the value of land when acquiring a site.

Where land is acquired based on the existing controls, any changes to costs, including increased contribution costs, will either need to be offset against cost savings in the development (i.e. reduce the cost of construction by reducing the specification of works) or directly reduce the profits anticipated by the developer potentially making the project unviable.

When development costs change (such as an increase in material costs or a new development contributions plan being put in place) this will influence the acquisition cost a developer can offer a landowner. Where these cost changes are known to a developer prior to a development site being purchased, this will impact on the purchase price a developer is willing to offer to acquire the land. However, owners may well not want to sell their land if they think it is undervalued based on recent or previous similar sales. Similarly, significant increases in contribution costs for developing in a particular area can result in investment dollars and development activity moving to other areas with a lower cost base until these are exhausted or the market improves in the original area to enable higher offers.

Alternatively, where development costs change after the site has been acquired but before the development has been approved, these can have significant impacts on the ongoing viability of the project. Either the developer has to find a way to offset the increase in costs either by an increase in revenue (i.e. an increase sales prices or increasing yield) or a reduction in costs (i.e. specifying a lower standard of materials for construction).

Conversely, any Planning Proposal which significantly increases the development yield on a site will result in a windfall gain to the developer as the project will achieve more dwellings at no additional acquisition costs, unless the developer has negotiated through an agreement a higher

⁵ This is based on the permissible uses within the B3 Commercial Core and B4 Mixed Use zones of Parramatta City Centre Local Environmental Plan 2011 and Section 4.3.3.2 of Parramatta Development Control Plan 2011.



purchase price if the Planning Proposal is successful. These Planning Proposals are the catalyst that typically requires re-evaluation and redirection of expenditure of the capital works programme to provide for an unanticipated new population, and on occasion the outcomes of broader planning studies, particularly traffic.

As the acquisition of the land is typically the first step in any project, any subsequent increases in costs including changes to the contributions need to be communicated early to existing and potential developers.

4.4 Development viability modelling

In order to understand the impacts of implementing a new contributions framework within the Parramatta CBD, a suite of models have been developed based on a hypothetical development of a site within the CBD. These models have been used to demonstrate the impact the various contribution regimes will have on the viability of the development (i.e. project IRR and Profit Margin) as well as the contributions Council could expect to collect from this development. In all cases, the modelling has assumed the same acquisition price for the site based on the existing planning controls.

The models provide for a base case development outcome (this is assumed to be 10:1 FSR is applied under the CBD Strategy) and Planning Proposal uplift outcome (assumed 15:1 FSR applied) for each of the different contribution frameworks that could be applied.

The different contribution frameworks modelled include:

- 3% section 94A applied to development permissible under the CBD Strategy
- 3% section 94A applied to a Planning Proposal seeking an additional 5:1 FSR on top of the CBD Strategy
- 4.5% section 94A applied to development permissible under the CBD Strategy
- 4.5% section 94A applied to a Planning Proposal seeking an additional 5:1 FSR on top of the CBD Strategy
- Section 94 contribution based on \$20,000 per unit applied to development permissible under the CBD Strategy
- Section 94 contribution based on \$20,000 per unit applied to Planning Proposal seeking an additional 5:1 FSR on top of the CBD Strategy

As the basis for applying the base case and Planning Proposal uplift scenarios, a site of 3,000m² was adopted and included the provision of 5,000m² of mixed use floor (non residential) space with the remainder of GFA being developed as a mix of apartments including 1, 2 and 3 bedroom units (that resulted in an overall average apartment size of 100m²).

A full list of the other guiding assumptions and development yields informing the viability assessments is provided at **Appendix B**.

We reiterate that in this hypothetical development, the inherent value that contributes to the profit generated for the project is the residential component. Given the current buoyancy of the market for mixed use development in Parramatta, the cost of delivering mixed use component (non residential) has been netted out to match the expected revenue that could be generated from the sale of this space (as discussed earlier).

The contributions able to be collected and impacts on the viability of the hypothetical project are provided in **Table 5**.



Scenario No.	Contributions option	Contribution generated	Developers profit	IRR	Margin
1	Section 94A @ 3% - Building controls as per CBD Strategy (base case)	\$3.17m	\$28.74m	20.0%	18.1%
2	Section 94A @ 3% - Planning Proposal for additional 5:1 FSR	\$4.69m	\$67.02m	30.1%	30.7%
3	Section 94A @ 4.5% - Building controls as per CBD Strategy	\$4.75m	\$27.13m	19.3%	17.0%
4	Section 94A @ 4.5% - Planning Proposal for additional 5:1 FSR	\$7.04m	\$64.63m	29.3%	29.3%
5	Section 94 (\$20,000 / unit cap) – Building controls as per CBD Strategy	\$5.46m	\$26.41m	18.9%	16.4%
6	Section 94 (\$20,000 / unit cap) – Planning Proposal for additional 5:1 FSR	\$8.66m	\$62.99m	28.7%	28.3%

Table 5 Development viability testing results

Key observations on the viability of the tested hypothetical project against the various contributions options include the following:

- If the section 94A levy increased to 4.5% or if the Council implemented a \$20,000 per dwelling section 94 contribution, developments that are within the floor space limits of the CBD Strategy would remain on the fringe of the viability benchmarks. In other words, some developments would have marginal viability. This would occur until there was a market adjustment to the underlying land values of development sites in the CBD due to the increased contributions payable.
- In the short term, this additional contribution cost would either translate to either lower offers being put to landowners to acquire sites (which may stagnate the market as motivation to sell will be reduced) or, alternatively, increase the need for developers to pursue a Planning Proposal to improve yields to offset the proportion contribution cost applied to the development that is permissible under the CBD Strategy to remain competitive in acquiring sites.
- The above observations are relevant to developers who purchase land today. In the event that developers have 'land-banked' for a period of time (assume at least a year reliance on commercial rents to offset holding costs), development proposals in accordance with the CBD Strategy are likely to achieve well above the viability benchmarks, provided they include a residential component. This is due to the significant growth in residential values that has occurred over the last few years.
- In the case of the particular hypothetical development that was tested, a section 94 contribution would yield a higher income than a higher section 94A levy. However, the overall analysis of the CBD in section 3 suggests that total section 94A levies from all CBD development would be higher than section 94, if the levy was increased to 4.5%.



4.5 Commercial development

The above development feasibility analysis focussed on the impact of new contribution regimes on mixed use development that includes dwellings. This focus was applied given the significant development interest and preference that has been shown within the Parramatta CBD to deliver a higher amount of residential and mixed use development compared to straight commercial development.

Given Parramatta's role as the second CBD of Sydney, the impact of increased contributions on straight commercial development still needs to be considered.

The Parramatta City Centre Plan is a section 94A contributions plan that authorises a contribution of 3% of the cost of construction to fund the delivery of a works program in the order \$211 million. As shown below, this rate is typically the minimum standard for commercial office development in Sydney suburban centres.

With the exception of Burwood Town Centre, the contributions for commercial space in Parramatta CBD are generally consistent with other centres whether levied under a section 94 or section 94A plan, as shown in **Table 6**.

Major Centre	Contribution Regime	Contribution/ 100m ² commercial GFA	
North Sydney	S94 levied per worker	\$13,164 ⁶	
St Leonards	S94 levied per m ² of new GFA	\$14,200 ⁷	
Chatswood Town Centre	S94A @ 3% cost of construction	\$13,335 ⁶	
Macquarie Park	S94 levied per m ² of new GFA	\$13,002 ⁷	
Burwood	S94A @ 4% cost of construction \$17,780 ⁶		
Parramatta	S94A @ 3% cost of construction	\$13,335 ⁸	

Table 6 Comparison of key centre contribution rates for commercial development

The feasibility of commercial development in different centres is influenced by a number of factors including:

- Business operational benefits Does the centre have any strategic advantages and service catchments, is it a speciality centre that supports the colocation of services
- Location attributes Does the location provide a high level of amenity, access to services, transport, distance from the CBD, is it appealing/accessible for staff to access?.
- Quality and size of space Is there variety in floorplates, what standard of finishes, how modern is the building?

⁸ Contribution value has been based on an assumption of cost for construction of new commercial GFA of \$3,300 per m² and cost of construction of new commercial office fitout of \$1,145/m² based on costs from *Australian Construction Handbook*, Rawlinsons 2015.



 ⁶ Based on contribution per 100m² identified in the 2015/16 Fees and Charges update for S94 Costs, North Sydney 2015.
 ⁷ Based on contribution per 100m² identified in the 2015 update of the *City of Ryde Section 94 Development Contribution Plan 2007*, City of Ryde 2015.

• Building costs and rental costs.

The Parramatta CBD has a number of strategic locational advantages and offers a range of commercial space types that makes it attractive in its own right compared to other centres.

The recent economic review prepared by Urbis⁹ illustrates that the Parramatta already provides a competitive rental offering compared to other centres as shown in **Table 7** below.

Table 7 Comparison of Centre market rents

Centre	A Grade Office Space Gross Face Rent (\$/m²)		
Sydney CBD	\$730 - \$1,020		
North Sydney	\$670 - \$840		
Macquarie Park	\$390 - \$445		
Chatswood	\$520 - \$610		
Parramatta CBD	A Grade \$450 - \$555 B Grade \$365 - \$460		

Source: Achieving A-Grade Office Space in the Parramatta CBD – Economic Review, Urbis, August 2015 as amended by GLN

The report also notes that compared to other centres such as Macquarie Park, commercial development in Parramatta attracts 20 to 30% higher rents. However it incurs 40% higher development costs likely due to land values and the additional cost of a 'traditional' office tower construction compared to low-rise campus-style development prevalent in business parks such as Macquarie Park. This may have been an issue in attracting significant commercial development to the Parramatta CBD. As such, the imposition of a greater contributions on commercial development would increase the overall building costs. From a development feasibility perspective, this increased cost would have to be offset by either an increase in rents or reduction in overall building costs. Both of these situations would lead to a reduction in the competitiveness of the Parramatta CBD in being able to attract more commercial development compared to other centres.

4.6 Potential equity issues for contributions on non residential development

Section 4.5 highlighted the need for caution in implementing a higher section 94A levy on commercial development. If the Council decides to pursue a 'hybrid' contributions approach for non residential development similar to options 4 or 5 discussed in section 3.3, then it should also consider the equity impacts of this approach.

The hybrid approach is a section 94 contribution applying to employment floor space in a mixed use employment and residential development, and a section 94A levy applying to employment floor space where no residential use is also in the development. With the hybrid approach there is a concern that a developer of a mixed use development might bring an unreasonableness appeal against the Council by showing that the per worker section 94 contribution rate is



⁹ Achieving A-Grade Office Space in the Parramatta CBD, Urbis August 2015.

significantly more than the rate applying to a straight non residential development under a section 94A scheme.

To reduce this risk Council should, as part of formulating any hybrid contributions strategy, calculate the contribution amounts that would apply to mixed use and straight non residential developments, and adjust the contribution rates so that similar amounts of non residential floor space are subject to similar contributions regardless of whether the requirements is by way of a section 94 contribution or a section 94A levy.

4.7 Summary

As with the imposition of a new cost or significant change in market conditions, the application of a new contributions framework with higher contributions will erode the viability of projects.

In the instance of Parramatta CBD, after the increase of the section 94A levy to 4.5% or a new section 94 contributions plan, projects would move from being viable to being on the cusp of achieving the necessary development benchmarks to make it worthwhile to proceed or alternatively force developers to accept a lower margin for the same risk profile.

This would occur until the underlying land market adjusts (through ongoing land exchanges) to take the new contribution regime into account. In the meantime, if market conditions remain buoyant (particularly the residential market), it is likely that Council would continue to receive Planning Proposals to extend development rights above those proposed in the CBD Strategy to offset the contribution costs and allow developers to remain competitive in acquiring sites from land owners.

However, the imposition of an increase in contributions on solely commercial development would impact on development feasibility of those projects and could reduce the competitive advantages of the Parramatta CBD in attracting new commercial development compared to other established centres.



5. Value sharing and development viability impacts

5.1 Introduction

This part investigates alternative funding models for funding the infrastructure required to meet the demands of future Parramatta CBD development.

The discussion addresses 'value sharing' as the primary alternative funding mechanism. Value sharing relies on increases to value of properties brought about by planning or infrastructure decisions. Value sharing is therefore tied to property ownership and / or development.

There are of course other ways that governments can fund infrastructure. These relate to general fees, charges and taxation policy of governments and are beyond the scope of this study.

5.2 What is value sharing?

A value sharing levy or betterment levy scheme may be described as a tax imposed on a land owner on the unearned increment in land price growth that accrues due to a planning or infrastructure decision.

A value sharing scheme would obtain a portion of the value released through new zoning and other public improvements so the communities that create this value share in the wealth it generates.¹⁰

5.3 Types of value sharing schemes

Demonstrating nexus, or the relationship between a development and the imposition of any contributions regime, is an important pre-condition for the value sharing schemes that have been implemented in jurisdictions locally and across the world.

Value sharing schemes are typically tied to a recent or proposed transport infrastructure project. Such infrastructure generates value uplift. The extent of uplift will vary depending on many factors including the level of improved accessibility provided to the land by the transport project and the distance of the land from the infrastructure. Value sharing programs linked to transport projects commonly hypothecate a portion of the land value increase to help pay for the infrastructure.¹¹

Australian examples of transport-linked value sharing schemes include the Gold Coast Light Rail and Melbourne CBD Underground Loop projects.

Value sharing schemes that are not linked to a transport infrastructure project can be linked to a planning decision. For instance where the planning authority makes a decision that provides development rights to land owners that generate an increase in the value of land. This could be rezoning of land that enables a more intense use of the land (for example, rezoning from rural to urban use), or that enables a more intense development of the land (such as an increase in the maximum height allowance). Bonus floor space schemes also fall into this category.

Local examples of value sharing schemes not explicitly linked to transport projects are discussed in the next section.

¹⁰ Langley J (2015), Value Capture Roadmap, p5 ¹¹ ibid., p7



5.3.1 Local examples

Value sharing as an infrastructure funding tool is not new, but examples of its successful implementation in Australia are few. There has not been much political appetite over the years for these types of schemes, with governments preferring instead to continue to rely on entrenched indirect taxation such as land tax and stamp duties on land transfers to fund capital works. Property and development groups typically resist further imposts on the development of land without some reform of the indirect taxes such as stamp duty.

There are no broad-based value sharing schemes operating in NSW that traverse local government boundaries, although a possible exception is the Special Infrastructure Contribution (or **SIC**). The SIC can be interpreted as a type of value sharing levy, as it is a tax flatly applied to residential or industrial zoned land at the time the land is first developed for those urban purposes.

The SIC currently applies to most greenfield development areas on Sydney's fringe. It does not apply to land that is used in accordance with its pre-urban zoned status (for example, for rural purposes), and only applies when development consent is granted for urban purposes. Different rates apply depending on whether the development is residential or industrial. The dollar rate is the same regardless of the underlying land value of the property – underlying land values between the South West and North West Growth Centres vary significantly. The value sharing aspect of the SIC is that some of the increase in value associated with the up-zoned land is captured through the developer paying a standard levy to the State Government, which is then applied to providing new road infrastructure and other purposes that are the responsibility of the State Government.¹²

There are some examples of value sharing schemes in NSW that have been connected with bonus floor space schemes in higher density housing and employment nodes in Sydney including Green Square (in City of Sydney LGA), Sydney Olympic Park (in Auburn LGA), Macquarie Park (in Ryde LGA), Waverley LGA, Randwick LGA and Rhodes (in Canada Bay LGA). These are locally based schemes in most cases implemented by the local council. The common elements of these schemes are that developers are can secure extra development rights in exchange for part of the uplift in site value being shared by the local community via land, works or cash contributions for a range of public purposes.

The public purposes envisaged in these schemes include roads, stormwater drainage, public domain and open space, community and cultural facilities, and affordable housing/ affordable housing is a particular focus of the schemes operating in the City of Sydney and Waverley LGAs.

These contributions are usually additional to the usual development contributions that are compulsorily required as part of the development consent (i.e. section 94 or 94A contributions).

Some other councils have tried to pursue similar types of schemes over the years but have met resistance by the Department of Planning and Environment when trying to effect these schemes through inserting provisions in LEPs.

More detail on the local value sharing schemes currently operating in Sydney development areas, and their potential transferability to development sites in the Parramatta CBD, is provided in section 5.6 of this report.

¹² Current State Government policy is that SIC funds are applied to roads infrastructure, land for emergency and justice purposes and regional open space



5.3.2 Tax increment financing and growth area bonds

Tax increment financing (**TIF**) and growth area bonds (**GAB**s) are types of value sharing schemes that have been promoted by developer and infrastructure lobby groups as a means of funding infrastructure for greenfield and renewal areas.

Both involve quarantining the revenue received from increased property taxes generated by rezoning and development in an area and applying the funds to service and pay out loans taken out to fund infrastructure provided to serve that development. This essentially involves determining a baseline land value for a precinct to determine the baseline revenue that would be generated by rates in the instance there were no changes to planning controls.

The increase in land value, and subsequent increase in the revenue stream from rates, is then determined for a specified period of time (say the anticipated length of development). Bonds, to raise money to forward fund the necessary infrastructure to support development generated by the change in planning controls, are then sold on the basis of guaranteeing a return at the maturity of the bonds, from the increase in rate / tax revenue as a result of the increase in land value. At the end of the bond period, the increased rate / tax revenue stream reverts back to rating / taxing authority.

TIF schemes have been successfully used in the United States where they are usually just one part of the funding mix.

A typical scheme includes the following elements:

- Identify a suitable growth area bond precinct and establish a growth area bond authority
- Prepare a plan for the area that describes its infrastructure needs, and the costs of the infrastructure
- Calculate the property tax revenues currently derived from the area
- Issue bonds to fund the infrastructure works for the area
- Repay the bonds from the incremental increase in property taxes (above the revenue previously collected) generated by the new infrastructure and development in the area
- Once the bonds are repaid, all property tax revenue for the area returns to the Government.

Figure 3 over page illustrates the concept.

The main advantage of this type of scheme is that it ties the increased Government revenue attributable to development to paying back loans for infrastructure provided in the early stages of the development process. Governments fund infrastructure with a bond, then repay it with the growth in property taxes generated by that investment. The forward-funding of infrastructure creates a positive feedback loop where development happens sooner because of the early infrastructure provision, thus generating a greater rate of increases in property taxes. In this scenario, property taxes can be viewed as 'working harder'.

At a local council level, a TIF-like scheme could be implemented based on a change in planning controls assuming an increase in land values. Alternatively, it could be supported with the support of a special rate variation. However, such a scheme would require leadership by State Government and partnership with local councils.





Note: GAB = Growth Area Bond

Sources:

Growth Area Bonds: New Solutions for Funding Infrastructure, Property Council of Australia (NSW), 2010 Realising New Funding for Infrastructure and Urban Renewal: Capturing Value, Consult Australia and SKM (2013)

Figure 3 Growth area bonds

5.4 **Opportunities for State government-led schemes**

These schemes require State government leadership and implementation.

Value sharing is recommended by Infrastructure NSW (**INSW**) as one of six strategies to help deliver infrastructure in the State Infrastructure Strategy.

INSW sees value sharing as a vehicle for taxpayers to recoup some of the value brought about by Government investment in major infrastructure. Some infrastructure projects, especially in transport, can increase the value of nearby landholdings and other assets over time. Where the taxpayer has made a financial contribution, it is desirable that a share of this value should be recovered by Government. INSW sees the challenges for value capture mechanisms as including identifying the beneficiaries, quantifying the gains and crystallising cashflows to Government. INSW however does not see value capture as a short term substitute for more conventional funding strategies.

The State Government has historically been hesitant in pursuing value sharing schemes in conjunction with major infrastructure projects. Some of the reasons for this are:

- Significant local land owner and developer resistance to what is seen as an additional tax on development and land transfer activity.
- Stamp duty already applies to the changing value of land on transfer. Increasing land values over time mean additional consolidated revenue for the State Government. State treasuries traditionally prefer to see these receipts accumulate in the pool of consolidated revenue rather than in a tied area-specific fund that would accompany a tax increment financing / growth area bond scheme for a particular area.

Because of this hesitancy there have been missed opportunities for implementation of State-led value sharing schemes. Value sharing is predicated on value uplift. Land value uplift generally commences when a firm decision is made to proceed with a transport infrastructure project. The formulation and implementation of a value capture scheme needs to occur well before the new infrastructure is operational, so as to not miss out on the value uplift. Some of the opportunity for value sharing income associated with the new rail links in north west and south west Sydney will


have already been lost because land sales along the routes have built in the value that will be derived from the yet-to-be-completed infrastructure.

There have however been some promising moves against this trend recently. There have been media reports that Commonwealth and State Governments are actively considering different potential value sharing mechanisms linked to major transport investment.

Speaking at a funding announcement on the Gold Coast light rail project the Prime Minister said in October 2015 that 'we have to look creatively at how we sharing the value that arises from the increase in property values and the improvement in the utility of adjacent land from the building of infrastructure like this'.¹³

The State Government is examining value capture levies as a key funding source for new light rail links connecting to the Parramatta CBD, particularly the route from Westmead to Sydney Olympic Park via Camellia. The mechanisms being considered include betterment levies, special infrastructure contributions and parking space levies.¹⁴

The Parramatta light rail project offers some promise of government action on a broad-based value capture scheme. Council should monitor these developments and, if a government-led scheme does materialise, the recommendations for a local scheme outlined in this report may need to be reviewed.

5.5 Why should Council extract some of the value uplift that results from planning decisions?

The value of land is directly related to the uses and densities permitted on that land.

Given the much higher value of residential floor space in highly accessible locations like Parramatta CBD, relative to industrial or business zoned land, rezoning sites to facilitate residential uses would result in a significant increase in the land value.

Similarly, decisions by planning authorities that enable development on land in high-demand locations to be carried out at a greater density can often also lead to a significant increase in land value.

There are philosophical public policy reasons why the landowner / developer should not receive all of this un-earned financial benefit - or 'windfall' gain – that has resulted from a government decision:

- These planning decisions will lead to developments that generate infrastructure impacts. The extra infrastructure required by development might not be able to be provided using conventional development contributions mechanisms. Sharing the value uplift from planning decisions between the developer and the community provides a valid funding source for the infrastructure needed by the extra development.
- Sharing value uplift can be justified on economic development grounds. The value uplift
 would not have occurred without the planning decision. It is reasonable for the prime
 beneficiary of this uplift (the land owner) to return some of the value to the community so that
 it can be reinvested in infrastructure that in turn adds value to other land in the area and
 allows more economic development of land and healthier local economy.

¹⁴ ibid, 21 October 2015



¹³ Australian Financial Review, 11 October 2015

Despite the State Government's previous reluctance to investigate or implement comprehensive value sharing schemes, several Sydney councils have acted to implement local schemes. These schemes are described in the next section of this report.

5.6 Review of value sharing schemes

GLN Planning investigated local value sharing schemes operating, or in the implementation stage, in the Sydney metropolitan area.

This section summarises the results of that review. The full analysis is shown in Appendix C.

The schemes that were reviewed are listed in Table 8 below:

Table 8Local value sharing schemes

Scheme	Applies to
Southern Employment Lands Affordable Housing	Certain land in Alexandria and Rosebery that is zoned B7 under Sydney Local Environmental Plan 2012
Green Square Urban Renewal Area Community Infrastructure Scheme	Certain land in Alexandria, Zetland and Beaconsfield zoned R1, B2, B4, B6 and B7 under Sydney Local Environmental Plan 2012
Green Square Town Centre Infrastructure Strategy	Land surrounding Green Square railway station that is zoned 4a General Industrial under the City of Sydney Planning Scheme Ordinance 1971
Macquarie Park Corridor Access and Open Space Infrastructure	Land in Macquarie Park that is currently zoned B3 and B4 under Ryde Local Environmental Plan 2014
Waverley Variation Floor Space Infrastructure Scheme	 Certain land zoned B3, B4 and R4 in Bondi Junction under Waverley Local Environmental Plan 2012 Certain land zoned B4, B1 and R3 in Bondi Beach under Waverley Local Environmental Plan 2012

5.6.1 How do the schemes work?

The schemes provide the opportunity for developers of land to voluntarily participate in an alternative land use control framework.

A person may seek approval for development under the zones, FSRs and height controls that currently apply to the land, or they may seek approval for a different development under an alternative set of controls. The presumption in all of the schemes is that the alternative set of controls allow different types or higher density of development that increase the value of the land.

If the developer wishes to utilise the alternative set of controls then that developer can do so if an offer is made to the council to make contributions of land, works and / or cash for provision of local infrastructure.



Local infrastructure includes new and widened roads, pedestrian and streetscape facilities, new parks, community facilities, stormwater management facilities. In some cases, the local infrastructure includes affordable housing managed by a community housing provider.

The offer of local infrastructure is secured through a VPA between the developer / land owner and the Council.

In some of the schemes the value uplift is triggered when the alternative planning controls are activated or 'switched on' after the statutory planning instrument (i.e. the LEP) is made. In other schemes (Green Square Urban Renewal Area and Waverley), the activation of value uplift is more aligned to the approval of the DA.

Table 9 summarises how value is captured in each of the schemes.

 Table 9
 Summary of how value is captured in the schemes

Scheme	How value is captured
Southern Employment Lands Affordable Housing	 A developer can have a Planning Proposal approved that allows: residential development where it is not currently permitted, and / or permits a greater intensity of development through increased floor space or height controls. The Planning Proposal proceeds only if the developer offers to share the 'planning gain' that accrues from the decision. The planning gain is calculated using a series of standard rates contained in a guidelines document. The rates, derived from an analysis by a land valuer, are a proxy for calculating the difference between value of the land before rezoning or before changes to height and FSR controls, and the value of the land following rezoning. The council requires 50% of the planning gain to be shared with the community by the developer providing works, land, or money up to that amount. The VPA that secures these arrangements is tied to the Planning Proposal. The amending LEP is not made until the VPA is entered into and is registered on the land title.
Green Square Urban Renewal Area Community Infrastructure Scheme	The LEP has a base and maximum FSR applying to all land in the renewal area. The maximum floor space can only be achieved in designated areas shown on the FSR maps and where the developer provides community infrastructure through direct provision and cash contributions. Additional floor space allowed is prescribed in clauses in the LEP. The value of the contribution is calculated based on standard rates published in a guidelines document and the amount of additional floor space above the base allowed in the LEP. The contribution may be provided in-kind or in cash. Regardless, the first \$100 per square metre of additional floor space must be made as a cash contribution towards Green Square Town Centre infrastructure. The VPA that secures these arrangements is tied to the DA.
Green Square Town Centre Infrastructure Strategy	The area is currently zoned Industrial under a 1971 planning ordinance. Two LEPs have been made that zones land for residential, retail, commercial and other non-industrial uses. The new B4 Mixed Use zone is 'deferred' and remains zoned Industrial until the developer offers to



Scheme	How value is captured
	enter into a VPA with the Council to deliver, or make an appropriate contribution toward, town centre infrastructure.
	The appropriate monetary contribution has been calculated for each site in the planning area; and has been based on the expected shortfall in funding available for the necessary upgrades. This amount is then broken up based on each site area's allowance of residential, retail and other non residential floor space, and standardised contribution rates that are included in the town centre infrastructure strategy. The VPA that secures these arrangements is tied to the Planning
	Proposal. The amending LEP is not made until the VPA is entered into and is registered on the land title.
Macquarie Park Corridor Access and Open Space Infrastructure	This scheme is predicated on the need to provide more open space and a more permeable and extensive local street network in the Macquarie Park employment area.
	The scheme operates in a similar way to the Green Square Town Centre model. Additional floor space allowed under the LEP is achieved through the consent authority being satisfied that there will be adequate provision for recreation areas and an access network.
	Adequate provision means the development making an equitable contribution (in cash, works or land) toward the provision of the recreation and access network described in the DCP.
	The types and amount of contribution would be confirmed in a VPA accompanying the DA.
	Monetary Contribution = total additional gross floor area x contribution rate. The contribution rate is \$250 per square metre of floor area, which is a proxy meant to reflect 50% of the increase in value assumed to result from the rezoning.
	The scheme also sets out the value of different works in kind and land dedications that may be offered by the applicant and offset against the total cash contribution.
Waverley Variation Floor Space Infrastructure Scheme	DAs in certain parts of Bondi Junction and Bondi Beach that propose up to 15% extra floor space above that permitted under the FSR controls in the LEP may be approved if:
	The development is acceptable on planning grounds, and
	• The developer enters into a VPA with the Council to provide public benefits (cash, works or land) in the surrounding area.
	On this basis, the planning controls are varied in the instance that the requirements of Clause 4.6 are satisfied. The contribution to be negotiated in the VPA will be 50% of the increase in net value to the development arising from the increase in FSR beyond that allowed under the LEP.
	The formula for calculating the value uplift from the bonus floor space is:
	Marginal net sale proceeds
	less
	Marginal cost to construct
	Council's VPA policy includes a step-by-step methodology that shows how the value uplift is calculated.



5.7 Calculation of value sharing contribution

As shown in **Table 9**, there are various ways a value sharing contribution can be based upon and calculated. On balance, schemes like the *Green Square Urban Renewal Area Community Infrastructure Scheme* and the *Macquarie Park Corridor Access and Open Space Infrastructure Scheme* that utilise a standard \$/m² rate for additional GFA provides the simplest and cost effective methodology to implement. This is primarily due to there not being a need to individually calculate the value uplift as a result of individual planning decisions.

It is also considered that setting a consistent and standard rate, rather than a site-by-site value analysis, would establish a clear signal for the market and developers to respond to. This would allow for a level playing field that the same contribution amount should be factored into future feasibility investigations for sites in the Parramatta CBD.

From a Council perspective, a standard rate also improves transparency in the levying of a value sharing contribution rather than by site by site negotiation. It would also allow Council to better project potential funding that could be obtained from value sharing contributions in the Parramatta CBD.

On this basis, GLN Planning was requested to recommend a reasonable m^2 of gross floor area (**GFA**) rate that could be used for a value sharing scheme in the Parramatta CBD.

The analysis initially drew on a sample of 10 sites zoned B4 Mixed Use with recently completed developments. Site areas ranged from 500m² to 6,000m² with apartment yields ranging from 19 dwellings to 591 dwellings. Land transaction date, including date of sale and sale amount was also provided. Land transactions had occurred for these sites between 2002 and 2015, some with development consent, some before development consent.

A \$/m² rate of approved GFA was determined for each site by dividing the land sales price by the approved GFA¹⁵. Based on this sample of sites and transaction data, which corresponded to a period of significant downturn during the GFC and a period of significant upswing around 2013/14, developers paid on average \$655/m² of GFA.

At the direction of the Council's Infrastructure Funding Review Committee, GLN Planning was asked to review more recent sales transactions to determine if there was any difference in the prices recently being paid for GFA entitlements identified in the CBD Strategy. On this basis, Council was able to provide sales transaction and development approval data for land identified as B4 Mixed Use in the CBD Strategy area for the 2 year period 2014 to 2015. This time period was chosen because it broadly corresponded to the time when information about the increased FSR limits in the CBD Strategy was publicly available.

The data was segmented into sites with the two main FSR ceilings permitted under the CBD Strategy (i.e. 6:1 and 10:1), as well as whether the sites had or did not have DA approvals for mixed use developments. Very small sites were excluded – a minimum lot size of 600m² was assumed to be representative of interest in consolidating sites to allow a reasonable size of development.

A total of 39 transactions comprised this data set.

The \$/m² rate paid for GFA under the existing controls or DA approvals in place was first reviewed. From this analysis, which showed values as high as \$2,850/m² of GFA, it was apparent that

¹⁵ Note the majority of these mixed use development only included token retail or commercial components to satisfy the requirement for 'mixed use development'. It is also assumed that the market for smaller retail and commercial space was not in high demand and as such, this GFA netted itself out in terms of cost to construct and revenue from the sale of this space.



development would be significantly constrained or unfeasible if only the GFA entitlement under the existing planning controls (or DAs approved under these controls) was constructed. As such, the \$/m² rate was adjusted to reflect the GFA entitlements that were identified under the CBD Strategy. The average, adjusted \$/m² rates are shown in **Table 10** below.

Site Circumstance	Average (\$/m² GFA)	Highest (\$/m² GFA)	Lowest (\$/m² GFA)
Sites with DA approval	\$823	\$1,167	\$532
Sites with no DA	\$805	\$2,132	\$104

Table 10Summary of \$/m² of GFA rates paid during 2014 and 2015

The data shows that average, adjusted site values reflective of recent market conditions and expectations were \$823/m² for DA approved sites and \$805/m² for sites with no DA.

It is considered that the 'no-DA' rate of \$805/m² of GFA rate would be the more appropriate rate to use for the purposes of modelling the impacts of various value sharing regimes on development viability.

5.8 **Potential value sharing schemes**

The draft CBD Planning Proposal being prepared by Council will significantly change the existing planning controls, including FSR. For instance, the FSR of certain sites will increase from 6:1 to 10:1 in the CBD core and from 2:1 to 6:1 on the periphery of the CBD area.

In addition to this increase in development right, Council's draft CBD Planning Proposal will also include 'incentive clause' provisions which will allow:

- (a) an additional 15% increase in FSR for 'Design Excellence'; and
- (b) an additional 0.5:1 FSR for 'High Performing Buildings' applicable to mixed use development on sites with a 10:1 FSR.

On top of this, the draft CBD Planning Proposal will also nominate a number of 'opportunity sites' that can benefit from an additional 3:1 FSR increase.

These changes will generate the potential for significant uplift in land values as a result of the increase in permitted development rights. In order to raise part of the funding required to deliver the Council's significant infrastructure program required to support development in the CBD, Council has identified a 2 phase approach for value sharing. In principle this approach is as follows:

- **Phase 1** value sharing is where the developer and the community share in the value uplift created by the extra floor space entitlements that will be permitted when the draft CBD Planning Proposal comes into effect;
- **Phase 2** value sharing is where the developer and the community share in the value uplift created by the extra floor space entitlements that will be permitted by the 'opportunity sites' provisions when the draft CBD Planning Proposal comes into effect.

In effect, the draft CBD Planning Proposal will usher in three maximum FSR outcomes:

6.9:1 FSR sites – Where the existing FSR control is increased to 6:1 plus an additional 0.9:1 (ie. 15%) from the 'design excellence' incentive. This outcome would only involve Phase 1



value sharing on the increase of GFA from the existing FSR to 6:1 FSR. Incentive FSR for 'High Performing Buildings' do not apply in this case.

- 12:1 FSR sites Where the existing FSR control is increased to 10:1 plus an additional 1.5:1 from the 'design excellence' incentive and an additional 0.5:1 from the 'High Performing Buildings' incentives. This outcome would only involve Phase 1 value sharing on the increase of GFA from the existing FSR to 10:1 FSR.
- **15:1 FSR sites** Where the existing FSR control is increased to 10:1 plus an additional 1.5:1 from the 'Design Excellence' incentive, an additional 0.5:1 from the 'High Performing Buildings' incentive and an additional 3:1 from the 'Opportunity Sites' provisions.

These outcomes are diagrammatically shown below.



DE = Design Excellence Incentive FSR HBP = High Performing Buildings Incentive FSR

Figure 4 Draft CBD Planning Proposal typical FSR outcomes

Based on these three typical FSR outcomes, following consideration of the contributions able to be generated by conventional means, Council identified three possible value sharing regimes mixed with a section 94A approach, as follows:

- Application of section 94A at 4.5% of the cost of construction over the full development delivered plus Phase 2 value sharing at 50% of the value on floor space delivered in excess of 12:1 FSR GFA.
- Application of section 94A at 3% of the cost of construction over the full development delivered plus Phase 1 value sharing at 10% of the value on floor space delivered between existing controls and the new controls (excluding incentive FSR) and Phase 2 value sharing at 50% of the value on floor space delivered in excess of 12:1 FSR GFA.
- Application of section 94A at 3% of the cost of construction over the full development delivered plus Phase 1 value sharing at 50% of the value on floor space delivered between existing controls and the new controls (excluding incentive FSR) and Phase 2 value sharing at 50% of the value on floor space delivered in excess of 12:1 FSR GFA.



5.9 Impact on development feasibility

5.9.1 Assumptions

The impact on development viability of the three different value sharing regimes was modelled for each of the typical maximum FSR GFA outcomes shown in Figure 4.

The potential revenue that would be generated under each regime was also calculated. Again, as per the previous modelling carried out for the conventional contribution approaches, a hypothetical development of a 3,000m² site was assumed for each FSR scenario.

This time, a standard value rate of \$805/m² of GFA was applied to the additional GFA above the existing controls as per the Phase 1 and Phase 2 approaches described earlier.

Additional allowances were made for the increase in the cost of construction that would likely result from the design excellence and building performance obligations. A full list of assumptions that underpin the hypothetical development model are included in **Appendix D**.

5.9.2 Results

A summary of the contributions and viability impacts is shown in Table 11.

Table 11	Summary of value sharing regimes at \$805/m ² of additional GFA
----------	--

	Scenario	Contribution Generated	Developer Profit	IRR	Margin
Base	e Case – 6.9:1 FSR with S94A @ 3% and no VS	\$2.21M	\$18.20M	20.0%	17.3%
Α	6.9:1 FSR with S94A @ 4.5% and no VS	\$3.31M	\$17.08M	19.2%	16.1%
в	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$3.17M	\$17.21M	19.3%	16.4%
С	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$4.14M	\$16.23M	18.6%	15.4%
D	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$7.04M	\$13.29M	16.4%	12.6%
Base	e Case – 12:1 FSR with S94A @ 3% and no VS	\$3.84M	\$35.22M	20.0%	18.3%
Е	12:1 FSR with S94A @ 4.5% and no VS	\$5.77M	\$33.34M	19.4%	17.1%
F	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$4.81M	\$34.31M	19.7%	17.8%
G	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$5.78M	\$33.33M	19.4%	17.3%
н	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$8.62M	\$30.44M	18.3%	15.8%



	Scenario	Contribution Generated	Developer Profit	IRR	Margin
I	15:1 FSR with S94A @ 4.5% and Phase 2 VS on 3:1 FSR @ 50%	\$10.76M	\$53.93M	25.9%	23.6%
J	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10% and Phase 2 VS on 3:1 FSR @ 50%	\$9.34M	\$55.34M	26.3%	24.5%
К	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50% and Phase 2 VS on 3:1 FSR @ 50%	\$13.21M	\$51.48M	25.0%	22.8%

The 2014-15 land transaction data demonstrated that the potential FSRs contained in the Architectus study (exhibited in late 2014) and adopted in the CBD Strategy have generally been factored into the price paid for land by developers and speculators, rather than the FSRs currently allowed under the existing LEP. As a result, any attempt by Council to extract a Phase 1 value sharing contribution on a development that did not exceed CBD Strategy FSR allowances will adversely affect viability.

This is shown in our base case where developments approved under the CBD Strategy FSR and paying only a 3% section 94A levy would achieve an IRR of 20%. This is a general minimum benchmark for lending authorities and represents the risk to reward appropriate for the development. If Council applied an additional 10% value sharing to Phase 1 floor space the IRR would reduce to 19.3% for the 6.9:1 FSR development scenario and to 19.8% for the 12:1 FSR development scenario. At this level, these developments would be considered to be on the cusp of being viable¹⁶.

Impacts on development viability with increased Phase 1 value sharing rates worsen to the point of being unviable as the development has to offset a greater contribution with the same revenue. These impacts will be most felt in the instance the residential property market flattens (i.e. static price growth) or declines for those that may have purchased sites.

In this instance, if the Council was to implement Phase 1 value sharing on developments that only achieve the CBD Strategy FSR, this:

- May render projects unviable and stall development unless a DA is already in place.
- Encourage developers to lodge Planning Proposals to exceed the CBD Strategy FSRs to offset the cost of the Phase 1 value sharing.
- Could stall future land transactions as developers would no longer be able to pay the current market rate and no longer provide the same incentives to landowners (unless they were under duress to sell). Although this would likely correct itself over time.

On the other hand, if the development site had been acquired and held for a lengthier period of time, say before the CBD Strategy was public knowledge, it is likely that the site had been acquired for less and as such, may be able to offset the imposition of a Phase 1 value sharing contribution.

¹⁶ This is ultimately dependent on the price a developer paid for land. If they had paid over the general average of \$805/m2 of GFA, the impacts would be more severe. Conversely, if they paid under that rate, the impacts would be reduced and development more viable.



In the instance where the Parramatta residential market continues to experience growth, it is likely that the viability of all projects will improve and offset the imposition of a Phase 1 value sharing contribution.

Developments that achieve FSR outcomes above the CBD Strategy allowances – i.e. Phase 2 value sharing - are not nearly as sensitive to value sharing imposts. This is on the basis a developer acquired a site based on the 12:1 FSR outcome intimated in the CBD Strategy. These results suggest greater room for the application of Phase 2 value sharing but it is also reasonable to assume that there would be higher viability benchmarks for a larger, more risky development that would have greater exposure to changes in market conditions. We suggest adopting a 25% IRR and Profit Margin for projects exceeding the CBD Strategy FSR would be reasonable.

The results for developments exceeding CBD Strategy FSR mask a reality that <u>any</u> Phase 1 value sharing on such development reduces IRR, but any Phase 2 value sharing still allows the development to meet or beat viability benchmarks. Again, this will depend on the price developers have paid for land.

From the above it is evident that Council should use caution in designing its value sharing scheme.

Projects that seek to utilise GFA up to but not exceeding CBD Strategy entitlements will have their viability adversely affected, although value sharing at the 10-20% level could probably be tolerated in the current buoyant economic environment.

It must be stressed however that developments of sites bought in recent times, at or above average rates, that are only able to achieve an FSR of 6.9:1 (including incentives) are far more sensitive to the impacts of Phase 1 value sharing and as such, these projects could likely only tolerate sharing at the 10% level in the current market.

Conversely, a development exceeding CBD Strategy GFA allowances could tolerate Phase 2 value sharing at or above 50%. Although Phase 2 value sharing on a development at or above the 50% level is technically feasible, there is no precedent for value sharing at levels above 50% and would bring into question the notion of equity in the risk and reward taken by developers.

5.10 Recommended floor space value and value sharing rates

If the Council proceeds to implement its value sharing scheme then we recommend that it comprise the following features:

• Adopt a floor space value in the range of \$700 to \$750 per square metre of GFA for the purpose of assessing value sharing

Whilst analysis of land transfers since 2014 identified a higher average \$/m² GFA rate has been paid, it is considered that the recommended rate is reflective of the average rate paid through the full property cycle and is based on a significant sample size. The period since 2014 reflects a time of significant growth in apartment prices in the property cycle. It also represents a time where allowances under the CBD Strategy were public knowledge and this could have helped drive speculation in the market.

The previous analysis of 10 key sites identified by Council reflected a sample of land transactions between 2002 and 2014 and provided an average rate of $655/m^2$ of GFA. This rate is considered to be more reflective of full property cycle movements however is limited in sample size. A more recent, larger sample revealed an average rate of $805/m^2$ of GFA As such, a mid point in the order of $700 - 750 / m^2$ should be more a more reasonable 'through the cycle' rate for the purposes of a value sharing scheme.



 Given the land transactions within the CBD over the last 2 years, a cautious approach should be taken to any Phase 1 value sharing rate – we recommend that the rate should be no more than 20% on larger projects, and 10% on smaller projects. Developments on sites that have been held for several years would more readily absorb a 20% or more Phase 1 value sharing rate, regardless of FSR.

As demonstrated in the modelling results, any application of Phase 1 value sharing will have an impact on development viability for sites purchased in the last two years. If applied at a lower rate (say 10%) for recently purchased sites, it is evident that whilst there will be impacts on development viability, the resultant IRR would still likely be considered to be feasible. This would be dependent upon the price that was ultimately paid for the land and if any opportunities to reduce construction costs. As such, it is considered that caution should be applied when considering value sharing to Phase 1 for sites purchased in the last two years given that value sharing at 10% delivers borderline results (19.8%) against the benchmarks used by developers and financiers). Developments on sites that have been held for several years would more readily absorb 20% or more Phase 1 value sharing, regardless of FSR.

A maximum 50% Phase 2 value sharing rate

From the modelling of developments that achieve an FSR of 15:1 or greater, even after adopting a higher benchmark for development viability (i.e. IRR of 25%), development would still remain viable if a value sharing of 50% was adopted. It is possible there could be a greater tolerance however a value sharing greater than 50% would be hard to politically justify and may disincentivise developers.

5.11 Implementation issues

5.11.1 Risks of value uplift not materialising

The potential for value sharing as an alternative or additional funding source needs to be tempered with an understanding of the current and future economic circumstances.

A value sharing scheme is predicated on there being positive movement in the value of land in a planning area. Rapid price escalation and land speculation have been features of the Sydney residential land market in recent times. This scenario is likely to be a factor that has driven the lodgement of recent Planning Proposals seeking substantially higher floor space development rights on numerous sites in the CBD.

The application of extra development rights to land does not automatically lead to higher land values. Wider economic circumstances and cycles (e.g. interest rates, taxation policy, international money markets) and complex forces of supply and demand in metropolitan development sub-markets all impact both positively and negatively on land values over time.

It is possible that by the time Council was to commence its value sharing scheme that the project pipeline for major developments in the Parramatta CBD for the next 10-15 years will have been established. As the value sharing scheme proposed by Council would only apply to sites that will be developed further down the track, it could be many years before significant infrastructure dividends from the scheme is realised. If land values stagnate or decline for a period rather than increase like they have done recently then this timeframe would extend further.

In other words, if the recent price escalation and developer activity does not continue, this would be a risk to the success of the value sharing scheme. There is a risk that a scheme introduced at an inopportune time in the property cycle may fail.



5.11.2 Probity

Another important issue is the need for any value sharing scheme to be transparent, equitable and applied consistently.

Developers and the community should be made familiar with the purposes and operation of the scheme in order for Council to avoid perceptions of 'cash for floor space' deals. This perception will be difficult to quell anyway, so the responsibility is with Council to work with stakeholders to establish the need for the scheme and the benefits to the wider community that will be achieved by the scheme.

The principles that underpin the draft Macquarie Park scheme are a good example for the Council to follow, that is:

- Nexus: That some of the benefit afforded to sites is captured by the community to provide essential infrastructure required as a result of increased densities in the area.
- Transparency: There is a clear understanding of what infrastructure is to be funded and how contribution rates and community benefit are calculated and applied to individual sites.
- Equity: Both infrastructure and incentives for development are based on equity and fairness.
- Practical: The implementation of the mechanism must be practical and occur in a timely fashion to avoid delays and provide certainty for commercial dealings.
- Feasibility: The contributions must not create development opportunities which are not economically viable.



6. Conclusion and recommendations

6.1 Recommended scheme using conventional mechanisms

There are various contributions tools that are available under the current NSW planning system to facilitate the delivery of development-generated infrastructure by developers.

The optimum contributions strategy for the anticipated future development in the Parramatta CBD, using only the current conventional development contributions mechanisms, is outlined below:

- Continue the current practice of requiring developers to carry out works on and near their development sites where it is reasonable to do so.
- Continue the current practice of consent authorities imposing a section 94A levy on all eligible development, but seek the support of the Minister for Planning to allow consent authorities to impose a 4.5% levy instead of the current 3%. This approach has the advantage of it being the contributions mechanism currently imposed on development (albeit at a higher rate), and is thus familiar to developers.
- Continue to negotiate VPAs with developers to extract additional public benefits associated with new developments.

If on the other hand the Council was unsuccessful in obtaining the Minister's support for a 4.5% levy, and it still wanted to achieve a higher financial return than the current 3% levy, it could pursue a hybrid contributions strategy that allowed:

- Section 94 contributions to be imposed on mixed use development, which is expected to be the majority of the development.
- A section 94A levy on wholly non residential (i.e. commercial and retail) development.

However the disadvantage of this approach is that it is a more complex system to set up and manage. The relationship (nexus test) has to be determined for all the anticipated different development types to be subject to section 94 contributions and the Council is potentially exposed to legal challenge on the reasonableness of its contributions in individual development approvals. These risks however can be managed.

We therefore recommend that Council prepare a draft section 94A contributions plan that authorises a levy of up to 4.5% on future CBD development and seek the Minister for Planning's endorsement to amend the EP&A Regulation to allow the higher levy to be imposed on developments.

This approach, if successful, will provide a robust mechanism for an increase in development contributions revenue with minimal risk of legal challenge. It would also be necessary for Council to consider its position on encouraging solely commercial development in the CBD and whether a levy of less than 4.5% on these developments is needed to achieve this goal.

If Council is unsuccessful in gaining the Minister's approval for an increased levy, we recommend that Council pursue a mixed or hybrid contribution strategy of:

- (a) Section 94 contributions applying to all residential and mixed use development
- (b) A 3% section 94A levy applying to all development that does not include a residential component.



It is recommended that Council take the following action to ensure a robust section 94 contributions plan is prepared that minimises risk of legal challenge:

- (a) Prepare a comprehensive infrastructure schedule and costs that is linked to a well researched demand analysis to justify the schedule.
- (b) Carry out a thorough demand assessment of worker and visitor use of facilities in the infrastructure schedule to provide a robust argument for costs apportionment.
- (c) Analysis of the contribution rates that would generally be attributable to the worker population under both the section 94 and section 94A plans to ensure it is generally aligned and equitable.

6.2 Potential for value sharing schemes as an alternative or additional infrastructure funding source

Council officers advised that the estimated costs for the infrastructure that will be required for the future Parramatta CBD development exceeds \$800 million.

They have also advised us that continuing with the business-as-usual approach of imposing a 3% section 94A levy on CBD development will only yield a small fraction of the funds needed to meet this cost. A change to a 4.5% levy will improve the result, but a substantial funds shortfall will still remain. Council therefore wants to explore alternative or additional mechanisms to bridge the funding gap. Value sharing is one such mechanism.

Although value sharing has become a topical urban issue in recent times, there has historically been little appetite for State or Commonwealth Governments for pursuing a revenue model that extracts some of the uplift in value from planning or infrastructure decisions. Governments have so far been reluctant to act because of the difficulty in defining beneficiaries of planning and infrastructure decisions, likely community resistance, and treasuries being reluctant to tie revenue to particular areas and projects.

The proposed Parramatta light rail network offers potential as a catalyst infrastructure project that could underpin a value capture scheme. Such a scheme is currently under active consideration by the State Government. Council should monitor developments in this space in terms of potential effects on any local value capture scheme it may wish to implement.

Some Sydney local councils are implementing their own value sharing schemes. Whereas international examples of value sharing schemes are usually linked to major transport infrastructure projects, the local schemes are linked to the value uplift on sites that emanates from decisions that allow more valuable development types or increased intensity of development.

We reviewed 5 local schemes. In all but the Waverley scheme the extra floor space opportunities are secured when the statutory plan that allows those opportunities is activated or 'switched on'. This is important in sending a clear signal to the market that the full site value can only be achieved once the property is rezoned or the developer takes up the opportunity for bonus floor space through a formal VPA. If this is not done then it is highly likely that sites will be bought and sold on the basis that 100% of the extra value is secured by the land owner, and not shared with the community.

It is necessary for any local value sharing scheme to be supported by legislation provisions or a statutory plan such as an LEP that invokes a general power to allow additional development floor space to be approved in exchange for the provision of infrastructure or payment of infrastructure contributions.



A statutory scheme that shows all the candidate sites for bonus floor space has maximum transparency. This is the approach followed in the City of Sydney (in the Green Square Urban Renewal Area and the Green Square Town Centre) and in the City of Ryde (in the Macquarie Park Corridor). These bonus floor space schemes provide a good model for Council in structuring a similar scheme for the Parramatta CBD.

6.3 Development viability testing of Council's value sharing proposals

Council officers provided us with a 2-phase concept for a value sharing scheme that could apply to Parramatta CBD development sites in the future. GLN Planning tested the impacts of different value sharing regimes in terms of the contributions generated and the viability of hypothetical developments on sites with the typical FSR outcomes anticipated by Council. The typical FSR outcomes include:

- (a) 6.9:1 FSR with only Phase 1 value sharing applied to FSR in excess of the current controls;
- (b) 12:1 FSR with only Phase 1 value sharing applied to FSR in excess of the current controls; and
- (c) 15:1 FSR with Phase 1 value sharing applied to FSR in excess of the current controls and Phase 2 value sharing applied to additional FSR allowed for 'opportunity sites'.

This analysis was also compared to the contributions generated and viability outcomes that would result from the Council imposing a traditional section 94A levy scheme on the same hypothetical developments instead of, or in combination with, Phase 1 and 2 value sharing.

Table 11 showed a summary of the contributions and development viability outcomes of the various tested scenarios with the adoption of a 'value sharing' rate of \$805/m² of GFA. In light of the impacts of the different value sharing regimes on the hypothetical development at this rate, a revised rate of \$750/m² has been recommended to be a more reasonable 'through the cycle' rate for the purposes of a value sharing scheme.

The following is a summary of the results of our testing with the adoption of \$750/m² of GFA as for the different value sharing regimes (detailed results are shown in **Appendix D**):

For the 6.9:1 FSR development:

- The total development contribution ranged between \$2.2 million and \$6.7 million.
- The highest contribution would be in the instance of the developer sharing 50% of the value uplift created by the draft CBD Planning Proposal (i.e. a Phase 1 value share). However, that development would not likely be seen as viable by lending authorities because of the low profit margin and diminished IRR well below industry benchmarks.
- Negative impacts on viability are reduced in instances where the contribution is either a section 94A levy at 4.5%, or where it is a combination of Phase 1 value sharing at the 10% level. These hypothetical developments would likely to be seen as viable in the current market, with the total contribution being \$3.3 million, \$3.1 million and \$4 million, respectively.

For the 12:1 FSR development:

- The total development contribution ranged between \$3.9 million and \$8.3 million.
- The highest contribution would be in the instance of the developer sharing 50% of the value uplift created by the draft CBD Planning Proposal (i.e. a Phase 1 value share). However,



being a larger and more risky development than the 6.9:1 FSR development, its viability would likely be seen as marginal.

 Negative impacts on viability are reduced for the other scenarios, with a similar viability benchmarks and contributions amounts being achieved where the contributions imposed were a 4.5% levy or a combination 3% levy / 20% Phase 1 value share. These hypothetical developments would likely to be seen as viable in the current market, with the total contribution being \$5.8 million and \$5.6 million, respectively.

For the 15:1 FSR development:

- The total development contribution ranged between \$9 million and \$12.6 million.
- While this hypothetical development is the largest and riskiest of the three, viability benchmarks are likely to be easily achieved where any contributions scheme involved both Phase 1 and 2 value sharing.
- The favourable viability results are due to the assumption that the site purchase price was based on development yield that would be achieved under the draft CBD Planning Proposal

In considering the parameters of a value sharing scheme, it will be necessary for Council to balance the feasibility impacts on development against the significant shortfall in funding that will be available to deliver necessary infrastructure support the significant population growth that is expected.

This will have to be assessed on CBD wide basis, not just on individual hypothetical development sites. In this regard, when the proportion of sites that will deliver a 6.9:1 FSR outcome versus a 12:1 FSR and above outcome is considered, the standard application of a section 94A contribution at 4.5% with only Phase 2 value sharing may deliver a lower contribution pool than the application of a section 94A contribution at 3% accompanied by a Phase 1 and Phase 2 value sharing regime.

It is our understanding that in progressing a potential value sharing scheme, Council will be considering the impacts on feasibility including impacts on the limited number of sites which have been recently purchased in the last two years, and on the larger number of sites that have been held for longer periods. Council will also be considering the overall cost of infrastructure required to be funded to support development in the CBD and potential pool of contributions to fund these works.

6.4 Way forward for Council to pursue a local value sharing scheme

Council is considering implementing a 2-phase value sharing scheme on developments in the Parramatta CBD area in order to help fund the significant CBD infrastructure upgrades.

If Council was to proceed with its value sharing scheme, based on our investigations of similar schemes and our assessment of viability impacts, we recommend that Council incorporate the following elements in the scheme:

- (a) Retain the 3% section 94A levy in preference to seeking the Minister's approval for a higher (4.5%) levy.
- (b) Develop in consultation with the Department of Planning and Environment a town planning scheme that provides a choice for the developer between pursuing an 'as-ofright' development under current general planning controls and infrastructure contributions; or pursuing a development under different planning controls that also have value sharing contributions attached to them.



- (c) Prepare a comprehensive infrastructure plan containing details of the different facilities and amenities that will be delivered using the proceeds from value sharing.
- (d) Prepare guidelines that contain details of the value sharing scheme and that show how developers can participate in the scheme.
- (e) Adopt a floor space value in the range of \$700 to \$750 per square metre of GFA for the purpose of assessing value to be shared.
- (f) Consider carefully the rate of Phase 1 value sharing that is implemented. Our investigations of hypothetical developments based on recent average sales data show that larger developments could absorb up to a 20% Phase 1 value share with the community and probably meet viability benchmarks. Smaller developments on sites purchased in the last two years could absorb a 10% Phase 1 value share. Developments on sites that have been held for several years would readily absorb 20% or more Phase 1 value share, regardless of FSR.
- (g) The Phase 2 value sharing rate be set at no more than 50%.
- (h) The arrangement for the developer to provide public benefits is achieved through the negotiation of a VPA between the developer and the Council.
- (i) Value sharing contributions that are paid by developers be held in a dedicated account that has accountability and reporting protocols that at least reflect the accounting requirements for section 94 and section 94A monies.



Glossary of terms and abbreviations

CBD	Central business district
DA	Development application
DPE	Department of Planning and Environment
EP&A Act	Environmental Planning and Assessment Act 1979
EP&A Regulation	Environmental Planning and Assessment Regulation 2000
FSR	Floor space ratio
GFA	Gross floor area
GAB	Growth area bonds
INSW	Infrastructure New South Wales
IPART	Independent Pricing and Regulatory Tribunal
IRR	Internal rate of return
LEP	Local Environmental Plan
LGA	Local Government Area
SIC	Special Infrastructure Contribution
TIF	Tax increment financing
VPA	Voluntary Planning Agreement
VS	Value sharing



References

Architectus (2014), *Parramatta City Centre Planning Framework Study*, prepared for Parramatta City Council, Draft Report, September

Architectus (2013), *Macquarie Park Plan Review Recommendations Paper*, prepared for City of Ryde Council

City of Ryde (2013), Macquarie Park – Draft Guideline for 'Un-deferrals' and Delivery of Public Infrastructure

City of Ryde (2015), City of Ryde Section 94 Development Contribution Plan 2007

Commonwealth of Australia (2015), *Transport infrastructure and land value uplift*, Information sheet 69, Department of Infrastructure and Regional Development, May

Consult Australia and Sinclair Knight Merz (2013), *Realising New Funding for Infrastructure and Urban Renewal: Capturing Value*

Council of the City of Sydney (2012), *Development Guidelines – Providing Infrastructure in Green Square*, December

Council of the City of Sydney (2015), Draft Southern Employment Lands Infrastructure Plan, March

Council of the City of Sydney (2015), Draft Guideline to Preparing Site Specific Planning Proposal Requests in the City of Sydney Employment Lands Investigation Areas

Hale C (2014), *City Rail Infrastructure Funding: International background and policy options for funding transit,* prepared for the City of Melbourne, May

Infrastructure Finance Working Group (2012), Infrastructure Finance and Funding Reform, April

Infrastructure NSW (2012), First Things First – The State Infrastructure Strategy 2012-2032, October

Langley J (2015), Value Capture Roadmap, prepared for Consult Australia and AECOM, June

NSW Government (2014), A Plan for Growing Sydney, December

Parramatta City Council (2008), *Planning Agreements Policy*, March, Reference 24689-002, Revision 3

Parramatta City Council (2014), Draft Auto Alley Planning Framework

Parramatta City Council (2015), Parramatta CBD Planning Strategy, adopted 27 April 2015

Productivity Commission (2013), Public Infrastructure, Issues Paper, November

Productivity Commission (2014), Public Infrastructure, Draft Inquiry Report, March.

Property Council of Australia (NSW) (2010), Growth Area Bonds: New Solutions for Funding Infrastructure

Rawlinsons (2015), Australian Construction Handbook



SGS Economics & Planning (2014), *Parramatta CBD Planning Framework: Economic analysis*, Draft Report, prepared for Parramatta City Council, August

Sydney Olympic Park Authority (unpublished), *Policy Paper - Proposed Infrastructure Contributions Framework*

Urbis (2015), Achieving A-Grade Office Space in the Parramatta CBD, August

Waverley Council (2014), Planning Agreement Policy



APPENDIX A Estimated Income Analysis (Standard schemes)



Scenario - Current Dwelling and Job Targets

Scenario Summary

Detail	Amount
Residential (floorspace m2)	750,000
Mixed Use (floorspace m2)	75,000
Commercial (floorspace m2)	573,000
No. Dwellings	7,500
No. Employees (Commercial)	23,875
No. Employees (Mixed Use)	3,125
Cost of works - Commercial (additional)	\$1,432,500,000
Cost of works - Commercial (replacement)	\$1,048,500,000
Cost of works - Commercial carparking	\$286,500,000
Cost of works - Mixed Use	\$187,500,000
Cost of works - Mixed Use carparking	\$37,500,000
Cost of works - Residential	\$2,137,500,000
Cost of works - Residential carparking	\$375,000,000

Contributions Generated			
S94	\$193,200,000		
\$20,000/Dwelling	\$150,000,000		
Employee	\$43,200,000		
Section 94A @ 3%	\$165,150,000		
Section 94A @ 4.5%	\$247,725,000		
Combination S94 and S94A @3%	\$206,570,000		
\$20,000/Dwelling	\$150,000,000		
Employees in Mixed Use	\$5,000,000		
Commercial S94A @ 3%	\$51,570,000		
Combination S94 and S94A @4.5%	\$232,355,000		
\$20,000/Dwelling	\$150,000,000		
Employees in Mixed Use	\$5,000,000		
Commercial S94A @ 4.5%	\$77,355,000		

Notes:

All contributions generated are indicative only for discussion purposes.

Commercial floorspace based on Council/SGS calculations for purposes of S94A contribution calculation, employee numbers calculation and S94 contribution calculation.

Residential floorspace based on Council/SGS calculations for purposes of S94A contribution calculation and dwelling numbers for S94 contribution calculation.

100m2 GFA per apartment assumed to determine dwelling numbers for S94 contribution calculation. Mixed use floorspace was based on 10% of the total residential floor space in order to deliver residential development in the B4 zone.

Number of employees based on Council/SGS rate of 1 employee per 24m2 of commercial GFA.

For purposes of determining a S94 contribution for commercial development, assumed 5 employees are equivalent to 1 resident in terms of demand/utilisation of local infrastructure as per City of Sydney approach/studies. A contribution of \$20,000 per dwelling and assumed occupancy rate of 2.5 people per dwelling was used to determine a contribution rate of \$8,000/resident. This equates to \$1,600/employee.

In order to achieve the total additional floorspace of 1,398,000 (as per Council/SGS data), assumed that there will need to be knockdown and replacement of existing residential and commercial developments to realise true potential of sites. As such, assumed that an additional 30% of the total additional floorspace would also be delivered as part of replacement of existing devleopment for the purposes of determining S94A contributions based on the total construction costs in the Parramatta CBD.

<u>Costs</u>

Commercial floorspace (new and replacement)at \$2,500/m2 (Rawlinsons 2015)

Residential floorspace at \$2,850/m2 (Rawlinsons 2015)

Mixed use floorspace at \$2,500/m2 (Rawlinsons 2015)

Underground carparking at \$50,000/space applied to Council's carparking controls rates to overall floorspace and dwellings (Rawlinsons 2015)

APPENDIX B

Standard contributions approach development feasibility analysis



Scenario 1 - S94A Levy 3% CBD Strategy Controls

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	10.0
Land Area	3,000
Site Value	79,762
Number of Dwellings	273

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$697,985	\$208,466,667
GST	62,168	16,971,970
Less Selling Costs	16,022	4,374,083
NET REVENUE	\$685,423	\$187,120,614
COSTS		
Land (including acquisition costs)	79,762	21,775,000
Acquisition costs	5,728	1,563,778
Construction	386,628	105,549,517
Consultants	15,465	4,221,981
Section 94A - Commercial	1,969	537,500
Section 94A - Residential	9,630	2,628,986
Statutory Fees & Contributions	7,807	2,131,430
On Costs	11,599	3,166,486
Marketing	15,272	4,169,333
Cost before Interest	533,861	145,744,009
Finance (incl Loan Est Fees)	46,287	12,636,271
TOTAL DEVELOPMENT COSTS	580,148	\$158,380,280
TOTAL PROJECT SURPLUS	18.1%	\$28,740,333
PROJECT IRR BEFORE INTEREST	20.0%	

Scenario 1 - S94A Levy 3% CBD Strategy Controls Input Sheet

SITE DETAILS				
Address	Representative			
Floor Space Ratio	10.0	10.0 :1		
Achievable FSR	30,000	m2		
Land Area	3,000	m2		
Site Value	79,762			
Any 1 parcel over \$3M	Yes			
LVR (Debt/Equity)	100%			
Date to start feasibility	Oct-15			



Cost Items	Cost (\$)	Period Commencing (month)	Term (months)	
LAND				
Land Purchase		21,775,000		
Loan Establishment Costs		22,825		
Stamp Duty		1,464,740		
Legals 0.4%		76,213		
LAND ACQUISITION COSTS - Deposit		2,177,500	1	1
LAND ACQUISITION COSTS - Settlement		21,161,278	2	1
CONSTRUCTION COSTS 2,000		105,549,517	12	20
PROFESSIONAL FEES 4%	4,221,981	1	31	
APPLICATION FEES				
DA Fees		502,018	3	1
CC Fees		194,920	12	1
Section 94		3,166,486	31	1
LPI Fees		37,275	31	1
LAND TAX/RATES				
Land Tax/Rates Year 1		465,739	10	1
Land Tax/Rates Year 2		465,739	22	1
Land Tax/Rates Year 3		465,739	34	1
ONCOSTS 3.0%	6 construction cost	3,166,486	2	30
MARKETING 2%	6 gross revenue	4,169,333	12	19
	TOTAL COST	145,744,009		

					Basement				
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
NON RESIDENTIAL	8					(1)	(1)		
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL	-							•	
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0						
		Target Floorspace	30,000	7					

Sheet 2 of 4

Scenario 1 - S94A Levy 3% CBD Strategy Controls Calculations

REVENUE		
GROSS REVENUE		208,466,667
GST		16,971,970
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	4,169,333
Legals (On Settlement)	\$750 per lot	204,750
TOTAL - SELLING COSTS		4,374,083
NET REVENUE		187,120,614



STAMP DUTY					
LAND VALUE THRESHOLD TAX					
Minimum	Maximum				
14,000		0			
14,000	30,000	0			
30,000	80,000	0			
80,000	300,000	0			
300,000	1,000,000	0			
1,000,000		FALSE			
3,000,000	Premium Property Tax	1,464,740			
	TOTAL	1,464,740			

CONTRIBUTIONS				
SECTION 94A	3%			
COSTS				
Retail	10,416,667	\$312,500		
Commercial	7,500,000	\$225,000		
Residential	87,632,850	\$2,628,986		
TOTAL VALUE	105,549,517			
TOTAL CONT	RIBUTION	\$3,166,486		

DA FEES		
CONSTRUCTION (COST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		132,595
LSL	0.35%	369,423
TOTAL DA FEES		502,018

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	194,920
TOTAL CC FEES	194,920

CONSTRUCTION COSTS					
Build Costs	Rates \$/m2	FSR +15%	UG Parking 50,000	Grade Parking	
Apartments	2,850	71,252,850	13,650,000	0	
Retail	2,500	6,250,000	4,166,667		
Commercial	2,500	6,250,000	1,250,000		
Visitor Parking			2,730,000	0	
TOTALS		83,752,850	21,796,667	0	
TOTAL CONSTRUCTION COST				105,549,517	

CAR PARKING				
Unit Type	No. Spaces			
1 bed/studio	1.0			
2 bed	1.0			
3+ bed	1.0			
Shops	30			
Commercial	100			
Visitor parking	In any case, min 1 visitor space is required			

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	37,275		

Sheet 3 of 4

Scenario 1 - S94A Levy 3% CBD Strategy Controls

Assumed Development Cashflow

DEVELOPMENT SUM	DEVELOPMENT SUMMARY		
IRR Before Interest	20.00%		
Surplus	\$28,740,333		
% Surplus on D.C	18.15%		
Max Loan Balance Debt/Equity	-\$157,096,331		
Debt/Equity	100.00%		

		ice Deb	t/Equity	-\$157,096,331	1																	nlang	ind
Debt/Eq	uity			100.00%	Numbers be	ellow expres	sed in nearest	\$,000														planr	nng
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,178	136	0	0	0	0	0	0	0	0	\$2,314	0	\$0	\$0	\$0	\$0	\$2,314	-\$2,314	\$0	-\$2,314	1
Budget	Dec-15	2	21,161	136	0	0	0	0	0	0	106	0	\$21,403	0	\$0	\$0	\$0	\$0	\$21,403	-\$21,403	-\$12	-\$23,729	2
Budget	Jan-16	3	0	136	502	0	0	0	0	0	106	0	\$744	0	\$0	\$0	\$0	\$0	\$744	-\$744	-\$124	-\$24,596	3
Budget	Feb-16	4	0	136	0	0	0	0	0	0	106	0	\$242	0	\$0	\$0	\$0	\$0	\$242	-\$242	-\$128	-\$24,966	4
Budget	Mar-16	5	0	136	0	0	0	0	0	0	106	0	\$242	0	\$0	\$0	\$0	\$0	\$242	-\$242	-\$130	-\$25,338	5
Budget	Apr-16	6	0	136	0	0	0	0	0	0	106	0	\$242	0	\$0	\$0	\$0	\$0	\$242	-\$242	-\$132	-\$25,711	6
Budget	May-16	7	0	136	0	0	0	0	0	0	106	0	\$242	0	\$0	\$0	\$0	\$0	\$242	-\$242	-\$134	-\$26,087	7
Budget	Jun-16	8	0	136	0	0	0	0	0	0	106	0	\$242	0	\$0	\$0	\$0	\$0	\$242	-\$242	-\$136	-\$26,465	8
Budget	Jul-16	9 10	0	136 136	0	0	0	0	0	0 466	106 106	0	\$242 \$707	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$242 \$707	-\$242 -\$707	-\$138 -\$140	-\$26,844 -\$27,692	9 10
Budget Budget	Aug-16 Sep-16	10	0	136	0	0	0	0	0	400	106	0	\$242	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0	\$242	-\$707	-\$140 -\$144	-\$28,078	10
Budget	Oct-16	12	0	136	0	195	5,277	0	0	0	106	219	\$5,934	0	\$0	\$0 \$0	\$0	\$0	\$5,934	-\$5,934	-\$146	-\$20,070	12
Budget	Nov-16	13	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0 \$0	\$0	\$0	\$5,739	-\$5,739	-\$178	-\$40,074	13
Budget	Dec-16	14	ů.	136	0	0	5,277	0 0	0	0 0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$209	-\$46,021	14
Budget	Jan-17	15	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$240	-\$52,000	15
Budget	Feb-17	16	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$271	-\$58,009	16
Budget	Mar-17	17	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$302	-\$64,050	17
Budget	Apr-17	18	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$334	-\$70,122	18
Budget	May-17	19	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$365	-\$76,226	19
Budget	Jun-17	20	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$397	-\$82,362	20
Budget	Jul-17	21	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$429	-\$88,529	21
Budget	Aug-17	22	0	136	0	0	5,277	0	0	466	106	219	\$6,204	0	\$0	\$0	\$0	\$0	\$6,204	-\$6,204	-\$461	-\$95,195	22
Budget	Sep-17	23	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$496	-\$101,429	23
Budget	Oct-17	24	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$528	-\$107,696	24
Budget Budget	Nov-17 Dec-17	25 26	0	136 136	0	0	5,277 5,277	0	0	0	106 106	219 219	\$5,739 \$5,739	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$5,739 \$5,739	-\$5,739 -\$5,739	-\$561 -\$594	-\$113,996 -\$120,328	25 26
Budget	Jan-18	20	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$627	-\$126,694	20
Budget	Feb-18	28	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$660	-\$133,092	28
Budget	Mar-18	29	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$693	-\$139,524	29
Budget	Apr-18	30	0	136	0	0	5,277	0	0	0	106	219	\$5,739	0	\$0	\$0	\$0	\$0	\$5,739	-\$5,739	-\$727	-\$145,989	30
Budget	May-18	31	0	136	0	0	5,277	3,166	37	0	106	0	\$8,723	0	\$0	\$0	\$0	\$0	\$8,723	-\$8,723	-\$760	-\$155,473	31
Budget	Jun-18	32	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$810	-\$156,282	32
Budget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$814	-\$157,096	33
Budget	Aug-18	34	0	0	0	0	0	0	0	466	0	0	\$466	70	\$208,467	\$16,972	\$4,374	\$187,121	\$466	\$186,655	-\$818	\$28,740	34
	1	1	23,339	\$4,222	\$502	\$195	\$105,550	\$3,166	\$37	\$1,397	\$3,166	\$4,169	\$145,744	70	\$208,467	\$16,972	\$4,374	\$187,121	\$145,744	\$41,377	-\$12,636	\$28,740	
			Land	Professional	DA Fees	CC Fees	Construction	Section 94	LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		
				Fees			Costs	Costs		Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

Scenario 2 - S94A Levy 3% CBD Strategy Controls + 5:1 FS

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	15.0
Land Area	3,000
Site Value	79,762
Number of Dwellings	433

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	62,480	27,053,788
Less Selling Costs	15,501	6,712,083
	10,001	0,1 12,000
NET REVENUE	\$659,586	\$285,600,795
COSTS		
Land (including acquisition costs)	50,289	21,775,000
Acquisition costs	3,611	1,563,778
Construction	361,307	156,446,017
Consultants	14,452	6,257,841
Section 94A - Commercial	1,241	537,500
Section 94A - Residential	9,598	4,155,881
Statutory Fees & Contributions	5,736	2,483,745
On Costs	10,839	4,693,381
Marketing	9,629	4,169,333
Cost before Interest	466,703	202,082,475
Finance (incl Loan Est Fees)	38,110	16,501,477
TOTAL DEVELOPMENT COSTS	504,813	\$218,583,952
TOTAL PROJECT SURPLUS	30.7%	\$67,016,844
PROJECT IRR BEFORE INTEREST	30.1%	
PROJECT IKK BEFORE INTEREST	30.1%	

Sheet 1 of 4

Scenario 2 - S94A Levy 3% CBD Strategy Controls with Planning Proposal +5:1 FSR Input Sheet

SITE DETAILS					
Address	Repr	Representative			
Floor Space Ratio	15.0	:1			
Achievable FSR	45,000	m2			
Land Area	3,000	m2			
Site Value	79,762				
Any 1 parcel over \$3M	Yes				
LVR (Debt/Equity)	100%				
Date to start feasibility	Oct-15				



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		21,775,000		
Loan Establishment Costs		22,825		
Stamp Duty		1,464,740		
Legals	0.4%	76,213		
LAND ACQUISITION COSTS	- Deposit	2,177,500	1	1
LAND ACQUISITION COSTS	- Settlement	21,161,278	2	1
CONSTRUCTION COSTS	2,000	156,446,017	12	20
PROFESSIONAL FEES	4% const costs	6,257,841	1	31
APPLICATION FEES				
DA Fees		740,723	3	1
CC Fees		287,459	12	1
Section 94		4,693,381	31	1
LPI Fees		58,347	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	465,739	10	1
Land Tax/Rates	Year 2	465,739	22	1
Land Tax/Rates	Year 3	465,739	34	1
ONCOSTS	3.0% construction cost	4,693,381	2	30
MARKETING	2% gross revenue	4,169,333	12	19
	TOTAL COST	202,082,475		

					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL						• •			
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL				-					
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80	•		•		-	•	-	
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0						
		Target Floorspace	45,000	1					

ADDITIONAL UNIT MIX AND SALES (based on additional 5:1 FSR after Planning Proposal Basement Unit Type No. Dwelling **Car Parking** Parking Price Total Actual Preferred Total Floor Space m2 FSR+15% Yes (\$) (\$) Mix Mix RESIDENTIAL 30 60 2,070 30 Yes 590,000 17,700,000 19% 15% 1 bedroom 120 80 11,040 120 Yes 700,000 84,000,000 75% 75% 2 bedroom 3 bedroom 10 120 1,380 10 Yes 920,000 9,200,000 6% 10.0% 4 bedroom 130 0 0 Yes 0 0 0.0% 110,900,000 100% 160 TOTAL 14,490 160 Yes 100% Average m2/unit 29 ADDITIONAL INCLUSIONS Other visitor parking 32 TOTAL YIELD 160 14,490 110,900,000 192 COMBINED YIELD 433 44,491 628 319,366,667 FSR 14.8 Target Floorspace 45,000

Scenario 2 - S94A Levy 3% CBD Strategy Controls + 5:1 FSR

Calculations

NET REVENUE		285,600,795
TOTAL - SELLING COSTS		6,712,083
Legals (On Settlement)	\$750 per lot	324,750
Sales Commission (On Settlement)	2% on gross rev	6,387,333
SELLING COSTS		
GST		27,053,788
GROSS REVENUE		319,366,667
REVENUE		



STAMP DUTY						
LAND VALUE	ΤΑΧ					
Minimum	Maximum					
14,000		0				
14,000	30,000	0				
30,000	80,000	0				
80,000	300,000	0				
300,000	1,000,000	0				
1,000,000		FALSE				
3,000,000	Premium Property Tax	1,464,740				
	TOTAL	1,464,740				

CONTRIBUTIONS						
SECTION 94A	3%					
COSTS						
Retail	10,416,667	\$312,500				
Commercial	7,500,000	\$225,000				
Residential	138,529,350	\$4,155,881				
TOTAL VALUE	156,446,017					
TOTAL CONT	\$4,693,381					

DA FEES							
CONSTRUCTION COST THRESHOLDS							
50,001	250,000	0					
250,001	500,000	0					
500,001	1,000,000	0					
1,000,001	10,000,000	0					
More than \$10,000,000		193,162					
LSL	0.35%	547,561					
TOTAL DA FEES		740,723					

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	287,459
TOTAL CC FEES	287,459

CONSTRUCTION COSTS								
Build Costs	· · · · · · · · · · · · · · · · · · ·							
	\$/m2		50,000	Parking				
Apartments	2,850	112,549,350	21,650,000	0				
Retail	2,500	6,250,000	4,166,667					
Commercial	2,500	6,250,000	1,250,000					
Visitor Parking			4,330,000	0				
TOTALS		125,049,350	31,396,667	0				
	156,446,017							

CAR PARKING							
Unit Type	No. Spaces						
1 bed/studio	1.0						
2 bed	1.0						
3+ bed	1.0						
Shops	30						
Commercial	100						
Visitor parking	In any case, min 1 visitor space is required						

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee 1,321	Per Dwg fee 132	
TOTAL LPI F	58,347		

Sheet 3 of 4

Scenario 2 - S94A Levy 3% CBD Strategy Controls with Planning Proposal +5:1 FSR

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	30.12%
Surplus	\$67,016,844
% Surplus on D.C	30.66%
Max Loan Balance Debt/Equity	-\$216,988,067
Debt/Equity	100.00%

IRR Before Interest 30.12%						
Surplus \$67,016,844					I	
% Surplus on D.C 30.66%			4			
Max Loan Balance Debt/Equity -\$216,988,067					nland	ine
Debt/Equity 100.00% Numbers bellow expressed in nearest \$,000	-		-		planr	iing
Budger/ Oct-15 Month Land Professional DA Fees CC Fees Construction Section 94 LPI Fees Statuory Oncosts Marketing Total Sales Gross GST Selling Costs Net	Monthly Net	Monthly		Monthly	Total	Period
			Cashflow		Funds	1 onloa
					Cashflow	
				6.25%		
Budget Oct-15 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$0 \$0	\$0	\$0	\$0	\$0	0
Budget Nov-15 1 2,178 202 0 0 0 0 0 0 0 0 0 \$ \$ \$ \$ \$ \$ \$ \$ \$	\$2,379 -\$2,379	\$2,379	-\$2,379	\$0	-\$2,379	1
			-\$21,520	-\$12	-\$23,911	2
Budget Jan-16 3 0 202 741 0 0 0 0 0 156 0 \$1,099 0 \$0 \$0 \$0 \$0 \$0	\$1,099 -\$1,099			-\$125	-\$25,135	3
Budget Feb-16 4 0 202 0 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 \$0 \$0	\$358 -\$358			-\$131	-\$25,624	4
Budget Mar-16 5 0 202 0 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 \$0	\$358 -\$358			-\$133	-\$26,116	5
Budget Apr-16 6 0 202 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 Budget May-16 7 0 202 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 \$0	\$358 -\$358 \$358 -\$358			-\$136 -\$139	-\$26,610 -\$27,107	6
Budget May-16 7 0 202 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 Budget Jun-16 8 0 202 0 0 0 0 156 0 \$358 0 \$0	\$358 -\$358			-\$139	-\$27,107	8
Budget Jul-16 9 0 202 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$358 -\$358			-\$144	-\$28,109	9
Budget Jun 5 5 202 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$824 -\$824			-\$146	-\$29,079	10
Budget Sep-16 11 0 202 0 0 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 \$0	\$358 -\$358			-\$151	-\$29,589	11
Budget Oct-16 12 0 202 0 287 7,822 0 0 0 156 219 \$8,688 0 \$0 \$0 \$0 \$0 0	\$8,688 -\$8,688	\$8,688	-\$8,688	-\$154	-\$38,431	12
Budget Nov-16 13 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 0	\$8,400 -\$8,400	\$8,400	-\$8,400	-\$200	-\$47,031	13
Budget Dec-16 14 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 0	\$8,400 -\$8,400			-\$245	-\$55,676	14
Budget Jan-17 15 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 0	\$8,400 -\$8,400			-\$290	-\$64,366	15
Budget Feb-17 16 0 202 0 0 7.822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0	\$8,400 -\$8,400			-\$335	-\$73,101	16
Budget Mar-17 17 0 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0	\$8,400 -\$8,400			-\$381 -\$426	-\$81,882	17 18
Budget April 7 18 0 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0	\$8,400 -\$8,400 \$8,400 -\$8,400			-\$426 -\$472	-\$90,708 -\$99,581	18
Budget Jun-17 20 0 202 0 0 7,622 0 0 0 1,622 0 0 0 350 219 30,400 0 30 30 30 30 30 30 30 30 30 30 30 30	\$8,400 -\$8,400			-\$519	-\$108,500	20
Budget Jul-17 21 0 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 \$0	\$8,400 -\$8,400			-\$565	-\$117,465	21
Budget Aug-17 22 0 202 0 0 7,822 0 0 466 156 219 \$8,866 0 \$0 \$0 \$0 \$0	\$8,866 -\$8,866	\$8,866	-\$8,866	-\$612	-\$126,942	22
Budget Sep-17 23 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 0	\$8,400 -\$8,400	\$8,400	-\$8,400	-\$661	-\$136,004	23
Budget Oct-17 24 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 0	\$8,400 -\$8,400			-\$708	-\$145,112	24
Budget Nov-17 25 0 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 \$0	\$8,400 -\$8,400			-\$756	-\$154,268	25
Budget Dec-17 26 0 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0	\$8,400 -\$8,400			-\$803	-\$163,471	26
	\$8,400 -\$8,400 \$8,400 -\$8,400			-\$851 -\$900	-\$172,723 -\$182,022	27 28
Budget Feb-18 28 0 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0	\$8,400 -\$8,400 \$8,400 -\$8,400			-\$900 -\$948	-\$182,022 -\$191,371	28
Budget Martin 29 0 202 0 0 7,622 0 0 0 1,622 0 0 0 30 20 30 30 30 30 30 30 30 30 30 30 30 30 30	\$8,400 -\$8,400			-\$997	-\$200,767	30
			-\$12,932	-\$1,046	-\$214,745	31
Budget Jun-18 32 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$0 \$0			-\$1,118	-\$215,864	32
Budget Jul-18 33 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 \$0 \$0 \$	\$0 \$0	\$0	\$0	-\$1,124	-\$216,988	33
Budget Aug-18 34 0 0 0 0 0 0 0 0 466 0 0 \$466 70 \$319,367 \$27,054 \$6,712 \$285,601	\$466 \$285,135	\$466	\$285,135	-\$1,130	\$67,017	34
23,339 \$6,258 \$741 \$287 \$156,446 \$4,693 \$58 \$1,397 \$4,693 \$4,169 \$202,082 70 \$319,367 \$27,054 \$6,712 \$285,601	\$202,082 \$83,518	\$202.082	\$83.518	-\$16,501	\$67,017	
	Monthly Net			Monthly		
	Outlays			Interest		

Sheet 4 of 4

Scenario 3 - S94A Levy 4.5% CBD Strategy Controls

Summary Sheet

SITE DETAILS					
Address	Representative				
Floor Space Ratio	10.0				
Land Area	3,000				
Site Value	79,762				
Number of Dwellings	273				

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$697,985	\$208,466,667
GST	62,168	16,971,970
Less Selling Costs	16,022	4,374,083
NET REVENUE	\$685,423	\$187,120,614
COSTS		
Land (including acquisition costs)	79,762	21,775,000
Acquisition costs	5,728	1,563,778
Construction	386,628	105,549,517
Consultants	15,465	4,221,981
Section 94A - Commercial	2,953	806,250
Section 94A - Residential	14,445	3,943,478
Statutory Fees & Contributions	7,807	2,131,430
On Costs	11,599	3,166,486
Marketing	15,272	4,169,333
Cost before Interest	539,660	147,327,252
Finance (incl Loan Est Fees)	46,378	12,661,139
TOTAL DEVELOPMENT COSTS	586,038	\$159,988,390
TOTAL PROJECT SURPLUS	17.0%	\$27,132,223
PROJECT IRR BEFORE INTEREST	19.3%	

Scenario 3 - S94A Levy 4.5% CBD Strategy Controls Input Sheet

SITE DETAILS						
Address	Repr	Representative				
Floor Space Ratio	10.0	:1				
Achievable FSR	30,000	m2				
Land Area	3,000	m2				
Site Value	79,762					
Any 1 parcel over \$3M	Yes					
LVR (Debt/Equity)	100%					
Date to start feasibility	Oct-15					



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)	
LAND		-			
Land Purchase		21,775,000			
Loan Establishment Costs		22,825			
Stamp Duty		1,464,740			
Legals	0.4%	76,213			
LAND ACQUISITION COSTS - De	posit	2,177,500	1	1	
LAND ACQUISITION COSTS - Set	tlement	21,161,278	2	1	
CONSTRUCTION COSTS	2,000	105,549,517	12	20	
PROFESSIONAL FEES	4% const costs	4,221,981	1	31	
APPLICATION FEES					
DA Fees		502,018	3	1	
CC Fees		194,920	12	1	
Section 94		4,749,728	31	1	
LPI Fees		37,275	31	1	
LAND TAX/RATES					
Land Tax/Rates Year	1	465,739	10	1	
Land Tax/Rates Year	2	465,739	22	1	
Land Tax/Rates Year	3	465,739	34	1	
ONCOSTS	3.0% construction cost	3,166,486	2	30	
MARKETING	2% gross revenue	4,169,333	12	19	
	TOTAL COST	147,327,252			

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80								-
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
	-	FSR	10.0			-			
		Target Floorspace	30,000						

Sheet 2 of 4

Scenario 3 - S94A Levy 4.5% CBD Strategy Controls Calculations

REVENUE		
GROSS REVENUE		208,466,667
GST		16,971,970
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	4,169,333
Legals (On Settlement)	\$750 per lot	204,750
TOTAL - SELLING COSTS		4,374,083
NET REVENUE		187,120,614



STAMP DUTY							
LAND VALUE THRESHOLD TAX							
Minimum	Maximum						
14,000		0					
14,000	30,000	0					
30,000	80,000	0					
80,000	300,000	0					
300,000	1,000,000	0					
1,000,000		FALSE					
3,000,000	Premium Property Tax	1,464,740					
	TOTAL	1,464,740					

CONTRIBUTIONS							
SECTION 94A	L	4.5%					
COSTS							
Retail	10,416,667	\$468,750					
Commercial	7,500,000	\$337,500					
Residential	87,632,850	\$3,943,478					
TOTAL VALUE							
TOTAL CONT	\$4,749,728						

DA FEES		
CONSTRUCTION O		SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		132,595
LSL	0.35%	369,423
TOTAL DA FEES		502,018

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	194,920
TOTAL CC FEES	194,920

CONSTRUCTION COSTS								
Build Costs	Rates \$/m2	FSR +15%	UG Parking 50,000	Grade Parking				
Apartments	2,850	71,252,850	13,650,000	0				
Retail	2,500	6,250,000	4,166,667					
Commercial	2,500	6,250,000	1,250,000					
Visitor Parking			2,730,000	0				
TOTALS		83,752,850	21,796,667	0				
	105,549,517							

CAR PARKING							
Unit Type	No. Spaces						
1 bed/studio	1.0						
2 bed	1.0						
3+ bed	1.0						
Shops	30						
Commercial	100						
Visitor parking	In any case, min 1 visitor space is required						

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	EES		37,275

Sheet 3 of 4

Scenario 3 - S94A Levy 4.5% CBD Strategy Controls

Assumed Development Cashflow

Max Loan Balan	ce Deb	t/Fauity	-\$158 696 109																	1		
Debt/Equity		- Lidarity	100.00%		ellow express	ed in nearest	\$,000												<		plann	ning
Budget/ Oct-15 Actual	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget Oct-15 Budget Nov-15 Budget Dec-15	0 1 2 3	0 2,178 21,161 0	0 136 136	0 0 0	0 0 0	0 0 0	0 0 0	0 0 0	0 0	0 0 106	0 0 0	\$0 \$2,314 \$21,403 \$744	0 0 0	\$0 \$0 \$0	\$0 \$0 \$0	\$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$2,314 \$21,403	\$0 -\$2,314 -\$21,403	\$0 \$0 -\$12	\$0 -\$2,314 -\$23,729	0 1 2 3
Budget Jan-16 Budget Feb-16 Budget Mar-16 Budget Apr-16	3 4 5 6 7	0	136 136 136 136 136	502 0 0 0	0	0 0 0	0 0 0	0	0	106 106 106 106 106	0 0 0 0 0	\$744 \$242 \$242 \$242 \$242 \$242	0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$744 \$242 \$242 \$242 \$242 \$242	-\$744 -\$242 -\$242 -\$242	-\$124 -\$128 -\$130 -\$132 -\$134	-\$24,596 -\$24,966 -\$25,338 -\$25,711 -\$26,087	3 4 5 6
Budget May-16 Budget Jun-16 Budget Jul-16 Budget Aug-16 Budget Sep-16	7 8 9 10 11	0	136 136 136 136 136	0	0 0 0	0 0 0	0 0 0	0 0 0 0	0 0 466 0	106 106 106 106 106	0 0 0 0 0	\$242 \$242 \$242 \$707 \$242	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$242 \$242 \$242 \$707 \$242	-\$242 -\$242 -\$242 -\$707 -\$242	-\$134 -\$136 -\$138 -\$140 -\$144	-\$26,465 -\$26,844 -\$27,692 -\$28,078	7 8 9 10 11
Budget Oct-16 Budget Nov-16 Budget Dec-16 Budget Jan-17	12 13 14 15	0 0 0	136 136 136 136 136	0 0 0	195 0 0 0	5,277 5,277 5,277 5,277 5,277	0 0 0	0 0 0	0 0 0	106 106 106 106	219 219 219 219 219	\$5,934 \$5,739 \$5,739 \$5,739	0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$5,934 \$5,739 \$5,739 \$5,739	-\$5,934 -\$5,739 -\$5,739 -\$5,739	-\$146 -\$178 -\$209 -\$240	-\$34,157 -\$40,074 -\$46,021 -\$52,000	12 13 14 15
BudgetFeb-17BudgetMar-17BudgetApr-17BudgetMay-17	16 17 18 19	0 0 0	136 136 136 136	0 0 0	0 0 0	5,277 5,277 5,277 5,277	0 0 0	0 0 0	0 0 0	106 106 106 106	219 219 219 219	\$5,739 \$5,739 \$5,739 \$5,739	0 0 0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$5,739	-\$5,739 -\$5,739 -\$5,739 -\$5,739	-\$271 -\$302 -\$334 -\$365	-\$58,009 -\$64,050 -\$70,122 -\$76,226	16 17 18 19
Budget Jun-17 Budget Jul-17 Budget Aug-17 Budget Sep-17	20 21 22 23	0 0 0 0	136 136 136 136	0 0 0	0 0 0	5,277 5,277 5,277 5,277	0 0 0	0 0 0	0 0 466 0	106 106 106 106	219 219 219 219 219	\$5,739 \$5,739 \$6,204 \$5,739	0 0 0 0 0 0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0	\$5,739 \$5,739 \$6,204 \$5,739	-\$5,739 -\$5,739 -\$6,204 -\$5,739	-\$397 -\$429 -\$461 -\$496	-\$82,362 -\$88,529 -\$95,195 -\$101,429	20 21 22 23
Budget Oct-17 Budget Nov-17 Budget Dec-17 Budget Jan-18 Budget Feb-18	24 25 26 27 28	0 0 0 0	136 136 136 136 136	0 0 0 0	0 0 0 0	5,277 5,277 5,277 5,277 5,277	0 0 0 0	0 0 0	0	106 106 106 106 106	219 219 219 219 219 219	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739	-\$5,739 -\$5,739 -\$5,739 -\$5,739 -\$5,739	-\$528 -\$561 -\$594 -\$627 -\$660	-\$107,696 -\$113,996 -\$120,328 -\$126,694 -\$133,092	24 25 26 27 28
Budget Feb-18 Budget Mar-18 Budget Apr-18 Budget May-18 Budget Jun-18	20 29 30 31 32	0 0 0 0 0	136 136 136 136 0	0 0 0 0	0 0 0	5,277 5,277 5,277 5,277 0	0 0 4,750 0	0 0 37 0	0 0 0	106 106 106 106 0	219 219 219 0 0	\$5,739 \$5,739 \$5,739 \$10,306 \$0	0 0 0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$10,306 \$0	-\$5,739 -\$5,739 -\$5,739 -\$10,306 \$0	-\$693 -\$727 -\$760 -\$818	-\$133,092 -\$139,524 -\$145,989 -\$157,056 -\$157,874	28 29 30 31 32
Budget Jul-18 Budget Aug-18	33 34	0 0 23,339	0 0 \$4,222	0 0 \$502	0 0 \$195	0 0 \$105,550	0 0 \$4,750	0 0 \$37	0 466 \$1,397	0 0 \$3,166	0 0 \$4,169	\$0 \$466 \$147,327	0 70 70	\$0 \$208,467 \$208,467	\$0 \$16,972 \$16,972	\$0 \$4,374 \$4,374	\$0 \$187,121 \$187,121	\$0 \$466 \$147,327	\$0 \$186,655 \$39,793	-\$822 -\$827 -\$12,661	-\$158,696 \$27,132 \$27,132	33 34
		Land	Professional Fees	DA Fees		Construction Costs		LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs		Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net	Monthly Interest		-

Sheet 4 of 4

Scenario 4 - S94A Levy 4.5% CBD Strategy Controls + 5:1 F

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	15.0
Land Area	3,000
Site Value	79,762
Number of Dwellings	433

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	62,480	27,053,788
Less Selling Costs	15,501	6,712,083
NET REVENUE	\$659,586	\$285,600,795
COSTS		
Land (including acquisition costs)	50,289	21,775,000
Acquisition costs	3,611	1,563,778
Construction	361,307	156,446,017
Consultants	14,452	6,257,841
Section 94A - Commercial	1,862	806,250
Section 94A - Residential	14,397	6,233,821
Statutory Fees & Contributions	5,736	2,483,745
On Costs	10,839	4,693,381
Marketing	9,629	4,169,333
Cost before Interest	472,123	204,429,165
Finance (incl Loan Est Fees)	38,195	16,538,336
TOTAL DEVELOPMENT COSTS	510,318	\$220,967,500
TOTAL PROJECT SURPLUS	29.3%	\$64,633,295
PROJECT IRR BEFORE INTEREST	29.3%	
Scenario 4 - S94A Levy 4.5% CBD Strategy Controls with Planning Proposal +5:1 FSR Input Sheet

SITE DETAILS					
Address	Repr	esentative			
Floor Space Ratio	15.0	:1			
Achievable FSR	45,000	m2			
Land Area	3,000	m2			
Site Value	79,762				
Any 1 parcel over \$3M	Yes				
LVR (Debt/Equity)	100%				
Date to start feasibility	Oct-15				



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		21,775,000		
Loan Establishment Costs		22,825		
Stamp Duty		1,464,740		
Legals	0.4%	76,213		
LAND ACQUISITION COST	S - Deposit	2,177,500	1	1
LAND ACQUISITION COST	S - Settlement	21,161,278	2	1
CONSTRUCTION COSTS	2,000	156,446,017	12	20
PROFESSIONAL FEES	4% const costs	6,257,841	1	31
APPLICATION FEES				
DA Fees		740,723	3	1
CC Fees		287,459	12	1
Section 94		7,040,071	31	1
LPI Fees		58,347	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	465,739	10	1
Land Tax/Rates	Year 2	465,739	22	1
Land Tax/Rates	Year 3	465,739	34	1
ONCOSTS	3.0% construction cost	4,693,381	2	30
MARKETING	2% gross revenue	4,169,333	12	19
	TOTAL COST	204,429,165		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	_	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL				•	-	-			-
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80			•	-	-	-	-	-
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0						
		Taxant Electronic and	45 000	7					

Target Floorspace 45,000

					Basement				
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
RESIDENTIAL									
1 bedroom	30	60	2,070	30	Yes	590,000	17,700,000	19%	15%
2 bedroom	120	80	11,040	120	Yes	700,000	84,000,000	75%	75%
3 bedroom	10	120	1,380	10	Yes	920,000	9,200,000	6%	10.0%
4 bedroom		130	0	0	Yes	0	0		0.0%
TOTAL	160		14,490	160	Yes		110,900,000	100%	100%
Average m2/unit	29								
ADDITIONAL INCLU	SIONS								
Other visitor parking				32					
TOTAL YIELD	160		14,490	192			110,900,000		
COMBINED YIELD	433	T T	44,491	628			319,366,667		
	-100	FSR	14.8				010,000,001		
		Target Floorspace	45,000						

Scenario 4 - S94A Levy 4.5% CBD Strategy Controls + 5:1 FSR

Calculations

REVENUE		
GROSS REVENUE		319,366,667
GST		27,053,788
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	6,387,333
Legals (On Settlement)	\$750 per lot	324,750
TOTAL - SELLING COSTS		6,712,083
NET REVENUE		285,600,795



STAMP DUTY						
LAND VALUE	THRESHOLD	ΤΑΧ				
Minimum	Maximum					
14,000		0				
14,000	30,000	0				
30,000	80,000	0				
80,000	300,000	0				
300,000	1,000,000	0				
1,000,000		FALSE				
3,000,000	Premium Property Tax	1,464,740				
	TOTAL	1,464,740				

CONTRIBUTIONS					
SECTION 94A	L	4.5%			
COSTS					
Retail	10,416,667	\$468,750			
Commercial	7,500,000	\$337,500			
Residential	138,529,350	\$6,233,821			
TOTAL VALUE 156,446,017					
TOTAL CONT	\$7,040,071				

DA FEES							
CONSTRUCTION C	CONSTRUCTION COST THRESHOLDS						
50,001	250,000	0					
250,001	500,000	0					
500,001	1,000,000	0					
1,000,001	10,000,000	0					
More than \$10,000,000		193,162					
LSL	0.35%	547,561					
TOTAL DA FEES		740,723					

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	287,459
TOTAL CC FEES	287,459

CONSTRUCTION COSTS						
Build Costs	Rates	FSR +15%	UG Parking	Grade		
	\$/m2		50,000	Parking		
Apartments	2,850	112,549,350	21,650,000	0		
Retail	2,500	6,250,000	4,166,667			
Commercial	2,500	6,250,000	1,250,000			
Visitor Parking			4,330,000	0		
TOTALS		125,049,350	31,396,667	0		
	156,446,017					

CAR PARKING				
Unit Type	No. Spaces			
1 bed/studio	1.0			
2 bed	1.0			
3+ bed	1.0			
Shops	30			
Commercial	100			
Visitor parking	In any case, min 1 visitor space is required			

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee 1,321	Per Dwg fee 132	
TOTAL LPI F	58,347		

Sheet 3 of 4

Scenario 4 - S94A Levy 4.5% CBD Strategy Controls with Planning Proposal +5:1 FSR

Assumed Development Cashflow

DEVELOPMENT SUM	DEVELOPMENT SUMMARY			
IRR Before Interest	29.30%			
Surplus	\$64,633,295			
% Surplus on D.C	29.25%			
Max Loan Balance Debt/Equity	-\$219,359,265			
Debt/Equity	100.00%			

Budget/ Oct-15 Actual	Month	Land	Professional	DA Fees	CC Fees	Construction	Section 94	LPI Fess	Statuory	0	Manufaction of			0				Manuality				-
			Fees			Costs	Costs	LFIF655	Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget Oct-15	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget Nov-15	1	2,178	202	0	0	0	0	0	0	0	0	\$2,379	0	\$0	\$0	\$0	\$0	\$2,379	-\$2,379	\$0	-\$2,379	1
Budget Dec-15	2	21,161	202	0	0	0	0	0	0	156	0	\$21,520	0	\$0	\$0	\$0	\$0	\$21,520	-\$21,520	-\$12	-\$23,911	2
Budget Jan-16	3	0	202	741	0	0	0	0	0	156	0	\$1,099	0	\$0	\$0	\$0	\$0	\$1,099	-\$1,099	-\$125	-\$25,135	3
Budget Feb-16	4	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$131	-\$25,624	4
Budget Mar-16	5	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$133	-\$26,116	5
Budget Apr-16	6	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$136	-\$26,610	6
Budget May-16	7	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$139	-\$27,107	7
Budget Jun-16	8	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$141	-\$27,607	8
Budget Jul-16	9	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$144	-\$28,109	9
Budget Aug-16	10	0	202	0	0	0	0	0	466	156	0	\$824	0	\$0	\$0	\$0	\$0	\$824	-\$824	-\$146	-\$29,079	10
Budget Sep-16	11	0	202	0	0	0	0	0	0	156	0	\$358	0	\$0	\$0	\$0	\$0	\$358	-\$358	-\$151	-\$29,589	11
Budget Oct-16	12	0	202	0	287	7,822	0	0	0	156	219	\$8,688	0	\$0	\$0	\$0	\$0	\$8,688	-\$8,688	-\$154	-\$38,431	12
Budget Nov-16	13	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$200	-\$47,031	13
Budget Dec-16	14	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$245	-\$55,676	14
Budget Jan-17	15	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$290	-\$64,366	15
Budget Feb-17	16	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$335	-\$73,101	16
Budget Mar-17	17	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$381	-\$81,882	17
Budget Apr-17	18	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$426	-\$90,708	18
Budget May-17	19	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$472	-\$99,581	19
Budget Jun-17	20	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$519	-\$108,500	20
Budget Jul-17	21	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$565	-\$117,465	21
Budget Aug-17	22	0	202	0	0	7,822	0	0	466	156	219	\$8,866	0	\$0	\$0	\$0	\$0	\$8,866	-\$8,866	-\$612	-\$126,942	22
Budget Sep-17	23	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$661	-\$136,004	23
Budget Oct-17	24	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$708	-\$145,112	24
Budget Nov-17	25	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$756	-\$154,268	25
Budget Dec-17	26	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$803	-\$163,471	26
Budget Jan-18	27	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$851	-\$172,723	27
Budget Feb-18	28	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$900	-\$182,022	28
Budget Mar-18	29	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$948	-\$191,371	29
Budget Apr-18	30	0	202	0	0	7,822	0	0	0	156	219	\$8,400	0	\$0	\$0	\$0	\$0	\$8,400	-\$8,400	-\$997	-\$200,767	30
Budget May-18	31	0	202	0	0	7,822	7,040	58	0	156	0	\$15,279	0	\$0	\$0	\$0	\$0	\$15,279	-\$15,279	-\$1,046	-\$217,092	31
Budget Jun-18	32	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$1,131	-\$218,223	32
Budget Jul-18	33	0	0	U	0	U	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$1,137	-\$219,359	33
Budget Aug-18	34	0	0	0	0	0	0	0	466	0	0	\$466	70	\$319,367	\$27,054	\$6,712	\$285,601	\$466	\$285,135	-\$1,142	\$64,633	34
	1	23,339	\$6,258	\$741	\$287	\$156,446	\$7,040	\$58	\$1,397	\$4,693	\$4,169	\$204,429	70	\$319,367	\$27,054	\$6,712	\$285,601	\$204,429	\$81,172	-\$16,538	\$64,633	
		Land	Professional	DA Fees	CC Fees	Construction		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
		1	Fees			Costs	Costs		Costs		5	Costs		Revenue		-	Revenue	Outlays		Interest		

planning

Sheet 4 of 4

Scenario 5 - S94 Capped at \$20,000/Unit CBD Strategy Contr

Summary Sheet

SITE DETAILS				
Address	Representative			
Floor Space Ratio	10.0			
Land Area	3,000			
Site Value	79,762			
Number of Dwellings	273			

gln planning

REVENUE		
	Average/	Development
	Unit (\$)	Development (\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$697,985	\$208,466,667
GST	62,168	16,971,970
Less Selling Costs	16,022	4,374,083
NET REVENUE	\$685,423	\$187,120,614
COSTS		
Land (including acquisition costs)	79,762	21,775,000
Acquisition costs	5,728	1,563,778
Construction	386,628	105,549,517
Consultants	15,465	4,221,981
Section 94 - Commercial	0	0
Section 94 - Residential	20,000	5,460,000
Statutory Fees & Contributions	7,807	2,131,430
On Costs	11,599	3,166,486
Marketing	15,272	4,169,333
Cost before Interest	542,262	148,037,523
Finance (incl Loan Est Fees)	46,419	12,672,294
TOTAL DEVELOPMENT COSTS	588,681	\$160,709,818
TOTAL PROJECT SURPLUS	16.4%	\$26,410,796
PROJECT IRR BEFORE INTEREST	18.9%	

Scenario 5 - S94 Capped at \$20,000/Unit CBD Strategy Controls

Input Sheet

SITE DETAILS					
Address	Repr	Representative			
Floor Space Ratio	10.0	:1			
Achievable FSR	30,000	m2			
Land Area	3,000	m2			
Site Value	79,762				
Any 1 parcel over \$3M	Yes				
LVR (Debt/Equity)	100%				
Date to start feasibility	Oct-15				



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		21,775,000		
Loan Establishment Costs		22,825		
Stamp Duty		1,464,740		
Legals 0.4%		76,213		
LAND ACQUISITION COSTS - Deposit		2,177,500	1	1
LAND ACQUISITION COSTS - Settlement	21,161,278	2	1	
CONSTRUCTION COSTS 2,000		105,549,517	12	20
PROFESSIONAL FEES 4%	const costs	4,221,981	1	31
APPLICATION FEES				
DA Fees		502,018	3	1
CC Fees		194,920	12	1
Section 94		5,460,000	31	1
LPI Fees		37,275	31	1
LAND TAX/RATES				
Land Tax/Rates Year 1		465,739	10	1
Land Tax/Rates Year 2		465,739	22	1
Land Tax/Rates Year 3		465,739	34	1
ONCOSTS 3.0	% construction cost	3,166,486	2	30
MARKETING 2	% gross revenue	4,169,333	12	19
	TOTAL COST	148,037,523		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80								-
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
	-	FSR	10.0			-			
		Target Floorspace	30,000						

Sheet 2 of 4

Scenario 5 - S94 Capped at \$20,000/Unit CBD Strategy Controls Calculations

REVENUE		
GROSS REVENUE		208,466,667
GST		16,971,970
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	4,169,333
Legals (On Settlement)	\$750 per lot	204,750
TOTAL - SELLING COSTS		4,374,083
NET REVENUE		187,120,614



STAMP DUTY							
LAND VALUE	LAND VALUE THRESHOLD TAX						
Minimum	Maximum						
14,000		0					
14,000	30,000	0					
30,000	80,000	0					
80,000	300,000	0					
300,000	1,000,000	0					
1,000,000		FALSE					
3,000,000	1,464,740						
	TOTAL	1,464,740					

CONTRIBUTIONS					
SECTION 94 (\$20K Cap/Unit) \$20,000					
Units					
Retail	0	\$0			
Commercial	0	\$0			
Residential	273	\$5,460,000			
TOTAL UNITS 273					
TOTAL CONTRIBUTION \$5,460,000					

DA FEES							
CONSTRUCTION COST THRESHOLDS							
50,001	250,000	0					
250,001	500,000	0					
500,001	1,000,000	0					
1,000,001	10,000,000	0					
More than \$10,000,000		132,595					
LSL	0.35%	369,423					
TOTAL DA FEES		502,018					

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	194,920
TOTAL CC FEES	194,920

CONSTRUCTION COSTS										
Build Costs	Rates \$/m2	FSR +15%	UG Parking 50,000	Grade Parking						
Apartments	2,850	71,252,850	13,650,000	0						
Retail	2,500	6,250,000	4,166,667							
Commercial	2,500	6,250,000	1,250,000							
Visitor Parking			2,730,000	0						
TOTALS		83,752,850	21,796,667	0						
	105,549,517									

CAR PARKING									
Unit Type	No. Spaces								
1 bed/studio	1.0								
2 bed	1.0								
3+ bed	1.0								
Shops	30								
Commercial	100								
Visitor parking	In any case, min 1 visitor space is required								

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	37,275		

Sheet 3 of 4

Scenario 5 - S94 Capped at \$20,000/Unit CBD Strategy Controls

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	18.92%
Surplus	\$26,410,796
% Surplus on D.C	16.43%
Max Loan Balance Debt/Equity	-\$159,413,799
Debt/Equity	100.00%

IRR Befo Surplus % Surplu Max Loar Debt/Equ	is on D. n Balan	.C	t/Equity	18.92% \$26,410,796 16.43% -\$159,413,799 100.00%	Numbers be	ellow express	sed in nearest	\$,000													5	planr	
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget Budget Budget Budget	Oct-15 Nov-15 Dec-15 Jan-16 Feb-16	0 1 2 3 4	0 2,178 21,161 0 0	0 136 136 136 136	0 0 502 0	0 0 0 0	0 0 0 0	0 0 0 0	0 0 0 0	0 0 0 0	0 0 106 106 106	0 0 0 0	\$0 \$2,314 \$21,403 \$744 \$242	0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$2,314 \$21,403 \$744 \$242	\$0 -\$2,314 -\$21,403 -\$744 -\$242	\$0 \$0 -\$12 -\$124 -\$128	\$0 -\$2,314 -\$23,729 -\$24,596 -\$24,966	0 1 2 3 4
Budget Budget Budget Budget Budget Budget	Mar-16 Apr-16 May-16 Jun-16 Jul-16 Aug-16	5 6 7 8 9 10	0 0 0 0 0	136 136 136 136 136 136 136	0 0 0 0 0	0 0 0 0 0	0 0 0 0 0	0 0 0 0 0	0 0 0 0 0	0 0 0 0 466	106 106 106 106 106 106	0 0 0 0 0	\$242 \$242 \$242 \$242 \$242 \$242 \$707	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$242 \$242 \$242 \$242 \$242 \$242 \$707	-\$242 -\$242 -\$242 -\$242 -\$242 -\$242 -\$707	-\$130 -\$132 -\$134 -\$136 -\$138 -\$140	-\$25,338 -\$25,711 -\$26,087 -\$26,465 -\$26,844 -\$27,692	5 6 7 8 9 10
Budget Budget Budget Budget Budget Budget	Sep-16 Oct-16 Nov-16 Dec-16 Jan-17 Feb-17	11 12 13 14 15 16	0 0 0 0 0	136 136 136 136 136 136	0 0 0 0 0	0 195 0 0 0 0	0 5,277 5,277 5,277 5,277 5,277 5,277	0 0 0 0 0	0 0 0 0 0	0 0 0 0 0	106 106 106 106 106 106	0 219 219 219 219 219 219	\$242 \$5,934 \$5,739 \$5,739 \$5,739 \$5,739	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$242 \$5,934 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739	-\$242 -\$5,934 -\$5,739 -\$5,739 -\$5,739 -\$5,739	-\$144 -\$146 -\$178 -\$209 -\$240 -\$271	-\$28,078 -\$34,157 -\$40,074 -\$46,021 -\$52,000 -\$58,009	11 12 13 14 15 16
Budget Budget Budget Budget Budget Budget	Mar-17 Apr-17 May-17 Jun-17 Jul-17 Aug-17	17 18 19 20 21 22		136 136 136 136 136 136	0 0 0 0	0 0 0 0 0	5,277 5,277 5,277 5,277 5,277 5,277 5,277	0 0 0 0	0 0 0 0	0 0 0 0 466	106 106 106 106 106 106	219 219 219 219 219 219 219	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$6,204	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$6,204	-\$5,739 -\$5,739 -\$5,739 -\$5,739 -\$5,739 -\$6,204	-\$302 -\$334 -\$365 -\$397 -\$429 -\$461	-\$64,050 -\$70,122 -\$76,226 -\$82,362 -\$88,529 -\$95,195	17 18 19 20 21 22
Budget Budget Budget Budget Budget	Sep-17 Oct-17 Nov-17 Dec-17 Jan-18	23 24 25 26 27	0 0 0 0 0	136 136 136 136 136	0 0 0 0	0 0 0 0	5,277 5,277 5,277 5,277 5,277 5,277	0 0 0 0 0	0 0 0 0	0 0 0 0	106 106 106 106 106	219 219 219 219 219 219	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$5,739 \$5,739 \$5,739	-\$5,739 -\$5,739 -\$5,739 -\$5,739 -\$5,739	-\$496 -\$528 -\$561 -\$594 -\$627	-\$101,429 -\$107,696 -\$113,996 -\$120,328 -\$126,694	23 24 25 26 27
Budget Budget Budget Budget Budget Budget	Feb-18 Mar-18 Apr-18 May-18 Jun-18 Jul-18	28 29 30 31 32 33	0 0 0 0 0	136 136 136 136 0 0	0 0 0 0 0	0 0 0 0 0	5,277 5,277 5,277 5,277 5,277 0 0	0 0 5,460 0 0	0 0 37 0	0 0 0 0	106 106 106 106 0 0	219 219 219 0 0 0	\$5,739 \$5,739 \$11,016 \$0 \$0	0 0 0 0 0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$0 \$0 \$0 \$0 \$0 \$0	\$5,739 \$5,739 \$5,739 \$11,016 \$0 \$0	-\$5,739 -\$5,739 -\$5,739 -\$11,016 \$0 \$0	-\$660 -\$693 -\$727 -\$760 -\$822 -\$826	-\$133,092 -\$139,524 -\$145,989 -\$157,766 -\$158,588 -\$159,414	28 29 30 31 32 33
Budget	Aug-18	34	0 23,339 Land	0 \$4,222 Professional	0 \$502 DA Fees	0 \$195 CC Fees	0 \$105,550 Construction	0 \$5,460 Section 94	0 \$37 LPI Fess	466 \$1,397 Statuory	0 \$3,166 Oncosts	0 \$4,169 Marketing	\$466 \$466 \$148,038 Total	70 70	\$208,467 \$208,467 Gross	\$16,972 \$16,972 GST	\$4,374 \$4,374 \$4,374 Selling Costs	\$187,121 \$187,121 \$187,121 Net	\$466 \$148,038 Monthly	\$186,655 \$39,083 Net	-\$830 -\$12,672 Monthly	\$26,411 \$26,411	34

Sheet 4 of 4

Scenario 6 - S94 Cap at \$20k/Unit CBD Strategy Controls +

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	15.0
Land Area	3,000
Site Value	79,762
Number of Dwellings	433

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	62,480	27,053,788
Less Selling Costs	15,501	6,712,083
NET REVENUE	\$659,586	\$285,600,795
COSTS		
Land (including acquisition costs)	50,289	21,775,000
Acquisition costs	3,611	1,563,778
Construction	361,307	156,446,017
Consultants	14,452	6,257,841
Section 94 - Commercial	0	0
Section 94 - Residential	20,000	8,660,000
Statutory Fees & Contributions	5,736	2,483,745
On Costs	10,839	4,693,381
Marketing	9,629	4,169,333
Cost before Interest	475,864	206,049,094
Finance (incl Loan Est Fees)	38,254	16,563,779
TOTAL DEVELOPMENT COSTS	514,117	\$222,612,873
TOTAL PROJECT SURPLUS	28.3%	\$62,987,922
PROJECT IRR BEFORE INTEREST	28.7%	

Sheet 1 of 4

Scenario 6 - S94 Cap at \$20k/Unit CBD Strategy Controls + 5:1 FSR Input Sheet

SITE DETAILS									
Address	Repr	resentative							
Floor Space Ratio	15.0	:1							
Achievable FSR	45,000	m2							
Land Area	3,000	m2							
Site Value	79,762								
Any 1 parcel over \$3M	Yes								
LVR (Debt/Equity)	100%								
Date to start feasibility	Oct-15								



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		21,775,000		
Loan Establishment Costs		22,825		
Stamp Duty		1,464,740		
Legals	0.4%	76,213		
LAND ACQUISITION COST	S - Deposit	2,177,500	1	1
LAND ACQUISITION COST	S - Settlement	21,161,278	2	1
CONSTRUCTION COSTS	2,000	156,446,017	12	20
PROFESSIONAL FEES	4% const costs	6,257,841	1	31
APPLICATION FEES				
DA Fees		740,723	3	1
CC Fees		287,459	12	1
Section 94		8,660,000	31	1
LPI Fees		58,347	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	465,739	10	1
Land Tax/Rates	Year 2	465,739	22	1
Land Tax/Rates	Year 3	465,739	34	1
ONCOSTS	3.0% construction cost	4,693,381	2	30
MARKETING	2% gross revenue	4,169,333	12	19
	TOTAL COST	206,049,094		

UNIT MIX AND SALE Basement Unit Type **Car Parking** Price Total No. Dwelling Total Parking Actual Preferred FSR+15% Floor Space m2 Yes (\$) Mix Mix (\$) NON RESIDENTIAL 2,500 2,500 10,416,667 10,416,667 Retail 83 Yes Commercial 2,500 2,500 25 Yes 7,500,000 7,500,000 RESIDENTIAL 1 bedroom 45 60 3,105 45 Yes 590,000 26,550,000 16% 15% 2 bedroom 208 80 19,136 208 Yes 700,000 145,600,000 76% 75% 3 bedroom 20 120 2,760 20 Yes 920,000 18,400,000 7% 10.0% 4 bedroom 0 130 0 Yes 0.0% 0 0 0 TOTAL 30,001 381 208,466,667 100% 273 Yes 100% Average m2/unit 80 ADDITIONAL INCLUSIONS Other visitor parking 55 208,466,667 TOTAL YIELD 273 30,001 436 FSR 10.0

Target Floorspace 45,000

					Basement				
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
RESIDENTIAL	-								
1 bedroom	30	60	2,070	30	Yes	590,000	17,700,000	19%	15%
2 bedroom	120	80	11,040	120	Yes	700,000	84,000,000	75%	75%
3 bedroom	10	120	1,380	10	Yes	920,000	9,200,000	6%	10.0%
4 bedroom		130	0	0	Yes	0	0		0.0%
TOTAL	160		14,490	160	Yes		110,900,000	100%	100%
Average m2/unit	29								
ADDITIONAL INCLUS	IONS								
Other visitor parking				32					
TOTAL YIELD	160		14,490	192			110,900,000		
COMBINED YIELD	433		44,491	628			319,366,667		
		FSR	14.8						•
		Target Floorspace	45,000						

Scenario 6 - S94 Cap at \$20k/Unit CBD Strategy Controls + 5:1 FSF Calculations

REVENUE		
GROSS REVENUE		319,366,667
GST		27,053,788
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	6,387,333
Legals (On Settlement)	\$750 per lot	324,750
TOTAL - SELLING COSTS		6,712,083
NET REVENUE		285,600,795



STAMP DUTY			
LAND VALUE	LAND VALUE THRESHOLD TAX		
Minimum	Maximum		
14,000		0	
14,000	30,000	0	
30,000	80,000	0	
80,000	300,000	0	
300,000	1,000,000	0	
1,000,000		FALSE	
3,000,000	Premium Property Tax	1,464,740	
	TOTAL	1,464,740	

CONTRIBUTIONS			
SECTION 94 (\$20K Cap/Unit) \$20,000			
Units			
Retail	0	\$0	
Commercial	0	\$0	
Residential 433 \$8,660,000			
TOTAL UNITS 433			
TOTAL CONTRIBUTION \$8,660,00			

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		193,162
LSL	0.35%	547,561
TOTAL DA FEES		740,723

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	287,459
TOTAL CC FEES	287,459

CONSTRUCTION COSTS				
Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	112,549,350	21,650,000	0
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			4,330,000	0
TOTALS		125,049,350	31,396,667	0
	TOTAL CONSTRUCTION COST 156,446,017			156,446,017

CAR PARKING		
Unit Type	No. Spaces	
1 bed/studio	1.0	
2 bed	1.0	
3+ bed	1.0	
Shops	30	
Commercial	100	
Visitor parking	In any case, min 1 visitor space is required	

RATES	
Council Rates	47,615
Land Tax	418,124
TOTAL RATES	465,739

LPI FEES			
Strata	Base Fee 1,321	Per Dwg fee 132	
TOTAL LPI F	EES		58,347

Sheet 3 of 4

Scenario 6 - S94 Cap at \$20k/Unit CBD Strategy Controls + 5:1 FSR

Assumed Development Cashflow

DEVELOPMENT SUM	DEVELOPMENT SUMMARY	
IRR Before Interest	28.73%	
Surplus	\$62,987,922	
% Surplus on D.C	28.29%	
Max Loan Balance Debt/Equity	-\$220,996,113	
Debt/Equity	100.00%	

planning Oct-15 Total Gross ction 9 Net Funds Cashflow Costs Costs Outlays Cashflow Interest Actual Fees Costs Costs Revenue Revenue 6.25% Oct-15 0 0 0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 Budget 0 0 0 0 0 0 0 0 Budget Nov-15 2,178 202 0 \$2,379 \$0 \$0 \$0 \$0 \$2,379 -\$2,379 \$0 -\$2,379 0 0 0 0 0 0 0 0 Budget Dec-15 2 21,161 202 0 0 156 \$21,520 \$0 \$0 \$0 \$0 \$21,520 -\$21,520 -\$12 -\$23,911 0 0 0 0 0 2 202 741 156 \$1,099 \$0 \$0 \$0 \$0 \$1,099 -\$1,099 -\$125 -\$25,135 Budget Jan-16 3 0 0 0 0 0 0 0 3 0 Budget Feb-16 4 202 0 0 0 0 0 0 156 0 \$358 0 \$0 \$0 \$0 \$0 \$358 -\$358 -\$131 -\$25.624 4 0 Budget Mar-16 202 156 \$358 \$0 \$0 \$0 \$0 \$358 -\$358 -\$133 -\$26,116 5 0 0 0 0 0 0 0 0 0 5 Apr-16 202 156 \$358 \$0 \$0 \$0 \$0 \$358 -\$358 -\$136 -\$26,610 Budget 6 0 0 0 0 0 0 0 0 0 6 \$0 \$0 Budget May-16 202 0 0 156 \$358 \$0 \$0 \$358 -\$358 -\$139 -\$27,107 7 0 0 0 0 0 0 0 7 \$0 \$0 156 \$358 \$0 \$0 -\$358 -\$27,607 Jun-16 202 0 \$0 \$358 -\$141 Budget 8 0 0 0 0 0 0 0 0 8 156 \$0 \$0 Budget Jul-16 9 0 202 0 0 0 0 0 0 0 \$358 0 \$0 \$358 -\$358 -\$144 -\$28 109 q Budget Aug-16 10 0 202 0 0 0 0 0 466 156 0 \$824 0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$824 -\$824 -\$146 -\$29.079 10 Budget Sep-16 11 0 202 0 0 0 0 0 0 156 0 \$358 0 \$0 \$358 -\$358 -\$151 -\$29,589 11 Budget Oct-16 12 202 0 287 7,822 0 0 156 219 \$8,688 \$0 \$0 \$0 \$0 \$8,688 -\$8,688 -\$154 -\$38,431 12 0 0 0 Budget Nov-16 13 202 0 7,822 156 219 \$8,400 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$200 -\$47,031 13 0 0 0 Budget Dec-16 14 202 7,822 0 156 219 \$8,400 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$245 -\$55,676 14 0 0 0 0 Budget Jan-17 15 0 202 0 0 7.822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$290 -\$64,366 15 Feb-17 202 7,822 156 \$8,400 \$0 \$0 \$0 \$0 -\$8,400 -\$335 -\$73,101 16 Budget 16 0 0 0 0 0 0 219 0 \$8,400 \$0 \$0 \$0 Budget Mar-17 17 202 0 0 7,822 0 0 156 219 \$8,400 0 \$0 \$8,400 -\$8,400 -\$381 -\$81,882 17 0 0 Budget Apr-17 18 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$426 -\$90,708 18 0 May-17 156 \$0 \$0 \$0 19 202 7.822 219 \$8,400 \$0 \$8,400 -\$8,400 -\$472 -\$99.581 19 Budget 0 0 0 0 0 0 Budget 7.822 156 \$8,400 \$0 \$0 \$0 \$0 -\$8,400 -\$519 -\$108,500 Jun-17 20 0 202 0 0 0 0 0 219 0 \$8,400 20 \$0 \$0 156 \$0 \$0 -\$117,465 Budget Jul-17 21 0 202 0 0 7.822 0 0 0 219 \$8,400 0 \$0 \$8,400 -\$8,400 -\$565 21 \$0 \$0 Budget Aug-17 22 0 202 0 0 7 822 0 0 466 156 219 \$8,866 0 \$0 \$8 866 -\$8 866 -\$612 -\$126 942 22 \$0 Budget Sep-17 23 202 0 0 7.822 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$661 -\$136.004 23 0 0 Budget Oct-17 24 202 0 0 7.822 0 0 0 156 219 \$8,400 0 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$708 -\$145.112 24 0 Budget Nov-17 25 202 0 7,822 0 0 156 219 \$8,400 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$756 -\$154,268 25 0 0 0 0 Budget Dec-17 26 202 0 0 7,822 0 0 156 219 \$8,400 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$803 -\$163,471 26 Budget Jan-18 27 202 7,822 156 219 \$8,400 \$0 \$0 \$0 \$0 \$8,400 -\$8,400 -\$851 -\$172,723 27 0 0 0 0 0 0 \$0 \$0 \$0 -\$182,022 Budget Feb-18 28 202 0 0 7,822 0 0 0 156 219 \$8,400 0 \$0 \$8,400 -\$8,400 -\$900 28 0 156 \$0 \$0 \$0 \$0 -\$8,400 -\$191,371 Budget Mar-18 29 0 202 0 0 7,822 0 0 0 219 \$8,400 0 \$8,400 -\$948 29 \$0 \$0 Budget Apr-18 30 202 7,822 0 156 219 \$8,400 0 \$0 \$0 \$8,400 -\$8,400 -\$997 -\$200,767 30 0 0 0 0 0 202 7,822 8.660 58 156 \$16,899 \$0 \$0 \$0 \$0 \$16,899 -\$16,899 -\$1.046 -\$218,712 31 Budget May-18 31 0 0 0 0 0 0 \$0 Budget Jun-18 32 0 0 0 0 0 0 0 0 0 0 \$0 0 \$0 \$0 \$0 \$0 \$0 -\$1,139 -\$219.851 32 \$0 \$0 Budget Jul-18 33 0 0 0 0 0 0 0 0 0 0 \$0 0 \$0 \$0 \$0 \$0 -\$1,145 -\$220,996 33 Budget Aug-18 34 0 0 0 0 0 0 0 466 0 0 \$466 70 \$319.367 \$27.054 \$6.712 \$285.601 \$466 \$285,135 -\$1.151 \$62.988 34 23.339 \$6.258 \$741 \$287 \$156,446 \$58 \$1,397 \$4.693 \$4,169 \$206,049 70 \$319,367 \$27,054 \$6,712 \$285.601 \$206,049 \$79,552 -\$16,564 \$62,988 \$8,660 Land Professional DA Fees CC Fees Construction Section 94 LPI Fess Statuory Oncosts Marketing Total Gross GST Selling Costs Net Monthly Net Monthly Fees Costs Costs Costs Costs Revenue Revenue Outlays Interest

Sheet 4 of 4

APPENDIX C

Review of value sharing schemes summary







Southern Employment Lands Affordable Housing

Area of application

Applies to certain land in Alexandria and Rosebery that is zoned B7 under Sydney Local Environmental Plan 2012

How does the scheme work?

The subject areas are zoned B7. No residential development is currently permitted.

A previous employment lands strategy had identified the areas as being potentially suitable for residential development.

Developers of land in these areas may obtain approval for a Planning Proposal that allows residential development (under a B4 zoning) and / or permits a greater intensity of development through increased floor space or height limits.

Where a change to planning controls through a Planning Proposal can be supported on its planning merits, the value uplift (or 'planning gain') will be equally shared between the landowner/developer and the public.

A VPA between the developer and Council that addresses the public benefit offer needs to have been negotiated, agreed, entered into and registered on the land title before the rezoning is made.

The incentive to provide affordable housing is provided at the point of rezoning, because it is at that point that the increased value associated with residential permissibility is absorbed into the sale price of land.

Scheme operation is described in the City's *Guideline to Preparing Site Specific Planning Proposal Requests in the City of Sydney Employment Lands Investigation Areas.*

The incentive scheme is part of the wider Employment Lands Affordable Housing Program administered by the Council.

The scheme differs from the City's Green Square value capture schemes in that the higher floor space allowances are not set out in the LEP.

What infrastructure / public benefits are delivered by the scheme?

Scheme is predicated on the idea that the southern employment lands are first and foremost an employment area. Increased land values caused by gentrification in the area will have an impact on housing affordability for workers in lower paid or 'key worker' jobs in the employment lands.

Planning studies undertaken by the Council identified a need for around 720 affordable housing units in the area.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

Council is acting on this need by providing incentives for affordable housing to be created by way of rezoning of and allowing greater intensity of development.

Contributions for affordable housing may be one or more of the following methods:

- The dedication of land
- The provision of works for public purposes
- The provision of completed affordable housing dwellings
- The paying of cash

How is the value that is to be captured by the Council calculated?

The planning gain is calculated using a series of rates to calculate the difference between value of the land before rezoning or before changes to height and FSR controls, and the value of the land following rezoning.

The rates are the result of land valuations that were undertaken by an independent consultant.

% of non-residential floor space retained in the rezoning request	Contribution rate for a change of zone request (R1) (\$ per m2)
76% -100%	\$0
51% -75%	\$200
26% -50%	\$650
0 -25%	\$1100

The rates for Planning Proposals to permit residential uses are as follows:

Additional floor space type	Contribution rate for additional or bonus floor space requested (R2) (\$ per m2)
Residential	\$2400
Non residential	\$600

Total value of planning gain is calculated by:



```
(% residential split * residential R2 rate * additional residential floor space'}
```

+

(% non-residential split * non-residential R2 rate * additional non-residential floor space)

+

The Council in the VPA requires 50% of the planning gain to be provided as a contribution.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

Other comments on the scheme

The scheme is underpinned by:

- A land use planning framework that hasn't yet given away value-generating development rights
- Research which justifies the need for the infrastructure scheme
- An infrastructure delivery strategy with priorities that will be provided from proceeds of the value capture scheme
- Clear guidelines concerning how a developer can achieve a rezoning, and how the developer contributions are equitably calculated.

The scheme is part of a suite of interrelated initiatives to achieve more affordable housing across the City of Sydney.

The scheme is in addition to provisions in the LEP that require or promote the provision of affordable housing by developers. These other provisions:

- Allow, without a rezoning, mixed use developments in which the residential component is affordable housing. Market-based housing is only allowed through a rezoning.
- Require an affordable housing levy contribution of up to 3% of the floor space proposed in a DA.

The scheme also is in addition to contributions required to be paid by the developer under section 94 of the EP&A Act.

Potential transferability to Parramatta CBD

Potential transferability is high, whether the Council decides to pursue a comprehensive, areawide scheme, or a scheme for site-specific developments.

The Southern Employment Lands Affordable Housing scheme is predicated on incremental Planning Proposals seeking additional development rights.

The infrastructure program for Parramatta CBD is yet to be determined. While the Southern Employment Lands scheme is focused on delivering affordable housing, a similar Parramatta scheme could relate other infrastructure such as public domain, traffic improvements, through site links and lanes.

Apart from the infrastructure program needing to be prepared, a similar Parramatta scheme would also need to address the following:

- Whether the value capture calculation would be based on standard rates
- Guidelines on how a developer can achieve a rezoning and can participate in the value capture scheme.



Green Square Urban Renewal Area Community Infrastructure Scheme

Area of application

Applies to land zoned R1, B2, B4, B6 and B7

How does the scheme work?

This scheme is enabled under clause 6.14 of Sydney Local Environmental Plan 2012.

The LEP allows for a base and a maximum FSR. Additional floor space (called 'community infrastructure floor space') up to the 'maximum' can only be achieved where 'community infrastructure' is provided.

'Community infrastructure' is recreation areas, recreation facilities (indoor), recreation facilities (outdoor), public roads, drainage or flood mitigation works.

The clause states that the additional floor space can only be approved if the development of the land *includes development for the above purposes*.

In practice the Council accepts offers of cash as well as 'development' for community infrastructure.

The Council has published Development Guidelines covering the scheme. Staff advise that this document will soon form a Schedule in the Sydney DCP.

The Guidelines explain the steps involved for an applicant to gain approval for additional floor space. The document also provides a contribution rate for each square metre of additional residential and non-residential floor space that is achieved over the base FSR.

The contribution is secured by a VPA during the DA process.

What infrastructure / public benefits are delivered by the scheme?

The scheme was introduced when the Green Square Urban Renewal Area was rezoned from predominantly industrial purposes in the late 1990s. The scheme has been critical in delivering essential infrastructure to support projected growth in Green Square.

Section 5.2 – Green Square of Sydney DCP 2012 details the type and location of community infrastructure needed to support the redevelopment and growth of the area.

The infrastructure includes new public streets, dedicated setbacks to facilitate future road widening, open space and parks, public transport amenities, pedestrian and bike links, parks, community facilities and stormwater management facilities to sustain the urban renewal project.

The scheme works in conjunction with the section 94 contributions plan applying to the area. That plan is not expected to generate sufficient funds to provide all the community infrastructure required in the urban renewal area.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

Developers are mandatorily levied section 94 contributions and affordable housing contributions in the consent. In addition, developers negotiate provision of community infrastructure contributions under a VPA.

How is the value that is to be captured by the Council calculated?

The contribution required is relative to the scale of the proposed development. The less 'additional' floor space sought by the landowner, the lower the contribution.

The contribution is calculated based on standard rates in the Development guidelines document and the amount of additional floor space above the base allowance, as follows:

	\$ rate / m ² incl. GST
Residential GFA	\$475
Retail GFA	\$275
Other non-residential GFA	\$200

The contribution may be provided 'in-kind' or by way of monetary contribution. Regardless, the first \$100 per square metre must be made as a cash contribution towards Green Square Town Centre infrastructure.

Other comments on the scheme

The Scheme does not allow or incentivise landowners to exceed what has been established as an appropriate built form outcome in consultation with the community, that is, there is no floor space 'bonus' for contribution. Floor space that can be achieved in addition to the 'base' floor space is clearly stated in the LEP.

This avoids any concerns or perceptions in the wider community that floor space is for sale in the planning system.

The Guidelines state that 'proposed development must be acceptable in terms of environmental capacity, compliance with development controls and must have little or no impact on adjoining properties and the surrounding area. The development proposal must be acceptable on a merit assessment before the City can agree to a package of community infrastructure associated with the development'.

Staff advise that the scheme is relatively easy to administer with clear guidelines to support the collection of funds and their direction to a suitable public benefit.

The scheme doesn't rely on a plan making process which can be costly and time consuming.

Potential transferability to Parramatta CBD

The scheme's potential for transferability to Parramatta CBD depends on whether the Council intends to pursue a comprehensive area-wide bonus floor space scheme, or whether it intends to pursue value capture on a site-by-site basis.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

The Green Square schemes all provide a good model for a comprehensive area-wide scheme. The scheme's standard \$/m rate for additional GFA provides the simplest and cost effective methodology to implement a value capture scheme. The standard rate approach allows for a level playing field that the same contribution amount should be factored into future site feasibility investigations.



Green Square Town Centre Infrastructure Strategy

Area of application

Applies to land surrounding Green Square railway station that is zoned 4a General Industrial under the City of Sydney Planning Scheme Ordinance 1971

How does the scheme work?

The following LEPs:

- Sydney Local Environmental Plan (Green Square Town Centre) 2013
- Sydney Local Environmental Plan (Green Square Town Centre—Stage 2) 2013

Both zone land in the centre B4 Mixed Use.

However the land is 'deferred' from a B4 zone and remains zoned Industrial until developers offer to enter into a VPA with the Council to deliver, or make an appropriate contribution toward, town centre infrastructure.

The 'appropriate' contribution is established by the Green Square Town Centre Infrastructure Strategy which identifies the required contribution for each site.

Until land is 'un-deferred', the planning controls for land are Zone 4a General Industrial under the City of Sydney Planning Scheme Ordinance 1971 which does not allow residential development.

The following steps are taken to un-defer land:

- Landowner approaches the council with a request to un-defer their site
- the council negotiates with the land owner on how the contribution is to be delivered (some in-kind, some monetary)
- a VPA is publicly exhibited, executed and registered on the title of the land
- the council requests the Minister 'un-defer' the site
- the maps are changed and once published on the Government's website the undeferral is complete.

What infrastructure / public benefits are delivered by the scheme?

The infrastructure includes:

- community and recreation facilities, such as the Green Square Town Centre Library and the aquatic centre at Epsom Park
- civic spaces, in particular the Green Square Town Centre Plaza
- stormwater management facilities
- public art

Council has been able to plan infrastructure delivery in parallel with development.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

The VPAs that are negotiated usually include a mixture of land, works and cash. Many developers are choosing to provide their contribution 'in-kind' by building roads and footpaths as they develop their own site.

How is the value that is to be captured by the Council calculated?

The contribution was established by valuing the required infrastructure and dividing the costs between sites based on their development potential under the 'deferred' controls.

For example, as site that allowed for 10% of the floor space across the precinct is required to pay 10% of the total value of the precinct's infrastructure. In this way, the scheme works in a similar way to section 94 apportioned contributions. Except that these are additional contributions to section 94.

The respective site contribution amounts are contained in Schedule 5 of the Green Square Town Centre Infrastructure Strategy. The total value of expected development contributions is \$73.2 million which equates to the assessed shortfall of funding for the infrastructure after section 94 contributions have been taken into account. This amount is then broken up based on each site area's allowance of residential, retail and other non-residential floor space, and standardised contribution rates as follows:

	\$ rate / m ² incl. GST
Residential GFA	\$475
Retail GFA	\$275
Other non-residential GFA	\$200

Other comments on the scheme

The value uplift is activated when the site zoning is un-deferred. It is the creation of the residential development opportunity which creates most of the value in the site. The un-deferral process therefore helps in managing land value and speculation.

The infrastructure strategy provides a clear indication of the contribution required for each individual site. This enables developers to 'factor-in' those costs before committing to development and/or purchasing the land.

The approach relies on a thorough master-planning of a precinct. Where there is uncertainty about what sites the planning controls should be for a site (such as zoning, height and FSR) then the appropriate contribution cannot be assigned to a site.

Potential transferability to Parramatta CBD

The scheme's potential for transferability to Parramatta CBD depends on whether the Council intends to pursue a comprehensive area-wide bonus floor space scheme, or whether it intends to pursue value capture on a site-by-site basis.

All the Green Square schemes provide a good model for a comprehensive area-wide scheme. The scheme's standard \$/m rate for additional GFA provides the simplest and cost effective methodology to implement a value capture scheme. The standard rate approach allows for a



APPENDIX - VALUE CAPTURE SCHEMES SUMMARY

level playing field that the same contribution amount should be factored into future site feasibility investigations.



Macquarie Park Corridor Access and Open Space Infrastructure

Area of application

Applies to land in Macquarie Park that is currently zoned B3 and B4 under Ryde Local Environmental Plan 2014.

How does the scheme work?

The Macquarie Park Corridor supports business park and commercial uses. The area however suffers from poor access at peak travel times and limited open space provision for the workers in the area.

These deficiencies are planned to be addressed by a more permeable street and pedestrian network, and through the provision of new open space.

The revised access and open space network is articulated in the Council's DCP.

The LEP allows sites to be developed with increased FSR and heights. For example, standard FSRs applying to land in Macquarie Park range between 1:1 and 3:1; and the LEP allows further FSR of between 0.5:1 and 2:1 depending on location.

Approval for increased FSR and / or height can only be granted where the consent authority is satisfied that:

- (a) there will be adequate provision for recreation areas and an access network, and
- (b) the configuration and location of the recreation areas will be appropriate for the recreational purposes of the precinct, and
- (c) the configuration and location of the access network will allow a suitable level of connectivity within the precinct.¹

Developers can satisfy these requirements by entering into a VPA with the Council to provide recreation or access infrastructure in accordance with the DCP, or make a cash contribution in lieu of provision.

What infrastructure / public benefits are delivered by the scheme?

The scheme provides for:

- The construction of new roads identified in the DCP
- The purchase of key sites to develop roads identified in the DCP
- The embellishment of open identified in the DCP
- The purchase of key sites for use as open space as identified in the DCP
- The embellishment of through-site links identified in the DCP

¹ Ryde Local Environmental Plan 2014, clause 6.9



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

- Additional new roads, through site links and open space in Macquarie Park at Councils discretion
- Administration of this scheme.

The scheme is designed to achieve the following:

- 4.1 km of 20m wide roads
- 3.6 km of 14m wide roads
- 3.4 km of 8m wide pedestrian links
- 111 m of road widening; and
- 3.5 hectares of open space.

The total cost of the required infrastructure was estimated at \$107 million in 2013. The funding strategy assumes that 85% of this amount will come from the increased floor space scheme.

How is the value that is to be captured by the Council calculated?

Monetary Contribution = total additional gross floor area x contribution rate

'Additional gross floor area' is any of the increased floor space permitted under the LEP.

The contribution rate is \$250 per square metre. This represents approximately 50% of the increased value projected to accrue to each square metre of extra floor space permitted under the LEP.

The total contribution amount will be reduced where the developer offers and the Council accepts land or works that are identified in the infrastructure scheme.

The assumed values used in the calculation of offsets are as follows:

- Land to be dedicated to Council for the purpose of road or open space \$250 per square metre.
- Embellishment of road, finished to Council's specifications \$380 per square metre.
- Embellishment of open space, finished to Council's satisfaction \$400 per square metre.

Other comments on the scheme

The scheme has the following desirable elements common to other schemes we have reviewed:

- A land use planning framework that hasn't yet given away value-generating development rights.
- An infrastructure delivery strategy including priorities that will be provided from the proceeds of the scheme.
- Guidelines for how developers participate in the scheme.
- There is built-in equity in that all landowners receiving the benefit of increased height and FSR will contribute whether there is infrastructure on their land or not.
- Both the infrastructure and the contributors are spread over a broad district, which emphasises the planning value of the scheme. This is distinct from a site by site approach.



APPENDIX – VALUE CAPTURE SCHEMES SUMMARY

The scheme is in addition to contributions required from the land developers under section 94 of the EP&A Act. These contributions go towards other infrastructure in the Ryde LGA which have a nexus with the Macquarie Park development. There is no double-dipping between the schemes.

The alternative of providing the DCP's recreation and access infrastructure solely through a conventional section 94 / 94A contributions scheme was rejected because:

- Much of the new development would be replacing existing floor space. Section 94 contributions would be limited because they can only be applied to the net increase in floor space.
- If all the net additional floor space was required to meet all of the cost of the infrastructure, then the contributions would be so high that it would make Macquarie Park unviable as a place to invest.

Section 94A levies would need to be 8% of the projected development costs in order to afford the infrastructure. This also would send investment elsewhere.

Potential transferability to Parramatta CBD

The scheme's potential for transferability to Parramatta CBD depends on whether the Council intends to pursue a comprehensive area-wide bonus floor space scheme, or whether it intends to pursue value capture on a site-by-site basis.

The Macquarie Park scheme provides a good model for a comprehensive area-wide scheme. The scheme's standard \$/m rate for additional GFA provides the simplest and cost effective methodology to implement a value capture scheme. The standard rate approach allows for a level playing field that the same contribution amount should be factored into future site feasibility investigations.

The Macquarie Park scheme also contains a set of principles that would be relevant to any local value capture scheme any council would implement:

Nexus: That some of the benefit afforded to sites is captured by the community to provide essential infrastructure required as a result of increased densities in the area.

Transparency: There is a clear understanding of what infrastructure is to be funded and how contribution rates and community benefit are calculated and applied to individual sites.

Equity: Both infrastructure and incentives for development are based on equity and fairness.

Practical: The implementation of the mechanism must be practical and occur in a timely fashion to avoid delays and provide certainty for commercial dealings.

Feasibility: The contributions must not create development opportunities which are not economically viable.



Waverley Variation Floor Space Infrastructure Scheme

Area of application

Applies to:

- land zoned B3, B4 and R4 in Bondi Junction
- certain land zoned B4, B1 and R3 in Bondi Beach

under Waverley Local Environmental Plan 2012

How does the scheme work?

The scheme applies to areas likely to experience substantial multi-storey redevelopment.

The scheme is described in the Waverley Council Planning Agreement Policy 2014.

DAs in these areas that propose up to 15% extra floor space above that permitted under the FSR controls in Waverley Local Environmental Plan may be approved if:

- The development is acceptable on planning grounds, and
- The developer enters into a VPA with the Council to provide public benefits (cash, works or land) in the surrounding area.

Significant variations (above 15%) are not encouraged but could be achieved through a Planning Proposal.

What infrastructure / public benefits are delivered by the scheme?

The infrastructure that is to be provided using the proceeds of value capture is addressed in Council's 'Complete Streets' program. Also, 10% of the contributions will be in the form of a monetary contribution for the purpose of providing affordable housing units managed by a community housing provider.

Complete Streets is a comprehensive plan to enhance Bondi Junction through greening, upgrading footpaths and public places and improving connections for cycling, walking and access to public transport.

More information on the Complete Streets program can be found at <u>http://www.waverley.nsw.gov.au/building/current projects/bondi junction projects/complete str</u> <u>eets</u>

Improving the street environment for residents and visitors is important because of the extra demands on the street environment generated by the additional development.

How is the value that is to be captured by the Council calculated?

The contribution to be negotiated will be 50% of the increase in net value to the development arising from an increase in FSR beyond that allowed under clause 4.4 of the LEP.

The formula for calculating the value uplift from the bonus floor space is:





Marginal net sale proceeds

less

Marginal cost to construct

'Marginal net sale proceeds' in general terms is the difference between the total value of the apartments without the bonus floor space and the total value of the apartments with the bonus floor space. This must be calculated by 2 registered valuers – the average of the 2 values will be used in the calculation.

A recommended method to calculate the 'Marginal cost to construct' is the pro-rata of the total building cost based on bonus floor space divided by total GFA plus a pro-rata of the car parking cost based on number of parking spaces allocated to the bonus units divided by total car parking spaces. Land costs and site costs such as landscaping and driveways are excluded (as these are needed regardless of the bonus floor space. GST is also excluded. Section 94A and interest costs are included.

Other comments on the scheme

The policy was adopted in late 2014, and has not yet been tested with an actual DA.

The scheme had its genesis with a DCP-based affordable housing incentive scheme that operated for over a decade until FSRs were transferred from the DCP to the LEP. That scheme set up the local practices of:

- linking increases in FSR controls to contributions for public amenities (i.e. affordable housing), and
- a contribution amount that reflected 50% of the value uplift generated by the FSR variation.

Council staff advised that the scheme was well supported by developers, and that the scheme achieved around 30 affordable housing units. The unit are managed by community housing providers.

The 15% benchmark for maximum additional floor space emanates from the average floor space variations approved under the DCP affordable housing program.

Potential transferability to Parramatta CBD

Potential transferability is high.

The scheme is predicated on incremental DAs seeking additional development rights. Incremental sites value capture is favoured by Parramatta City Council.

There is however a concern that, because the VPA is tied to a DA then it may unravel because of the applicant's appeal rights.

A developer could, for example, lodge and have approved a DA on a site that has 15% extra floor space and an accompanying value capture VPA. Soon after that, the developer or anyone else could lodge another DA that was more or less identical except without a VPA offer. Council



APPENDIX - VALUE CAPTURE SCHEMES SUMMARY

would likely refuse the DA without the VPA but the developer could argue in an appeal that the clause 4.6 variation was acceptable regardless of the VPA. In other words the development was acceptable on planning rounds, and the VPA was not crucial to the DA being approved.

Such a scenario is less likely if the VPA offer is secured at the Planning Proposal, provided the rezoning which grants the extra floor space rights is not made until the VPA is signed and entered onto the land title.

APPENDIX D

Value sharing development feasibility analysis and assumptions



1.0 Introduction

GLN was previously instructed to model the impacts of development feasibility that would likely be experienced under different contribution regimes that could be applied under Section 94 and Section 94A of the EP& Act (See **Appendix B**). Following review of the different contribution regimes, as an alternative method to fund part of the significant infrastructure works required in the Parramatta CBD, Council identified a series of different scenarios for the application of a Value Sharing scheme to the development of B4 Mixed Use land within the Parramatta CBD. Council essentially identified two opportunities for Value Sharing including:

- Phase 1 Imposition of Value Sharing where a developer proposes to exceed the existing controls (ie. 6:1 FSR GFA) to achieve the new floorspace ratio (ie. 10:1) in the draft Planning Proposal. This scale of development would also benefit from an additional 2:1 FSR as a result of incentive clauses however Council does not intend to apply Value Sharing to this GFA.
- Phase 2 Imposition of Value Sharing where a developer proposes to exceed the draft Planning Proposal controls (including incentive GFA) up to a maximum of 15:1 for 'Opportunity Sites'.

In response to these proposed approaches, the following scenarios were modelled:

- Base case 1 6.9:1 FSR Development of the hypothetical site to achieve a development outcome achieving a 6.9:1 FSR GFA outcome with S94A contributions levied at 3% of the cost of construction. This GFA outcome represent the 6:1 FSR entitlement shown in Council's draft Planning Proposal. The land acquisition value determined in this scenario was then used for Scenarios A, B, C and D as a constant to reflect the situation where developers have purchased land and new contribution regimes have been put in place.
- Scenario A Development that achieves an 6.9:1 FSR GFA outcome with the application of S94A contributions at 4.5% of the cost of construction of the entire development.
- Scenario B Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 10% of the value of the additional GFA above the draft Planning Proposal controls (ie. 4:1 FSR GFA). This assumes that the base FSR for the site before the draft Planning Proposal was 2:1. Council considers this to be reasonable representation of the likely FSR uplifts in the CBD for this scenario.
- Scenario C Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 20% of the value of the additional GFA above the draft Planning Proposal controls (ie. 4:1 FSR GFA). This assumes that the base FSR for the site before the draft Planning Proposal was 2:1. Council considers this to be reasonable representation of the likely FSR uplifts in the CBD for this scenario.
- Scenario D Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 50% of the value of the additional GFA above the draft Planning Proposal controls (ie. 4:1 FSR GFA). This assumes that the base FSR for the site before the draft Planning Proposal was 2:1. Council considers this to be reasonable representation of the likely FSR uplifts in the CBD for this scenario.

- Base case 2 12:1 FSR Development of the hypothetical site to achieve a development outcome achieving a 12:1 FSR GFA outcome with S94A contributions levied at 3% of the cost of construction. This GFA outcome represent the 10:1 FSR entitlement shown in Council's draft Planning Proposal as well as the additional 2:1 FSR that would be achieved under the incentive provisions for 'Design Excellence' and 'High Performing Buildings'. The land acquisition value determined in this scenario was then used for each of the other scenarios as a constant to reflect the situation where developers have purchased land and new contribution regimes have been put in place.
- Scenario E Development that achieves an 12:1 FSR GFA outcome with the application of S94A contributions at 4.5% of the cost of construction of the entire development.
- Scenario F Development that achieves an 12:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 10% of the value of additional GFA above the existing controls (ie. 4:1 FSR GFA).
- Scenario G Development that achieves an 12:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 20% of the value of additional GFA above the existing controls (ie. 4:1 FSR GFA).
- Scenario H Development that achieves an 12:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 50% of the value of additional GFA above the existing controls (ie. 4:1 FSR GFA).
- Scenario I Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 4.5% of the cost of construction of the entire development plus Value Sharing on Phase 2 at 50% of the value of the additional GFA above the draft Planning Proposal controls (ie. 3:1 FSR GFA).
- Scenario J Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 10% of the value of additional GFA above the existing controls (ie. 4:1 FSR GFA) and Phase 2 at 50% of the value of the additional GFA above the draft Planning Proposal controls (ie. 3:1 FSR GFA).
- Scenario K Development that achieves an 15:1 FSR GFA outcome with the application of S94A contributions at 3% of the cost of construction of the entire development plus Value Sharing on Phase 1 at 50% of the value of additional GFA above the existing controls (ie. 4:1 FSR GFA) and Phase 2 at 50% of the value of the additional GFA above the draft Planning Proposal controls (ie. 3:1 FSR GFA).

These scenarios were modelled as a hypothetical development of a 3,000m² site within the Parramatta CBD with various assumptions informing the feasibility analysis (see **Appendix B**). An analysis of land transactions and DA approvals in the Parramatta CBD within the 2014 to 2015 period was carried out to identify an average dollar rate that had been paid for per m2 of GFA in the Parramatta CBD on B4 land (this rate is discussed earlier in the report). For modelling purposes, a rate of \$805/m² of GFA was adopted to determine the likely contributions generated in each of the scenarios as well as to model the impact on development feasibility following the imposition of Value Sharing schemes. Increments of approximately \$50/m² GFA were then modelled for consideration by Council in determining the appropriate rate for any Value Sharing scheme that was pursued.



Appendix D –Value Sharing development feasibility analysis and assumptions Parramatta Infrastructure Funding Models Study

The results of the application of these different rates on contribution income generated and impacts on resultant development feasibility are summarised in the tables below. From review of the tables, it is evident that the application of different m^2 GFA rates do not have the greatest impact on development achieving the feasibility benchmarks but rather the imposition of Phase 1 Value Sharing

The summary sheets from each of the modelled scenarios follow these tables in Section 2.0.



1.1 Contribution income and development feasibility at \$805/m² of additional GFA

	Contributions scenario	Contribution generated	Developers profit	IRR	Margin
	Case 1 – 6.9:1 FSR with S94A @ nd no VS	\$2.21M	\$18.20M	20.0%	17.3%
A	6.9:1 FSR with S94A @ 4.5% and no VS	\$3.31M	\$17.08M	19.2%	16.1%
В	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$3.17M	\$17.21M	19.3%	16.4%
С	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$4.14M	\$16.23M	18.6%	15.4%
D	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$7.04M	\$13.29M	16.4%	12.6%
Base 3%	Case 2 – 12:1 FSR with S94A @	\$3.84M	\$35.22M	20.0%	18.3%
E	12:1 FSR with S94A @ 4.5% and no VS	\$5.77M	\$33.34M	19.4%	17.1%
F	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @10%	\$4.81M	\$34.31M	19.7%	17.8%
G	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @20%	\$5.78M	\$33.33M	19.4%	17.3%
н	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @50%	\$8.62M	\$30.44M	18.3%	15.8%
I	15:1 FSR with S94A @ 4.5% and Phase 2 VS on 3:1 FSR @ 50%	\$10.76M	\$53.93M	25.9%	23.6%
J	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10% and Phase 2 VS on 3:1 FSR @ 50%	\$9.34M	\$55.34M	26.3%	24.5%
к	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50% and Phase 2 VS on 3:1 FSR @ 50%	\$13.21M	\$51.48M	25.0%	22.8%



1.2 Contribution income and development feasibility at \$750/m² of additional GFA

	Contributions scenario	Contribution generated	Developers profit	IRR	Margin
	Case – 6.9:1 FSR with S94A @ nd no VS	\$2.21M	\$18.20M	20.0%	17.3%
A	6.9:1 FSR with S94A @ 4.5% and no VS	\$3.31M	\$17.08M	19.2%	16.1%
В	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$3.11M	\$17.28M	19.4%	16.4%
С	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$4.01M	\$16.34M	18.7%	15.6%
D	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$6.71M	\$13.62M	16.7%	12.9%
Base 3%	Case – 12:1 FSR with S94A @	\$3.84M	\$35.22M	20.0%	18.3%
E	12:1 FSR with S94A @ 4.5% and no VS	\$5.77M	\$33.34M	19.4%	17.1%
F	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @10%	\$4.74M	\$34.37M	19.8%	17.9%
G	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @20%	\$5.64M	\$33.46M	19.4%	17.4%
н	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @50%	\$8.29M	\$30.77M	18.4%	16.0%
I	15:1 FSR with S94A @ 4.5% and Phase 2 VS on 3:1 FSR @ 50%	\$10.51M	\$54.18M	25.9%	23.7%
J	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10% and Phase 2 VS on 3:1 FSR @ 50%	\$9.03M	\$55.66M	26.4%	24.6%
к	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50% and Phase 2 VS on 3:1 FSR @ 50%	\$12.63M	\$52.06M	25.2%	23.0%



1.3 Contribution income and development feasibility at \$700/m² of additional GFA

	Contributions scenario	Contribution generated	Developers profit	IRR	Margin
	Case – 6.9:1 FSR with S94A @ nd no VS	\$2.21M	\$18.20M	20.0%	17.3%
A	6.9:1 FSR with S94A @ 4.5% and no VS	\$3.31M	\$17.08M	19.2%	16.1%
В	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$3.05M	\$17.34M	19.4%	16.5%
С	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$3.89M	\$16.49M	18.8%	15.7%
D	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$6.41M	\$13.93M	16.9%	13.2%
Base 3%	Case – 12:1 FSR with S94A @	\$3.84M	\$35.22M	20.0%	18.3%
E	12:1 FSR with S94A @ 4.5% and no VS	\$5.77M	\$33.34M	19.4%	17.1%
F	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @10%	\$4.68M	\$34.44M	19.8%	17.9%
G	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @20%	\$5.52M	\$33.58M	19.5%	17.4%
н	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @50%	\$7.99M	\$31.08M	18.5%	16.1%
I	15:1 FSR with S94A @ 4.5% and Phase 2 VS on 3:1 FSR @ 50%	\$10.28M	\$54.40M	26.0%	23.8%
J	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10% and Phase 2 VS on 3:1 FSR @ 50%	\$8.75M	\$55.94M	26.5%	24.7%
к	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50% and Phase 2 VS on 3:1 FSR @ 50%	\$12.11M	\$52.58M	25.4%	23.3%



1.4 Contribution income and development feasibility at \$655/m² of additional GFA

	Contributions scenario	Contribution generated	Developers profit	IRR	Margin
Base Case – 6.9:1 FSR with S94A @ 3% and no VS		\$2.21M	\$18.20M	20.0%	17.3%
A	6.9:1 FSR with S94A @ 4.5% and no VS	\$3.31M	\$17.08M	19.2%	16.1%
В	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10%	\$2.99M	\$17.40M	19.4%	16.5%
С	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 20%	\$3.78M	\$16.60M	18.9%	15.8%
D	6.9:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50%	\$6.14M	\$14.20M	17.1%	13.5%
Base 3%	Case – 12:1 FSR with S94A @	\$3.84M	\$35.22M	20.0%	18.3%
E	12:1 FSR with S94A @ 4.5% and no VS	\$5.77M	\$33.34M	19.4%	17.1%
F	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @10%	\$4.63M	\$34.49M	19.8%	17.9%
G	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @20%	\$5.42M	\$33.69M	19.5%	17.5%
н	12:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @50%	\$7.72M	\$31.35M	18.6%	16.6%
I	15:1 FSR with S94A @ 4.5% and Phase 2 VS on 3:1 FSR @ 50%	\$10.08M	\$54.61M	26.1%	23.9%
J	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 10% and Phase 2 VS on 3:1 FSR @ 50%	\$8.49M	\$56.20M	26.6%	24.9%
к	15:1 FSR with S94A @ 3% and Phase 1 VS on 4:1 FSR @ 50% and Phase 2 VS on 3:1 FSR @ 50%	\$11.63M	\$53.05M	25.6%	23.5%



2.0 Scenario Model Summary Sheets

This section includes the model summary sheets for each of the two base case and each of 11 Value Sharing scenarios proposed by Council. Diagrams representing the application of the different contribution and Value Sharing regimes are provided followed by a feasibility Summary sheet, Key Inputs sheet, Calculations sheet and Cashflow sheet. The Model Summary sheets utilise the \$805/m2 of GFA rate (discussed in the main body of this report) for the purposes of determining the different value sharing contributions. This rate is then modified by percentage rates (ie. 10%, 20%, 50%) for the purposes of modelling the different scenarios.


2.1 6.9:1 FSR Base Scenario

Hypothetical development to achieve 6.9:1 FSR GFA outcome with application of S94A at 3% of cost of construction. Whilst the draft Planning Proposal will typically apply a 6:1 FSR control, it will also provide for additional incentive FSR of 0.9:1 FSR for 'Design Excellence' that is required for any building in excess of 40 metres in height.

Council considers it unlikely that developers will not choose to take up the additional FSR incentives given the likelihood that 6.9:1 developments would exceed the 40 metre height limit and that the costs of achieving their 'Design Excellence' criteria are not overly onerous and are offset by the increase in GFA permitted.



planning

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	6.9
Land Area	3,000
Site Value	66,110
Number of Dwellings	172

gln planning

REVENUE		
REVENUE	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$696,802	\$137,766,667
GST	66,805	11,490,517
Less Selling Costs	16,769	2,884,333
NET REVENUE	\$717,394	\$123,391,816
COSTS		
Land (including acquisition costs)	66,110	11,371,000
Acquisition costs	4,610	792,839
Construction	428,343	73,675,042
Consultants	17,134	2,947,002
Section 94A - Commercial	3,125	537,500
Section 94A - Residential	9,699	1,668,251
Value Sharing Contribution	0	0
Statutory Fees & Contributions	6,953	1,195,882
On Costs	12,850	2,210,251
Marketing	16,019	2,755,333
Cost before Interest	564,844	97,153,101
Finance (incl Loan Est Fees)	46,761	8,042,955
TOTAL DEVELOPMENT COSTS	611,605	\$105,196,055
TOTAL PROJECT SURPLUS	17.3%	\$18,195,783
PROJECT IRR BEFORE INTEREST	20.0%	

Input Sheet

SITE DETAILS				
Address	Repr	Representative		
Floor Space Ratio	6.9	6.9 :1		
Achievable FSR	20,700	m2		
Land Area	3,000	m2		
Site Value	66,110			
Any 1 parcel over \$3M	Yes			
LVR (Debt/Equity)	100%			
Date to start feasibility	Oct-15			



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		11,370,978		
Loan Establishment C	osts	16,583		
Stamp Duty		736,458		
Legals	0.4%	39,798		
LAND ACQUISITION C	COSTS - Deposit	1,137,098	1	1
LAND ACQUISITION C	COSTS - Settlement	11,026,719	2	1
CONSTRUCTION COS	TS	73,675,042	12	20
PROFESSIONAL FEE	6 4% const costs	2,947,002	1	31
APPLICATION FE	ES			
DA Fees		352,527	3	1
CC Fees		136,966	12	1
Section 94		2,205,751	31	1
Value Sharing		0	31	1
LPI Fees		1,321	31	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	235,023	10	1
Land Tax/Rates	Year 2	235,023	22	1
Land Tax/Rates	Year 3	235,023	34	1
ONCOSTS	3.0% construction cost	2,210,251	2	30
MARKETING	2% gross revenue	2,755,333	12	19
	TOTAL COST	97,153,078		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL	-								
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL	-					-	-		
1 bedroom	29	60	2,001	29	Yes	590,000	17,110,000	17%	15%
2 bedroom	131	80	12,052	131	Yes	700,000	91,700,000	76%	75%
3 bedroom	12	120	1,656	12	Yes	920,000	11,040,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	172		20,709	280	Yes		137,766,667	100%	100%
Average m2/unit	79								
ADDITIONAL INCLU	SIONS								
Other visitor parking				34					
TOTAL YIELD	172		20,709	315			137,766,667		
		FSR	6.9			_			
		Target Floorspace	20,700						

Calculations

REVENUE		
GROSS REVENUE		137,766,667
GST		11,490,517
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	2,755,333
Legals (On Settlement)	\$750 per lot	129,000
TOTAL - SELLING COSTS		2,884,333
NET REVENUE		123,391,816



STAMP DUTY			
LAND VALUE	THRESHOLD	ТАХ	
Minimum	Maximum		
14,000		0	
14,000	30,000	0	
30,000	80,000	0	
80,000	300,000	0	
300,000	1,000,000	0	
1,000,000		FALSE	
3,000,000	Premium Property Tax	736,458	
	TOTAL	736,458	

CONTRIBUTIONS		
SECTION 94A		3.0%
COSTS		
Retail	10,416,667	\$312,500
Commercial	7,500,000	\$225,000
Residential	55,608,375	\$1,668,251
TOTAL VALUE 73,525,042		
TOTAL CONT	\$2,205,751	

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		94,665
LSL	0.35%	257,863
TOTAL DA FEES		352,527

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	136,966
TOTAL CC FEES	136,966

CONSTRUCT	ION COSTS	3		
Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	44,770,650	8,600,000	
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			1,720,000	0
ESD Costs	0	0		
Design Comp		150,000		
Design Cost	25	517,725		
TOTALS		57,938,375	15,736,667	0
	TOTAL CO	NSTRUCTIC	ON COST	73,675,042

Unit Turne No. Creases		
Unit Type	No. Spaces	
1 bed/studio	1.0	
2 bed	1.0	
3+ bed	1.0	
Shops	30	
Commercial	100	
Visitor parking	In any case, min 1 visitor space is required	

RATES	
Council Rates	24,979
Land Tax	210,044
TOTAL RATES	235,023

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	1,321		

VALUE SHARING		
GFA Element	GFA	Base Value
4:1 CBD Strategy	0	\$0
3:1 Above Strategy	0	\$0
Calculation		
Rate/additional m2 GFA		\$805
CBD Strategy Rate		10%
Above CBD Strategy Rate	50%	
TOTAL CONTRIBUTION	\$0	

Sheet 3 of 4

Assumed Development Cashflow

DEVELOPMENT SUMMARY						
IRR Before Interest	20.04%					
Surplus	\$18,195,783					
% Surplus on D.C	17.30%					
Max Loan Balance Debt/Equity	-\$104,417,171					
Debt/Equity	100.00%	Numbers b				

% Surplu				17.30%																	4	-		-
Max Loa Debt/Eq		ice Debi	t/Equity	-\$104,417,171 100.00%		ellow expres	sed in neares	t \$,000													<		plann	ing
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget Budget	Oct-15 Nov-15 Dec-15 Jan-16 Feb-16 Mar-16 Jun-16 Jun-16 Jun-16 Sep-16 Oct-16 Nov-18 Dec-16 Nov-18 Dec-16 Nov-18 Dec-17 Jan-17 Apr-17 Apr-17 Apr-17 Jun-17 Jun-17 Sep-17 Oct-17 Jun-17 Nov-17 Dec-17 Jan-18 Feb-18 Mar-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-18 Apr-17 Nov-17 Dec-17 Jun-17 Nov-17 Dec-17 Jun-17 Nov-17 Dec-17 Jun-17 Apr-17 Nov-17 Dec-17 Jun-17 Apr-17 Nov-17 Dec-17 Jun-17 Apr-17 Apr-17 Apr-17 Apr-17 Apr-17 Apr-17 Apr-17 Apr-17 Apr-17 Nov-17 Dec-17 Jun-17 Apr-17 Nov-17 Dec-17 Jun-17 Apr-18 Apr-18 Ap	0 1 2 3 4 5 6 7 8 9 10 11 12 13 4 15 6 7 8 9 10 11 12 13 14 15 16 17 18 9 20 21 22 24 26 27 28 29 30 31 22 23 24 25 26 27 28 29 20 31 22 23 24 25 26 27 28 29 20 20 21 22 23 24 25 26 27 28 29 20 20 20 20 20 20 20 20 20 20	0 1,137 11,027 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 95 95 95 95 95 95 95 95 95 95 95 95 95	0 0 353 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 74 74 74 74 74 74 74 74 74 74 74 74 74	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$0 \$11,232 \$11,195 \$521 \$169 \$169 \$169 \$169 \$4,134 \$3,998		\$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$	\$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$	\$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$	\$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$0 \$	\$0 \$1,232 \$11,195 \$521 \$169 \$169 \$169 \$169 \$169 \$169 \$4,04 \$169 \$4,134 \$169 \$4,134 \$169 \$3,998 \$3,99	\$0 -\$1,232 -\$11,232 -\$169 -\$169 -\$169 -\$169 -\$169 -\$410 -\$4,134 -\$3,998 -	\$0 \$0 -\$65 -\$665 -\$669 -\$772 -\$773 -\$774 -\$775 -\$778 -\$779 -\$101 -\$122 -\$1445 -\$187 -\$122 -\$1465 -\$187 -\$122 -\$1465 -\$187 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$201 -\$209 -\$200 -\$20	\$0 \$1,232 \$12,434 \$13,020 \$13,257 \$13,494 \$13,257 \$13,494 \$13,277 \$13,494 \$14,215 \$14,458 \$14,937 \$15,184 \$19,397 \$23,466 \$27,616 \$27,616 \$27,616 \$27,616 \$27,616 \$27,616 \$27,616 \$27,616 \$27,615 \$27,625 \$66,910 \$774,614 \$79,000 \$87,025 \$74,614 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$79,000 \$87,814 \$19,0338 \$10,0338 \$10,0338 \$10,0376 \$10,0417 \$18,196	0 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 17 19 20 21 22 24 25 26 29 30 31 22 33 34
Budget Budget Budget	Aug-18 Sep-18 Oct-18	34 35 36	0	0	0	0	0	0	0	0	235 0 0	0	0 0	\$235 \$0 \$0	0	\$137,767 \$0 \$0	\$11,491 \$0 \$0	\$2,884 \$0 \$0	\$123,392 \$0 \$0	\$235 \$0 \$0	\$123,157 \$0 \$0	-\$544 \$0 \$0	\$18,196 \$18,196 \$18,196	34 35 36
L			12,164	\$2,947	\$353	\$137	\$73,675	\$2,206	\$0	\$1	\$705	\$2,210	\$2,755	\$97,153	70	\$137,767	\$11,491	\$2,884	\$123,392	\$97,153	\$26,239	-\$8,043	\$18,196	
			Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs		LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs		Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net	Monthly Interest		

Sheet 4 of 4

รเป

2.2 Scenario A

Hypothetical development to achieve 6.9:1 FSR GFA outcome with application of S94A at 4.5% of cost of construction. Whilst the draft Planning Proposal will typically apply a 6:1 FSR control, it will also provide for additional incentive FSR of 0.9:1 FSR for 'Design Excellence' that is required for any building in excess of 40 metres in height.



Scenario A - S94A Levy 4.5% Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	6.9
Land Area	3,000
Site Value	66,110
Number of Dwellings	172

g lin planning

REVENUE		
	Average/	_
	Unit	Development
	(\$) Ex Retail	(\$) Inc Retail
	EX Retail	inc Retail
GROSS REVENUE	\$696,802	\$137,766,667
	· · ·	
GST	66,805	11,490,515
Less Selling Costs	16,769	2,884,333
	A747.004	
NET REVENUE	\$717,394	\$123,391,818
COSTS		
Land (including acquisition costs)	66,110	11,371,000
Acquisition costs	4,610	792,841
Construction	428,343	73,675,042
Consultants	17,134	2,947,002
Consultants	17,134	2,347,002
Section 94A - Commercial	4,688	806,250
Section 94A - Residential	14,549	2,502,377
	•	
Value Sharing Contribution	0	0
Statutory Fees & Contributions	6,953	1,195,883
,	- ,	,,
On Costs	12,850	2,210,251
Marketing	16,019	2,755,333
Cost before Interest	571 256	98,255,979
	571,256	90,200,979
Finance (incl Loan Est Fees)	46,862	8,060,282
	- ,	-,
TOTAL DEVELOPMENT COSTS	618,118	\$106,316,261
TOTAL PROJECT SURPLUS	16.1%	\$17,075,557
PROJECT IRR BEFORE INTEREST	19.2%	

Scenario A - S94A Levy 4.5% Input Sheet

SITE DETAILS						
Address	Repre	Representative				
Floor Space Ratio	6.9 :'	1				
Achievable FSR	20,700 n	n2				
Land Area	3,000 n	n2				
Site Value	66,110					
Any 1 parcel over \$3M	Yes					
LVR (Debt/Equity)	100%					
Date to start feasibility	Oct-15					



			Period	
Cost Items		Cost (\$)	Commencing (month)	Term (months)
LAND				
Land Purchase		11,371,000		
Loan Establishment Costs	i	16,583		
Stamp Duty		736,460		
Legals	0.4%	39,799		
LAND ACQUISITION COST	rS - Deposit	1,137,100	1	1
LAND ACQUISITION COST	rs - Settlement	11,026,741	2	1
CONSTRUCTION COSTS		73,675,042	12	20
PROFESSIONAL FEES	4% const costs	2,947,002	1	31
APPLICATION FEES				
DA Fees		352,527	3	1
CC Fees		136,966	12	1
Section 94		3,308,627	31	1
Value Sharing		0	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	235,023	10	1
Land Tax/Rates	Year 2	235,023	22	1
Land Tax/Rates	Year 3	235,023	34	1
ONCOSTS	3.0% construction cost	2,210,251	2	30
MARKETING	2% gross revenue	2,755,333	12	19
	TOTAL COST	98,255,979		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL					-				
1 bedroom	29	60	2,001	29	Yes	590,000	17,110,000	17%	15%
2 bedroom	131	80	12,052	131	Yes	700,000	91,700,000	76%	75%
3 bedroom	12	120	1,656	12	Yes	920,000	11,040,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	172		20,709	280	Yes		137,766,667	100%	100%
Average m2/unit	79								-
ADDITIONAL INCLU	SIONS								
Other visitor parking				34					
TOTAL YIELD	172		20,709	315			137,766,667		
		FSR	6.9		-	-			-
		Target Floorspace	20,700]					

Sheet 2 of 4

Scenario A - S94A Levy 4.5% Calculations

REVENUE		
GROSS REVENUE		137,766,667
GST		11,490,515
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	2,755,333
Legals (On Settlement)	\$750 per lot	129,000
TOTAL - SELLING COSTS		2,884,333
NET REVENUE		123,391,818



STAMP DUTY							
LAND VALUE	ΤΑΧ						
Minimum	Maximum						
14,000		0					
14,000	30,000	0					
30,000	80,000	0					
80,000	300,000	0					
300,000	1,000,000	0					
1,000,000		FALSE					
3,000,000	Premium Property Tax	736,460					
TOTAL 736,460							

CONTRIBUTIONS							
SECTION 94A	SECTION 94A						
COSTS							
Retail	10,416,667	\$468,750					
Commercial	7,500,000	\$337,500					
Residential	55,608,375	\$2,502,377					
TOTAL VALUE	73,525,042						
TOTAL CONT	RIBUTION	\$3,308,627					

DA FEES		
CONSTRUCTION C	COST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		94,665
LSL	0.35%	257,863
TOTAL DA FEES		352,527

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	136,966
TOTAL CC FEES	136,966

Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	44,770,650	8,600,000	
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			1,720,000	C
ESD Costs	0	0	-	
Design Comp		150,000		
Design Cost	25	517,725		
TOTALS		57,938,375	15,736,667	C
	73,675,042			

Unit Type	No. Spaces							
1 bed/studio	1.0							
2 bed	1.0							
3+ bed	1.0							
Shops	30							
Commercial	100							
Visitor parking	In any case, min 1 visitor space is required							

RATES	
Council Rates	24,979
Land Tax	210,044
TOTAL RATES	235,023

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	1,321		

VALUE SHARING							
GFA Element	GFA	Base Value					
4:1 CBD Strategy	0	\$0					
3:1 Above Strategy	\$0						
Calculation							
Rate/additional m2 GFA		\$0					
CBD Strategy Rate		0%					
Above CBD Strategy Rate	0%						
TOTAL CONTRIBUTION	\$0						

Sheet 3 of 4

Scenario A - S94A Levy 4.5%

Assumed Development Cashflow

DEVELOPMENT SUM	DEVELOPMENT SUMMARY					
IRR Before Interest	19.21%					
Surplus	\$17,075,557					
% Surplus on D.C	16.06%					
Max Loan Balance Debt/Equity	-\$105,531,594					
Debt/Equity	100.00%					

Max Loai Debt/Equ		ce Deb	t/Equity	-\$105,531,594 100.00%		ellow express	sed in nearest	\$,000													~		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	1,137	95	0	0	0	0	0	0	0	0	0	\$1,232	0	\$0	\$0	\$0	\$0	\$1,232	-\$1,232	\$0	-\$1,232	1
Budget	Dec-15	2	11,027	95	0	0	0	0	0	0	0	74	0	\$11,195	0	\$0	\$0	\$0	\$0	\$11,195	-\$11,195	-\$6	-\$12,434	2
Budget	Jan-16	3	0	95	353	0	0	0	0	0	0	74	0	\$521	0	\$0	\$0	\$0	\$0	\$521	-\$521	-\$65	-\$13,020	3
Budget	Feb-16	4	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$68	-\$13,257	4
Budget	Mar-16	5	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$69	-\$13,494	5
Budget	Apr-16	6	0	95	0	0	0	0	0	0	0	74 74	0	\$169	0	\$0 \$0	\$0 \$0	\$0	\$0	\$169	-\$169	-\$70	-\$13,733	6
Budget Budget	May-16 Jun-16	8	0	95 95	0	0	0	0	0	0	0	74	0	\$169 \$169	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$169 \$169	-\$169 -\$169	-\$72 -\$73	-\$13,974 -\$14,215	8
Budget	Jul-16	9	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0 \$0	\$0	\$0	\$0	\$169	-\$169	-\$73	-\$14,458	9
Budget	Aug-16	10	ő	95	0	0	ő	ő	ŏ	ő	235	74	ő	\$404	ő	\$0	\$0	\$0	\$0	\$404	-\$404	-\$75	-\$14,937	10
Budget	Sep-16	11	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$78	-\$15,184	11
Budget	Oct-16	12	0	95	0	137	3,684	0	0	0	0	74	145	\$4,134	0	\$0	\$0	\$0	\$0	\$4,134	-\$4,134	-\$79	-\$19,397	12
Budget	Nov-16	13	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$101	-\$23,496	13
Budget	Dec-16	14	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$122	-\$27,616	14
Budget	Jan-17	15	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$144	-\$31,757	15
Budget	Feb-17	16	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$165	-\$35,920	16
Budget	Mar-17	17	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$187	-\$40,104	17
Budget	Apr-17	18 19	0	95 95	0	0	3,684 3,684	0	0	0	0	74 74	145 145	\$3,998 \$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998 \$3,998	-\$3,998 -\$3,998	-\$209 -\$231	-\$44,311 -\$48,539	18 19
Budget	May-17 Jun-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998	-\$3,998	-\$253	-\$46,539 -\$52,789	20
Budget Budget	Jul-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$233	-\$57,062	20
Budget	Aug-17	22	0	95	0	0	3,684	0	0	ő	235	74	145	\$4,233	0	\$0	\$0	\$0	\$0	\$4,233	-\$4,233	-\$297	-\$61,592	22
Budget	Sep-17	23	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$321	-\$65,910	23
Budget	Oct-17	24	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$343	-\$70,251	24
Budget	Nov-17	25	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$366	-\$74,614	25
Budget	Dec-17	26	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$389	-\$79,000	26
Budget	Jan-18	27	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$411	-\$83,409	27
Budget	Feb-18	28	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$434	-\$87,841	28
Budget	Mar-18	29	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$458	-\$92,296	29
Budget	Apr-18	30 31	0	95 95	0	0	3,684	0	0	0	0	74 74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0	\$3,998	-\$3,998	-\$481	-\$96,774	30
Budget Budget	May-18 Jun-18	31 32	0	95	0	0	3,684	3,309	0	1	0	74 0	0	\$7,162 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,162 \$0	-\$7,162 \$0	-\$504 -\$544	-\$104,441 -\$104,985	31 32
Budget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0 \$0	\$0	\$0	\$0	\$0 \$0	\$0	-\$547	-\$105,532	32
Budget	Aug-18	34	0	ő	0	ŏ	0	0	0	ő	235	0	ő	\$235	70	\$137,767	\$11,491	\$2,884	\$123,392	\$235	\$123,157	-\$550	\$17,076	34
Budget	Sep-18	35	ő	ő	0 0	0	ő	0	õ	ő	0	0	ő	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$17,076	35
Budget	Oct-18	36	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$17,076	36
	1	1	12,164	\$2,947	\$353	\$137	\$73,675	\$3,309	\$0	\$1	\$705	\$2,210	\$2,755	\$98,256	70	\$137,767	\$11,491	\$2,884	\$123,392	\$98,256	\$25,136	-\$8,060	\$17,076	
			Land	Professional	DA Fees	CC Fees	Construction			LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

Scenario A - Parramatta_S94A 4.5%.xlsx

2.3 Scenario B

Hypothetical development to achieve 6.9:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 10%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 2:1 control to a proposed 6:1 FSR, resulting in an FSR uplift of 4:1.



Scenario B - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	6.9
Land Area	3,000
Site Value	66,110
Number of Dwellings	172



REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$696,802	\$137,766,667
GST	66,805	11,490,515
Less Selling Costs	16,769	2,884,333
NET REVENUE	\$717,394	\$123,391,818
COSTS		
Land (including acquisition costs)	66,110	11,371,000
Acquisition costs	4,610	792,841
Construction	428,343	73,675,042
Consultants	17,134	2,947,002
Section 94A - Commercial	3,125	537,500
Section 94A - Residential	9,699	1,668,251
Value Sharing Contribution	5,616	966,000
Statutory Fees & Contributions	6,953	1,195,883
On Costs	12,850	2,210,251
Marketing	16,019	2,755,333
Cost before Interest	564,844	97,153,104
Finance (incl Loan Est Fees)	46,850	8,058,132
TOTAL DEVELOPMENT COSTS	611,693	\$105,211,236
TOTAL PROJECT SURPLUS	16.4%	\$17,214,583
PROJECT IRR BEFORE INTEREST	19.3%	

Sheet 1 of 4

Scenario B - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy

Input Sheet

SITE DETAILS		
Address	Repr	resentative
Floor Space Ratio	6.9	:1
Achievable FSR	20,700	m2
Land Area	3,000	m2
Site Value	66,110	
Any 1 parcel over \$3M	Yes	
LVR (Debt/Equity)	100%	
Date to start feasibility	Oct-15	



DEVELOPMENT COST

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		11,371,000		
Loan Establishment Co	osts	16,583		
Stamp Duty		736,460		
Legals	0.4%	39,799		
LAND ACQUISITION C	OSTS - Deposit	1,137,100	1	1
LAND ACQUISITION C	OSTS - Settlement	11,026,741	2	1
CONSTRUCTION COST	rs	73,675,042	12	20
PROFESSIONAL FEES	4% const costs	2,947,002	1	31
APPLICATION FE	ES			
DA Fees		352,527	3	1
CC Fees		136,966	12	1
Section 94		2,205,751	31	1
Value Sharing		966,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES	3			
Land Tax/Rates	Year 1	235,023	10	1
Land Tax/Rates	Year 2	235,023	22	1
Land Tax/Rates	Year 3	235,023	34	1
ONCOSTS	3.0% construction cost	2,210,251	2	30
MARKETING	2% gross revenue	2,755,333	12	19
	TOTAL COST	98,119,104		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	29	60	2,001	29	Yes	590,000	17,110,000	17%	15%
2 bedroom	131	80	12,052	131	Yes	700,000	91,700,000	76%	75%
3 bedroom	12	120	1,656	12	Yes	920,000	11,040,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	172		20,709	280	Yes		137,766,667	100%	100%
Average m2/unit	79								
ADDITIONAL INCLU	SIONS								
Other visitor parking				34					
TOTAL YIELD	172		20,709	315			137,766,667		
		FSR	6.9						-
		Target Floorspace	20,700						

Sheet 2 of 4

Scenario B - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		137,766,667
GST		11,490,515
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	2,755,333
Legals (On Settlement)	\$750 per lot	129,000
TOTAL - SELLING COSTS		2,884,333
NET REVENUE		123,391,818



STAMP DUTY				
LAND VALUE THRESHOLD TAX				
Minimum	Maximum			
14,000		0		
14,000	30,000	0		
30,000	80,000	0		
80,000	300,000	0		
300,000	1,000,000	0		
1,000,000		FALSE		
3,000,000	Premium Property Tax	736,460		
	TOTAL	736,460		

CONTRIBUTIONS				
SECTION 94A	3.0%			
COSTS				
Retail	10,416,667	\$312,500		
Commercial	7,500,000	\$225,000		
Residential	55,608,375	\$1,668,251		
TOTAL VALUE 73,525,042				
TOTAL CONT	\$2,205,751			

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		94,665
LSL	0.35%	257,863
TOTAL DA FEES		352,527

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	136,966
TOTAL CC FEES	136,966

CONSTRUCTION COSTS						
Build Costs	Rates	FSR +15%	UG Parking	Grade		
	\$/m2		50,000	Parking		
Apartments	2,850	44,770,650	8,600,000			
Retail	2,500	6,250,000	4,166,667			
Commercial	2,500	6,250,000	1,250,000			
Visitor Parking			1,720,000	C		
ESD Costs	0	0	-			
Design Comp		150,000				
Design Cost	25	517,725				
TOTALS		57,938,375	15,736,667	0		
TOTAL CONSTRUCTION COST			73,675,042			

CAR PARKING			
Unit Type	No. Spaces		
1 bed/studio	1.0		
2 bed	1.0		
3+ bed	1.0		
Shops	30		
Commercial	100		
Visitor parking	In any case, min 1 visitor space is required		

RATES	
Council Rates	24,979
Land Tax	210,044
TOTAL RATES	235,023

LPI FEES			
Strata	Base Fee	Per Dwg fee 132	
TOTAL LPI F	1,321 EES	132	1,321

VALUE SHARING						
GFA Element	GFA	Base Value				
4:1 CBD Strategy	12,000	\$9,660,000				
3:1 Above Strategy	0	\$0				
Calculation						
Rate/additional m2 GFA		\$805				
CBD Strategy Rate		10%				
Above CBD Strategy Rate	0%					
TOTAL CONTRIBUTION	\$966,000					

Sheet 3 of 4

Scenario B - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy PLUS

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	19.31%
Surplus	\$17,214,583
% Surplus on D.C	16.21%
Max Loan Balance Debt/Equity	-\$105,393,289
Debt/Equity	100.00%

Max Loa Debt/Equ		ice Deb	t/Equity	-\$105,393,289 100.00%		ellow expres	sed in neares	t \$,000													<		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	1,137	95	0	0	0	0	0	0	0	0	0	\$1,232	0	\$0	\$0	\$0	\$0	\$1,232	-\$1,232	\$0	-\$1,232	1
Budget	Dec-15	2	11,027	95	0	0	0	0	0	0	0	74	0	\$11,195	0	\$0	\$0	\$0	\$0	\$11,195	-\$11,195	-\$6	-\$12,434	2
Budget	Jan-16	3	0	95	353	0	0	0	0	0	0	74	0	\$521	0	\$0	\$0	\$0	\$0	\$521	-\$521	-\$65	-\$13,020	3
Budget	Feb-16	4	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$68	-\$13,257	4
Budget	Mar-16	5	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$69	-\$13,494	5
Budget	Apr-16	6	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$70	-\$13,733	6
Budget	May-16	7	0	95 95	0	0	0	0	0	0	0	74 74	0	\$169 \$169	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$169 \$169	-\$169 -\$169	-\$72 -\$73	-\$13,974	7
Budget Budget	Jun-16 Jul-16	9	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$169	-\$169	-\$73 -\$74	-\$14,215 -\$14,458	9
Budget	Aug-16	10	0	95	0	0	0	0	0	0	235	74	0	\$404	0	\$0	\$0	\$0	\$0	\$404	-\$404	-\$75	-\$14,937	10
Budget	Sep-16	11	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$78	-\$15,184	11
Budget	Oct-16	12	0	95	0	137	3,684	0	0	0	0	74	145	\$4,134	0	\$0	\$0	\$0	\$0	\$4,134	-\$4,134	-\$79	-\$19,397	12
Budget	Nov-16	13	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$101	-\$23,496	13
Budget	Dec-16	14	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$122	-\$27,616	14
Budget	Jan-17	15	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$144	-\$31,757	15
Budget	Feb-17	16	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$165	-\$35,920	16
Budget	Mar-17	17	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$187	-\$40,104	17
Budget	Apr-17	18 19	0	95 95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$3,998	-\$3,998	-\$209	-\$44,311	18
Budget Budget	May-17 Jun-17	20	0	95	0	0	3,684 3,684	0	0	0	0	74 74	145 145	\$3,998 \$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998 \$3,998	-\$3,998 -\$3,998	-\$231 -\$253	-\$48,539 -\$52,789	19 20
Budget	Jul-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$3,998	-\$3,998	-\$255	-\$57,062	20
Budget	Aug-17	22	0	95	0	0	3,684	0	0	0	235	74	145	\$4,233	0	\$0	\$0	\$0	\$0	\$4,233	-\$4,233	-\$297	-\$61,592	22
Budget	Sep-17	23	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$321	-\$65,910	23
Budget	Oct-17	24	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$343	-\$70,251	24
Budget	Nov-17	25	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$366	-\$74,614	25
Budget	Dec-17	26	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$389	-\$79,000	26
Budget	Jan-18	27	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$411	-\$83,409	27
Budget	Feb-18	28	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$434	-\$87,841	28
Budget	Mar-18	29	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$458	-\$92,296	29
Budget	Apr-18 May-18	30 31	0	95 95	0	0	3,684 3,684	0 2,206	0 966	0	U	74 74	145 0	\$3,998 \$7,026	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998 \$7,026	-\$3,998 -\$7,026	-\$481 -\$504	-\$96,774 -\$104,304	30 31
Budget Budget	Jun-18	31 32	0	95	0	0	3,684	2,206	966	1	0	74 0	0	\$7,026 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,026	-\$7,026 \$0	-\$504 -\$543	-\$104,304 -\$104,847	31 32
Budget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0 \$0	-\$546	-\$105,393	33
Budget	Aug-18	34	ő	ő	ő	0	0 0	ő	õ	ő	235	ő	ő	\$235	70	\$137,767	\$11,491	\$2,884	\$123,392	\$235	\$123,157	-\$549	\$17,215	34
Budget	Sep-18	35	0	0	0	0	0	0	0	0	0	ō	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$17,215	35
Budget	Oct-18	36	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$17,215	36
		1	12,164	\$2,947	\$353	\$137	\$73,675	\$2,206	\$966	\$1	\$705	\$2,210	\$2,755	\$98,119	70	\$137,767	\$11,491	\$2,884	\$123,392	\$98,119	\$25,273	-\$8,058	\$17,215	
			Land	Professional	DA Fees	CC Fees	Construction			LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

2.4 Scenario C

Hypothetical development to achieve 6.9:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 20%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 2:1 control to a proposed 6:1 FSR, resulting in an FSR uplift of 4:1.



Scenario C - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	6.9
Land Area	3,000
Site Value	66,110
Number of Dwellings	172

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$) Ex Retail	(\$) Inc Retail
	Exretai	Retail
GROSS REVENUE	\$696,802	\$137,766,667
	<i>•••••</i> ,••=	•••••••••••
GST	66,805	11,490,515
Less Selling Costs	16,769	2,884,333
NET REVENUE	\$717,394	\$123,391,818
00070		
COSTS		
Land (including acquisition costs)	66,110	11,371,000
	00,110	11,371,000
Acquisition costs	4,610	792,841
	.,	,
Construction	428,343	73,675,042
Consultants	17,134	2,947,002
	_	
Section 94A - Commercial	3,125	537,500
Section 044 Posidential	0 600	1 669 251
Section 94A - Residential	9,699	1,668,251
Value Sharing Contribution	11,233	1,932,000
Value channy contribution	11,200	1,002,000
Statutory Fees & Contributions	6,953	1,195,883
On Costs	12,850	2,210,251
	10.010	0.755.000
Marketing	16,019	2,755,333
Cost before Interest	564,844	97,153,104
	507,077	37,133,104
Finance (incl Loan Est Fees)	46,938	8,073,304
		-,,
TOTAL DEVELOPMENT COSTS	611,781	\$105,226,408
TOTAL PROJECT SURPLUS	15.4%	\$16,233,410
PROJECT IRR BEFORE INTEREST	18.6%	

Scenario C - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy Input Sheet

SITE DETAILS				
Address Representative				
Floor Space Ratio	6.9	:1		
Achievable FSR	20,700	m2		
Land Area	3,000	m2		
Site Value	66,110			
Any 1 parcel over \$3M	Yes			
LVR (Debt/Equity)	100%			

Oct-15



DEVELOPMENT COSTS

Date to start feasibility

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		11,371,000		
Loan Establishment Co	osts	16,583		
Stamp Duty		736,460		
Legals	0.4%	39,799		
LAND ACQUISITION C	OSTS - Deposit	1,137,100	1	1
LAND ACQUISITION C	OSTS - Settlement	11,026,741	2	1
CONSTRUCTION COST	rs	73,675,042	12	20
PROFESSIONAL FEES	4% const costs	2,947,002	1	31
APPLICATION FE	ES			
DA Fees		352,527	3	1
CC Fees		136,966	12	1
Section 94		2,205,751	31	1
Value Sharing		1,932,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES	6			
Land Tax/Rates	Year 1	235,023	10	1
Land Tax/Rates	Year 2	235,023	22	1
Land Tax/Rates	Year 3	235,023	34	1
ONCOSTS	3.0% construction cost	2,210,251	2	30
MARKETING	2% gross revenue	2,755,333	12	19
	TOTAL COST	99,085,104		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	Ū.	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL	-								
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL							-		
1 bedroom	29	60	2,001	29	Yes	590,000	17,110,000	17%	15%
2 bedroom	131	80	12,052	131	Yes	700,000	91,700,000	76%	75%
3 bedroom	12	120	1,656	12	Yes	920,000	11,040,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	172		20,709	280	Yes		137,766,667	100%	100%
Average m2/unit	79			•					
ADDITIONAL INCLU	SIONS								
Other visitor parking				34					
TOTAL YIELD	172		20,709	315			137,766,667		
	-	FSR	6.9						-
		Target Floorspace	20,700	1					

Scenario C - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		137,766,667
GST		11,490,515
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	2,755,333
Legals (On Settlement)	\$750 per lot	129,000
TOTAL - SELLING COSTS		2,884,333
NET REVENUE		123,391,818



STAMP DUTY						
LAND VALUE THRESHOLD TAX						
Minimum	Maximum					
14,000		0				
14,000	30,000	0				
30,000	80,000	0				
80,000	300,000	0				
300,000	1,000,000	0				
1,000,000		FALSE				
3,000,000	Premium Property Tax	736,460				
	TOTAL	736,460				

CONTRIBUTIONS					
SECTION 94A	3.0%				
COSTS					
Retail	10,416,667	\$312,500			
Commercial	7,500,000	\$225,000			
Residential	55,608,375	\$1,668,251			
TOTAL VALUE	73,525,042				
TOTAL CONT	\$2,205,751				

DA FEES							
CONSTRUCTION COST THRESHOLDS							
50,001	250,000	0					
250,001	500,000	0					
500,001	1,000,000	0					
1,000,001	10,000,000	0					
More than \$10,000,000		94,665					
LSL	0.35%	257,863					
TOTAL DA FEES		352,527					

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	136,966
TOTAL CC FEES	136,966

CONSTRUCTION COSTS										
Build Costs	Rates	FSR +15%	UG Parking	Grade						
	\$/m2		50,000	Parking						
Apartments	2,850	44,770,650	8,600,000							
Retail	2,500	6,250,000	4,166,667							
Commercial	2,500	6,250,000	1,250,000							
Visitor Parking			1,720,000	0						
ESD Costs	0	0	-							
Design Comp		150,000								
Design Cost	25	517,725								
TOTALS		57,938,375	15,736,667	0						
	N COST	73,675,042								

CAR PARKING							
Unit Type	No. Spaces						
1 bed/studio	1.0						
2 bed	1.0						
3+ bed	1.0						
Shops	30						
Commercial	100						
Visitor parking	In any case, min 1 visitor space is required						

RATES	
Council Rates	24,979
Land Tax	210,044
TOTAL RATES	235,023

LPI FEES			
Strata	Base Fee	Per Dwg fee 132	
TOTAL LPI F	1,321 EES	132	1,321

VALUE SHARING							
GFA Element	GFA	Base Value					
4:1 CBD Strategy	12,000	\$9,660,000					
3:1 Above Strategy	0	\$0					
Calculation							
Rate/additional m2 GFA		\$805					
CBD Strategy Rate		20%					
Above CBD Strategy Rate		0%					
TOTAL CONTRIBUTION	\$1,932,000						

Sheet 3 of 4

Scenario C - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	18.59%
Surplus	\$16,233,410
% Surplus on D.C	15.15%
Max Loan Balance Debt/Equity	-\$106,369,378
Debt/Equity	100.00%

Budget/	Oct-15	Month	Land	Professional	DA Fees	CC Fees	Construction	Section 94	Value	LPI Fess	Statuory	Oncosts	Marketing	Total	Sales	Gross	GST	Selling Costs	Net	Monthly	Net	Monthly	Total	Perio
Actual	Oct-15	Month	Lano	Fees	DA Fees	CC Fees	Costs	Costs	Sharing	LFIFess	Costs	Uncosts	marketing	Costs	Sales	Revenue	651	Selling Costs	Revenue	Outlays	Cashflow		Funds Cashflow	Perio
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	1,137	95	ő	ő	ő	0	õ	0	ő	ő	ő	\$1,232	0	\$0	\$0	\$0	\$0	\$1,232	-\$1,232	\$0	-\$1,232	1
Budget	Dec-15	2	11,027	95	0	0	0	0	0	0	0	74	0	\$11,195	0	\$0	\$0	\$0	\$0	\$11,195	-\$11,195	-\$6	-\$12,434	2
Budget	Jan-16	3	0	95	353	0	0	0	0	0	0	74	0	\$521	0	\$0	\$0	\$0	\$0	\$521	-\$521	-\$65	-\$13,020	3
Budget	Feb-16	4	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$68	-\$13,257	4
Budget	Mar-16	5	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$69	-\$13,494	5
Budget	Apr-16	6	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$70	-\$13,733	6
Budget	May-16	7	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$72	-\$13,974	7
udget	Jun-16	8	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$73	-\$14,215	8
udget	Jul-16	9	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$74	-\$14,458	9
udget	Aug-16	10	0	95	0	0	0	0	0	0	235	74	0	\$404	0	\$0	\$0	\$0	\$0	\$404	-\$404	-\$75	-\$14,937	10
udget	Sep-16	11	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$78	-\$15,184	11
udget	Oct-16	12 13	0	95 95	0	137	3,684	0	0	0	0	74 74	145	\$4,134	0	\$0 \$0	\$0 \$0	\$0	\$0	\$4,134	-\$4,134	-\$79 -\$101	-\$19,397	12
udget	Nov-16	13	0	95	0	0	3,684 3.684	0	0	0	0	74	145 145	\$3,998 \$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998 \$3,998	-\$3,998 -\$3,998	-\$101	-\$23,496 -\$27,616	14
udget	Dec-16 Jan-17	14	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998	-\$3,998	-\$122	-\$27,616	1
idget idget	Feb-17	15	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998	-\$3,998	-\$144	-\$35,920	1
udget	Mar-17	17	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$187	-\$40,104	1
udget	Apr-17	18	ő	95	ő	0	3.684	ő	0	0	ő	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$209	-\$44.311	1
udget	May-17	19	ő	95	ő	ő	3,684	ő	ő	0	ő	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$231	-\$48,539	1
udget	Jun-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$253	-\$52,789	2
udget	Jul-17	21	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$275	-\$57,062	2
udget	Aug-17	22	0	95	0	0	3,684	0	0	0	235	74	145	\$4,233	0	\$0	\$0	\$0	\$0	\$4,233	-\$4,233	-\$297	-\$61,592	2
udget	Sep-17	23	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$321	-\$65,910	2
udget	Oct-17	24	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$343	-\$70,251	2
udget	Nov-17	25	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$366	-\$74,614	2
udget	Dec-17	26	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$389	-\$79,000	2
idget	Jan-18	27	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$411	-\$83,409	2
idget	Feb-18	28	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$434	-\$87,841	2
dget	Mar-18	29	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$458	-\$92,296	2
idget	Apr-18	30	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$481	-\$96,774	3
idget	May-18	31	0	95	0	0	3,684	2,206	1,932	1	0	74	0	\$7,992	0	\$0	\$0	\$0	\$0	\$7,992	-\$7,992	-\$504	-\$105,270	3
Idget	Jun-18	32	0	0	0	0	0	0	0	0	U	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$548	-\$105,818	3
udget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$551	-\$106,369	3
udget	Aug-18 Sep-18	34 35	0	0	0	0	0	0	0	0	235	0	0	\$235 \$0	70 0	\$137,767 \$0	\$11,491 \$0	\$2,884 \$0	\$123,392 \$0	\$235 \$0	\$123,157 \$0	-\$554 \$0	\$16,233 \$16,233	3
udget udget	Oct-18	35	0	0	0	0	0	0	0	0	0	0	0	\$0 \$0	0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$16,233 \$16,233	36
	l	l	12,164	\$2,947	\$353	\$137	\$73,675	\$2,206	\$1,932	\$1	\$705	\$2,210	\$2,755	\$99,085	70	\$137,767	\$11,491	\$2,884	\$123,392	\$99,085	\$24,307	-\$8,073	\$16,233	+
			Land	Professional	DA Fees	CC Fees	Construction			LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
			1	Fees			Costs	Costs			Costs		5	Costs		Revenue		-	Revenue	Outlays		Interest		

Sheet 4 of 4

gln

2.5 Scenario D

Hypothetical development to achieve 6.9:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 50%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 2:1 control to a proposed 6:1 FSR, resulting in an FSR uplift of 4:1.



Scenario D - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	6.9
Land Area	3,000
Site Value	66,110
Number of Dwellings	172

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$696,802	\$137,766,667
	****;**=	÷···;···;···
GST	66,805	11,490,515
Less Selling Costs	16,769	2,884,333
NET REVENUE	\$717,394	\$123,391,818
COSTS		
Land (including acquisition costs)	66,110	11,371,000
	00,110	11,371,000
Acquisition costs	4,610	792,841
	.,	
Construction	428,343	73,675,042
Consultants	17,134	2,947,002
Section 94A - Commercial	3,125	537,500
Section 94A - Residential	0 600	1 669 254
Section 94A - Residential	9,699	1,668,251
Value Sharing Contribution	28,081	4,830,000
Value channy contribution	20,001	4,000,000
Statutory Fees & Contributions	6,953	1,195,883
On Costs	12,850	2,210,251
Manufaction of	40.040	0.755.000
Marketing	16,019	2,755,333
Cost before Interest	564,844	97,153,104
	····	37,100,104
Finance (incl Loan Est Fees)	47,202	8,118,822
TOTAL DEVELOPMENT COSTS	612,046	\$105,271,926
TOTAL PROJECT SURPLUS	12.6%	\$13,289,893
PROJECT IRR BEFORE INTEREST	16.4%	

Scenario D - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy Input Sheet

SITE DETAILS		
Address	Repi	resentative
Floor Space Ratio	6.9	:1
Achievable FSR	20,700	m2
Land Area	3,000	m2
Site Value	66,110	
Any 1 parcel over \$3M	Yes	
LVR (Debt/Equity)	100%	
Date to start feasibility	Oct-15	



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		11,371,000		
Loan Establishment C	osts	16,583		
Stamp Duty		736,460		
Legals	0.4%	39,799		
LAND ACQUISITION C	COSTS - Deposit	1,137,100	1	1
LAND ACQUISITION C	OSTS - Settlement	11,026,741	2	1
CONSTRUCTION COS	TS	73,675,042	12	20
PROFESSIONAL FEE	S 4% const costs	2,947,002	1	31
APPLICATION FE	ES			
DA Fees		352,527	3	1
CC Fees		136,966	12	1
Section 94		2,205,751	31	1
Value Sharing		4,830,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	235,023	10	1
Land Tax/Rates	Year 2	235,023	22	1
Land Tax/Rates	Year 3	235,023	34	1
ONCOSTS	3.0% construction cost	2,210,251	2	30
MARKETING	2% gross revenue	2,755,333	12	19
	TOTAL COST	101,983,104		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	29	60	2,001	29	Yes	590,000	17,110,000	17%	15%
2 bedroom	131	80	12,052	131	Yes	700,000	91,700,000	76%	75%
3 bedroom	12	120	1,656	12	Yes	920,000	11,040,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	172		20,709	280	Yes		137,766,667	100%	100%
Average m2/unit	79								
ADDITIONAL INCLUS	SIONS								
Other visitor parking				34					
TOTAL YIELD	172		20,709	315			137,766,667		
		FSR	6.9			_			
		Target Floorspace	20,700	1					

Scenario D - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		137,766,667
GST		11,490,515
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	2,755,333
Legals (On Settlement)	\$750 per lot	129,000
TOTAL - SELLING COSTS		2,884,333
NET REVENUE		123,391,818



STAMP DUT	Y	
LAND VALU	E THRESHOLD	ΤΑΧ
Minimum	Maximum	
14,000		0
14,000	30,000	0
30,000	80,000	0
80,000	300,000	0
300,000	1,000,000	0
1,000,000		FALSE
3,000,000	Premium Property Tax	736,460
	TOTAL	736,460

CONTRIBUTIO	ONS	
SECTION 94A		3.0%
COSTS		
Retail	10,416,667	\$312,500
Commercial	7,500,000	\$225,000
Residential	55,608,375	\$1,668,251
TOTAL VALUE	73,525,042	
TOTAL CONT	RIBUTION	\$2,205,751

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		94,665
LSL	0.35%	257,863
TOTAL DA FEES		352,527

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	136,966
TOTAL CC FEES	136,966

CONSTRUCT	ION COSTS	3		
Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	44,770,650	8,600,000	
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			1,720,000	C
ESD Costs	0	0	-	
Design Comp		150,000		
Design Cost	25	517,725		
TOTALS		57,938,375	15,736,667	0
	TOTAL CO	NSTRUCTIC	ON COST	73,675,042

CAR PARK	NG
Unit Type	No. Spaces
1 bed/studio	1.0
2 bed	1.0
3+ bed	1.0
Shops	30
Commercial	100
Visitor parking	In any case, min 1 visitor space is required

RATES	
Council Rates	24,979
Land Tax	210,044
TOTAL RATES	235,023

LPI FEES			
Strata	Base Fee	Per Dwg fee 132	
TOTAL LPI F	1,321 EES	132	1,321

VALUE SHARING		
GFA Element	GFA	Base Value
4:1 CBD Strategy	12,000	\$9,660,000
3:1 Above Strategy	0	\$0
Calculation		
Rate/additional m2 GFA		\$805
CBD Strategy Rate		50%
Above CBD Strategy Rate		0%
TOTAL CONTRIBUTION		\$4,830,000

Sheet 3 of 4

Scenario D - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	16.40%
Surplus	\$13,289,893
% Surplus on D.C	12.07%
Max Loan Balance Debt/Equity	-\$109,297,644
Debt/Equity	100.00%

Budget/	Oct-15	Month	Land	Professional	DA Fees	CC Fees	Construction	Section 94	Value	LPI Fess	Statuory	Oncosts	Marketing	Total	Sales	Gross	GST	Selling Costs	Net	Monthly	Net	Monthly	Total	Period
Actual	Oct-15	Month	Lano	Fees	DA rees	CC Fees	Costs	Costs	Sharing	LFIFess	Costs	Uncosts	Marketing	Costs	Sales	Revenue	651	Selling Costs	Revenue	Outlays	Cashflow		Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	1,137	95	0	0	0	0	0	0	0	0	0	\$1,232	0	\$0 \$0	\$0 \$0	\$0	\$0	\$1,232	-\$1,232	\$0	-\$1,232	1
Budget	Dec-15	2	11,027	95	0	0	0	0	0	0	0	74	0	\$11.195	0	\$0	\$0	\$0	\$0	\$11.195	-\$11,195	-\$6	-\$12,434	2
Budget	Jan-16	3	0	95	353	ő	ő	ŏ	ŏ	0 0	ŏ	74	ŏ	\$521	ŏ	\$0	\$0	\$0	\$0	\$521	-\$521	-\$65	-\$13,020	3
Budget	Feb-16	4	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$68	-\$13,257	4
Budget	Mar-16	5	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$69	-\$13,494	5
Budget	Apr-16	6	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$70	-\$13,733	6
Budget	May-16	7	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$72	-\$13,974	7
Budget	Jun-16	8	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$73	-\$14,215	8
Budget	Jul-16	9	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$74	-\$14,458	9
Budget	Aug-16	10	0	95	0	0	0	0	0	0	235	74	0	\$404	0	\$0	\$0	\$0	\$0	\$404	-\$404	-\$75	-\$14,937	10
Budget	Sep-16	11	0	95	0	0	0	0	0	0	0	74	0	\$169	0	\$0	\$0	\$0	\$0	\$169	-\$169	-\$78	-\$15,184	11
Budget	Oct-16	12	0	95	0	137	3,684	0	0	0	0	74	145	\$4,134	0	\$0	\$0	\$0	\$0	\$4,134	-\$4,134	-\$79	-\$19,397	12
Budget	Nov-16	13	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$101	-\$23,496	13
Budget	Dec-16	14	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$122	-\$27,616	14
Budget	Jan-17	15 16	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0 \$0	\$0	\$0	\$3,998	-\$3,998	-\$144	-\$31,757	15
Budget	Feb-17	16	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0 \$0	\$0	\$0	\$3,998	-\$3,998	-\$165	-\$35,920	16
Budget	Mar-17	17	0	95	0	0	3,684 3,684	0	0	0	0	74 74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0	\$0	\$3,998	-\$3,998	-\$187	-\$40,104 -\$44,311	17
Budget Budget	Apr-17 May-17	10	0	95 95	0	0	3,684	0	0	0	0	74	145 145	\$3,998 \$3,998	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998 \$3,998	-\$3,998 -\$3,998	-\$209 -\$231	-\$48,539	18 19
Budget	Jun-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0 \$0	\$0 \$0	\$0	\$0	\$3,998	-\$3,998	-\$253	-\$46,535	20
Budget	Jul-17	20	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$3,998	-\$3,998	-\$275	-\$57.062	20
Budget	Aug-17	22	0	95	ő	0	3,684	0	0	0	235	74	145	\$4,233	0	\$0	\$0	\$0	\$0	\$4,233	-\$4,233	-\$297	-\$61,592	22
Budget	Sep-17	23	ő	95	ő	0	3.684	ő	0	0	0	74	145	\$3,998	ő	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$321	-\$65,910	23
Budget	Oct-17	24	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$343	-\$70,251	24
Budget	Nov-17	25	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$366	-\$74,614	25
Budget	Dec-17	26	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$389	-\$79,000	26
Budget	Jan-18	27	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$411	-\$83,409	27
Budget	Feb-18	28	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$434	-\$87,841	28
Budget	Mar-18	29	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$458	-\$92,296	29
Budget	Apr-18	30	0	95	0	0	3,684	0	0	0	0	74	145	\$3,998	0	\$0	\$0	\$0	\$0	\$3,998	-\$3,998	-\$481	-\$96,774	30
Budget	May-18	31	0	95	0	0	3,684	2,206	4,830	1	0	74	0	\$10,890	0	\$0	\$0	\$0	\$0	\$10,890	-\$10,890	-\$504	-\$108,168	31
Budget	Jun-18	32	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$563	-\$108,731	32
Budget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$566	-\$109,298	33
udget	Aug-18	34	0	0	0	0	0	0	0	0	235	0	0	\$235	70	\$137,767	\$11,491	\$2,884	\$123,392	\$235	\$123,157	-\$569	\$13,290	34
Budget	Sep-18	35	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$13,290	35
Budget	Oct-18	36	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$13,290	36
		1	12,164	\$2,947	\$353	\$137	\$73,675	\$2,206	\$4,830	\$1	\$705	\$2,210	\$2,755	\$101,983	70	\$137,767	\$11,491	\$2,884	\$123,392	\$101,983	\$21,409	-\$8,119	\$13,290	
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
			1	Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

gln

2.6 12:1 FSR Base Scenario

Hypothetical development to achieve 12:1 FSR GFA outcome with application of S94A at 3% of cost of construction. Whilst the draft Planning Proposal will typically apply a 10:1 FSR control over the majority of the CBD area, it will also provide for additional incentive FSR of:

- 1.5:1 FSR for 'Design Excellence' that is required for any building in excess of 40 metres in height; and
- 0.5:1 for 'High Performance Buildings'.

Council considers it unlikely that developers will not choose to take up the additional FSR incentives given the likelihood that 10:1 developments would exceed the 40 metre height limit and that the costs of achieving their 'High Performance Building' criteria are not overly onerous and are offset by the increase in GFA permitted.



Base Scenario - S94A Levy 3% on 12:1 FSR GFA

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	12.0
Land Area	3,000
Site Value	78,551
Number of Dwellings	338

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$) Ex Retail	(\$) Inc Retail
	EX Retail	Inc Retail
GROSS REVENUE	\$698,047	\$253,856,667
	• • • • • •	
GST	61,137	20,664,210
Less Selling Costs	15,771	5,330,633
NET REVENUE	\$674,147	\$227,861,823
COSTS		
Land (including convicition costs)	70 554	20 550 254
Land (including acquisition costs)	78,551	26,550,354
Acquisition costs	5,673	1,917,631
	5,015	1,317,031
Construction	379,488	128,266,993
Consultants	15,180	5,130,680
Section 94A - Commercial	1,590	537,500
Section 94A - Residential	9,781	3,306,010
Value Sharing Contribution	٥	0
Value Sharing Contribution	0	0
Statutory Fees & Contributions	7,771	2,626,602
	- ,	_,,
On Costs	11,385	3,848,010
Marketing	15,021	5,077,133
Cost before Interest	524,441	177,260,913
Finance (incl.) con Est Esco)	45 404	45 270 004
Finance (incl Loan Est Fees)	45,494	15,376,981
TOTAL DEVELOPMENT COSTS	569,935	\$192,637,893
TOTAL DEVELOPMENT COSTS		
TOTAL PROJECT SURPLUS	18.3%	\$35,223,930
	10.0 /0	
PROJECT IRR BEFORE INTEREST	20.0%	

Base Scenario - S94A Levy 3% on 12:1 FSR GFA Input Sheet

SITE DETAILS						
Address	Represer	ntative				
Floor Space Ratio	12.0 :1					
Achievable FSR	36,000 m2					
Land Area	3,000 m2					
Site Value	78,551					
Any 1 parcel over \$3M	Yes					
LVR (Debt/Equity)	100%					
Date to start feasibility	Oct-15					



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,354		
Loan Establishment C	osts	25,690		
Stamp Duty		1,799,015		
Legals	0.4%	92,926		
LAND ACQUISITION C	OSTS - Deposit	2,655,035	1	1
LAND ACQUISITION C	OSTS - Settlement	25,812,950	2	1
CONSTRUCTION COS	TS	128,266,993	12	20
PROFESSIONAL FEES	6 4% const costs	5,130,680	1	31
APPLICATION FE	ES			
DA Fees		608,563	3	1
CC Fees		236,224	12	1
Section 94		3,843,510	31	1
Value Sharing		0	31	1
LPI Fees		66,908	31	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	571,636	10	1
Land Tax/Rates	Year 2	571,636	22	1
Land Tax/Rates	Year 3	571,636	34	1
ONCOSTS	3.0% construction cost	3,848,010	2	30
MARKETING	2% gross revenue	5,077,133	12	19
	TOTAL COST	177,260,913		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	Ū.	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	56	60	3,864	56	Yes	590,000	33,040,000	17%	15%
2 bedroom	257	80	23,644	257	Yes	700,000	179,900,000	76%	75%
3 bedroom	25	120	3,450	25	Yes	920,000	23,000,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	338		35,958	446	Yes		253,856,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				68					
TOTAL YIELD	338		35,958	514			253,856,667		
		FSR	12.0						
		Target Floorspace	36,000	1					

Base Scenario - S94A Levy 3% on 12:1 FSR GFA

Calculations

REVENUE		
GROSS REVENUE		253,856,667
GST		20,664,210
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	5,077,133
Legals (On Settlement)	\$750 per lot	253,500
TOTAL - SELLING COSTS		5,330,633
NET REVENUE		227,861,823



STAMP DUTY							
LAND VALUE THRESHOLD TAX							
Minimum							
14,000		0					
14,000	30,000	0					
30,000	80,000	0					
80,000	300,000	0					
300,000	1,000,000	0					
1,000,000		FALSE					
3,000,000	Premium Property Tax	1,799,015					
TOTAL 1,799,015							

CONTRIBUTIONS							
SECTION 94A		3.0%					
COSTS							
Retail	10,416,667	\$312,500					
Commercial	7,500,000	\$225,000					
Residential	110,200,326	\$3,306,010					
TOTAL VALUE							
TOTAL CONT	\$3,843,510						

DA FEES							
CONSTRUCTION COST THRESHOLDS							
50,001	250,000	0					
250,001	500,000	0					
500,001	1,000,000	0					
1,000,001	10,000,000	0					
More than \$10,000,000		159,629					
LSL	0.35%	448,934					
TOTAL DA FEES		608,563					

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	236,224
TOTAL CC FEES	236,224

CONSTRUCTION COSTS								
Build Costs	Rates	FSR +15%	UG Parking	Grade				
	\$/m2		50,000	Parking				
Apartments	2,850	88,230,300	16,900,000					
Retail	2,500	6,250,000	4,166,667					
Commercial	2,500	6,250,000	1,250,000					
Visitor Parking			3,380,000	0				
ESD Costs	22	791,076						
Design Comp		150,000						
Design Cost	25	898,950						
TOTALS		102,570,326	25,696,667	0				
	128,266,993							

CAR PARKING								
Unit Type	No. Spaces							
1 bed/studio	1.0							
2 bed	1.0							
3+ bed	1.0							
Shops	30							
Commercial	100							
Visitor parking	In any case, min 1 visitor space is required							

RATES	
Council Rates	58,004
Land Tax	513,631
TOTAL RATES	571,636

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	66,908		

VALUE SHARING									
GFA Element	GFA	Base Value							
4:1 CBD Strategy	0	\$0							
3:1 Above Strategy	0	\$0							
Calculation									
Rate/additional m2 GFA		\$805							
CBD Strategy Rate		10%							
Above CBD Strategy Rate	50%								
TOTAL CONTRIBUTION		\$0							

Sheet 3 of 4

Base Scenario - S94A Levy 3% on 12:1 FSR GFA

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	20.10%
Surplus	\$35,223,930
% Surplus on D.C	18.29%
Max Loan Balance Debt/Equity	-\$191,071,096
Debt/Equity	100.00%

Max Loa Debt/Equ	n Balan		t/Equity	-\$191,071,096 100.00%				* 000															planr	ning
DeptrEqu	uity			100.00 %	Numbers be	ellow expres	sed in nearest	\$,000															plain	iiiig
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	166	0	0	0	0	0	0	0	0	0	\$2,821	0	\$0	\$0	\$0	\$0	\$2,821	-\$2,821	\$0	-\$2,821	1
Budget	Dec-15	2	25,813	166	0	0	0	0	0	0	0	128	0	\$26,107	0	\$0	\$0	\$0	\$0	\$26,107	-\$26,107	-\$15	-\$28,942	2
Budget	Jan-16	3	0	166	609	0	0	0	0	0	0	128	0	\$902	0	\$0	\$0	\$0	\$0	\$902	-\$902	-\$151	-\$29,995	3
Budget	Feb-16	4	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$156	-\$30,445	4
Budget	Mar-16	5	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$159	-\$30,897	5
Budget	Apr-16	6	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$161	-\$31,352	6
Budget	May-16	7	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$163	-\$31,809	7
Budget	Jun-16	8	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$166	-\$32,269	8
Budget	Jul-16	9 10	0	166 166	0	0	0	0	0	0	0 572	128 128	0	\$294 \$865	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$865	-\$294 -\$865	-\$168 -\$170	-\$32,730	9 10
Budget	Aug-16 Sep-16	10	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$865	-\$865	-\$170	-\$33,766 -\$34,236	10
Budget Budget	Oct-16	12	0	166	0	236	6,413	0	0	0	0	128	267	\$294 \$7,211	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$7,211	-\$294 -\$7,211	-\$176	-\$34,236	12
Budget	Nov-16	13	0	166	0	230	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$178	-\$48,816	12
Budget	Dec-16	14	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$254	-\$56,045	14
Budget	Jan-17	15	0	166	ő	0	6.413	ő	ő	0	ő	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6.974	-\$6,974	-\$292	-\$63,311	15
Budget	Feb-17	16	ő	166	ő	ő	6,413	ŏ	ŏ	0 0	ŏ	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$330	-\$70,615	16
Budget	Mar-17	17	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$368	-\$77,957	17
Budget	Apr-17	18	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$406	-\$85,337	18
Budget	May-17	19	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$444	-\$92,756	19
Budget	Jun-17	20	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$483	-\$100,214	20
Budget	Jul-17	21	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$522	-\$107,710	21
Budget	Aug-17	22	0	166	0	0	6,413	0	0	0	572	128	267	\$7,546	0	\$0	\$0	\$0	\$0	\$7,546	-\$7,546	-\$561	-\$115,817	22
Budget	Sep-17	23	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$603	-\$123,394	23
Budget	Oct-17	24	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$643	-\$131,011	24
Budget	Nov-17	25	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$682	-\$138,668	25
Budget	Dec-17	26	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$722	-\$146,365	26
Budget	Jan-18	27	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$762	-\$154,101	27
Budget	Feb-18	28	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$803	-\$161,878	28
Budget	Mar-18	29 30	0	166	0	0	6,413	0	U	0	0	128 128	267 267	\$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$843	-\$169,696	29 30
Budget	Apr-18 May-18	30	0	166 166	0	0	6,413 6,413	3,844	0	67	0	128	267	\$6,974 \$10,618	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974 \$10,618	-\$6,974 -\$10,618	-\$884 -\$925	-\$177,554 -\$189,096	30
Budget Budget	May-18 Jun-18	31 32	0	100	0	0	6,413	3,844	0	67	0	128	0	\$10,618 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$10,618	-\$10,618 \$0	-\$925 -\$985	-\$189,096 -\$190,081	31 32
Budget	Jul-18	32	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	-\$990	-\$190,081	32
Budget	Aug-18	34	0	0	0	0	0	0	0	0	572	0	0	\$572	70	\$253.857	\$20,664	\$5,331	\$227.862	\$572	\$227,290	-\$995	\$35.224	33
Budget	Sep-18	35	0	0	ő	0	0	0	ŏ	0	0	ő	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$35,224	35
Budget	Oct-18	36	0	0	0	0	ő	0	0	ō	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$35,224	36
	1	I	28,468	\$5,131	\$609	\$236	\$128,267	\$3,844	\$0	\$67	\$1,715	\$3,848	\$5,077	\$177,261	70	\$253,857	\$20,664	\$5,331	\$227,862	\$177,261	\$50,601	-\$15,377	\$35,224	<u> </u>
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

12-1 FSR Base Scenario - Parramatta_S94A 3% on 12-1 FSR GFAxlsx

2.7 **Scenario E**

Hypothetical development to achieve 12:1 FSR GFA outcome with application of S94A at 4.5% of cost of construction.



planning



15

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	12.0
Land Area	3,000
Site Value	78,551
Number of Dwellings	338

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$698,047	\$253,856,667
	φ030,04 <i>1</i>	\$233,030,001
GST	61,137	20,664,210
Less Selling Costs	15,771	5,330,633
	10,111	0,000,000
NET REVENUE	\$674,147	\$227,861,823
COSTS		
Land (including acquisition costs)	78,551	26,550,354
A a mulaitian a a sta	E 070	4 047 004
Acquisition costs	5,673	1,917,631
Construction	379,488	128,266,993
Construction	579,400	120,200,995
Consultants	15,180	5,130,680
Consultants	15,100	3,130,000
Section 94A - Commercial	2,385	806,250
	_,	,
Section 94A - Residential	14,672	4,959,015
	·	
Value Sharing Contribution	0	0
Statutory Fees & Contributions	7,577	2,561,015
	44.005	0.040.040
On Costs	11,385	3,848,010
Marketing	15,021	5,077,133
Marketing	15,021	5,077,155
Cost before Interest	529,932	179,117,081
	020,002	
Finance (incl Loan Est Fees)	45,580	15,406,135
	,	,
TOTAL DEVELOPMENT COSTS	575,512	\$194,523,216
TOTAL PROJECT SURPLUS	17.1%	\$33,338,608
PROJECT IRR BEFORE INTEREST	19.4%	

Input Sheet

SITE DETAILS								
Address	Representative							
Floor Space Ratio	12.0	:1						
Achievable FSR	36,000	m2						
Land Area	3,000	m2						
Site Value	78,551							
Any 1 parcel over \$3M	Yes							
LVR (Debt/Equity)	100%							
Date to start feasibility	Oct-15							



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,354		
Loan Establishment C	osts	25,690		
Stamp Duty		1,799,015		
Legals	0.4%	92,926		
LAND ACQUISITION C	COSTS - Deposit	2,655,035	1	1
LAND ACQUISITION C	COSTS - Settlement	25,812,950	2	1
CONSTRUCTION COS	TS	128,266,993	12	20
PROFESSIONAL FEE	S 4% const costs	5,130,680	1	31
APPLICATION FE	ES			
DA Fees		608,563	3	1
CC Fees		236,224	12	1
Section 94		5,765,265	31	1
Value Sharing		0	31	1
LPI Fees		1,321	31	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	571,636	10	1
Land Tax/Rates	Year 2	571,636	22	1
Land Tax/Rates	Year 3	571,636	34	1
ONCOSTS	3.0% construction cost	3,848,010	2	30
MARKETING	2% gross revenue	5,077,133	12	19
	TOTAL COST	179,117,081		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	-	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL	-								
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	56	60	3,864	56	Yes	590,000	33,040,000	17%	15%
2 bedroom	257	80	23,644	257	Yes	700,000	179,900,000	76%	75%
3 bedroom	25	120	3,450	25	Yes	920,000	23,000,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	338		35,958	446	Yes		253,856,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				68					
TOTAL YIELD	338		35,958	514			253,856,667		
		FSR	12.0						
		Target Floorspace	36,000						

Calculations

REVENUE		
GROSS REVENUE		253,856,667
GST		20,664,210
SELLING COSTS		-
Sales Commission (On Settlement)	2% on gross rev	5,077,133
Legals (On Settlement)	\$750 per lot	253,500
TOTAL - SELLING COSTS		5,330,633
NET REVENUE		227,861,823



STAMP DUTY									
LAND VALUE	ΤΑΧ								
Minimum									
14,000		0							
14,000	30,000	0							
30,000	0,000 80,000								
80,000	80,000 300,000								
300,000	1,000,000	0							
1,000,000		FALSE							
3,000,000	1,799,015								
	3,000,000 Premium Property Ta> TOTAL								

CONTRIBUTIONS								
SECTION 94A		4.5%						
COSTS								
Retail	\$468,750							
Commercial	7,500,000	\$337,500						
Residential	\$4,959,015							
TOTAL VALUE								
TOTAL CONT	\$5,765,265							

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		159,629
LSL	0.35%	448,934
TOTAL DA FEES		608,563

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	236,224
TOTAL CC FEES	236,224

CONSTRUCTION COSTS												
Build Costs	Rates	FSR +15%	UG Parking	Grade								
	\$/m2		50,000	Parking								
Apartments	2,850	88,230,300	16,900,000									
Retail	2,500	6,250,000	4,166,667									
Commercial	2,500	6,250,000	1,250,000									
Visitor Parking			3,380,000	0								
ESD Costs	22	791,076										
Design Comp		150,000										
Design Cost	25	898,950										
TOTALS		102,570,326	25,696,667	0								
	128,266,993											

Unit Type	No. Spaces							
1 bed/studio	1.0							
2 bed	1.0							
3+ bed	1.0							
Shops	30							
Commercial	100							
Visitor parking	In any case, min 1 visitor space is required							

RATES	
Council Rates	58,004
Land Tax	513,631
TOTAL RATES	571,636

LPI FEES							
Strata	Base Fee	Per Dwg fee					
	1,321	132					
TOTAL LPI F	TOTAL LPI FEES						

VALUE SHARING		
GFA Element	GFA	Base Value
4:1 CBD Strategy	0	\$0
3:1 Above Strategy	0	\$0
Calculation		
Rate/additional m2 GFA		\$0
CBD Strategy Rate	0%	
Above CBD Strategy Rate	0%	
TOTAL CONTRIBUTION		\$0

Sheet 3 of 4

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	19.38%
Surplus	\$33,338,608
% Surplus on D.C	17.14%
Max Loan Balance Debt/Equity	-\$192,946,650
Debt/Equity	100.00%

	a Balan		/Equity	-\$192,946,650																				
ebt/Equ		ce Debi	UEquity	100.00%		ellow express	sed in nearest	\$,000													<		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	166	0	0	ő	ő	0	ő	ő	ő	ő	\$2,821	0	\$0	\$0	\$0	\$0	\$2,821	-\$2,821	\$0	-\$2,821	1
Budget	Dec-15	2	25,813	166	0	0	0	0	0	0	0	128	0	\$26,107	0	\$0	\$0	\$0	\$0	\$26,107	-\$26,107	-\$15	-\$28,942	2
Budget	Jan-16	3	0	166	609	0	0	0	0	0	0	128	0	\$902	0	\$0	\$0	\$0	\$0	\$902	-\$902	-\$151	-\$29,995	3
Budget	Feb-16	4	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$156	-\$30,445	4
Budget	Mar-16	5	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$159	-\$30,897	5
Budget	Apr-16	6	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$161	-\$31,352	6
Budget	May-16	7	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$163	-\$31,809	7
Budget	Jun-16	8	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$166	-\$32,269	8
Budget	Jul-16	9	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$168	-\$32,730	9
Budget	Aug-16	10	0	166	0	0	0	0	0	0	572	128	0	\$865	0	\$0	\$0	\$0	\$0	\$865	-\$865	-\$170	-\$33,766	10
Budget	Sep-16	11	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$176	-\$34,236	11
Budget	Oct-16	12 13	0	166	0	236 0	6,413	0	0	0	0	128	267 267	\$7,211	0	\$0	\$0	\$0	\$0	\$7,211	-\$7,211	-\$178	-\$41,625	12
Budget	Nov-16	13	0	166 166	0	0	6,413 6,413	0	0	0	0	128 128	267	\$6,974 \$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974 \$6,974	-\$6,974 -\$6,974	-\$217 -\$254	-\$48,816	13 14
Budget	Dec-16 Jan-17	14	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$254 -\$292	-\$56,045 -\$63.311	14
Budget Budget	Feb-17	15	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$292	-\$63,311	15
Budget	Mar-17	17	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0 \$0	\$0	\$6,974	-\$6,974	-\$368	-\$77,957	17
Budget	Apr-17	18	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$406	-\$85.337	18
Budget	May-17	19	0	166	0	0	6,413	0	0	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$444	-\$92,756	19
Budget	Jun-17	20	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$483	-\$100,214	20
Budget	Jul-17	21	0	166	0	0	6.413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$522	-\$107,710	21
Budget	Aug-17	22	0	166	0	0	6,413	0	0	0	572	128	267	\$7,546	0	\$0	\$0	\$0	\$0	\$7,546	-\$7,546	-\$561	-\$115,817	22
Budget	Sep-17	23	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$603	-\$123,394	23
Budget	Oct-17	24	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$643	-\$131,011	24
Budget	Nov-17	25	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$682	-\$138,668	25
Budget	Dec-17	26	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$722	-\$146,365	26
Budget	Jan-18	27	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$762	-\$154,101	27
Budget	Feb-18	28	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$803	-\$161,878	28
Budget	Mar-18	29	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$843	-\$169,696	29
Budget	Apr-18	30	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$884	-\$177,554	30
Budget	May-18	31	0	166	0	0	6,413	5,765	0	1	0	128	0	\$12,474	0	\$0	\$0	\$0	\$0	\$12,474	-\$12,474	-\$925	-\$190,952	31
Budget	Jun-18	32	0	U	0	U	0	0	0	0	0	0	0	\$0 \$0	0	\$0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$0 ©0	-\$995	-\$191,947	32
Budget	Jul-18	33	0	U	0	0	0	U	0	0	-	0	0	\$0	0 70	\$0	\$0	\$0	\$0	\$0 \$570	\$0	-\$1,000	-\$192,947	33
Budget Budget	Aug-18 Sep-18	34 35	0	0	0	0	0	0	0	0	572 0	0	0	\$572 \$0	70	\$253,857 \$0	\$20,664 \$0	\$5,331 \$0	\$227,862 \$0	\$572 \$0	\$227,290 \$0	-\$1,005 \$0	\$33,339 \$33,339	34 35
Budget	Oct-18	36	0	0	0	0	0	0	0	0	0	0	0	\$0 \$0	0	\$0 \$0	\$0	\$0 \$0	\$0	\$0 \$0	\$0	\$0	\$33,339	36
		1	28,468	\$5,131	\$609	\$236	\$128,267	\$5,765	\$0	\$1	\$1,715	\$3,848	\$5,077	\$179,117	70	\$253,857	\$20,664	\$5,331	\$227,862	\$179,117	\$48,745	-\$15,406	\$33,339	
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

2.8 Scenario F

Hypothetical development to achieve 12:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 10%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 6:1 control to a proposed 10:1 FSR, resulting in an FSR uplift of 4:1.




Scenario F - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	12.0
Land Area	3,000
Site Value	78,550
Number of Dwellings	338

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
	¢600.047	¢050.050.007
GROSS REVENUE	\$698,047	\$253,856,667
GST	61,137	20 664 242
Less Selling Costs	15,771	20,664,242 5,330,633
Less Senning Costs	15,771	5,550,055
NET REVENUE	\$674,147	\$227,861,791
COSTS		
Land (including acquisition costs)	78,550	26,550,000
Acquisition costs	5,673	1,917,605
Construction	379,488	128,266,993
Consultants	15,180	5,130,680
Section 94A - Commercial	1,590	537,500
Section 94A - Residential	9,781	3,306,010
Value Sharing Contribution	2,858	966,000
Statutory Fees & Contributions	7,577	2,560,992
On Costs	11,385	3,848,010
Marketing	15,021	5,077,133
Cost before Interest	524,245	177,194,922
Finance (incl Loan Est Fees)	45,536	15,391,053
TOTAL DEVELOPMENT COSTS	569,781	\$192,585,975
TOTAL PROJECT SURPLUS	17.8%	\$34,309,816
PROJECT IRR BEFORE INTEREST	19.7%	

Scenario F - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy Input Sheet

SITE DETAILS			
Address	Repr	Representative	
Floor Space Ratio	12.0	12.0 :1	
Achievable FSR	36,000	m2	
Land Area	3,000	m2	
Site Value	78,550		
Any 1 parcel over \$3M	Yes		
LVR (Debt/Equity)	100%		
Date to start feasibility	Oct-15		



Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment Co	sts	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION CO	STS - Deposit	2,655,000	1	1
LAND ACQUISITION CO	STS - Settlement	25,812,605	2	1
CONSTRUCTION COST	5	128,266,993	12	20
PROFESSIONAL FEES	4% const costs	5,130,680	1	31
APPLICATION FEE	S			
DA Fees		608,563	3	1
CC Fees		236,224	12	1
Section 94		3,843,510	31	1
Value Sharing		966,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	34	1
ONCOSTS	3.0% construction cost	3,848,010	2	30
MARKETING	2% gross revenue	5,077,133	12	19
	TOTAL COST	178,160,922		

UNIT MIX AND SALE	ES								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	Ū.	Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL						-	•		
1 bedroom	56	60	3,864	56	Yes	590,000	33,040,000	17%	15%
2 bedroom	257	80	23,644	257	Yes	700,000	179,900,000	76%	75%
3 bedroom	25	120	3,450	25	Yes	920,000	23,000,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	338		35,958	446	Yes		253,856,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				68					
TOTAL YIELD	338		35,958	514			253,856,667		
	-	FSR	12.0			-			-
		Target Floorspace	36,000	1					

Sheet 2 of 4

Scenario F - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		253,856,667
GST		20,664,242
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	5,077,133
Legals (On Settlement)	\$750 per lot	253,500
TOTAL - SELLING COSTS		5,330,633
NET REVENUE		227,861,791



STAMP DUTY			
LAND VALUE	E THRESHOLD	ΤΑΧ	
Minimum	Maximum		
14,000		0	
14,000	30,000	0	
30,000	80,000	0	
80,000	300,000	0	
300,000	1,000,000	0	
1,000,000		FALSE	
3,000,000	Premium Property Tax	1,798,990	
	TOTAL	1,798,990	

CONTRIBUTIO	ONS	
SECTION 94A		3.0%
COSTS		
Retail	10,416,667	\$312,500
Commercial	7,500,000	\$225,000
Residential	110,200,326	\$3,306,010
TOTAL VALUE		
TOTAL CONTRIBUTION		\$3,843,510

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		159,629
LSL	0.35%	448,934
TOTAL DA FEES		608,563

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	236,224
TOTAL CC FEES	236,224

CONSTRUCTION COSTS				
Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	88,230,300	16,900,000	C
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			3,380,000	C
ESD Costs	22	791,076	-	
Design Comp		150,000		
Design Cost	25	898,950		
TOTALS		102,570,326	25,696,667	0
	TOTAL CONSTRUCTION COST			128,266,993

CAR PARKING		
Unit Type	No. Spaces	
1 bed/studio	1.0	
2 bed	1.0	
3+ bed	1.0	
Shops	30	
Commercial	100	
Visitor parking	In any case, min 1 visitor space is required	

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	1,321		

VALUE SHARING		
GFA Element	GFA	Base Value
4:1 CBD Strategy	12,000	\$9,660,000
3:1 Above Strategy	0	\$0
Calculation		
Rate/additional m2 GFA		\$805
CBD Strategy Rate		10%
Above CBD Strategy Rate		0%
TOTAL CONTRIBUTION	\$966,000	

Sheet 3 of 4

Scenario F - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA of CBD Strategy

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	19.75%
Surplus	\$34,309,816
% Surplus on D.C	17.73%
Max Loan Balance Debt/Equity	-\$191,980,449
Debt/Equity	100.00%

/lax Loai Debt/Equ		ce Deb	t/Equity	-\$191,980,449 100.00%	Numbers be	ellow express	sed in nearest	\$,000													~		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	166	0	0	0	0	0	0	0	0	0	\$2,821	0	\$0	\$0	\$0	\$0	\$2,821	-\$2,821	\$0	-\$2,821	1
Budget	Dec-15	2	25,813	166	0	0	0	0	0	0	0	128	0	\$26,106	0	\$0	\$0	\$0	\$0	\$26,106	-\$26,106	-\$15	-\$28,942	2
Budget	Jan-16	3	0	166	609	0	0	0	0	0	0	128	0	\$902	0	\$0	\$0	\$0	\$0	\$902	-\$902	-\$151	-\$29,995	3
Budget	Feb-16	4	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$156	-\$30,445	4
Budget	Mar-16	5	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$159	-\$30,897	5
Budget	Apr-16	6	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$161	-\$31,352	6
Budget	May-16	7	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$163	-\$31,809	7
Budget	Jun-16	8	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$166	-\$32,268	8
Budget	Jul-16	9 10	0	166 166	0	0	0	0	0	0	0 572	128 128	0	\$294 \$865	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$865	-\$294 -\$865	-\$168 -\$170	-\$32,730 -\$33,766	9 10
Budget Budget	Aug-16 Sep-16	11	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$294	-\$294	-\$176	-\$34,236	11
Budget	Oct-16	12	0	166	0	236	6.413	0	0	0	0	128	267	\$7,211	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$7.211	-\$7,211	-\$178	-\$41,624	12
Budget	Nov-16	13	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$217	-\$48,816	13
Budget	Dec-16	14	0	166	0	0	6,413	ő	ő	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$254	-\$56,044	14
Budget	Jan-17	15	0	166	0	0	6.413	ő	ő	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$292	-\$63,310	15
Budget	Feb-17	16	ō	166	ō	ō	6,413	0	ō	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$330	-\$70,614	16
Budget	Mar-17	17	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$368	-\$77,957	17
Budget	Apr-17	18	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$406	-\$85,337	18
Budget	May-17	19	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$444	-\$92,756	19
Budget	Jun-17	20	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$483	-\$100,213	20
Budget	Jul-17	21	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$522	-\$107,709	21
Budget	Aug-17	22	0	166	0	0	6,413	0	0	0	572	128	267	\$7,546	0	\$0	\$0	\$0	\$0	\$7,546	-\$7,546	-\$561	-\$115,816	22
Budget	Sep-17	23	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$603	-\$123,394	23
Budget	Oct-17	24	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$643	-\$131,011	24
Budget	Nov-17	25	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$682	-\$138,668	25
Budget	Dec-17	26	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$722	-\$146,364	26
Budget	Jan-18	27	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$762	-\$154,101	27
Budget	Feb-18	28	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$803	-\$161,878	28
Budget	Mar-18	29	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$843	-\$169,695	29
Budget	Apr-18	30 31	0	166	0	0	6,413	-	0	0	0	128	267 0	\$6,974	0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$884 -\$925	-\$177,553	30
Budget Budget	May-18 Jun-18	31 32	0	166	0	0	6,413 0	3,844	966	1	0	128	0	\$11,518 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$11,518 \$0	-\$11,518 \$0	-\$925 -\$990	-\$189,996 -\$190,986	31 32
Budget	Jul-18	32	0	0	0	0	0	0	0	0	0	0	0	\$0 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	-\$990	-\$190,986	32
Budget	Aug-18	33	0	0	0	0	0	0	0	0	572	0	0	\$572	70	\$253,857	\$20,664	\$0 \$5,331	\$0 \$227,862	\$0 \$572	\$0 \$227,290	-\$995	\$34,310	33
Budget	Sep-18	35	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$253,857	\$20,004	\$0	\$0	\$0	\$227,290	\$0	\$34,310	34
Budget	Oct-18	36	0	0	0	0	0	0	õ	ō	0	0	0	\$0	ō	\$0	\$0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0	\$34,310	36
		l	28,468	\$5,131	\$609	\$236	\$128,267	\$3,844	\$966	\$1	\$1,715	\$3,848	\$5,077	\$178,161	70	\$253,857	\$20,664	\$5,331	\$227,862	\$178,161	\$49,701	-\$15,391	\$34,310	-
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

2.9 Scenario G

Hypothetical development to achieve 12:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 20%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 6:1 control to a proposed 10:1 FSR, resulting in an FSR uplift of 4:1.



planning

17

Scenario G - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	12.0
Land Area	3,000
Site Value	78,550
Number of Dwellings	338

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$) Ex Retail	(\$) Inc Retail
	EX Retail	Retail
GROSS REVENUE	\$698,047	\$253,856,667
	<i><i><i>vvvvvvvvvvvvv</i></i></i>	+=00,000,000
GST	61,137	20,664,242
Less Selling Costs	15,771	5,330,633
, i i i i i i i i i i i i i i i i i i i		
NET REVENUE	\$674,147	\$227,861,791
COSTS		
Land (including acquisition costs)	78,550	26,550,000
Acquisition costs	5,673	1,917,605
Construction	270 499	128 266 002
Construction	379,488	128,266,993
Consultants	15,180	5,130,680
oonsulants	10,100	0,100,000
Section 94A - Commercial	1,590	537,500
	.,	
Section 94A - Residential	9,781	3,306,010
Value Sharing Contribution	5,716	1,932,000
Statutory Fees & Contributions	7,577	2,560,992
On Costa	44.205	2 848 848
On Costs	11,385	3,848,010
Marketing	15,021	5,077,133
Marketing	13,021	5,077,155
Cost before Interest	524,245	177,194,922
	0_1,_10	,
Finance (incl Loan Est Fees)	45,581	15,406,225
TOTAL DEVELOPMENT COSTS	569,826	\$192,601,147
TOTAL PROJECT SURPLUS	17.3%	\$33,328,644
PROJECT IRR BEFORE INTEREST	19.4%	

Scenario G - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy Input Sheet

input Sheet

SITE DETAILS					
Address	Representative				
Floor Space Ratio	12.0	:1			
Achievable FSR	36,000	m2			
Land Area	3,000	m2			
Site Value	78,550				
Any 1 parcel over \$3M	Yes				
LVR (Debt/Equity)	100%				
Date to start feasibility	Oct-15				



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment Cos	sts	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION CO	STS - Deposit	2,655,000	1	1
LAND ACQUISITION CO	STS - Settlement	25,812,605	2	1
CONSTRUCTION COSTS	3	128,266,993	12	20
PROFESSIONAL FEES	4% const costs	5,130,680	1	31
APPLICATION FEE	S			
DA Fees		608,563	3	1
CC Fees		236,224	12	1
Section 94		3,843,510	31	1
Value Sharing		1,932,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	34	1
ONCOSTS	3.0% construction cost	3,848,010	2	30
MARKETING	2% gross revenue	5,077,133	12	19
	TOTAL COST	179,126,922		

UNIT MIX AND SALE	S								
		_			Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%		Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL	-						-		
1 bedroom	56	60	3,864	56	Yes	590,000	33,040,000	17%	15%
2 bedroom	257	80	23,644	257	Yes	700,000	179,900,000	76%	75%
3 bedroom	25	120	3,450	25	Yes	920,000	23,000,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	338		35,958	446	Yes		253,856,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				68					
TOTAL YIELD	338		35,958	514			253,856,667		
		FSR	12.0						
		Target Floorspace	36,000						

Scenario G - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		253,856,667
GST		20,664,242
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	5,077,133
Legals (On Settlement)	\$750 per lot	253,500
TOTAL - SELLING COSTS		5,330,633
NET REVENUE		227,861,791



STAMP DUTY							
LAND VALUE THRESHOLD TAX							
Minimum	Maximum						
14,000		0					
14,000	30,000	0					
30,000	80,000	0					
80,000	300,000	0					
300,000	1,000,000	0					
1,000,000		FALSE					
3,000,000	Premium Property Tax	1,798,990					
	TOTAL	1,798,990					

CONTRIBUTIONS								
SECTION 94A		3.0%						
COSTS								
Retail	10,416,667	\$312,500						
Commercial	7,500,000	\$225,000						
Residential	110,200,326	\$3,306,010						
TOTAL VALUE 128,116,993								
TOTAL CONT	\$3,843,510							

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		159,629
LSL	0.35%	448,934
TOTAL DA FEES		608,563

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	236,224
TOTAL CC FEES	236,224

CONSTRUCT	ION COSTS	5		
Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	88,230,300	16,900,000	0
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			3,380,000	0
ESD Costs	22	791,076	-	
Design Comp		150,000		
Design Cost	25	898,950		
TOTALS		102,570,326	25,696,667	0
	128,266,993			

CAR PARKING								
Unit Type	No. Spaces							
1 bed/studio	1.0							
2 bed	1.0							
3+ bed	1.0							
Shops	30							
Commercial	100							
Visitor parking	In any case, min 1 visitor space is required							

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	1,321		

VALUE SHARING						
GFA Element	GFA	Base Value				
4:1 CBD Strategy	12,000	\$9,660,000				
3:1 Above Strategy	\$0					
Calculation						
Rate/additional m2 GFA		\$805				
CBD Strategy Rate		20%				
Above CBD Strategy Rate	0%					
TOTAL CONTRIBUTION	\$1,932,000					

Sheet 3 of 4

Scenario G - S94A Levy 3% PLUS Value Sharing @ 20% of 4:1 GFA of CBD Strategy

Assumed Development Cashflow

DEVELOPMENT SUM	EVELOPMENT SUMMARY				
IRR Before Interest	19.37%				
Surplus	\$33,328,644				
% Surplus on D.C	17.13%				
Max Loan Balance Debt/Equity	-\$192,956,537				
Debt/Equity	100.00%				

Max Loai Debt/Equ		ce Deb	t/Equity	-\$192,956,537 100.00%	Numbers be	ellow express	sed in nearest	\$,000													~		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	166	0	0	0	0	0	0	0	0	0	\$2,821	0	\$0	\$0	\$0	\$0	\$2,821	-\$2,821	\$0	-\$2,821	1
Budget	Dec-15	2	25,813	166	0	0	0	0	0	0	0	128	0	\$26,106	0	\$0	\$0	\$0	\$0	\$26,106	-\$26,106	-\$15	-\$28,942	2
Budget	Jan-16	3	0	166	609	0	0	0	0	0	0	128	0	\$902	0	\$0	\$0	\$0	\$0	\$902	-\$902	-\$151	-\$29,995	3
Budget	Feb-16	4	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$156	-\$30,445	4
Budget	Mar-16	5	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$159	-\$30,897	5
Budget	Apr-16	6	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$161	-\$31,352	6
Budget	May-16	7	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$163	-\$31,809	7
Budget	Jun-16	8	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$166	-\$32,268	8
Budget	Jul-16	9 10	0	166 166	0	0	0	0	0	0	0 572	128 128	0	\$294 \$865	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$865	-\$294 -\$865	-\$168 -\$170	-\$32,730 -\$33,766	9 10
Budget Budget	Aug-16 Sep-16	10	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$294	-\$294	-\$176	-\$34,236	10
Budget	Oct-16	12	0	166	0	236	6.413	0	0	0	0	128	267	\$7,211	0	\$0 \$0	\$0 \$0	\$0	\$0 \$0	\$7.211	-\$7,211	-\$178	-\$41,624	12
Budget	Nov-16	12	0	166	0	230	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$178	-\$48,816	12
Budget	Dec-16	14	0	166	0	0	6,413	ő	0	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$254	-\$56,044	14
Budget	Jan-17	15	0	166	0	0	6.413	ő	0	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$292	-\$63,310	15
Budget	Feb-17	16	ō	166	ō	ō	6,413	0	ō	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$330	-\$70,614	16
Budget	Mar-17	17	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$368	-\$77,957	17
Budget	Apr-17	18	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$406	-\$85,337	18
Budget	May-17	19	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$444	-\$92,756	19
Budget	Jun-17	20	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$483	-\$100,213	20
Budget	Jul-17	21	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$522	-\$107,709	21
Budget	Aug-17	22	0	166	0	0	6,413	0	0	0	572	128	267	\$7,546	0	\$0	\$0	\$0	\$0	\$7,546	-\$7,546	-\$561	-\$115,816	22
Budget	Sep-17	23	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$603	-\$123,394	23
Budget	Oct-17	24	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$643	-\$131,011	24
Budget	Nov-17	25	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$682	-\$138,668	25
Budget	Dec-17	26	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$722	-\$146,364	26
Budget	Jan-18	27	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$762	-\$154,101	27
Budget	Feb-18	28	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$803	-\$161,878	28
Budget	Mar-18	29	0	166	0	0	6,413	0	U	0	0	128	267	\$6,974	0	\$0 \$0	\$0 ©0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$843	-\$169,695	29
Budget	Apr-18	30	0	166	0	0	6,413	-	0	0	0	128	267 0	\$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$6,974	-\$884 -\$925	-\$177,553	30
Budget Budget	May-18 Jun-18	31 32	0	166	0	0	6,413 0	3,844	1,932	1	0	128 0	0	\$12,484 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$12,484 \$0	-\$12,484 \$0	-\$925 -\$995	-\$190,962 -\$191,957	31 32
Budget	Jul-18	32	0	0	0	0	0	0	0	0	0	0	0	\$0 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	-\$995	-\$191,957	32
Budget	Aug-18	33	0	0	0	0	0	0	0	0	572	0	0	\$0 \$572	70	\$253,857	\$0 \$20,664	\$0 \$5,331	\$0 \$227,862	\$0 \$572	\$0 \$227,290	-\$1,000	\$33,329	33
Budget	Sep-18	34	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$255,857	\$20,004	\$0	\$0	\$0	\$227,290	\$0	\$33,329	34
Budget	Oct-18	36	0	ō	0	0	0	0	0	ō	0	0	0	\$0	0	\$0	\$0	\$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0	\$33,329	36
		1	28,468	\$5,131	\$609	\$236	\$128,267	\$3,844	\$1,932	\$1	\$1,715	\$3,848	\$5,077	\$179,127	70	\$253,857	\$20,664	\$5,331	\$227,862	\$179,127	\$48,735	-\$15,406	\$33,329	+
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		-
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

2.10 Scenario H

Hypothetical development to achieve 12:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 50%. In this scenario, the draft CBD Planning Proposal intends to increase the FSR from the current 6:1 control to a proposed 10:1 FSR, resulting in an FSR uplift of 4:1.



planning

Scenario H - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	12.0
Land Area	3,000
Site Value	78,550
Number of Dwellings	338

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$698,047	\$253,856,667
	φ030,041	\$200,000,001
GST	61,137	20,664,242
Less Selling Costs	15,771	5,330,633
3	-,	- , ,
NET REVENUE	\$674,147	\$227,861,791
COSTS		
Land (including acquisition costs)	78,550	26,550,000
	_	_
Acquisition costs	5,673	1,917,605
O an atmostic an	070 400	
Construction	379,488	128,266,993
Consultants	15,180	5,130,680
Consultants	15,100	5,150,080
Section 94A - Commercial	1,590	537,500
	1,550	557,500
Section 94A - Residential	9,631	3,255,309
	-,	-,
Value Sharing Contribution	14,290	4,830,000
-	-	
Statutory Fees & Contributions	7,577	2,560,992
On Costs	11,385	3,848,010
Manlastina	45.004	E 077 400
Marketing	15,021	5,077,133
Cost before Interest	524,095	177,144,221
	027,000	177,177,221
Finance (incl Loan Est Fees)	45,713	15,450,946
	,	,
TOTAL DEVELOPMENT COSTS	569,808	\$192,595,167
TOTAL PROJECT SURPLUS	15.8%	\$30,436,623
PROJECT IRR BEFORE INTEREST	18.3%	

Scenario H - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy Input Sheet

SITE DETAILS				
Address	Repr	Representative		
Floor Space Ratio	12.0	12.0 :1		
Achievable FSR	36,000	m2		
Land Area	3,000	m2		
Site Value	78,550			
Any 1 parcel over \$3M	Yes			
LVR (Debt/Equity)	100%			
Date to start feasibility	Oct-15			



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment Cost	S	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION COS	TS - Deposit	2,655,000	1	1
LAND ACQUISITION COS	TS - Settlement	25,812,605	2	1
CONSTRUCTION COSTS		128,266,993	12	20
PROFESSIONAL FEES	4% const costs	5,130,680	1	31
APPLICATION FEES	3			
DA Fees		608,563	3	1
CC Fees		236,224	12	1
Section 94		3,792,809	31	1
Value Sharing		4,830,000	31	1
LPI Fees		1,321	31	1
LAND TAX/RATES				
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	34	1
ONCOSTS	3.0% construction cost	3,848,010	2	30
MARKETING	2% gross revenue	5,077,133	12	19
	TOTAL COST	181,974,221		

UNIT MIX AND SALE	S								
					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%		Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL						-	-		
1 bedroom	56	60	3,864	56	Yes	590,000	33,040,000	17%	15%
2 bedroom	257	80	23,644	257	Yes	700,000	179,900,000	76%	75%
3 bedroom	25	120	3,450	25	Yes	920,000	23,000,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	338		35,958	446	Yes		253,856,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				68					
TOTAL YIELD	338		35,958	514			253,856,667		
		FSR	12.0						-
		Target Floorspace	36,000	1					

Scenario H - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy Calculations

REVENUE		
GROSS REVENUE		253,856,667
GST		20,664,242
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	5,077,133
Legals (On Settlement)	\$750 per lot	253,500
TOTAL - SELLING COSTS		5,330,633
NET REVENUE		227,861,791



STAMP DUTY				
LAND VALUE THRESHOLD TAX				
Minimum	Ainimum Maximum			
14,000		0		
14,000	30,000	0		
30,000	80,000	0		
80,000	300,000	0		
300,000	1,000,000	0		
1,000,000		FALSE		
3,000,000	1,798,990			
	TOTAL	1,798,990		

CONTRIBUTIONS				
SECTION 94A	3.0%			
COSTS				
Retail	10,416,667	\$312,500		
Commercial	7,500,000	\$225,000		
Residential	108,510,300	\$3,255,309		
TOTAL VALUE 126,426,967				
TOTAL CONT	\$3,792,809			

DA FEES				
CONSTRUCTION COST THRESHOLDS				
50,001	250,000	0		
250,001	500,000	0		
500,001	1,000,000	0		
1,000,001	10,000,000	0		
More than \$10,000,000		159,629		
LSL	0.35%	448,934		
TOTAL DA FEES		608,563		

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	236,224
TOTAL CC FEES	236,224

CONSTRUCTION COSTS						
Build Costs	Rates	FSR +15%	UG Parking	Grade		
	\$/m2		50,000	Parking		
Apartments	2,850	88,230,300	16,900,000	C		
Retail	2,500	6,250,000	4,166,667			
Commercial	2,500	6,250,000	1,250,000			
Visitor Parking			3,380,000	C		
ESD Costs	22	791,076	-			
Design Comp		150,000				
Design Cost	25	898,950				
TOTALS		102,570,326	25,696,667	0		
	ON COST	128,266,993				

CAR PARKING				
Unit Type	No. Spaces			
1 bed/studio	1.0			
2 bed	1.0			
3+ bed	1.0			
Shops	30			
Commercial	100			
Visitor parking	In any case, min 1 visitor space is required			

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	1,321		

VALUE SHARING								
GFA Element	GFA	Base Value						
4:1 CBD Strategy	12,000	\$9,660,000						
3:1 Above Strategy	0	\$0						
Calculation								
Rate/additional m2 GFA		\$805						
CBD Strategy Rate		50%						
Above CBD Strategy Rate	0%							
TOTAL CONTRIBUTION	\$4,830,000							

Sheet 3 of 4

New Scenario H - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA of CBD Strategy

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	18.27%
Surplus	\$30,436,623
% Surplus on D.C	15.42%
Max Loan Balance Debt/Equity	-\$195,833,573
Debt/Equity	100.00%

lax Loai)ebt/Equ		ce Deb	t/Equity	-\$195,833,573 100.00%	Numbers be	ellow express	sed in nearest	\$,000													<		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	166	0	0	0	0	0	0	0	0	0	\$2,821	0	\$0	\$0	\$0	\$0	\$2,821	-\$2,821	\$0	-\$2,821	1
Budget	Dec-15	2	25,813	166	0	0	0	0	0	0	0	128	0	\$26,106	0	\$0	\$0	\$0	\$0	\$26,106	-\$26,106	-\$15	-\$28,942	2
Budget	Jan-16	3	0	166	609	0	0	0	0	0	0	128	0	\$902	0	\$0	\$0	\$0	\$0	\$902	-\$902	-\$151	-\$29,995	3
Budget	Feb-16	4	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$156	-\$30,445	4
Budget	Mar-16	5	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$159	-\$30,897	5
Budget	Apr-16	6	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$161	-\$31,352	6
Budget	May-16	7	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$163	-\$31,809	7
Budget	Jun-16	8	0	166	0	0	0	0	0	0	0	128	0	\$294	0	\$0	\$0	\$0	\$0	\$294	-\$294	-\$166	-\$32,268	8
Budget	Jul-16	9 10	0	166 166	0	0	0	0	0	0	0 572	128 128	0	\$294	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$865	-\$294 -\$865	-\$168 -\$170	-\$32,730	9
Budget	Aug-16 Sep-16	10	0	166	0	0	0	0	0	0	0	128	0	\$865 \$294	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$805	-\$665	-\$170	-\$33,766 -\$34,236	10 11
Budget	Oct-16	12	0	166	0	236	6.413	0	0	0	0	128	267	\$294 \$7,211	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$294 \$7.211	-\$294 -\$7,211	-\$176 -\$178	-\$34,236 -\$41,624	12
Budget Budget	Nov-16	12	0	166	0	230	6,413	0	0	0	0	128	267	\$6,974	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$6,974	-\$7,211	-\$178	-\$48,816	12
Budget	Dec-16	14	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$254	-\$56,044	14
Budget	Jan-17	15	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$292	-\$63,310	15
Budget	Feb-17	16	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0 \$0	\$6,974	-\$6,974	-\$330	-\$70,614	16
Budget	Mar-17	17	ő	166	0	0	6,413	ő	0	ő	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$368	-\$77,957	17
Budget	Apr-17	18	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$406	-\$85,337	18
Budget	May-17	19	0	166	ō	ō	6,413	0	ō	0	ō	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$444	-\$92,756	19
Budget	Jun-17	20	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$483	-\$100,213	20
Budget	Jul-17	21	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$522	-\$107,709	21
Budget	Aug-17	22	0	166	0	0	6,413	0	0	0	572	128	267	\$7,546	0	\$0	\$0	\$0	\$0	\$7,546	-\$7,546	-\$561	-\$115,816	22
Budget	Sep-17	23	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$603	-\$123,394	23
Budget	Oct-17	24	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$643	-\$131,011	24
Budget	Nov-17	25	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$682	-\$138,668	25
Budget	Dec-17	26	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$722	-\$146,364	26
Budget	Jan-18	27	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$762	-\$154,101	27
Budget	Feb-18	28	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$803	-\$161,878	28
Budget	Mar-18	29	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$843	-\$169,695	29
Budget	Apr-18	30	0	166	0	0	6,413	0	0	0	0	128	267	\$6,974	0	\$0	\$0	\$0	\$0	\$6,974	-\$6,974	-\$884	-\$177,553	30
Budget	May-18	31	0	166	0	0	6,413	3,793	4,830	1	0	128	0	\$15,331	0	\$0	\$0	\$0	\$0	\$15,331	-\$15,331	-\$925	-\$193,809	31
Budget	Jun-18	32	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$1,009	-\$194,819	32
Budget	Jul-18	33	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	-\$1,015	-\$195,834	33
Budget	Aug-18	34 35	0	0	0	0	0	0	U	0	572 0	0	0	\$572	70 0	\$253,857	\$20,664	\$5,331	\$227,862	\$572	\$227,290	-\$1,020	\$30,437	34
Budget Budget	Sep-18 Oct-18	35	0	0	0	0	0	0	0	0	0	0	0	\$0 \$0	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$30,437 \$30,437	35 36
	I	I	28,468	\$5,131	\$609	\$236	\$128,267	\$3,793	\$4,830	\$1	\$1,715	\$3,848	\$5,077	\$181,974	70	\$253,857	\$20,664	\$5,331	\$227,862	\$181,974	\$45,888	-\$15,451	\$30,437	
			Land	Professional	DA Fees	CC Fees	Construction			LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		
			1	Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

2.11 Scenario I

Hypothetical development to achieve 15:1 FSR GFA outcome with application of S94A at 4.5% of cost of construction PLUS Value Sharing on Phase 2 (3:1 FSR GFA) at 50%. In this scenario, the draft CBD Planning Proposal identifies the site as an 'Opportunity Site', eligible for an additional 3:1 FSR above the proposed 10:1 FSR.



planning

Scenario I - S94A Levy 4.5% PLUS Value Sharing @ 50% on 3:1 FSR GFA

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	15.0
Land Area	3,000
Site Value	97,253
Number of Dwellings	433

gln planning

REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$)
	Ex Retail	Inc Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	61,477	26,619,697
Less Selling Costs	15,501	6,712,083
NET REVENUE	\$660,589	\$286,034,886
COSTS		
Land (including acquisition costs)	61,316	26,550,000
Acquisition costs	4,429	1,917,605
Construction	366,483	158,687,094
Consultants	14,659	6,347,484
Section 94A - Commercial	1,862	806,250
Section 94A - Residential	14,614	6,327,919
Value Sharing Contribution	8,366	3,622,500
Statutory Fees & Contributions	6,503	2,815,997
On Costs	10,994	4,760,613
Marketing	9,629	4,169,333
Cost before Interest	490,490	212,382,295
Finance (incl Loan Est Fees)	37,182	16,099,968
TOTAL DEVELOPMENT COSTS	527,673	\$228,482,263
TOTAL PROJECT SURPLUS	23.6%	\$53,930,124
PROJECT IRR BEFORE INTEREST	25.9%	

Scenario I - S94A Levy 4.5% PLUS Value Sharing @ 50% on 3:1 FSR GFA Input Sheet

SITE DETAILS						
Address	Representative					
Floor Space Ratio	15.0	:1				
Achievable FSR	45,000	m2				
Land Area	3,000	m2				
Site Value	97,253					
Any 1 parcel over \$3M	Yes					
LVR (Debt/Equity)	100%					
Date to start feasibility	Oct-15					



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment (Costs	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION	COSTS - Deposit	2,655,000	1	1
LAND ACQUISITION	COSTS - Settlement	25,812,605	2	1
CONSTRUCTION CO	STS 2,000	158,687,094	15	22
PROFESSIONAL FEE	S 4% const costs	6,347,484	1	36
APPLICATION FE	ES			
DA Fees		751,234	3	1
CC Fees		291,533	15	1
Section 94		7,134,169	36	1
Value Sharing		3,622,500	36	1
LPI Fees		58,347	36	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	36	1
ONCOSTS	3.0% construction cost	4,760,613	2	35
MARKETING	2% gross revenue	4,169,333	15	22
	TOTAL COST	216,004,795		

					Basement				
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL							-		·•
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80								
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0				-		
		Target Floorspace	45,000						

					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%	•	Yes	(\$)	(\$)	Mix	Mix
RESIDENTIAL	•								
1 bedroom	30	60	2,070	30	Yes	590,000	17,700,000	19%	15%
2 bedroom	120	80	11,040	120	Yes	700,000	84,000,000	75%	75%
3 bedroom	10	120	1,380	10	Yes	920,000	9,200,000	6%	10.0%
4 bedroom		130	0	0	Yes	0	0		0.0%
TOTAL	160		14,490	160	Yes		110,900,000	100%	100%
Average m2/unit	29								
ADDITIONAL INCLUS	IONS								
Other visitor parking				32					
TOTAL YIELD	160		14,490	192			110,900,000		
COMBINED YIELD	433		44,491	628			319,366,667		
		FSR	14.8						
		Target Floorspace	45,000	1					

Scenario I - S94A Levy 4.5% PLUS Value Sharing @ 50% on 3:1 FSR GFA Calculations

REVENUE		
GROSS REVENUE		319,366,667
GST		26,619,697
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	6,387,333
Legals (On Settlement)	\$750 per lot	324,750
TOTAL - SELLING COSTS		6,712,083
NET REVENUE		286,034,886



STAMP DUTY									
LAND VALU	E THRESHOLD	ΤΑΧ							
Minimum	1inimum Maximum								
14,000		0							
14,000	30,000	0							
30,000	80,000	0							
80,000	300,000	0							
300,000	1,000,000	0							
1,000,000		FALSE							
3,000,000	Premium Property Tax	1,798,990							
	TOTAL	1,798,990							

CONTRIBUTIONS							
SECTION 94A		4.5%					
COSTS							
Retail	10,416,667	\$468,750					
Commercial	7,500,000	\$337,500					
Residential	140,620,427	\$6,327,919					
TOTAL VALUE							
TOTAL CONT	\$7,134,169						

DA FEES		
CONSTRUCTION C	OST THRE	SHOLDS
50,001	250,000	0
250,001	500,000	0
500,001	1,000,000	0
1,000,001	10,000,000	0
More than \$10,000,000		195,829
LSL	0.35%	555,405
TOTAL DA FEES		751,234

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	291,533
TOTAL CC FEES	291,533

Build Costs	Rates	FSR +15%	UG Parking	Grade
	\$/m2		50,000	Parking
Apartments	2,850	112,549,350	21,650,000	(
Retail	2,500	6,250,000	4,166,667	
Commercial	2,500	6,250,000	1,250,000	
Visitor Parking			4,330,000	(
ESD Costs	22	978,802	-	
Design Comp		150,000		
Design Cost	25	1,112,275		
TOTALS		127,290,427	31,396,667	(
	158,687,094			

CAR PARKING								
Unit Type	No. Spaces							
1 bed/studio	1.0							
2 bed	1.0							
3+ bed	1.0							
Shops	30							
Commercial	100							
Visitor parking	In any case, min 1 visitor space is required							

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	EES		58,347

VALUE SHARING		
GFA Element	GFA	Base Value
4:1 CBD Strategy	0	\$0
3:1 Above Strategy	9,000	\$7,245,000
Calculation		
Rate/additional m2 GFA		\$805
CBD Strategy Rate		0%
Above CBD Strategy Rate		50%
TOTAL CONTRIBUTION		\$3,622,500

Scenario I - S94A Levy 4.5% PLUS Value Sharing @ 50% on 3:1 FSR GFA

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	25.85%
Surplus	\$53,930,124
% Surplus on D.C	23.24%
Max Loan Balance Debt/Equity	-\$211,899,574
Debt/Equity	100.00%

Max Loa Debt/Equ	n Balan		t/Equity	-\$211,899,574 100.00%		ellow expres	sed in nearest	\$,000													~		plann	ing
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	176	0	0	0	0	0	0	0	0	0	\$2,831	0	\$0	\$0	\$0	\$0	\$2,831	-\$2,831	\$0	-\$2,831	1
Budget	Dec-15	2	25,813	176	0	0	0	0	0	0	0	136	0	\$26,125	0	\$0	\$0	\$0	\$0	\$26,125	-\$26,125	-\$15	-\$28,971	2
Budget	Jan-16	3	0	176	751	0	0	0	0	0	0	136	0	\$1,064	0	\$0	\$0	\$0	\$0	\$1,064	-\$1,064	-\$151	-\$30,185	3
Budget	Feb-16	4	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$157	-\$30,655	4
Budget	Mar-16	5	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$160	-\$31,127	5
Budget	Apr-16	6	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$162	-\$31,601	6
Budget	May-16	7	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$165	-\$32,078	7
Budget	Jun-16	8	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$167	-\$32,558	8
Budget	Jul-16	9 10	0	176 176	0	0	0	0	0	0	0 572	136 136	0	\$312 \$884	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$312 \$884	-\$312 -\$884	-\$170 -\$172	-\$33,040 -\$34,096	9 10
Budget Budget	Aug-16 Sep-16	10	0	176	0	0	0	0	0	0	0	136	0	\$664 \$312	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$004 \$312	-\$664 -\$312	-\$172	-\$34,096	10
Budget	Oct-16	12	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0 \$0	\$312	-\$312	-\$178	-\$35,078	12
Budget	Nov-16	13	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$183	-\$35,573	13
Budget	Dec-16	14	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$185	-\$36,071	14
Budget	Jan-17	15	0	176	0	292	7,213	0	0	ō	0	136	190	\$8,006	0	\$0	\$0	\$0	\$0	\$8,006	-\$8,006	-\$188	-\$44,265	15
Budget	Feb-17	16	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$231	-\$52,211	16
Budget	Mar-17	17	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$272	-\$60,197	17
Budget	Apr-17	18	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$314	-\$68,226	18
Budget	May-17	19	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$355	-\$76,296	19
Budget	Jun-17	20	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$397	-\$84,408	20
Budget	Jul-17	21	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$440	-\$92,563	21
Budget	Aug-17	22	0	176	0	0	7,213	0	0	0	572	136	190	\$8,287	0	\$0	\$0	\$0	\$0	\$8,287	-\$8,287	-\$482	-\$101,331	22
Budget	Sep-17	23	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0 \$0	\$0	\$0	\$7,715	-\$7,715	-\$528	-\$109,574	23
Budget	Oct-17	24 25	0	176 176	0	0	7,213	0	0	0	0	136 136	190 190	\$7,715 \$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715	-\$7,715 -\$7,715	-\$571 -\$614	-\$117,860	24 25
Budget Budget	Nov-17 Dec-17	25	0	176	0	0	7,213 7,213	0	0	0	0	136	190 190	\$7,715 \$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715 \$7,715	-\$7,715	-\$614 -\$657	-\$126,189 -\$134,561	25 26
Budget	Jan-18	20	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0 \$0	\$7,715	-\$7,715	-\$037	-\$142,976	20
Budget	Feb-18	28	0	176	0	0	7,213	ő	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$745	-\$151,436	28
Budget	Mar-18	29	0	176	0	0	7,213	0	0	ő	õ	136	190	\$7,715	ő	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$789	-\$159,940	29
Budget	Apr-18	30	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$833	-\$168,488	30
Budget	May-18	31	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$878	-\$177,080	31
Budget	Jun-18	32	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$922	-\$185,717	32
Budget	Jul-18	33	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$967	-\$194,399	33
Budget	Aug-18	34	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$1,012	-\$203,127	34
Budget	Sep-18	35	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$1,058	-\$211,900	35
Budget	Oct-18	36	0	176	0	0	7,213	7,134	3,623	58	572	136	190	\$19,102	70	\$319,367	\$26,620	\$6,712	\$286,035	\$19,102	\$266,933	-\$1,104	\$53,930	36
		•	28,468	\$6,347	\$751	\$292	\$158,687	\$7,134	\$3,623	\$58	\$1,715	\$4,761	\$4,169	\$216,005	70	\$319,367	\$26,620	\$6,712	\$286,035	\$216,005	\$70,030	-\$16,100	\$53,930	
			Land	Professional	DA Fees	CC Fees	Construction Costs	Section 94 Costs		LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs		Gross Revenue	GST	Selling Costs	Net Revenue	Monthly	Net	Monthly		
			I	Fees			COSIS	COSIS			COSIS			COSIS		Revenue			Revenuê	Outlays		Interest		

Sheet 4 of 4

2.12 Scenario J

Hypothetical development to achieve 15:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 10% Phase 2 (3:1 FSR GFA) at 50%. In this scenario, the draft CBD Planning Proposal identifies the site to benefit from an increase in FSR from the current 6:1 to 10:1. It is also identified as an 'Opportunity Site', eligible for an additional 3:1 FSR above the proposed 10:1 FSR.





Scenario J - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA

Summary Sheet

SITE DETAILS	
Address	Representative
Floor Space Ratio	15.0
Land Area	3,000
Site Value	97,253
Number of Dwellings	433



REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	61,477	26,619,697
Less Selling Costs	15,501	6,712,083
NET REVENUE	\$660,589	\$286,034,886
COSTS		
Land (including acquisition costs)	61,316	26,550,000
Acquisition costs	4,429	1,917,605
Construction	366,483	158,687,094
Consultants	14,659	6,347,484
Section 94A - Commercial	1,241	537,500
Section 94A - Residential	9,743	4,218,613
Value Sharing Contribution	10,597	4,588,500
Statutory Fees & Contributions	6,503	2,815,997
On Costs	10,994	4,760,613
Marketing	9,629	4,169,333
Cost before Interest	484,998	210,004,239
Finance (incl Loan Est Fees)	37,182	16,099,968
TOTAL DEVELOPMENT COSTS	522,181	\$226,104,206
TOTAL PROJECT SURPLUS	24.5%	\$55,342,180
PROJECT IRR BEFORE INTEREST	26.3%	

Scenario J - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA Input Sheet

SITE DETAILS		
Address	Repr	esentative
Floor Space Ratio	15.0	:1
Achievable FSR	45,000	m2
Land Area	3,000	m2
Site Value	97,253	
Any 1 parcel over \$3M	Yes	
LVR (Debt/Equity)	100%	
Date to start feasibility	Oct-15	



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment C	Costs	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION	COSTS - Deposit	2,655,000	1	1
LAND ACQUISITION	COSTS - Settlement	25,812,605	2	1
CONSTRUCTION COS	STS 2,000	158,687,094	15	22
PROFESSIONAL FEE	S 4% const costs	6,347,484	1	36
APPLICATION FE	ES			
DA Fees		751,234	3	1
CC Fees		291,533	15	1
Section 94		4,756,113	36	1
Value Sharing		4,588,500	36	1
LPI Fees		58,347	36	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	36	1
ONCOSTS	3.0% construction cost	4,760,613	2	35
MARKETING	2% gross revenue	4,169,333	15	22
	TOTAL COST	214,592,739		

UNIT MIX AND SALE	S								
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Basement Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
NON RESIDENTIAL						(+)	(+)		
Retail	1	2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL						-			÷
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80	•		•	-	-	-	-	-
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0						
		Target Floorspace	45,000						

ADDITIONAL UNIT MIX AND SALES (based on additional 5:1 FSR after Planning Proposal Basement Unit Type No. Dwelling Total Car Parking Parking Price Total Actual Preferred Floor Space m2 FSR+15% Yes (\$) (\$) Mix Mix RESIDENTIAL 17,700,000 19% 1 bedroom 30 120 2.070 590.000 60 30 Yes 15% 80 11,040 120 Yes 700,000 84,000,000 75% 75% 2 bedroom 10.0% 3 bedroom 10 120 1,380 10 Yes 920,000 9,200,000 6% 4 bedroom 130 0 0 Yes 0 0 0.0% 160 14,490 110,900,000 100% TOTAL 160 Yes 100% Average m2/unit 29 ADDITIONAL INCLUSIONS Other visitor parking 32 TOTAL YIELD 160 14,490 192 110,900,000 COMBINED YIELD 433 44,491 628 319,366,667 FSR 14.8 Target Floorspace 45,000

Scenario J - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA Calculations

REVENUE		
GROSS REVENUE		319,366,667
GST		26,619,697
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	6,387,333
Legals (On Settlement)	\$750 per lot	324,750
TOTAL - SELLING COSTS		6,712,083
NET REVENUE		286,034,886



STAMP DUTY				
LAND VALUE THRESHOLD TAX				
Minimum	Maximum			
14,000		0		
14,000	30,000	0		
30,000	80,000	0		
80,000	300,000	0		
300,000	1,000,000	0		
1,000,000		FALSE		
3,000,000 Premium Property Tax 1,798,990				
	TOTAL	1,798,990		

CONTRIBUTIONS					
SECTION 94A	SECTION 94A 3.0%				
COSTS					
Retail	10,416,667	\$312,500			
Commercial	7,500,000	\$225,000			
Residential	140,620,427	\$4,218,613			
TOTAL VALUE	158,537,094				
TOTAL CONT	\$4,756,113				

DA FEES				
CONSTRUCTION COST THRESHOLDS				
50,001	250,000	0		
250,001	500,000	0		
500,001	1,000,000	0		
1,000,001	10,000,000	0		
More than \$10,000,000		195,829		
LSL	0.35%	555,405		
TOTAL DA FEES		751,234		

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	291,533
TOTAL CC FEES	291,533

CONSTRUCTION COSTS						
Build Costs	Rates	FSR +15%	UG Parking	Grade		
	\$/m2		50,000	Parking		
Apartments	2,850	112,549,350	21,650,000	0		
Retail	2,500	6,250,000	4,166,667			
Commercial	2,500	6,250,000	1,250,000			
Visitor Parking			4,330,000	0		
ESD Costs	22	978,802				
Design Comp		150,000				
Design Cost	25	1,112,275				
TOTALS		127,290,427	31,396,667	0		
	158,687,094					

CAR PARKING				
Unit Type	No. Spaces			
1 bed/studio	1.0			
2 bed	1.0			
3+ bed	1.0			
Shops	30			
Commercial	100			
Visitor parking	In any case, min 1 visitor space is required			

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	58,347		

VALUE SHARING				
GFA Element	GFA	Base Value		
4:1 CBD Strategy	12,000	\$9,660,000		
3:1 Above Strategy	9,000	\$7,245,000		
Calculation				
Rate/additional m2 GFA	\$805			
CBD Strategy Rate	10%			
Above CBD Strategy Rate	50%			
TOTAL CONTRIBUTION		\$4,588,500		

Scenario J - S94A Levy 3% PLUS Value Sharing @ 10% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA

Assumed Development Cashflow

DEVELOPMENT SUM	MARY
IRR Before Interest	26.32%
Surplus	\$55,342,180
% Surplus on D.C	23.99%
Max Loan Balance Debt/Equity	-\$211,899,574
Debt/Equity	100.00%

Max Loa			+/Equity	-\$211,899,574																				-
Debt/Equ		ce Deb	vEquity	100.00%		ellow express	sed in nearest	\$,000													<		planr	ning
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	176	0	0	ō	0	0	0	0	0	0	\$2,831	0	\$0	\$0	\$0	\$0	\$2,831	-\$2,831	\$0	-\$2,831	1
Budget	Dec-15	2	25,813	176	0	0	0	0	0	0	0	136	0	\$26,125	0	\$0	\$0	\$0	\$0	\$26,125	-\$26,125	-\$15	-\$28,971	2
Budget	Jan-16	3	0	176	751	0	0	0	0	0	0	136	0	\$1,064	0	\$0	\$0	\$0	\$0	\$1,064	-\$1,064	-\$151	-\$30,185	3
Budget	Feb-16	4	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$157	-\$30,655	4
Budget	Mar-16	5	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$160	-\$31,127	5
Budget	Apr-16	6	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$162	-\$31,601	6
Budget	May-16	7	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$165	-\$32,078	7
Budget	Jun-16	8	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$167	-\$32,558	8
Budget	Jul-16	9	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$170	-\$33,040	9
Budget	Aug-16	10	0	176	0	0	0	0	0	0	572	136	0	\$884	0	\$0	\$0	\$0	\$0	\$884	-\$884	-\$172	-\$34,096	10
Budget	Sep-16	11	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$178	-\$34,586	11
Budget	Oct-16	12 13	0	176	0	0	0	0	0	0	0	136 136	0	\$312	0	\$0	\$0	\$0	\$0 \$0	\$312	-\$312	-\$180	-\$35,078	12
Budget	Nov-16	13	0	176 176	0	0	0	0	0	0	0	136	0	\$312 \$312	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$312 \$312	-\$312	-\$183 -\$185	-\$35,573	13 14
Budget	Dec-16 Jan-17	14	0	176	0	292	7,213	0	0	0	0	136	190	\$312	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$312	-\$312 -\$8,006	-\$165	-\$36,071 -\$44,265	14
Budget Budget	Feb-17	15	0	176	0	292	7,213	0	0	0	0	136	190	\$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$8,006	-\$8,006	-\$100	-\$44,265 -\$52,211	15
Budget	Mar-17	17	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0 \$0	\$0	\$7,715	-\$7,715	-\$272	-\$60,197	17
Budget	Apr-17	18	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$314	-\$68,226	18
Budget	May-17	19	0	176	ő	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$355	-\$76,296	19
Budget	Jun-17	20	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$397	-\$84,408	20
Budget	Jul-17	21	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$440	-\$92,563	21
Budget	Aug-17	22	0	176	0	0	7,213	0	0	0	572	136	190	\$8,287	0	\$0	\$0	\$0	\$0	\$8,287	-\$8,287	-\$482	-\$101,331	22
Budget	Sep-17	23	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$528	-\$109,574	23
Budget	Oct-17	24	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$571	-\$117,860	24
Budget	Nov-17	25	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$614	-\$126,189	25
Budget	Dec-17	26	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$657	-\$134,561	26
Budget	Jan-18	27	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$701	-\$142,976	27
Budget	Feb-18	28	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$745	-\$151,436	28
Budget	Mar-18	29	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$789	-\$159,940	29
Budget	Apr-18	30	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$833	-\$168,488	30
Budget	May-18	31	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$878	-\$177,080	31
Budget	Jun-18	32	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$922	-\$185,717	32
Budget	Jul-18	33	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$967	-\$194,399	33
Budget	Aug-18	34	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$1,012	-\$203,127	34
Budget Budget	Sep-18 Oct-18	35 36	0	176 176	0	0	7,213 7,213	0 4,756	0 4.589	0 58	0 572	136 136	190 190	\$7,715 \$17,689	0 70	\$0 \$319,367	\$0 \$26,620	\$0 \$6,712	\$0 \$286,035	\$7,715 \$17.689	-\$7,715 \$268,345	-\$1,058 -\$1,104	-\$211,900 \$55,342	35 36
Dungor	00.10		Ŭ			0			,						-									
			28,468 Land	\$6,347 Professional	\$751	\$292	\$158,687	\$4,756	\$4,589	\$58 I BI Ecco	\$1,715 Statuany	\$4,761	\$4,169 Marketing	\$214,593	70	\$319,367	\$26,620	\$6,712 Solling Costs	\$286,035 Net	\$214,593	\$71,442	-\$16,100	\$55,342	4
			Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs		LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs		Gross Revenue	GST	Selling Costs	Revenue	Monthly Outlays	Net	Monthly Interest		
			L																	,5				

Sheet 4 of 4

2.13 Scenario K

Hypothetical development to achieve 15:1 FSR GFA outcome with application of S94A at 3% of cost of construction PLUS Value Sharing on Phase 1 (4:1 FSR GFA) at 50% Phase 2 (3:1 FSR GFA) at 50%. In this scenario, the draft CBD Planning Proposal identifies the site to benefit from an increase in FSR from the current 6:1 to 10:1. It is also identified as an 'Opportunity Site', eligible for an additional 3:1 FSR above the proposed 10:1 FSR.





Scenario K - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA

Summary Sheet

SITE DETAILS					
Address	Representative				
Floor Space Ratio	15.0				
Land Area	3,000				
Site Value	97,253				
Number of Dwellings	433				



REVENUE		
	Average/	
	Unit	Development
	(\$)	(\$) Inc
	Ex Retail	Retail
GROSS REVENUE	\$696,189	\$319,366,667
GST	61,477	26,619,697
Less Selling Costs	15,501	6,712,083
NET REVENUE	\$660,589	\$286,034,886
COSTS		
Land (including acquisition costs)	61,316	26,550,000
Acquisition costs	4,429	1,917,605
Construction	366,483	158,687,094
Consultants	14,659	6,347,484
Section 94A - Commercial	1,241	537,500
Section 94A - Residential	9,743	4,218,613
Value Sharing Contribution	19,521	8,452,500
Statutory Fees & Contributions	6,503	2,815,997
On Costs	10,994	4,760,613
Marketing	9,629	4,169,333
Cost before Interest	484,998	210,004,239
Finance (incl Loan Est Fees)	37,182	16,099,968
TOTAL DEVELOPMENT COSTS	522,181	\$226,104,206
TOTAL PROJECT SURPLUS	22.8%	\$51,478,180
PROJECT IRR BEFORE INTEREST	25.0%	

Scenario K - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA Input Sheet

SITE DETAILS							
Address	Representative						
Floor Space Ratio	15.0	:1					
Achievable FSR	45,000	m2					
Land Area	3,000	m2					
Site Value	97,253						
Any 1 parcel over \$3M	Yes						
LVR (Debt/Equity)	100%						
Date to start feasibility	Oct-15						



DEVELOPMENT COSTS

Cost Items		Cost (\$)	Period Commencing (month)	Term (months)
LAND				
Land Purchase		26,550,000		
Loan Establishment C	Costs	25,690		
Stamp Duty		1,798,990		
Legals	0.4%	92,925		
LAND ACQUISITION	COSTS - Deposit	2,655,000	1	1
LAND ACQUISITION	COSTS - Settlement	25,812,605	2	1
CONSTRUCTION COS	STS 2,000	158,687,094	15	22
PROFESSIONAL FEE	S 4% const costs	6,347,484	1	36
APPLICATION FE	ES			
DA Fees		751,234	3	1
CC Fees		291,533	15	1
Section 94		4,756,113	36	1
Value Sharing		8,452,500	36	1
LPI Fees		58,347	36	1
LAND TAX/RATE	S			
Land Tax/Rates	Year 1	571,628	10	1
Land Tax/Rates	Year 2	571,628	22	1
Land Tax/Rates	Year 3	571,628	36	1
ONCOSTS	3.0% construction cost	4,760,613	2	35
MARKETING	2% gross revenue	4,169,333	15	22
	TOTAL COST	218,456,739		

					Basement				
Unit Type	No.	Dwelling	Total	Car Parking	Parking	Price	Total	Actual	Preferred
		Floor Space m2	FSR+15%		Yes	(\$)	(\$)	Mix	Mix
NON RESIDENTIAL									
Retail		2,500	2,500	83	Yes	10,416,667	10,416,667		
Commercial		2,500	2,500	25	Yes	7,500,000	7,500,000		
RESIDENTIAL									
1 bedroom	45	60	3,105	45	Yes	590,000	26,550,000	16%	15%
2 bedroom	208	80	19,136	208	Yes	700,000	145,600,000	76%	75%
3 bedroom	20	120	2,760	20	Yes	920,000	18,400,000	7%	10.0%
4 bedroom	0	130	0	0	Yes	0	0		0.0%
TOTAL	273		30,001	381	Yes		208,466,667	100%	100%
Average m2/unit	80			-	-	-	•	-	-
ADDITIONAL INCLU	SIONS								
Other visitor parking				55					
TOTAL YIELD	273		30,001	436			208,466,667		
		FSR	10.0						
		Target Floorspace	45,000						

					Basement				
Unit Type	No.	Dwelling Floor Space m2	Total FSR+15%	Car Parking	Parking Yes	Price (\$)	Total (\$)	Actual Mix	Preferred Mix
RESIDENTIAL									
1 bedroom	30	60	2,070	30	Yes	590,000	17,700,000	19%	15%
2 bedroom	120	80	11,040	120	Yes	700,000	84,000,000	75%	75%
3 bedroom	10	120	1,380	10	Yes	920,000	9,200,000	6%	10.0%
4 bedroom		130	0	0	Yes	0	0		0.0%
TOTAL	160		14,490	160	Yes		110,900,000	100%	100%
Average m2/unit	29								-
ADDITIONAL INCLUSI	ONS								
Other visitor parking				32					
TOTAL YIELD	160		14,490	192			110,900,000		
COMBINED YIELD	433		44,491	628			319,366,667		
		FSR	14.8				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
		Target Floorspace	45,000						

Scenario K- S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA Calculations

REVENUE		
GROSS REVENUE		319,366,667
GST		26,619,697
SELLING COSTS		
Sales Commission (On Settlement)	2% on gross rev	6,387,333
Legals (On Settlement)	\$750 per lot	324,750
TOTAL - SELLING COSTS		6,712,083
NET REVENUE		286,034,886



STAMP DUTY									
LAND VALUE THRESHOLD TAX									
Minimum	Maximum								
14,000		0							
14,000	30,000	0							
30,000	80,000	0							
80,000	300,000	0							
300,000	1,000,000	0							
1,000,000		FALSE							
3,000,000	Premium Property Tax	1,798,990							
	TOTAL	1,798,990							

CONTRIBUTIONS								
SECTION 94A	L	3.0%						
COSTS								
Retail	10,416,667	\$312,500						
Commercial	7,500,000	\$225,000						
Residential	140,620,427	\$4,218,613						
TOTAL VALUE	158,537,094							
TOTAL CONT	\$4,756,113							

DA FEES									
CONSTRUCTION COST THRESHOLDS									
50,001	250,000	0							
250,001	500,000	0							
500,001	1,000,000	0							
1,000,001	10,000,000	0							
More than \$10,000,000		195,829							
LSL	0.35%	555,405							
TOTAL DA FEES		751,234							

CC FEES	
CONSTRUCTION COSTS	
Exceeding \$200,000	291,533
TOTAL CC FEES	291,533

CONSTRUCT	CONSTRUCTION COSTS													
Build Costs	Rates	FSR +15%	UG Parking	Grade										
	\$/m2		50,000	Parking										
Apartments	2,850	112,549,350	21,650,000	0										
Retail	2,500	6,250,000	4,166,667											
Commercial	2,500	6,250,000	1,250,000											
Visitor Parking			4,330,000	0										
ESD Costs	22	978,802												
Design Comp		150,000												
Design Cost	25	1,112,275												
TOTALS		127,290,427	31,396,667	0										
	TOTAL CONSTRUCTION COST													

CAR PARKING											
Unit Type	No. Spaces										
1 bed/studio	1.0										
2 bed	1.0										
3+ bed	1.0										
Shops	30										
Commercial	100										
Visitor parking	In any case, min 1 visitor space is required										

RATES	
Council Rates	58,004
Land Tax	513,624
TOTAL RATES	571,628

LPI FEES			
Strata	Base Fee	Per Dwg fee	
	1,321	132	
TOTAL LPI F	58,347		

VALUE SHARING				
GFA Element	GFA	Base Value		
4:1 CBD Strategy	12,000	\$9,660,000		
3:1 Above Strategy	9,000	\$7,245,000		
Calculation				
Rate/additional m2 GFA		\$805		
CBD Strategy Rate		50%		
Above CBD Strategy Rate		50%		
TOTAL CONTRIBUTION		\$8,452,500		

Scenario K - S94A Levy 3% PLUS Value Sharing @ 50% of 4:1 GFA + Value Sharing @ 50% of 3:1 GFA

Assumed Development Cashflow

DEVELOPMENT SUM	DEVELOPMENT SUMMARY								
IRR Before Interest	25.03%								
Surplus	\$51,478,180								
% Surplus on D.C	21.95%								
Max Loan Balance Debt/Equity	-\$211,899,574								
Debt/Equity	100.00%								

Maxioa			t/Fauity	-\$211,899,574	1																			
Debt/Equ			u Equity	100.00%		ellow express	sed in nearest	\$,000													<		planr	ing
Budget/ Actual	Oct-15	Month	Land	Professional Fees	DA Fees	CC Fees	Construction Costs	Section 94 Costs	Value Sharing	LPI Fess	Statuory Costs	Oncosts	Marketing	Total Costs	Sales	Gross Revenue	GST	Selling Costs	Net Revenue	Monthly Outlays	Net Cashflow	Monthly Interest 6.25%	Total Funds Cashflow	Period
Budget	Oct-15	0	0	0	0	0	0	0	0	0	0	0	0	\$0	0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	0
Budget	Nov-15	1	2,655	176	0	0	0	0	0	0	0	0	0	\$2,831	0	\$0	\$0	\$0	\$0	\$2,831	-\$2,831	\$0	-\$2,831	1
Budget	Dec-15	2	25,813	176	0	0	0	0	0	0	0	136	0	\$26,125	0	\$0	\$0	\$0	\$0	\$26,125	-\$26,125	-\$15	-\$28,971	2
Budget	Jan-16	3	0	176	751	0	0	0	0	0	0	136	0	\$1,064	0	\$0	\$0	\$0	\$0	\$1,064	-\$1,064	-\$151	-\$30,185	3
Budget	Feb-16	4	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$157	-\$30,655	4
Budget	Mar-16	5	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$160	-\$31,127	5
Budget	Apr-16	6	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$162	-\$31,601	6
Budget	May-16	7	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$165	-\$32,078	7
Budget	Jun-16	8	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0	\$0	\$0	\$0	\$312	-\$312	-\$167	-\$32,558	8
Budget	Jul-16	9 10	0	176 176	0	0	0	0	0	0	0 572	136 136	0	\$312 \$884	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$312 \$884	-\$312 -\$884	-\$170 -\$172	-\$33,040 -\$34,096	9 10
Budget	Aug-16	10	0	176	0	0	0	0	0	0	0	136	0	\$004 \$312	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$664	-\$664 -\$312	-\$172	-\$34,586	10
Budget Budget	Sep-16 Oct-16	12	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0 \$0	\$0	\$0 \$0	\$0	\$312	-\$312	-\$178	-\$35,078	12
Budget	Nov-16	12	0	176	0	0	0	0	0	0	0	136	0	\$312	0	\$0 \$0	\$0	\$0 \$0	\$0	\$312	-\$312	-\$183	-\$35,573	12
Budget	Dec-16	14	ő	176	ő	0	0	ő	0	0	ő	136	0 0	\$312	ő	\$0	\$0	\$0	\$0	\$312	-\$312	-\$185	-\$36,071	14
Budget	Jan-17	15	ő	176	ő	292	7,213	ő	0	0	ő	136	190	\$8,006	ő	\$0	\$0	\$0	\$0	\$8.006	-\$8,006	-\$188	-\$44,265	15
Budget	Feb-17	16	0	176	ō	0	7,213	0	ō	0	ō	136	190	\$7,715	õ	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$231	-\$52,211	16
Budget	Mar-17	17	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$272	-\$60,197	17
Budget	Apr-17	18	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$314	-\$68,226	18
Budget	May-17	19	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$355	-\$76,296	19
Budget	Jun-17	20	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$397	-\$84,408	20
Budget	Jul-17	21	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$440	-\$92,563	21
Budget	Aug-17	22	0	176	0	0	7,213	0	0	0	572	136	190	\$8,287	0	\$0	\$0	\$0	\$0	\$8,287	-\$8,287	-\$482	-\$101,331	22
Budget	Sep-17	23	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$528	-\$109,574	23
Budget	Oct-17	24	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$571	-\$117,860	24
Budget	Nov-17	25	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$614	-\$126,189	25
Budget	Dec-17	26	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$657	-\$134,561	26
Budget	Jan-18	27	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$701	-\$142,976	27
Budget	Feb-18 Mor 18	28	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715	-\$7,715	-\$745	-\$151,436	28
Budget	Mar-18 Apr-18	29 30	0	176 176	0	0	7,213 7,213	0	0	0	0	136 136	190 190	\$7,715 \$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715 \$7,715	-\$7,715 -\$7,715	-\$789 -\$833	-\$159,940 -\$168,488	29 30
Budget Budget	Apr-18 May-18	30 31	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715 \$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715	-\$7,715	-\$833 -\$878	-\$168,488 -\$177,080	30
Budget	Jun-18	31	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0 \$0	\$0 \$0	\$0 \$0	\$0 \$0	\$7,715	-\$7,715	-\$676 -\$922	-\$185.717	31
Budget	Jul-18	32	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0 \$0	\$0	\$7,715	-\$7,715	-\$967	-\$194,399	32
Budget	Aug-18	34	0	176	0	0	7,213	0	0	0	0	136	190	\$7,715	0	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$1,012	-\$203,127	34
Budget	Sep-18	35	0	176	ő	0	7,213	0	0	0	0	136	190	\$7,715	ő	\$0	\$0	\$0	\$0	\$7,715	-\$7,715	-\$1,058	-\$211,900	35
Budget	Oct-18	36	0	176	0	0	7,213	4,756	8,453	58	572	136	190	\$21,553	70	\$319,367	\$26,620	\$6,712	\$286,035	\$21,553	\$264,481	-\$1,104	\$51,478	36
	1	I	28,468	\$6,347	\$751	\$292	\$158,687	\$4,756	\$8,453	\$58	\$1,715	\$4,761	\$4,169	\$218,457	70	\$319,367	\$26,620	\$6,712	\$286,035	\$218,457	\$67,578	-\$16,100	\$51,478	<u> </u>
			Land	Professional	DA Fees	CC Fees	Construction	Section 94		LPI Fess	Statuory	Oncosts	Marketing	Total		Gross	GST	Selling Costs	Net	Monthly	Net	Monthly		
				Fees			Costs	Costs			Costs			Costs		Revenue			Revenue	Outlays		Interest		

Sheet 4 of 4

3.0 Model Assumptions

Parramatta Council has identified some \$835 million worth of infrastructure improvements as a result of the greatly increased development potential to be implemented through the CBD Planning Proposal. The Planning Strategy to increase the heights and floor space ratios have made it clear that part of the funding for this infrastructure will come from a review of Section 94A contributions and/or use of a Value Share mechanism for those development utilising the additional floor space bonuses. This Paper summarises the assumptions undertaken in the modelling to test the impact of these contributions on the developers. The results of the modelling have ben separately provided to Council.

3.1 Approach

Feasibility modelling has been used to test the impacts on developer returns by varying the Section 94A and Value Sharing regimes applicable to existing and additional floor space being considered as part of the Parramatta CBD Planning Proposal.

Two base scenarios were prepared to reflect the different FSR outcomes typically envisaged by Council for the CBD area being:

- **6.9:1 FSR Development** Application of 6:1 FSR to B4 land outside of the CBD core with an additional 0.9:1 FSR as a result of the proposed incentive clause for 'Design Excellence'.
- **12:1 FSR Development** Application of 10:1 FSR to B4 land in the CBD core with additional 2:1 FSR as a result of the proposed incentive clauses for 'Design Excellence' (1.5:1) and 'High Performance Buildings' (0.5:1 FSR).

For each base scenario, the Residual Land Value (being the price a developer could afford to pay for the land) was determined by modelling the cash flow of all revenue generated from the hypothetical development less associated costs of development (such as finance, construction costs, contributions, marketing etc) to achieve the development and finance industry benchmarks of Internal Rate of Return and Profit Margin of a minimum of 20%. These acquisition costs were then adopted as a constant for each of the scenarios that were modelled.

3.2 Benchmarks

The key benchmarks used in the feasibility assessment include **Developer Profit**, **Development Margin** and **Internal Rate of Return** (IRR). These benchmarks are reflective of the risk and reward to incentivise developers to commit to (and banks to finance) a development proposal. Typically, a benchmark of 20% for IRR and Development Margin is considered to be representative of a feasible development opportunity.

This benchmark of 20% was applied to all scenarios delivering an FSR outcome of 12:1 or below. When considering the impact of the different contribution scenarios of development in excess of a 12:1 FSR outcome, the discussion around development feasibility sought to adopt a higher benchmark (IRR of 25%) to represent the additional risk involved with larger scale developments.

3.3 Development Scheme Assumptions

In preparing the feasibility the following development assumptions were made:



- Delivery of 2,500m² Retail GFA and 2,500m² Commercial GFA to meet the planning controls requiring a mixed use development outcome. It is often the case that these components are not profitable unless significant pre-commitment is secured. As such, developers will seek only to ensure that the cost of construction is covered by revenue generated by the sale of these components of development.
- Apartment mix achieves (as close as possible):
- 15% of 1 Bedroom Apartments with an average area of 60m²
- 75% of 2 Bedroom Apartments with an average area of 80m²
- 10% of 3 Bedroom Apartments with an average of 120m²
- The area of apartments have been increased by an additional 15% to reflect common areas, circulation space and balcony areas.
- Car parking is provided as basement levels to the development.
- Car parking is at one space per apartment. This allows flexibility in the marketing and configuration of apartment products to meet a variety of price points.
- Car parking for commercial GFA is provided at 1 space per 100m² GFA.
- Car parking for retail GFA is provided at 1 space per 30m².
- Visitor car parking is provided at 1 space per 5 apartments.
- Margin Scheme using the purchase price applied for GST for the sale of apartments.

3.4 **Revenue Assumptions**

In preparing the feasibility the following revenue assumptions were made:

- Net revenue equals gross revenue less GST, less sales commission and legals.
- Revenue generated from commercial and retail development is netted off against the cost of construction.
- Average sales price of 1 Bedroom Apartments of \$590,000 (\$9,800/m²) based on 6 listed off the plan apartments in Parramatta CBD in September/October 2015. See attached Sales Summary Sheet.
- Average sales price 2 Bedroom Apartments of \$700,000 (\$8,750/m²) based on 13 listed off the plan and new apartments in Parramatta CBD in September/October 2015. See attached Sales Summary Sheet.
- Average sales price 2 Bedroom Apartments of \$920,000 (\$7,660/m²) based on 7 listed off the plan and new apartments in Parramatta CBD in September/October 2015. See attached Sales Summary Sheet.
- No allowance for escalation in sales prices or costs has been made over life of the project.
- Any increase in sales prices will likely improve feasibility outcomes. Conversely, any decrease in sales prices will have a negative impact on feasibility outcomes.
- Assumed that 100% of apartments are 'sold' prior to registration of Subdivision Plan.
- GST is included on residential sales.
- Sales commission of 2% paid at settlement.
- Legal costs paid on settlement of \$750 per apartment.

3.5 Cost Assumptions

In preparing the feasibility the following cost assumptions were made:

- Based on 100% financing (debt) at 6.25%. Considered to be a more conservative assumption to reflect worst case for interest.
- Loan establishment costs based on fixed fee and 0.06% of land acquisition cost paid at settlement in Month 2.
- Land acquisition legal costs at 0.4% of land acquisition paid at settlement in Month 2.
- Stamp Duty paid at settlement in Month 2.
- Apartment cost of construction at \$2,850/m² (Rawlinsons 2015). This cost includes a development margin to cover Tier 1 Developers entering into Design and Construct contracts. Inclusive of contingency and allowance for site works and servicing.
- Retail and commercial cost of construction at \$2,500/m² (Rawlinsons 2015).
- Basement carparking cost at \$50,000/space (Rawlinsons 2015).
- High Performance Building cost of construction at \$22/m² (Council advice provided).
- Design Excellence costs of \$150,000 for design competition (Council advice provided) and allowance of \$25/m².
- No escalation for construction costs have been assumed.
- DA, LSL and CC Fees as per Council's Fees and Charges.
- Professional Fees at 4% of cost of construction.
- Land Tax over 3 year period at Statutory Rate.
- S94 costs as per scenario descriptions.
- Strata Fees based on base fee of \$1,321 plus one drawing per apartment at \$132/drawing.
- GST is included on residential sales.
- Sales commission of 2% paid at settlement.
- Legal costs paid on settlement of \$750 per apartment.



Recent Apartment Sales/Prices September - October 2015

1 Bedroom Apartm	ents							
Address	Developer	Condition	Bedrooms	Bathrooms	Parking	Sale/Asking Price	Sales Date	Source
330 Church Street	Meriton	Off Plan	1	1	0	\$662,000	Current	http://www.domain.com.au/project/690/altitude-parramatta-nsw
330 Church Street	Meriton	Off Plan	1	1	1	\$683,000	Current	http://www.domain.com.au/project/690/altitude-parramatta-nsw
2 Morton Street		Off Plan	1	1	1	\$533,000	Current	http://www.domain.com.au/project/993/promenade-parramatta-nsw
2 Morton Street		Off Plan	1	1	1	\$555,000	Current	http://www.domain.com.au/project/993/promenade-parramatta-nsw
805/118 Church Street		Off Plan	1	1	1	\$570,000	Current	http://www.domain.com.au/805-118-church-street-parramatta-nsw-2150-2012386042?sp=12
806/118 Church Street		New	1	1	1	\$556,500	9/09/201	5 http://www.domain.com.au/806-118-church-street-parramatta-nsw-2150-2012194673?sp=13
		Average	1	1	0.8333	\$593,250		
		P	Adopted Sales Price			\$590,000		

2 Bedroom Apartm	ents							
Address	Developer	Condition	Bedrooms	Bathrooms	Parking	Sale/Asking Price	Sales Date	Source
330 Church Street	Meriton	Off Plan	2	2 2	1	\$797,000	Current	http://www.domain.com.au/project/690/altitude-parramatta-nsw
2 Morton Street		Off Plan	2	2 2	1	\$725,000	Current	http://www.domain.com.au/project/993/promenade-parramatta-nsw
2 Morton Street		Off Plan	2	2 2	1	\$685,000	Current	http://www.domain.com.au/project/993/promenade-parramatta-nsw
710/2 Morton Street		Off Plan	2	2 2	1	\$690,000	Current	http://www.domain.com.au/710-2-morton-street-parramatta-nsw-2150-2012332144?sp=10
35/105 Church Street		Modern	2	2 2	1	\$655,000	19/10/2015	http://www.domain.com.au/35-105-church-street-parramatta-nsw-2150-2012312667?sp=3
19/51A High Street		New	2	2 2	2	\$600,000	2/10/2015	http://www.domain.com.au/19-51a-high-street-parramatta-nsw-2150-2012241149?sp=13
810/36 Cowper Street		Modern	2	2 2	1	\$675,000	26/09/2015	http://www.domain.com.au/810-36-cowper-street-parramatta-nsw-2150-2012216244?sp=18
5/8 Elizabeth Street		New	2	2 2	1	\$670,000	26/09/2015	http://www.domain.com.au/5-8-elizabeth-street-parramatta-nsw-2150-2012225576?sp=0
905/8 Cowper Street		Modern	2	2 2	1	\$650,000	23/09/2015	http://www.domain.com.au/905b-8-cowper-street-parramatta-nsw-2150-2012235251?sp=3
3/11-13 Hunter Street		New	2	2 2	1	\$770,000	18/09/2015	http://www.domain.com.au/3-11-13-hunter-street-parramatta-nsw-2150-2012156620?sp=6
7/4589 Church Street			2	2 2	1	\$630,000	8/09/2015	http://www.domain.com.au/7-459-church-parramatta-nsw-2150-2012026879?sp=14
1/69 High Street		New	2	2 1	1	\$651,000	4/09/2015	http://www.domain.com.au/1-69-high-street-parramatta-nsw-2150-2012209174?sp=15
51/20 Victoria Road		New	2	2 2	2	\$790,000	2/09/2015	http://www.domain.com.au/51-20-victoria-road-parramatta-nsw-2150-2012184147?sp=16
		Average	2	1.92307692	1.1538	\$691,385		
			Adopted S	ales Price		\$700,000		

3 Bedroom Apartm	ents							
Address	Developer	Condition	Bedrooms	Bathrooms	Parking	Sale/Asking Price	Sales Date	Source
330 Church Street	Meriton	Off Plan	3	2	2	\$1,299,000	Current	http://www.domain.com.au/project/690/altitude-parramatta-nsw
20-24 Kendall Street		Off Plan	3	2	2	\$850,000	Current	http://www.domain.com.au/20-24-kendall-street-parramatta-nsw-2150-2012357595?sp=8
Macquarie Street		Off Plan	3	2	2	\$1,050,000	Current	http://www.domain.com.au/parramatta-nsw-2150-2011485687?sp=17
14/9-11 Cowper Street		Modern	3	2	1	\$710,000	2/10/2015	http://www.domain.com.au/14-9-11-cowper-street-parramatta-nsw-2150-2012222334?sp=14
46/20 Victoria Road		New	3	2	1	\$833,000	23/09/2015	http://www.domain.com.au/46-20-victoria-road-parramatta-nsw-2150-2012259448?sp=1
6/27 Stewart Street		New	3	2	2	\$850,000	17/09/2015	http://www.domain.com.au/6-27-stewart-street-parramatta-nsw-2150-2012226820?sp=8
604/36 Cowper Street			3	2	2	\$838,000	14/09/2015	http://www.domain.com.au/604-36-cowper-st-parramatta-nsw-2150-2012133555?sp=11
		Average	3	2	1.7143	\$918,571		
			Adopted S	ales Price		\$920,000		

aurecon



City of Parramatta

Discussion Paper

Infrastructure Planning and Funding in the Parramatta CBD

March 2017







Executive summary

Sitting at the heart of Greater Sydney's 'Central City', Parramatta CBD will grow significantly over the coming decades. This means city infrastructure such as green spaces, recreational facilities, arts and cultural destinations, and community spaces will face greater demands than ever before, necessitating clear infrastructure planning that aligns with City of Parramatta Council's ('Council') vision of being a liveable, sustainable, productive, and leading city.

Council has developed a draft list of the local infrastructure that the city will need in order to provide high-quality spaces and services that meet the needs of the CBD's growing community of residents, workers and visitors. Council is now seeking feedback on this needs analysis which, led by Council's *Statement of Vision and Priorities*, responds to the following themes:

- Managing growth and transport to improve accessibility, navigation and connectivity, which will provide a better city experience for pedestrians and active transport users.
- Promoting green spaces and the environment by creating and maintaining green spaces and transitioning towards a resilient city. The focus will be on developing Parramatta River as a key green public space for residents, workers and visitors, serving as a 'green' trail throughout the city.
- Providing opportunities for recreation and leisure activities to promote healthy and active lifestyles and maintaining a reputation as a premier sporting destination.
- Creating a strong economy with a strong city centre, which relies on improving the city's public domain backbone of streets and laneways, as well as flood management. There is also is a push towards becoming a Smart City.

- Having a community focus that fosters and celebrates a sense of community through the new Civic Centre, community centres and spaces, and childcare facilities.
- Supporting arts, culture, celebrations and destinations which enrich people's lives by creating a collective sense of identity for the community and spaces to spark new ideas and imagination.

The local infrastructure identified to support growth in the CBD and the realisation of these priorities will cost approximately \$1 billion over the next 40 years. This exceeds Council's estimated income from user charges (section 94a developer contributions), taxes (rates), and potential Government grants. The total estimated income from these sources will likely fall in the range of \$449 - \$605 million between 2016 and 2056. This means there is an anticipated funding gap of between \$394 - \$549 million.

This paper describes potential options for Council to reduce the funding gap, and feedback is welcomed on the funding options considered - in particular whether they are fair, equitable and transparent. Some of these options are currently used by Council (such as rates, developer contributions and grants), while others would represent new funding opportunities (such as City Deals and a planning uplift value share (PUVS) mechanism).

In particular, this paper focuses on funding options that Council can readily influence, and highlights the difference between user charges (i.e. meaning that those who benefit from something should pay for it) and taxes. Value sharing is a type of user / beneficiary charge that is currently not being implemented in Parramatta CBD, and this paper will provide more detail on value sharing and how it might apply in the CBD.

While the funding options considered may be able to raise a significant amount of funds, there remains

a funding gap. The identified local infrastructure needs of the Parramatta CBD represent an unprecedented level of investment, and finding the right mix of funding mechanisms will require innovative thinking and approaches.

Council welcomes your feedback on this discussion paper and the funding options considered, as well as on the draft infrastructure list. All submissions will be considered in Council's decision making process relating to CBD infrastructure planning and funding.

If you would like to provide feedback, visit www.cityofparramatta.nsw.gov.au.
Contents

1	Intro	oduction	1
	1.1	What this paper is about	1
	1.2	How this paper was written	1
	1.3	How to get involved	1
2	Plan	ning for growth in Parramatta CBD	2
	2.1	Frameworks guiding growth in Parramatta CBD	2
	2.2	Planning Proposal for the Parramatta CBD	5
3	Infra	structure Needs in the Parramatta CBD	8
	3.1	What the community has told us about the CBD, growth and infrastructure	8
	3.2	State-led infrastructure projects	9
	3.3	Local infrastructure needs of the CBD	10
	3.4	Funding the CBD's infrastructure needs	12
	3.5	Chapter 3 discussion questions	12
4	Infra	structure Funding Options	13
	4.1	Introduction	13
	4.2	Beneficiary charges	13
	4.3	Principles of beneficiary charges	13
	4.4	Types of beneficiary charges	14
	4.5	Taxes	15
	4.6	Government support	18
	4.7	Summary: establishing the funding gap	19
5	Valu	e Sharing: Harnessing the Benefits of Growth for all	22
	5.1	Introduction	22
	5.2	What is value sharing?	22
	5.3	Strategic endorsement of value sharing	23
	5.4	Why value sharing is not a tax	26
	5.5	The benefits of density	26
	5.6	Conclusion	26
6	Cou	ncil's Proposed Value Sharing Mechanism	27
	6.1	Introduction	27
	6.2	Introducing Council's proposed mechanism	27
	6.3	Modelling Council's proposed approach	30
	6.4	Council's proposed implementation of the PUVS	31
	6.5	Peer Review of Council's work on value sharing	31
	6.6	Value sharing's contribution to closing the funding gap	33
	6.7	Chapter 6 discussion questions	35
7	Disc	ussion Questions and Next Steps	36

- 7.1 Discussion questions
- 7.2 Next steps

Figures

Figure 1: Summary of Strategic Planning for Parramatta CBD

- Figure 2: Extract from Council's Statement of Vision and Priorities, 2016
- Figure 3: Expanded CBD footprint under the CBD Planning Proposal
- Figure 4: Land use zoning changes under the CBD Planning Proposal
- Figure 5: Estimated growth in dwellings and jobs the Parramatta CBD to 2056
- Figure 6: Infrastructure Funding Needs for Parramatta CBD (Low estimated potential income)
- Figure 7: Infrastructure Funding Needs for Parramatta CBD (High estimated potential income)
- Figure 8: Example of Base and Incentive FSR maps
- Figure 9: Example of Opportunity Site Map

Figure 10: Summary of FSR controls and PUVS mechanism

- Figure 11: Value sharing's potential impact on the funding gap (low estimated income)
- Figure 12: Value sharing's potential impact on the funding gap (high estimated income)

Tables

Table 1: Summary of Council rate types and their potential to help fund local infrastructure in the CBD

Table 2: Chapter 4 Summary table - funding options

- Table 3: Revenue potential under a Phase 1 and Phase 2 value sharing mechanism
- Table 4: Revenue potential under funding option (i)
- Table 5: Revenue potential under funding option (ii)

Boxes

Box 1: Greater Sydney Commission's vision for Parramatta CBD

- Box 2: Former Parramatta City Council's vision for Parramatta CBD
- Box 3: ACT Taxation Reform Plan Example
- Box 4: Case study in value sharing Infrastructure funding at Green Square
- Box 5: Case study in value sharing Infrastructure funding at Macquarie Park

Appendices

Appendix A: *Peer Review: Parramatta CBD Infrastructure Funding* Appendix B: *Draft Parramatta CBD Infrastructure Needs Analysis*

1 Introduction

1.1 What this paper is about

This paper identifies that Parramatta CBD will grow significantly over the next 40 years, and will need additional local infrastructure of all kinds costing approximately \$1 billion. Council cannot meet this requirement under its current revenue stream, creating a funding gap.

This paper introduces the additional infrastructure needs and assesses several options to reduce the funding gap. It highlights the challenges and opportunities of these options, and recommends a value sharing mechanism as a potential way forward.

This paper aims to provide the community with an understanding of infrastructure planning and funding in the CBD by:

- Providing an overview of the plans and strategies put in place by Council, the State Government and the Greater Sydney Commission, which all identify Parramatta's strategic importance.
- Highlighting that Parramatta CBD is set to grow significantly over the coming decades and identifying what infrastructure is needed to support this growth.
- Assessing several funding options, identifying challenges and opportunities, and demonstrating that a value share mechanism shows strong potential to help reduce the funding gap.
- Determining how a value share mechanism would work in a fair, equitable and transparent manner.
- Outlining a way forward and inviting the community and industry to provide feedback on Council's work to date.

1.2 How this paper was written

This paper is the result of a highly collaborative effort between City of Parramatta Council and Aurecon Australasia Pty Ltd. Aurecon also undertook an independent review of Council's original analysis around value sharing; this is provided as **Appendix A** to this paper, and provides further information on assumptions, market feasibility analysis and results. Council staff have undertaken a draft needs analysis identifying what local infrastructure will be needed in the CBD and this is included at **Appendix B**.

Parts of this work were also informed by discussions with staff of the Department of Planning and Environment and Greater Sydney Commission, and interviews with developers and real estate agents.

1.3 How to get involved

Council welcomes your feedback on this discussion paper, especially on the infrastructure list and the funding options considered. All submissions will be considered in Council's decision making process relating to CBD infrastructure planning and funding.

If you would like to provide feedback, visit www.cityofparramatta.nsw.gov.au.

2 Planning for growth in Parramatta CBD

The NSW Government has identified Greater Parramatta to the Olympic Peninsula (GPOP) and surrounds as Sydney's Central City³⁴, with employment and population forecast to grow significantly over the coming decades. Parramatta CBD is one of four important precincts within GPOP.

Council and the NSW State Government are preparing for this growth and have developed several plans and frameworks outlining the vision for Parramatta. This vision seeks to transform the city into a sustainable, liveable, and productive world class city.

2.1 Frameworks guiding growth in Parramatta CBD

Great cities need a plan for growth. There are many examples of cities that have expanded too quickly without any kind of planning. The results are chaotic at best, and greatly threaten the quality of life in that city⁵. In order to transform Sydney's Central City (with Parramatta CBD at its heart) into a world class city, a series of plans and frameworks have been developed by the Greater Sydney Commission, NSW State Government and Council, which all agree on the need to grow the city while also addressing the impacts arising from that growth. The following sections outline this clear strategic line of sight in more detail.



Figure 1: Summary of Strategic Planning for Parramatta CBD

³West Central District Plan (<u>http://greater.sydney/west-central-district</u>)

⁴ Greater Sydney Commission Greater Parramatta to Olympic Peninsula Vision (http://www.greater.sydney/gpop)

⁵ How to make a great city, McKinsey&Company (2013)

A Plan for Growing Sydney⁶

The NSW State Government envisions the transformation of Sydney's metropolitan area over the next 20 years in *A Plan for Growing Sydney* (The Metro Plan). The Metro Plan provides key directions and actions that will make Sydney more productive, liveable and sustainable.

The Metro Plan identifies Parramatta as a new priority growth area and a second CBD⁷. The NSW State Government commits in the Metro Plan to working with Council to review expansion opportunities in Parramatta CBD such as updating building height controls and removing other barriers to growth.

The result will be a city with a strong commercial centre supported by infrastructure together creating a dynamic and diverse place to work, live and play.

Draft West Central District Plan

The Greater Sydney Commission (GSC) has released draft District Plans for the six regions that make up the Sydney Metropolitan area. City of Parramatta sits within the West Central District, which also includes Blacktown, Cumberland and The Hills Local Government Areas. This area is forecasted to grow from 971,000 residents in 2016 to 1.5 million in 2036 (roughly 27,500 more people every year from now until 2036).

The *Draft West Central District Plan*, like the Metro Plan, has the priority of developing Parramatta CBD as the GSC seeks to "collaborate to create, own and deliver GPOP" [Greater Parramatta and the Olympic Peninsula; refer next section].

Greater Parramatta and the Olympic Peninsula Vision Document

The Greater Sydney Commission has also released a Vision Document for Greater Parramatta and the Olympic Peninsula ("GPOP")⁹. In line with the Metro and District plans, this document positions Parramatta as Sydney's "Central City", and identifies GPOP as "the geographic and demographic centre of Sydney". The Parramatta CBD and Westmead form one of four key precincts within GPOP.

BOX 1: Greater Sydney Commission's vision for Parramatta CBD

"The revitalised Parramatta CBD will be GPOP's commercial and civic centre. It will grow with a strong commercial core, an identifiable CBD skyline, a sound mix of finance, insurance, accountancy, legal, real estate, convention, public administration and IT services and a lively night-time economy. The revitalised Parramatta River will be the CBD's centrepiece and will connect to the prestigious commercial address of Parramatta Square via the Civic Link. Parramatta CBD will be designed as our central '30-minute city'"

GPOP Vision (Pg. 30)

⁹ http://www.greater.sydney/gpop

 ⁶ http://www.planning.nsw.gov.au/Plans-for-yourarea/Sydney/A-Plan-for-Growing-Sydney
 ⁷ Direction 1.2 of the plan being to "Grow Greater Parramatta – Sydney's second CBD"

Community Strategic Planning: *Parramatta* 2038

Council's current long-term Community Strategic Plan is *Parramatta 2038¹⁰*. This plan closely aligns with metropolitan and regional plans in terms of planning for major growth in Parramatta CBD. It envisions four major transformational opportunities for Parramatta:

- Development of Parramatta CBD, Westmead, Camellia and Rydalmere
- A light rail network and local and regional ring roads
- Parramatta River entertainment precinct
- A connected series of parks and recreation spaces

Community Strategic Planning: City of Parramatta Council's *Statement of Vision and Priorities*

City of Parramatta Council released a *Statement of Vision and Priorities* in December 2016, following its formation through the amalgamation of parts of the former Parramatta City Council and The Hills, Hornsby, Auburn and Holroyd Councils.

While the Statement outlines the Council and community's vision and priorities for the area as a whole, its guidance for the CBD is clear. It supports a strategic vision for a Parramatta CBD which includes a strong city centre, effective transport and a focus on sustainability and equity. Council's new *Vision and Priorities* are referenced in Figure 2.

Parramatta CBD Planning Strategy and background work

In 2015, the former Parramatta City Council adopted the *Parramatta CBD Planning Strategy*¹¹, which, through careful consideration of urban design and economic outcomes, envisioned Parramatta's CBD as a world class city.

It concluded that significant changes to local planning controls would be required to drive change and growth in the CBD, and provided a clear implementation plan for delivery of a new planning framework for the Parramatta CBD. This Strategy envisioned the preparation of a major Planning Proposal to change planning controls in the CBD to allow for significant growth and development, and help to realise the strategic vision laid out across all levels of planning for the CBD. This Planning Proposal is described in more detail in the next section.

BOX 2: Former Parramatta City Council's vision for Parramatta CBD

"Parramatta will be Australia's next great city, defined by landmark buildings and high quality public spaces with strong connections to regional transport. It will respect its heritage, be an exemplar in design excellence, facilitate job growth and ensure its streets are well activated."

Parramatta CBD Planning Strategy 2015

¹⁰ https://www.cityofparramatta.nsw.gov.au/sites/ council/files/inline-

files/Community%20Strategic%20Plan%202038.pdf ¹¹ https://www.cityofparramatta.nsw.gov.au/sites/council/ files/inline-

files/Appendix_2_Parramatta_CBD_Planning_Strategy.pdf



Figure 2: Extract from Council's Statement of Vision and Priorities, 2016

2.2 Planning Proposal for the Parramatta CBD

The clear strategic line of sight for Parramatta CBD described in the previous section is given statutory (legal) expression through a major Planning Proposal for the Parramatta CBD. Planning Proposals are the legal mechanism for changing planning controls, like land use zoning, height and floor-space ratio (FSR) controls. This section explains in more detail the changes proposed under the Parramatta CBD Planning Proposal, which was endorsed by the former Parramatta City Council in April 2016, and is currently awaiting a "Gateway Determination" from the Department of Planning and Environment to allow public exhibition and community consultation.

An expanding CBD

The CBD Planning Proposal proposes to expand the boundary of Parramatta's statutory "City Centre" as shown in the red shaded areas of Figure 3 below.

The Planning Proposal will not make any changes to the existing controls around the Park Edge Highly Sensitive Area adjacent to the world heritage-listed Old Government House and Domain Parramatta Park, the Stadium/Sports and Leisure Precinct and surrounds. However, these areas will still remain as part of the City Centre boundary.

Consideration of a further expansion of the city centre boundary will be undertaken as part of future planning studies in the CBD.

Figure 3: Expanded CBD footprint under the CBD Planning Proposal

A changing CBD

The CBD Planning Proposal proposes changes to existing land use zones to protect and grow a strong commercial core, surrounded with vibrant mixed use areas.

These changes include expansion of the commercial core (bright blue) zone and changing some low and medium density residential zones surrounding the core to mixed use (purple) zones. These changes are shown in more detail at Figure 4 below.

In addition, the proposal will respond to several key issues facing Parramatta CBD in its new role as the heart of Greater Sydney's Central City. These issues include lifting environmental performance of buildings, creating active streets, protecting solar access to key public spaces, facilitating design excellence, responding to airspace operations issues, maintaining heritage protections, and managing flood risks.



Figure 4: Land use zoning changes under the CBD Planning Proposal

A growing CBD

Along with changes to land use controls, the CBD Planning Proposal also proposes to amend the existing planning controls of height and density. The key purpose of these changes is to meet the employment and housing targets outlined in the Parramatta CBD Planning Strategy.

As shown in Figure 5 below, the CBD Planning Proposal releases capacity for about **48,700 additional workers** and **42,600 additional residents**¹². This estimate is based on two-thirds take-up of the total floor space area released under the Planning Proposal, and is estimated to occur over the period from 2016-2056 (40 years). This growth means that all types of infrastructure in the CBD will experience new pressures and demands.



Figure 5: Estimated growth in dwellings and jobs the Parramatta CBD to 2056

The Planning Proposal will allow for significant development of the built environment in Parramatta's CBD, but this will generate a need for significant investment in new and upgraded local infrastructure. Without this investment in infrastructure, growth in the CBD will not occur in a well-managed and appropriately-serviced way, and the strategic vision for the heart of Greater Sydney's Central City will not be achieved. Chapter 3 will focus in more detail on this topic.

¹² Based on 2.1 persons/dwelling.

3 Infrastructure Needs in the Parramatta CBD

3.1 What the community has told us about the CBD, growth and infrastructure

Council's *Statement of Vision and Priorities* identifies 'creating a strong economy with a strong city centre', as a key priority that drives the conversation around infrastructure needs of the growing centre of a sustainable, liveable and productive city.

Addressing this priority will specifically involve:

- Creating a city centre that generates jobs and attracts business and investment
- Creating a well-connected, efficient city that attracts knowledge intensive jobs and promotes Parramatta as a centre for ideas
- Ensuring that Parramatta Square is a key economic driver for the CBD
- Working with key partners to create a high valueadding, employment hub and driving force behind the generation of new wealth for Western Sydney.

However, growing a city centre that is economically strong is only part of the picture. The community has clearly told Council that growth in Parramatta must be well-managed, and that we must harness the benefits of growth for all. This acknowledges that all stakeholders in the CBD – whether residents, businesses, workers, visitors, landowners or developers – will benefit when appropriate infrastructure is provided to service the growth in the CBD. Servicing growth in the Parramatta CBD will include providing better spaces, experiences and transport as follows:

Better Spaces

 Create innovative spaces and places for the community

- Create a place that encourages social connectivity and is inclusive and accessible for all
- Ensure that green and open spaces are created, protected and maintained in line with population growth

Better Experiences

- Provide a variety of cultural experiences and attractions unique to Parramatta which make it a destination of choice for residents and visitors
- Create a green city by creating and maintaining green spaces, bushland and waterways for residents and visitors to enjoy
- Create a resilient city that uses less energy and water as the city grows – doing more with less

Better Transport

- Create more active travel options and maintaining accessible and high quality facilities to promote healthy and active lifestyles
- Work with government partners to improve connections and traffic flow
- Manage the parking and transport needs of residents, visitors and workers

These actions will benefit the CBD's growing community of businesses, workers, residents and visitors. These actions will also benefit landowners and developers, as they will improve the Parramatta CBD's attractiveness, competitiveness and land values, and because the increased densities that benefit developers and landowners will not be supportable without these improvements to infrastructure.

After briefly touching on important regional infrastructure projects, this chapter will lay out the local infrastructure which is necessary to allow increased densities and growth in the CBD to occur.

3.2 State-led infrastructure projects

Due to its metropolitan and regional importance Parramatta is and will continue to be the focus of significant State-led infrastructure investment.

Many of these State-led infrastructure projects are outlined in the *Parramatta Strategic Framework*¹³, and regional infrastructure investment in Parramatta CBD will allow the city to grow over the coming decades. Key regional projects include:

- Major transport projects (like Parramatta Light Rail and Sydney Metro West),
- Major investment in new and upgraded educational facilities (like the Parramatta Schools projects),
- Major Cultural Facilities (like the relocation of the Museum of Applied Arts and Sciences),
- Major sporting facilities (Western Sydney Stadium), and
- Major utility upgrades.

These State-led projects are important city-shaping infrastructure elements which will help to effectively service and manage the growth of Parramatta CBD.

Council remains an important partner and stakeholder on these activities. However, it is not generally the responsibility of Council to fund and deliver major regional infrastructure projects. Therefore, this paper focuses on **local** infrastructure, and a few locally-led aspects of State-led projects (for example, works led by Council resulting from the Parramatta Light Rail project).

¹³ http://www.greater.sydney/news/parramatta-strategicframework - a joint endeavour by Infrastructure NSW, the Department of Planning and Environment, Office of the Government Architect, the former Parramatta City Council and consultants Terroir.

3.3 Local infrastructure needs of the CBD

The growing Parramatta CBD will require all kinds of local infrastructure projects, including both new infrastructure and upgrades to existing assets.

Six of the eight priorities identified in Council's *Statement of Vision and Priorities,* have direct implications for infrastructure provision in the CBD, Therefore, this Discussion Paper organises the projects around these six priorities as follows:

- Managing Growth and Transport
- Promoting Green Spaces and the Environment
- Providing Opportunities for Recreation and Leisure
- Creating a Strong Economy with a Strong City Centre
- Having a Community Focus
- Supporting Arts and Culture, Celebrations and Destinations

(Please note that many of the local infrastructure projects identified are related to more than one of these priorities.)

The other two priorities identified in the *Statement* are less directly related to infrastructure provision in the CBD, but still underpin all of Council's actions relating to planning for growth in the CBD. These two priorities are *Building a Stronger, More Innovative Council for our Community's Future* and *Creating Vibrant Neighbourhoods and Precincts.*

The following sections will describe the infrastructure needs under each of the above six priorities. **Appendix B** of this Discussion Paper contains the *Draft Parramatta CBD Infrastructure Needs Analysis,* including projects descriptions, estimated costs and timelines.

Managing Growth and Transport

Managing growth and transport in Parramatta CBD will require improvements to accessibility, navigation and connectivity. Of particular importance will be improving the experience of pedestrians and active transport users.

Many projects needed for the Parramatta CBD have already been identified through the City Ring Road¹⁴ vision. The City Ring Road is mainly comprised of Victoria Road, and O'Connell, Parkes and Harris Streets, and will help create an identifiable boundary to the city centre with entrances and thresholds, while also simplifying traffic flow, and reducing through-traffic and congestion in the CBD. Projects related to the City Ring Road include pedestrian improvement projects, intersection upgrades, median islands and road widenings.

A program of other road widenings will also be needed in association with further development of public and private transport options throughout the CBD.

New and upgraded active transport connections are also needed to span the Parramatta River and improve connectivity. These include improvements to Gasworks Bridge and Barry Wilde Bridge, as well as a new pedestrian bridge at Morton Street.

More detail on each of the required projects associated with *Managing Growth and Transport* can be found in **Appendix B**.

Promoting Green Spaces and the Environment

Green spaces and the environment will be promoted throughout the CBD as it develops, with a particular focus around the Parramatta River as a key green public space for residents, workers and visitors to enjoy.

Under the *Parramatta City River Strategy*¹⁵, Council is planning for upgrades to every section of the Parramatta River foreshore from the Marsden Street Weir to Parramatta Quay. This includes major projects at River Square and Parramatta Quay (on which Council will partner with key State agencies), public domain upgrades to the river foreshore throughout the CBD precinct, and improved active transport links to and along the foreshore.

A naturalisation project at Brickfields Creek and a river pool have been identified as requirements for Council to further its aim of returning swimming to Parramatta's natural waterways and to broaden recreational options in the CBD.

Upgraded parking and access paths at Lake Parramatta and significant investment in the CBD portion of the Parramatta Ways project will also help Council work to this priority, as will a major street tree planting project to help green the CBD.

More detail on each of the required projects associated with *Promoting Green Spaces and the Environment* can be found in **Appendix B**.

¹⁴ http://www.designparramatta.com.au/projects/

¹⁵ https://www.cityofparramatta.nsw.gov.au/ sites/council/files/2016-11/River%20City%20 Strategy%20Volume%2001%20Report%20Part1.pdf

Providing Opportunities for Recreation and Leisure

As the CBD grows, Council will need to provide new and improved opportunities for recreation and leisure to the community.

A new aquatic and leisure centre is needed to support the growing CBD community and to provide a range of accessible, sustainable, attractive and safe recreation opportunities.

An increased number of residents and the shortage of potential new recreation sites in the high-density CBD mean that our existing green open spaces must be significantly upgraded to meet new demands. These upgrades include improvements to sportsgrounds – like all-weather playing surfaces, better lighting and improved amenities – and provision of new multi-purpose outdoor spaces as well as playgrounds.

More detail on each of the required projects associated with *Providing Opportunities for Recreation and Leisure* can be found in **Appendix B**.

Creating a Strong Economy with a Strong City Centre

All of the projects described in this paper underpin this priority, but of particular importance are three key needs: improving the public domain, streets and laneways of the CBD, addressing flooding issues and undertaking Smart City initiatives.

The backbone of any city is its streets and major public spaces, which in Parramatta CBD are the river foreshore (see 3.3.2), the Civic Link, and Parramatta Square. The public domain of Parramatta Square will serve as the heart of the new city centre, and the upcoming Civic Link project will connect Parramatta Square and the river to complete the CBD's core of world-class public spaces. Major CBD streets and laneways also need upgrades to create great places for the community, cope with additional foot traffic and elevate them to the standards of a world-class city centre.

Major infrastructure works will be needed to address both localised overland flooding issues, as well as flooding risks associated with the river itself. These works will help to protect people and property in the CBD, as will installation of an early flood warning system. This system also connects to Council's vision for Parramatta as a Smart City, which raises several needs for infrastructure like improved CCTV, multi-function street poles, and utilities rationalisation. More detail on each of the required projects associated with *Creating a Strong Economy with a Strong City Centre* can be found in **Appendix B**.

Having a Community Focus

The local community of the Parramatta CBD will grow by an estimated 42,600 additional residents over the next 40 years. This growing community will need all sorts of new and upgraded community facilities.

A key need of the growing Parramatta CBD is the new Civic Centre at 5 Parramatta Square, which has been the subject of a recent international design competition, and will provide state of the art library and community facilities.

Aside from the major facility at Parramatta Square, a wide variety of new community spaces are needed throughout the CBD. These include a new community centre, new childcare centres, and flexible community rooms of various sizes located throughout the CBD and CBD fringe. Because the growing CBD will impact on community facilities in North Parramatta, provision is also made here for a contribution towards those facilities.

As the CBD develops, we will also need better infrastructure for assisting disadvantaged community members with food and amenities.

More detail on each of the required projects associated with *Having a Community Focus* can be found in **Appendix B**.

Supporting Arts and Culture, Celebrations and Destinations

Many cultural infrastructure projects and facilities will be required to support arts, culture, celebrations and destinations in the new Parramatta CBD.

A key major cultural project will be modernising and expanding Riverside Theatres, to accompany the public domain river foreshore projects outlined in a previous section. Expansion of Parramatta Artists Studio and a new world class art exhibition and gallery space will elevate the place of the arts in the Parramatta CBD, providing new and improved spaces for production and presentation of diverse art forms. Aboriginal cultural infrastructure will showcase the local Darug peoples' sites of significance, history and contemporary connections to Parramatta. An incubator in the new Civic Link will house cultural organisations focusing in media, digital and creative industries, and provision is also made for appropriate storage of and public access to cultural and archaeological materials (as

Council's collections grow due to increased construction in the CBD).

More detail on each of the required projects associated with *Supporting Art and Culture, Celebrations and Destinations* can be found in **Appendix B.**

3.4 Funding the CBD's infrastructure needs

The preceding sections – along with the detailed *Draft Parramatta CBD Infrastructure Needs Analysis* at **Appendix B** – lay out what is needed to transform Parramatta CBD into a world class city centre over the next 40 years. This list presents exciting opportunities and challenges for Council and the community.

The total cost of local infrastructure needs in the CBD is currently estimated at approximately **\$1 billion**.

It is important to remember that infrastructure planning for a growing centre is a dynamic process. The estimates contained in the *Draft Infrastructure Needs Analysis* reflect Council's current knowledge about the needs and costs of infrastructure in the CBD, and will continue to be refined as Council and its partners progress the planning and delivery of CBD infrastructure projects.

The *Draft Infrastructure Needs Analysis* reflects what will be an unprecedented level of investment in local infrastructure for the Parramatta CBD, and finding the right mix of mechanisms to fund this infrastructure will require innovative thinking and approaches. This will be the focus of the following chapters of this paper.

3.5 Chapter 3 discussion questions

- Has Council considered the right types of local* infrastructure projects in its *Draft Parramatta CBD Infrastructure Needs Analysis* at Appendix B?
- 2. In your mind, what are the most important local* infrastructure priorities for Parramatta CBD?
- 3. Is there anything you feel is missing from the draft list?

*Remember that Council is not directly responsible for infrastructure like schools, hospitals and public transport. While Council advocates for the community and partners on projects where appropriate, provision of these types of infrastructure are generally the responsibility of the NSW State Government.

4 Infrastructure Funding Options

4.1 Introduction

This chapter will explore different funding options that Council might use in meeting the approximately \$1 billion local infrastructure funding requirement laid out in the last chapter.

First, it is helpful to clearly establish the difference between infrastructure funding and financing. In 2012, Infrastructure Australia, an independent Federal Government body that has the mandate to prioritise and progress nationwide infrastructure, commissioned a paper entitled Infrastructure *Finance and Funding Reform*¹⁶. It clearly explains the difference between financing and funding of infrastructure projects. The term funding, as defined in this paper, refers to sourcing an amount of money to fund infrastructure. However, financing refers to the way in which debt and/or equity is raised for the delivery and operation of an infrastructure project. with the expectation to 'pay back' the borrowed amount with interest.

The Infrastructure Australia report also presents a useful framework for considering who should pay for infrastructure and how those arrangements should be structured. The idea proposed in this report was supported and improved in the recent 'Value Capture' discussion paper released by the Australian Government in November 2016¹⁷. Together, these papers highlight an Australia-wide infrastructure investment backlog, mainly as a result of funding constraints from its two main sources user charges and taxes. This chapter will address each of these funding source types in turn, beginning with user or "beneficiary" charges.

4.2 **Beneficiary charges**

Beneficiary charges can be explained as, 'those who benefit from infrastructure should pay for it'. Beneficiary charges include both direct and indirect user charges.

Infrastructure projects have both direct and indirect beneficiaries. Take, for example, a museum, which has direct users/beneficiaries in the form of ticket holders who may be residents, workers or visitors and who directly benefit due to increased utility (satisfaction) from visiting the museum. However, the museum also has indirect users/beneficiaries including property owners; for example, store owners may benefit from increased shoppers due to increased foot traffic from people visiting the museum, and home owners may see an uplift in the value of their property due to their proximity to the museum. Indirect users may never set foot in the museum but may benefit from positive externalities (an economic term for benefits enjoyed by a third party).

This paper suggests that the most appropriate way to fund the infrastructure projects listed in Appendix B is by moving towards a 'beneficiary-pays' (charges) model to demonstrate stronger links between infrastructure funding sources and those who benefit.

4.3 **Principles of beneficiary** charges

The core challenge of a beneficiary charges model is defining who the direct and indirect users are, their 'willingness-to-pay' for the associated benefits, and how much they should be charged.

There are many ways of pricing shared costs or benefits; prices might be developed with reference to the following methods:

¹⁶ http://infrastructureaustralia.gov.au/policypublications/publications/Infrastructure-Finance-Reform-Issues-Paper-Report-prepared-by-the-Infrastructure-FWG-2012.aspx ¹⁷ http://investment.infrastructure.gov.au/whatis/Value-

Capture-Discussion-Paper.pdf

- Market prices: Prices in existing markets are the best indicator of consumer demand and supplier provision. Market prices may be changed by new demand or preferences due to an infrastructure initiative under consideration.
- Hedonic prices: In the context of land use planning, hedonic prices are derived from factors that determine land values such as changes to site accessibility or changes to existing planning controls. These factors ultimately drive changes in land values which may be realised when the infrastructure project under consideration is planned or implemented.
- Revealed preference: This method analyses consumer's purchasing habits to uncover their preferences. Travel cost methods use information about how much people are willing to pay to visit locations, to infer to how much they value changes in those attributes. (For example, the costs that people are willing to incur to visit Parramatta CBD for recreation is an indication of the benefits provided by the CBD infrastructure.)
- Stated preference: This principle explains the assumed choices and sacrifices of beneficiaries by using questionnaires to obtain a value based on the results. This method indirectly includes the inputs of the project users or beneficiaries in its planning process.

The application of these principles and concepts must be within the established regulatory framework or other governmental requirements to estimate a value to the benefits of the infrastructure project. The use of these principles and concepts also vary from project to project and the nature of the project will essentially determine which principle or combination of principles to use. This paper recommends that the collective use of the above principles and concepts is fundamental in developing beneficiary charging.

4.4 Types of beneficiary charges

Council already has a few beneficiary charges in place, such as developer contributions and a variety of other service fees and charges, where the collected charges are typically reinvested into infrastructure or facilities.

Developer contributions

Developer contributions are a form of *direct* beneficiary charge, in that the developer benefits from being able to develop in an area and, in return, is required to contribute towards infrastructure, in either monetary terms or works in kind, as a way of remediating the impact of their development on the community. Essentially, developer contributions reflect a 'user charges' system in that those who create the demand for infrastructure help pay for the provision of that infrastructure. Developer contributions are levied on developers for efficiency, but are passed through to property owners (on sale of property) and ultimately residents (through ownership or rent). These contributions are collected by Council through Section 94 and 94A of NSW Environmental Planning and Assessment Act (EP&A) 1979.

Section 94 Contributions

Section 94 contributions require a direct connection (a "nexus") to be established between the development and the infrastructure it is contributing towards (through monetary contributions and/or works in kind).

Section 94 contributions are limited by "caps" set by the NSW State Government. The relevant cap for Parramatta CBD is \$20,000 per dwelling (for dwellings in in-fill development areas), which has been set by a Ministerial Direction. Any upward changes to Section 94 contributions as a funding source are likely to be controversial across government and industry, and would require support at the State level to progress amendment.

City of Parramatta Council does not currently use Section 94 contributions in the CBD. Should Council attempt to prepare a section Section 94 contributions plan that authorises a Section 94 contributions above the cap, the contributions plan would need to be reviewed by IPART and the contributions collected could only be directed towards items on the "essential works" list. In order for the Parramatta CBD to achieve its potential as the heart of Greater Sydney's Central City, a broader range of infrastructure is needed than that of the "essential works" list.

Section 94A Contribution

Section 94A contributions do not require a direct nexus to be established, and instead take the form of a levy based on a percentage of the total cost of development.

Currently in the Parramatta CBD the Section 94A levy is 3% of the cost of development (where the cost of development exceeds \$250,000; the levy is not progressive). The cost of development is determined in accordance with Clause 25J of the Environmental and Planning Assessment Regulation 2000 (EP&A Reg.). Despite the significant additional capacity offered under the Parramatta CBD Planning Proposal, the current 3% levy will be insufficient on its own to fund the local infrastructure requirements associated with Parramatta's new role as the heart of Greater Sydney's Central City. Council has estimated the potential range of Section 94A contributions under the Parramatta CBD Planning Proposal at \$207 -\$323 million.¹⁸ This range reflects two site consolidation scenarios – the low range being minimal site consolidation and the high range being a greater amount of site consolidation.

Summary of Council's Past Work on Section 94 and 94A Contributions

Council has considered increasing developer contributions, and with the input of consultants GLN Planning modelled the impacts this would have on market feasibility and ability to raise funding for local infrastructure. The main finding was that increasing Section 94A contributions to 4.5% would only marginally increase funding, and that changes to Section 94 contributions are complex as a relationship (nexus) needs to be formed for all different types of developments and brings in a number of legal issues.¹⁹

Fees and charges

Fees and charges are (mostly) direct charges for providing services. Fees and charges are levied by Council on all chargeable services, requests, applications, approvals, licences, hire bookings and memberships. Fees and charges tend to have relatively fixed levy-base, so these sources are unlikely to grow substantially and support the funding requirement of major infrastructure projects. Furthermore, they would not relate generally directly to the provision of local infrastructure.

4.5 Taxes

Taxes are mandatory payments that usually have no direct link with infrastructure, though tax revenue can go towards funding projects that benefit the community as a whole. Types of taxes include income tax, GST and capital gains tax, land tax, transfer of assets duty, and Council rates; these taxes are governed by different levels of governments.

Federal taxes

Income tax, GST and capital gains tax are collected by the Federal Government. Tax payers rarely know specifically how their tax dollars are being spent. It is unlikely that federal taxes will go directly to spending the types of local infrastructure outlined in Appendix B.

Land tax

Land tax is collected by the NSW State Government, and applies to land regardless of whether income is earned from the land. Payment of land tax generally arises when the sale or transfer of land occurs. Land tax would not generally be directed towards local infrastructure projects.

Transfer of land or business duty

Transfer of land or business duty (formerly known as stamp duty) is a duty levied by the NSW State Government on the sale or transfer of land, including improvements and, business assets, and a declaration of trust over dutiable property in NSW. The buyer or seller is liable to pay the duty, and must be paid within three months from the transfer arising. When purchasing property 'off the plan', the duty must be paid within a three month period from the date of completion of the agreement, the assignment of the whole or any part of the purchaser's interest under the agreement, the expiration of 12 months after the date of the agreement, whichever occurs first. Stamp duty is not currently directed towards local infrastructure provision, nor is a change towards this outcome expected.

Box 3: ACT Taxation Reform Plan Example

The ACT Taxation Reform Plan 2012 makes taxes fairer, simpler and more efficient. On July 1 2012, the ACT Government announced it would phase out stamp duty and increase reliance on rates. This is part of key tax reforms undertaken by the ACT Government to generate wider economic benefits. This option was open to the ACT Government since it collects both rates and stamp duty – unlike City of Parramatta Council.

20

Council Rates

Rates are an important source of Council revenue that can be used to provide essential infrastructure

¹⁸ Refer to Council Business Papers 27 June 2016, Item 7.4 for further discussion.

¹⁹ Refer to Council Business Paper 27 June 2016, Item

^{7.4,} Attachment 2 for further discussion.

²⁰ http://www.cmd.act.gov.au/open_government/inform/ act_government_media_releases/barr/2012/fairer,_simpler _and_more_efficient_taxes

and services such as waste collection, drainage maintenance, public parks, and building inspections and construction certificates. These rates are determined in accordance with the provisions of the Local Government Act 1993, and in NSW, are calculated by land value²¹, multiplied by a rate-inthe-\$, (that is the rate multiplied by value of the land). The Act also restricts rate revenue growth by rate pegging that is set by the Independent Pricing and Regulatory Tribunal (IPART) which sets the maximum increase allowed in each Council's general revenue for the financial year. It applies to general income in total, not specifically to individual ratepayers, and means these rates may vary by higher or lower than the rate peg. (Note rate pegging does not apply to storm water, waste collection, water and sewerage charges.)

IPART sets the rate peg each year and in doing so, they consider the Local Government Cost Index (LGCI), which measures price changes over the previous year for the goods and labour an average Council will use, as well as productivity changes over the same period. The increase approved by IPART for 2016/17 is 1.8%, on the basis that LGCI was 1.78% and no adjustment for productivity.

Notwithstanding the above, Council can apply for a special variation to the rate peg, which if accepted, allows Council to increase their general revenue by more than the rate peg. Note that IPART accepted a special variation requested by the former Parramatta City Council in 2011. As part of the request, Council sought to replace an existing time-limited special variation of 4.9% from 2013/14 onwards with a special variation of a similar size to be incorporated into its rate base permanently. This was approved, and the 4.9% increase in general income is to be allocated as such:

- 2 percentage points applying to all ratepayers as part of the Council's financial sustainability strategy
- 2.9 percentage points applying to certain business ratepayers for the CBD Infrastructure and Economic Development special rates.

Table 1 summarises the types of rates that Council currently implements that could potentially fund local infrastructure, and their potential to raise revenue, stability and limitations.

The rate at which Council revenue grows is generally a reflection of (Council) budgetary requirements. However, 'rate pegging' limits how much the Council can spend on local infrastructure and other facilities. When there is a positive revenue growth through general growth, it can be used to fund local infrastructure and other assets.

Council has estimated the potential for rates growth under the Parramatta CBD Planning Proposal which is likely to be directed towards funding infrastructure at \$111 - \$151 million over the expected build-out period (2016-2056). This estimate is based on the rate types laid out in Table 1, and represents a combination of projecting a proportion of revenue from general rates towards infrastructure, as well as applying the current special infrastructure rates to the growth envisioned under the Planning Proposal. The range represents different possibilities for how much of general rates revenue might be directed towards capital works. This growth in rates will make an important contribution to funding local infrastructure, but would not address the entire \$1 billion funding requirement.

²¹ Land values are issued every four years and are defined under Section 64 of the Valuation of Land Act 1916.

Rate	Potential to Raise Revenue*	Stability and Predictability	Limitation
Ordinary Rate	High	Stable and predictable Depends on external economic	Ad valorem** subject to a minimum
		conditions	Only a proportion of ordinary rates will go to funding infrastructure
Special Rates for Open Space	High	Stable and predictable Depends on external economic conditions	Part ad valorem and part base amount (fixed amount)
Special Rates for CBD Infrastructure	Medium	Stable and predictable Depends on external economic conditions	Limited tax base
Special Rates for Economic Development	Medium	Stable and predictable Depends on external economic conditions	Limited tax base
Annual Charges for Storm Water & Waste Management services	Low	Stable and predictable Depends on external economic conditions	Limited tax base Directed towards storm water and waste management

Table 1: Summary of Council rate types and their potential to help fund local infrastructure in the CBD

*High is defined as having a tendency to expand annually, is stable and has been a predictable source of revenue over the last five years

Medium is defined as having a slight tendency to expand and has been a predictable source of revenue over the last five years Low is defined as less stable and a less predictable source of revenue

** Ad valorem rates are not fixed, but depend on property value as determined by the NSW Valuer General.

Limitations of Council funding

Council has limited taxation powers beyond rates, which limits tax as a source of funding for local infrastructure. Furthermore, taxes are not cost-less – increased taxes reduce the money available to spend on other goods and services – representing a deadweight loss²² to the economy.

The likelihood of borrowing is also limited due to increases in the Council's net debt positions, which will generally have an adverse effect on their ability to maintain good credit ratings. Even if Council were to take on debt to fund the infrastructure projects described here, it would have to have an appropriate funding stream to pay back the debt. Any increase in Council revenue will provide a stronger capacity for Council to borrow funds and subsequently deliver local infrastructure.

While Council does hold some reserves, this infrastructure plan identifies new infrastructure needs that are generally *beyond* Council's current funding and operational planning. Council's reserves are largely already earmarked for particular projects. In reality, Council only has an approximately \$1-2m yearly operating surplus.

4.6 Government support

Generally, both federal and state governments can be a source of limited funding for local infrastructure.

Direct Funding by the State Government

As part of its *State Infrastructure Strategy*²³, the NSW State Government made a proposal to directly fund infrastructure projects in Parramatta that are seen as having an influence on a regional scale, such as light rail, Western Sydney Stadium, Museum of Applied Arts & Sciences, and public school upgrades. Much of the funding for these projects will come from taxes and levies imposed by the NSW Government through its Office of State Revenue. These projects have been excluded from the *Draft Infrastructure Needs Analysis* at Appendix B, as Council is not generally responsible for funding these larger-scale projects.

The NSW State Government is also making indirect contributions to the economy by locating State Government agencies in Parramatta CBD, providing revenue to building owners in the form of rent, and additional pedestrian foot traffic that supports local businesses and employment. Council benefits from these economic activities.

Federal

The Federal Government has not directly funded local infrastructure in Parramatta – other than the National Broadband Network (NBN), which operates under a beneficiary charges model. However, similar to State Government, the Federal Government has indirectly contributed to the economy by locating its agencies such as the Australian Taxation Office (ATO) and courts in Parramatta CBD.

Grants

Government grants and subsidies are made under various circumstances to support community initiatives that achieve goals and objectives consistent with government policy. Grants may be covered by legislation or regulation, or be subject to cabinet, ministerial or administrative discretion. They range in their accountability requirements from highly complex arrangements to the relatively informal.

For local councils in Australia, grants are awarded by both Federal and State governments. For example, the Roads to Recovery Programme, is a grant from the Federal Government to local councils through the state local government grants commission. The State Government also provides grants and smaller-scale funding on a project-byproject basis. However, such grant funding arrangements vary from project to project and year to year, and are therefore not predictable.

Private grant / gift funding is rare in Australia, especially for local infrastructure. Therefore, grants, if any, have to come from State or Federal governments for specific projects that have an impact on a region wide scale and are less likely to be directed to local infrastructure. This means it is unlikely that grants will be able to fund most of the projects listed in Appendix B, though they may contribute towards individual projects on an ad-hoc basis.

Council has estimated the potential contribution from State grants towards individual infrastructure projects listed in Appendix B at around \$131 million over the expected build-out period of 2016-2056 – only a small part of the total need of approximately \$1 billion.

http://library.bsl.org.au/jspui/bitstream/1/611/1/
 Costs_of_taxation.pdf
 https://www.nsw.gov.au/improving-nsw/projects-and-

²³ https://www.nsw.gov.au/improving-nsw/projects-andinitiatives/state-infrastructure-strategy/#resource-allocation

City Deals

City Deals are partnerships between the Australian Government and State, Local and Territory governments for investments or projects that enhance quality of life and the knowledge-based economy. These deals provide cities of all sizes with a level playing field to receive funding for investments through coordinated governance, strategic planning, investment and reform.

The Australian Government has so far committed to early City Deals for Townsville, Launceston, and Western Sydney. As of 21 October 2016, the Prime Minister, Malcolm Turnbull, and the then-NSW Premier, Mike Baird, signed a Memorandum of Understanding to formalise a partnership to work together on the Western Sydney City Deal. It will involve "deliver[ing] a Western Sydney Airport and leverage[ing] other key infrastructure investments to catalyse jobs growth and better transport links".²⁴

Further details on how the Australian Government intends to roll out a City Deals program will be released in 2017. At this stage, the Australian Government has yet to indicate support for a City Deals partnership involving GPOP or Parramatta CBD.

4.7 Summary: establishing the funding gap

This chapter has reviewed potential funding sources which could contribute to funding the approximately \$1 billion local infrastructure requirement outlined in Chapter 3. The potential impact of these sources is summarised in Table 2 below, which demonstrates that the combined expected funding sources are insufficient to meet the anticipated \$1 billion local infrastructure requirement. As shown in Figures 6 and 7 a funding gap of roughly \$394 - \$549 million remains.

All levels of governments are experiencing budgetary constraints. It is unlikely, for example, that the NSW State Government will take on additional borrowings to help fund local infrastructure, as its increased debt position will impact its current AAA credit rating. To service the gap, alternative funding sources need to be considered.

This funding gap must be considered by landowners and developers in the CBD, as this infrastructure is required to service the needs of a rapidly growing and densifying CBD. An appropriate mix of funding sources must be identified in order for the densities proposed under the Parramatta CBD Planning Proposal to proceed.

The next chapter explores another type of beneficiary charge that could help address the local infrastructure funding gap. Chapter 5 will introduce the definition, forms and benefits of planning uplift value sharing.

²⁴ https://cities.dpmc.gov.au/western-sydney-citydeal/documents/44846/download

Table 2: Chapter 4 Summary table – Funding options

	Chapter 4 St	ummary Table: Funding options	5	
Type of funding		Likelihood of contributing to Local Infrastructure	Estimated Contribution	
Beneficiary	Section 94	Not recommended due to capping and scope of works	n/a	
Charges	Section 94a	High likelihood	\$207 - \$323 million	
	Fees and Charges	No	n/a	
	Federal Taxes	Unlikely	n/a	
Tayaa	Land Taxes	Unlikely	n/a	
Taxes	Stamp Duty	Unlikely	n/a	
	Council Rates	High likelihood	\$111 - \$151 million	
	Direct State Funding	No (more likely for regional infrastructure projects)	n/a	
Government	Federal Funding	Unlikely	n/a	
Support	Grants	High likelihood (only for some projects)	\$131 million	
	City Deals	Unknown	n/a	



Figure 6: Infrastructure Funding Needs for Parramatta CBD (Low estimated potential income)

Figure 7: Infrastructure Funding Needs for Parramatta CBD (High estimated potential income)



5 Value Sharing: Harnessing the Benefits of Growth for all

5.1 Introduction

Value sharing is a funding strategy that seeks to raise contributions towards the capital cost of infrastructure from those who benefit from planning decisions or from the provision of infrastructure. Value sharing could be an important way for Council to alleviate (though not fully resolve) the funding gap for infrastructure provision.

Value sharing is not a tax. It is a mechanism that offers the potential to generate new funding streams by tapping into the value created by infrastructure or planning decisions for beneficiaries. This in turn can allow governments to deliver new infrastructure which they would not otherwise be able to fund, or to enable borrowings (financing) to bring forward planned infrastructure ahead of time. Delivering projects earlier also brings forward the benefits of those projects. Implementing value sharing could ensure that projects would be in a better position to deliver an adequate level of return.

Even though value sharing can help alleviate funding constraints, it is unlikely to provide enough funding for all required infrastructure – it is not a "silver bullet" to fund all infrastructure.²⁵

5.2 What is value sharing?

Value sharing is an alternative way of funding infrastructure projects. It is an approach that seeks to fund the planned infrastructure project by capturing some of the benefits that are generated from implementing the project or from related planning decisions. Well-planned public infrastructure such as the Parramatta CBD projects can create benefits for a diverse range of stakeholders, including property owners, developers and occupiers, public transport users and operators, businesses and employees, and government. Value generally accrues:

- Directly to those who use the infrastructure through improved quality of service or
- Indirectly to those in proximity to the infrastructure.

Infrastructure projects exist within broader networks of economic activity, which means that the infrastructure will create benefits for individuals, businesses and governments beyond those who directly use the infrastructure. This can lead to increased value of residential and commercial properties and land surrounding the new infrastructure; growth in commercial activities for businesses; and improved connections between individuals and businesses, encouraging greater market competition, and opening up new employment and labour market opportunities.

Broad forms of value sharing arising from infrastructure investment include the following:

User charging, which is a targeted way of ensuring users who derive benefits from infrastructure investment, such as a new motorway, rail line or utility asset, make a contribution to the provision, maintenance and operation of that asset. A charge is applied for the use of a specific asset each time the asset is used. The price charged typically depends on the use made of the asset and the length of time the asset is being used. Examples of user charges include various toll roads, access charges for ports, retail electricity, gas, water and telecommunications network usage charges, and public transport fares. As will be discussed further in this chapter (see Sections 5.4 - 5.5), value sharing based on a density-bonus scheme is also a type of user charge, as planning decisions to increase density create benefits for particular groups of users.

²⁵ https://blogs.crikey.com.au/theurbanist/2016/02/22/is-value-capture-the-silver-bullet-for-funding-infrastructure/

Sale or rent of a public asset, such as government-owned land or development rights. This form of value sharing occurs when government owns or acquires land in the vicinity of the infrastructure project, and after construction, the land, air rights, or rights to develop the land are sold or leased.

The key principle of value sharing is that increased land values arising from government decisions should be shared with the government and public in order to defray the resulting infrastructure costs.

Growing acceptance of user charges

As mentioned above, value sharing is one example of a user charge. There has been growing acceptance of the "user pays" principle as technological advancements have reduced the transaction costs of exclusion and charging for use.²⁶ Funding sources, therefore, should reflect benefits to users, with public funding (taxes) making up the shortfall between user charges and the overall costs of the infrastructure (construction and operation).

Value Sharing in Australia

Value sharing as a method of funding infrastructure has been around for well over 100 years in Australia – notably to fund the Sydney Harbour Bridge²⁷. In NSW, value sharing as a method of funding infrastructure projects is still not widely used, though there are examples as discussed in this paper.

Recent examples of value sharing across Australia include:

- The upgrades to Chatswood station in Sydney and Melbourne Central were a joint development form of value sharing between the developer and the local Council. The developer provided the infrastructure, in return for air rights.
- In Queensland, the Gold Coast City Council established a land value sharing scheme, which was based on a charge applied to certain areas that benefited from the Gold Coast Rapid Transit Stage 1 project, and helped fund the project.
- In Western Australia, a Metropolitan Regional Improvement Tax has been in place since 1959 to help fund the cost of land for roads, public spaces and other public facilities in greater Perth. This levy is an additional 0.14% charge on the aggregate taxable value of all metropolitan properties in excess of \$300,000. The revenue

from the levy is hypothecated (legally directed) to fund the acquisition of land by government for parks and transport corridors.

- The Macquarie Park Corridor Access and Open Space infrastructure scheme allows sites to be developed with increased floor space and heights provided that there will be adequate provision for recreation areas, and an access network. Similar to the Green Square Scheme discussed later in this paper, this mechanism operates in addition to the Section 94 contributions plan and the provisions are formally satisfied under a Voluntary Planning Agreement (VPA).
- The Waverley Variation Floor Space Infrastructure Scheme allows developments in certain areas to achieve up to 15% additional floor space above that permitted under existing planning controls, provided the developer enters a VPA with Council to provide affordable housing units.

5.3 Strategic endorsement of value sharing

The Australian Government's discussion paper²⁵ sets out the strategic justification for value capture (or value sharing) – at least from the point of view of value uplift arising from land transport infrastructure investment. Similar arguments apply to other infrastructure, to the extent that there are beneficiaries of local infrastructure.

Infrastructure Australia's paper *Capturing Value* – *Advice on making value capture work in Australia*²⁵ discussed the role value sharing can play in infrastructure funding. Value sharing can be used "as part of both a project funding mix and a broader policy agenda". It allows for more equitable and sustainable funding while creating an incentive structure to engage the local community throughout the planning, project development and delivery process.

Value sharing mechanisms have recently received a great deal of attention with the following statements from significant public sector bodies. It is noted that these statements more often relate to value sharing associated with transport investments, rather than planning uplifts.

Department of Prime Minister and Cabinet: Supports the concept and notes that all levels of

²⁶ http://www.pc.gov.au/research/supporting/publicinfrastructure-financing/public-infrastructure-financing.pdf
²⁷ https://bitre.gov.au/publications/2015/files/is_069.pdf

²⁵ http://infrastructureaustralia.gov.au/policypublications/publications/files/Capturing_Value-Advice_on_making_value_capture_work_in_Australiaacc.pdf

government can do more to realise the potential benefits of value capture.³⁰

Commonwealth Department of Infrastructure and Regional Development:

Assessment of proposals for public funding of transport projects should include consideration of what proportion of the project can be funded by the beneficiaries of the infrastructure through targeted contributions and what proportion of the project should be funded by the broader community.³¹

Infrastructure Australia:

Infrastructure Australia in 2016 said that:

"... value capture can work in Australia and should be regularly considered for all public infrastructure projects, but with realistic expectations about the role it can play in funding the infrastructure we need." 32

"All governments should routinely consider land value capture in public infrastructure investments".33

Federal Productivity Commission:

The Australian Government should encourage direct user charging and value capture measures (such as betterment levies and property development charges) where justified. When the benefits from infrastructure accrue to more than users, governments should also consider value capture initiatives — such as betterment levies and property development - so that wider beneficiaries contribute to funding.34

The then-NSW Minister for Planning:

"Councils should be able to capture a reasonable share of the uplift in value from a rezoning, to help pay for community facilities and amenities."35

Infrastructure New South Wales:

Infrastructure NSW supports the use of targeted value capture mechanisms, including special purpose property levies, in situations where there is a clear link to new infrastructure.³⁶

Greater Sydney Commission – Western Central draft District Plan Nov 2016:

"We will continue to work across government on the amount, mechanisms and purpose of value capture to create a more consistent approach to capturing value for public benefit, complementary with other existing mechanisms".37

IPART:

IPART recommends that councils capture 50% of the uplift in land value from a rezoning decision through negotiations with the developers. These funds can be used to fund community benefits in the local government area.38

Infrastructure Victoria:

"...individuals and businesses who receive significant financial benefits from planning decisions made by government should also contribute to providing infrastructure the community needs". 39

³⁰ Better Cities, Department of Prime Minister and Cabinet, May 2016.

Department of Infrastructure and Regional Development, Principles for Innovative Financing, March 2016. http://investment.infrastructure.gov.au/whatis/ Principles_for_Innovative_Financing_Mar2016.pdf

³² Infrastructure Australia 2016, Capturing Value - Advice on making Value Capture Work in Australia, from http://infrastructureaustralia.gov.au/policy-

publications/publications/Capturing-Value.aspx ³³ Infrastructure Australia: Australian infrastructure Plan February 2016.

⁽Chapter 4 - Funding Mechanisms Infrastructure report 2014)

⁽Media Release 4 November 2016)

³⁶ (State Infrastructure Strategy 2014)

³⁷ (Section 1.2.3 - Infrastructure funding and delivery) 38 https://www.ipart.nsw.gov.au/files/sharedassets/

website/shared-files/local-government-tribunal-briefingsfull-tribunal-2016/ipart-submission-to-the-draft-voluntaryplanning-agreement-policy-22-december-2016.pdf

http://www.infrastructurevictoria.com.au/node/84

Box 4: Case study in value sharing - Infrastructure Funding at Green Square

The Green Square Urban Renewal Area is subject to significant transformation from a heavy industrial area to a high density residential area surrounding the Green Square Town Centre. This renewal has so-far progressed over 20 years, originally starting in 1996.

As a consequence of the transformation, an array of new public infrastructure is being provided or funded by the new development, including new roads and traffic improvements, recreation and community facilities, drainage and flood mitigation works. Some of these are being funded Section 94 developer contributions under the *City of Sydney Development Contributions Plan 2015* and its predecessor plans. Other works may involve land dedication or works provided in-kind, formalised by Voluntary Planning Agreements.

Sydney Local Environmental Plan 2012 enables additional floor space at Green Square to be sought above the amounts set out on the Floor Space Ratio Map if the development includes Green Square community infrastructure. "Green Square community infrastructure" is specifically defined within the LEP and includes recreation areas, indoor and outdoor recreation facilities, public roads, drainage or flood mitigation works. The provisions of the LEP are supplemented by Sydney Development Control Plan 2012 and, more specifically, a Development Guideline "Providing Community Infrastructure in Green Square".

This Development Guideline sets out the processes involved if a proponent seeks to access the additional floor space within Green Square. The development guideline also sets out a dollar value per square metre of additional floor space which is used to calculate the value of any additional floor space depending on the type of development proposed. At the time of publication of this Discussion paper, the rates were \$475/sqm (residential floorspace), \$275/sqm (retail floorspace) and \$200 (other non-residential floorspace).

These values are applied to the additional floorspace sought by a developer to formulate the total value of the contribution. This value is then used to identify particular works that could be delivered by the developer either physically on-site in accordance with the Development Control Plan, or off-site by way of a monetary contribution. A proportion of the contribution always forms a monetary payment towards infrastructure within the Green Square Town Centre. The outcome of this process is formalised by a Voluntary Planning Agreement with an offer from the developer to provide certain Green Square community infrastructure and/or a monetary payment towards the infrastructure.

Box 5: Case study in value sharing - Infrastructure Funding at Macquarie Park

Macquarie Park is undergoing transformation from a traditional large campus-style business park to a higher density commercial and mixed use area. The traditional large lots in private ownership historically limit the permeability of the area and, consequently, rely mainly on private car as the dominant form of transport.

As a consequence of the transformation, new infrastructure – primarily new roads and open space areas – will be provided. Some of these are being funded Section 94 developer contributions under the Ryde Section 94 Development Contributions Plan 2007 (as amended). Other works may involve land dedication or works provided inkind, formalised by Voluntary Planning Agreements.

Ryde Local Environmental Plan 2014 enables additional building height and floor space at Macquarie Park to be sought above the amounts set out on the Floor Space Ratio Map if the development includes adequate provision for recreation areas and an access network. The amount of additional building height and floor space available throughout the corridor is prescribed on the "Macquarie Park Corridor Precinct Incentive Height of Buildings Map" and "Macquarie Park Corridor Precinct Incentive Floor Space Ratio Map"

The provisions of the LEP are supplemented by Ryde Development Control Plan 2014 and, more specifically, provisions in Part 4.5 of the DCP. The DCP sets out the type and preferred location of the access network and recreation areas throughout the Macquarie Park corridor which would be subject to dedication or provision by the developer in accordance with the DCP. The DCP also sets out the procedure to implement the planning incentives mechanism.

A dollar value per square metre of additional floor space is used to calculate the value of any additional floor space depending on the type of development proposed, and this amount is set out in the annual Fees and Charges document published by the council. This value is then used to identify particular works that could be delivered by the developer either physically on-site in accordance with the Development Control Plan, or off-site by way of a monetary contribution.

The outcome of this process will be formalised by a Voluntary Planning Agreement with an offer from the developer to provide certain infrastructure on-site where relevant and in accordance with the DCP and/or a monetary payment towards the infrastructure where the infrastructure is located elsewhere in Macquarie Park.

5.4 Why value sharing is not a tax

Value sharing mechanisms only apply to parties who benefit (albeit indirectly) from proposed infrastructure, so should not be regarded as a tax. If a scheme has a threshold before which value sharing is not applicable (i.e. in a density bonus scheme) developers also would have the option of not exceeding this threshold. This also supports the proposition that such a scheme is not a tax.

Economic theory states that the primary purpose of a user charge is to recoup the costs of a good or a service. This is in contrast to taxes which are used to raise revenue irrespective of benefits. Charges also have a clear and direct nexus to benefits. Taxes do not. When thinking about value sharing as a charge, therefore, it is necessary to identify the service being provided and the benefits that accrue; this highlights the importance of clearly linking value sharing mechanisms with infrastructure planning.

The benefits of density 5.5

A key group of benefits accruing from the planning context in the Parramatta CBD are the benefits of densitv.

When governments undertake planning changes that create density (i.e. a "service"), a charge can be used to recoup the costs of providing this service, such as the costs of stronger local infrastructure. This recognises the economic opportunity of providing this service and allows property owners to contribute to the locale. Density allows for economies of agglomeration or the benefits that arise when households and firms locate near each other.40 These benefits include the following:

Amenity

High density areas benefit from increased amenity as providers of goods and services can enjoy economies of scale. Social infrastructure and public spaces will attract more funding if it is likely to impact a greater number of people. Public services also tend to be better in more populated areas. Moreover, density increases demand for products and services creating a viable environment for shops, restaurants, bars and cafes. Empirically, higher-amenity cities are also higher-growth cities.41

Convenience

Density creates convenience. When personal services are within a short distance, residents and workers enjoy a higher quality of life. Increased density reduces the need for driving, which reduces petrol spend, eases traffic congestions and cuts air pollution.

Speed

Speed is a function of location. The accessibility of a location determines how mobile residents and workers can be and how quickly goods and services can reach that area. Higher-density areas are more fast-paced than lower-density areas, whether in terms of information sharing or travel.

5.6 Conclusion

Local infrastructure is required to support a certain level of density, such as that being established in the Parramatta CBD through the Parramatta CBD Planning Proposal. As the population grows, it will trigger the need for additional infrastructure, which can be partly funded by a value sharing mechanism (even though the resulting infrastructure will be used by all).

The next chapter will examine in more specific detail how Council has sought to apply value sharing under the Parramatta CBD Planning Proposal, in an effort to partly fund the infrastructure requirements outlined in this paper.

⁴⁰ National Bureau of Economic Research 2007 ⁴¹ http://scholar.harvard.edu/files/glaeser/files/ consumer_city.pdf

6 Council's Proposed Value Sharing Mechanism

6.1 Introduction

As outlined in Chapter 2, the former Parramatta City Council endorsed the draft Parramatta CBD Planning Proposal in April 2016 to be submitted for "Gateway Determination" from the NSW Department of Planning and Environment, so as to enable public exhibition.

An important part of the CBD Planning Proposal is a proposed Planning Uplift Value Share (PUVS) mechanism to suit the infrastructure needs of the CBD. This PUVS mechanism is essentially a density bonus scheme, in which incentive density (FSR) controls are achievable, provided that development makes an appropriate contribution to Community Infrastructure (i.e. local infrastructure projects such as those listed in Appendix B).

Without providing appropriate infrastructure to meet the needs of the growing CBD, the proposed densities will not be able to be supported. Furthermore, developers stand to gain much from the provision of this infrastructure, as it will make the CBD a functional and attractive place to live, work, do business and play; this will only serve to further increase the competitiveness and land values of the Parramatta CBD.

In summary, and as outlined in the previous chapter, the proposed densities and accompanying infrastructure directly and indirectly benefits developers.

Before proceeding any further with the PUVS mechanism in the Parramatta CBD Planning Proposal, Council has resolved to undertake further work on its proposed approach to value sharing, including an independent peer review of Council's work on value sharing so far, as well as community consultation on the matter, which is the key purpose of this discussion paper. This chapter will introduce and explain Council's past work on value sharing (Sections 6.2 - 6.4), and share the results of the

independent peer review undertaken by Aurecon and Land Econ Group (Section 6.5); please also refer to the peer review report at **Appendix A**.

6.2 Introducing Council's proposed mechanism

As introduced in the case studies in the previous chapter, the principle of value sharing by way of a density bonus scheme is in place in several council areas across Greater Sydney. Locations where value sharing is currently used include the City of Sydney at Green Square; City of Ryde at Macquarie Park; Burwood Council in Burwood Town Centre; Inner West (former Leichhardt) Council and Waverley Council. Some of these programs have been established for over 10 years (e.g. Green Square) while others have recently commenced within the last two years (e.g. Macquarie Park).

In these cases, the application of value sharing has been localised to a limited area and not applied at the scale of a full CBD environment. The Parramatta CBD Planning Proposal contains provisions seeking to apply a value sharing mechanism based on incremental density across the entire Parramatta CBD area.

The proposed value sharing mechanism is intended to apply to new residential development within the Parramatta CBD seeking to develop beyond the current density controls. Non-residential (i.e. commercial) development is excluded from the value sharing mechanism, in order to promote commercial uses in the CBD.

The proposed value sharing mechanism is based on sharing a portion of the uplift in density controls (and therefore land value) proposed under the Parramatta CBD Planning Proposal. Such a mechanism can be referred to as a "Planning Uplift Value Share" (PUVS) mechanism (as opposed to value uplift related to transport or other major infrastructure provision). The purpose of the proposed PUVS mechanism is to share part of the economic value gained from the increase in development rights with the community. The monetary contributions generated through the PUVS mechanism would then assist Council in providing necessary infrastructure in the Parramatta CBD over the next 40 years (refer to the *Draft Parramatta CBD Infrastructure Needs Analysis* at **Appendix B**).

The following subsections explain the general approach of the proposed PUVS mechanism in more detail.

Phase 1 Value Sharing

The Parramatta CBD Planning Proposal seeks to introduce "Base" and "Incentive" FSR controls for sites within the CBD. Base controls are generally the current Floor Space Ratio (FSR) controls under Parramatta Local Environmental Plan (LEP) 2011. Incentive controls are generally (though not always) increased FSR controls, and are achievable provided that a contribution to Community Infrastructure is made. An example of the Base and Incentive FSR maps from the draft LEP maps are shown in Figure 8 below.

Figure 8: Example of Base and Incentive FSR maps⁴²



Under the proposed PUVS mechanism, a value sharing contribution is made based on the uplift between the Base and Incentive FSR controls. This contribution is referred to as "Phase 1 value sharing".

Phase 2 Value Sharing

The Parramatta CBD Planning Proposal also proposes amendments to the existing planning

controls through the identification of "Opportunity Sites". Opportunity Site FSR controls are applied to a significant portion of land zoned B4 (Mixed Use) within the City Core area, with the intent to allow additional residential development within this zone. An example of the Opportunity Site FSR controls is shown in Figure 9 below.

Identification as an Opportunity Site means that a site is eligible for an additional 3:1 FSR on top of the Incentive FSR, provided that an additional contribution to Community Infrastructure is made. This contribution is made based on the uplift between the Incentive and Opportunity Site FSR controls, and is referred to as "Phase 2 value sharing".

Figure 9: Example of Opportunity Site Map⁴³



Summary

All development – including residential development – can achieve the Base FSR without being subject to the PUVS mechanism. The PUVS mechanism would only apply when a residential development seeks to develop beyond the Base FSR controls, with value sharing contributions made based on the difference between the Base and Incentive FSRs (otherwise known as Phase 1 value sharing), and on the difference between the Incentive and Opportunity Site FSR controls (otherwise known as

⁴² Note: numbers on this example image indicate FSRs (e.g. 6.0 signifies 6:1 FSR).

⁴³ Note: "OS" on this example map indicates an Opportunity Site area.

Phase 2 value sharing, and only if the site is identified as an Opportunity Site).

Figure 10, below, summarises the proposed PUVS mechanism in terms of floor space. Other FSR schemes in the Planning Proposal include a Design Excellence bonus scheme (up to 15% bonus FSR on Incentive FSR), and High Performing Buildings bonus scheme (additional 0.5:1 FSR for meeting

certainly environmental sustainability requirements). Both of these operate separately to the proposed PUVS scheme, so are not discussed in further detail here. However, they are shown in Figure 10 to demonstrate how all of the FSR schemes are proposed to work together in the CBD.

Figure 10: Summary of Proposed FSR controls and PUVS mechanism



6.3 Modelling Council's proposed approach

During 2015-2016, Council and consultants GLN Planning undertook development feasibility testing of a number of hypothetical development scenarios to determine the effect of introducing a Phase 1 and Phase 2 PUVS mechanism as outlined in the previous section.

Given that sales data over the preceding two years indicated that property transactions generally occurred on the basis of the potential controls outlined in the Parramatta CBD Planning Strategy (rather than existing statutory controls under the Parramatta LEP 2011) the base case used in the testing was modelled to reflect this.

The results of the development feasibility modelling undertaken by GLN and Council were based on a conservative land value uplift rate of \$750/sqm.⁴⁴

Table 3, below, shows the funding potential under the proposed PUVS, using multiple rate scenarios for Phase 1 and Phase 2 (with rates being a set percentage of the \$750/sqm value rate). As previously mentioned in Section 4.4, this low and high range reflects two site consolidation scenarios – the low range being minimal site consolidation and the high range being a greater amount of site consolidation.

Table 3: Revenue potential under a Phase 1 and Phase 2 value sharing mechanism

Phase Scenario	Rate (% of \$750/sqm)	Potential revenue (low range)	Potential revenue (high range)	
Phase 1				
1	50%	\$483 million	\$589 million	
2	40%	\$387 million	\$471 million	
3	30%	\$290 million	\$353 million	
4	20%	\$193 million	\$235 million	
5	10%	\$97 million	\$118 million	
Phase 2				
1	50%	\$44 million	\$133 million	

Further Options Analysis

In early 2016, Council formed an Infrastructure Funding Review Committee to further review its work on the proposed PUVS mechanism. The Committee recommended that Council concurrently analyse two options for infrastructure funding as part of its review of the Parramatta CBD Planning Framework:

- 4.5% section 94A levy to apply to the whole development (subject to Ministerial approval), plus 50% value capture for Phase 2 (being \$375/m²); and
- ii. 3% section 94A levy to apply to the whole development, plus 20% value capture for Phase 1 uplift (being \$150/m²) and 50% value capture for Phase 2 uplift (being \$375/m²); and

The financial implications for Council of each of these two Options are shown in Table 4 and 5 below.

Table 4: Revenue potential under funding option (i)Option (i) No Phase 1, Phase 2 @50%, S94A @4.5%

Source	Rate	Low Range	High Range
S94A	4.5%	\$310.5 million	\$484.5 million
Phase 2 50%		\$44 million	\$133 million
Revenue p	otential	\$354.5 million	\$617.5 million

Table 5: Revenue potential under funding option (ii) Option (ii) Phase 1 @20%, Phase 2 @50%, S94A @ 3%

Option (II)	Phase 1	@20%,	Phase 2	2@50%,	S94A @ 3%

Source Rate		Low Range	High Range
S94A 3%		\$207 million	\$323 million
Phase 1 20%		\$193 million	\$235 million
Phase 2 50%		\$44 million	\$133 million
Revenue p	otential	\$444 million	\$691 million

The above tables demonstrate that Phase 1 value sharing generates significantly more revenue than a 1.5% increase in the section 94A levy. Therefore, Option (ii) - which uses Phase 1 value sharing - would generate significantly more income than Option (i) - which uses the 4.5% section 94A levy.

Development Feasibility Testing

GLN Planning and Council also undertook development feasibility testing of various scenarios of value sharing and section 94A charges.⁴⁵ The key outcomes of this testing are summarised as follows:

 10-20% Phase 1 value sharing could likely be tolerated in the current market by those who have purchased land at above average rates, while a higher Phase 1 value sharing rate could likely be tolerated by those who have purchased land at below average rates.

⁴⁴ See Council Business Papers 27 June 2016, Item 7.4 for further discussion of this rate.

⁴⁵ See Council Business Paper 27 June 2016, Item 7.4 for further discussion of this feasibility testing.

- Sites which were acquired and held for a lengthier time would be more likely to be able to absorb a higher Phase 1 value sharing cost.
- Phase 2 value sharing at 50% will still enable higher (i.e. more risky) developments to meet lending authority benchmarks.

In addition, since late 2016, eight recent Voluntary Planning Agreements in the CBD have been negotiated on the basis of 20% Phase 1 value sharing. This further underscores the feasibility of Council's proposed approach, and how it is being applied in practice.

6.4 Council's proposed implementation of the PUVS

As part of the Parramatta CBD Planning Proposal, Council has proposed three key implementation mechanisms which would work together to formalise the proposed PUVS mechanism. These three mechanisms are provisions in Council's Local Environmental Plan (LEP), a Development Guideline containing the value sharing rates and Voluntary Planning Agreements.

Local Environmental Plan (LEP) provisions

As outlined previously in this chapter, the proposed LEP provisions to enact the PUVS mechanism would include Base, Incentive and Opportunity Site FSR controls. These controls would be contained in statutory maps as part of the LEP. The maps would be accompanied by an LEP clause which outlines that the Incentive and Opportunity Site FSR controls are only achievable if Community Infrastructure is provided. The conditions by which Community Infrastructure can be provided will be set out in more specific detail in a Development Guideline (see next section). This is similar to the approach used by the City of Sydney for Green Square.

Development Guideline

Alongside the LEP provisions, a separate Development Guideline would lay out in clear detail the value sharing rates on a "per square metre" basis for both Phase 1 and Phase 2 value sharing. The Development Guideline would highlight how community infrastructure is to be provided, which could be through dedication of land, monetary contributions, construction of infrastructure, provision of materials for public benefit and/or use, or a combination of these.²⁷ The main purpose of this Development Guideline would be to ensure certainty, transparency and fairness for Council and developers.

Voluntary Planning Agreements

The third key implementation component for the proposed PUVS mechanism is an update to Council's Voluntary Planning Agreement (VPA) policy. This update would articulate that VPAs are the intended way of formalising the PUVS mechanism for each individual development. Each individual development's contribution under the PUVS mechanism would be formally described and agreed to in a VPA for that site. All contributions collected under Voluntary Planning Agreements must be spent or utilised for the specific purpose they were levied and any interest applicable to unspent funds must be attributed to remaining funds.

6.5 Peer Review of Council's work on value sharing

As part of this project, Aurecon working with Land Econ Group undertook an independent Peer Review of Council's work on value sharing to date (as described in Sections 6.2 - 6.4). The peer review report is included at **Appendix A** of this Discussion Paper.

The peer review covered interviews with selected real estate agents and developers that are active in the Parramatta and greater Sydney market, and critically examined core documents:

- Parramatta CBD Planning Framework: Economic Analysis (2014) by SGS Economics and Planning
- Parramatta CBD Infrastructure Funding Models Study (2016) by GLN Planning
- Council staff reports and other relevant materials

The review highlighted that real estate development is a highly cyclical business, where developers are eager to apply for additional floor space during strong market conditions, but tend to hold back when the market conditions are weak as building higher may not necessarily translate to more profits. Because of the unpredictability of income flow, the review recommends that Council views the value sharing mechanism source as an important supplemental rather than primary source of funding for the construction of local infrastructure and

²⁷ https://www.planningportal.nsw.gov.au/understandingplanning/voluntary-planning-agreements

amenities in the Parramatta CBD. It also recommends a method for reviewing the rate, based on a selected residential market index (see more detail in recommendations below).

Market Feasibility

As noted in the peer review, several interviews were conducted with selected real estate agents and developers active in the Parramatta and greater Sydney market. The responses indicate some slowing of the Western Sydney apartment market but with no expectation of serious oversupply and resulting dramatic downturn. The reasons cited include:

- Population growth pushing out from the more developed portions of the Sydney region.
- Government policy support and planned infrastructure investment in and around Parramatta.
- Central location of the Parramatta CBD.
- High housing cost in the Sydney region.
- Major development projects being constructed that will elevate the market perception of the Parramatta CBD when completed.

However, there are early signs of a slowing apartment market due to some foreign governments beginning to slow capital outflow, which has been a factor in the Parramatta apartment market, making local banks more restrictive on financing for apartment investment, combined with expectations of higher global interest rates due to USA election results.

In addition, developers expressed their concerns on a value sharing mechanism needing to be easily predictable, flexible to market conditions, and that Council implement speedy development approval processes (which in some cases can take up to 18 months).

Review of Developer Responses

The interviews undertaken with three real estate agents and two developers active in the Parramatta and Greater Sydney market indicated some slowing of the Western Sydney apartment market, but with no expectation of serious oversupply resulting in a dramatic downturn. The peer review also reviewed and responded directly to past comments from the development community regarding Council's proposed value sharing strategy; these comments are addressed in turn in Section 2.4 of the Peer Review (Appendix B).

Recommendations of the Peer Review

Council is generally supportive of the peer review recommendations, and subject to consultation, proposes to adopt them as part of the CBD Planning Proposal. Balancing the continued local market optimism with the need for caution due to macroeconomic considerations, the recommendations based upon the peer review are as follows:

- Implement the PUVS mechanism as promptly as possible to provide Council with an additional source of funding for community infrastructure during this up-market cycle and the development community with cost predictability as the market moves toward less certain times.
- Provide the developer community with cost predictability through smoothing the implementation of PUVS over five years,
 - Set the Phase 1 (Incentive) contribution to a maximum of \$150 per square metre for new residential developments that seek to develop beyond the existing planning controls up to the incentive controls.
 - Set the Phase 2 (Opportunity) contribution to a maximum of \$375 per square metre for new residential developments that seek to develop beyond the incentive controls up to the opportunity site controls, applicable to certain areas in Parramatta CBD.

This will allow developers to internalise this contribution into their pro forma calculations, and the impact will be on the amount they are willing to pay for land going forward. For future major projects, removing the uncertainty of the amount of PUVS contribution and the time required for negotiations will allow Council to continue to communicate its "open for business" attitude essential for the continued rapid transition into a world class city.

- Re-evaluate the PUVS process after five years of implementation to make sure the contributions reflect market conditions, and are on track to assist in meeting funding requirements for community infrastructure.
- Build flexibility into the PUVS mechanism that provides Council with the option to either suspend or reduce the dollar per square metre contributions for a finite one to two year period should a selected residential market index decline in five of six successive quarters. In the event of a severe real estate recession, this provides Council with an efficient tool to temporarily lower development cost and therefore reduce the mechanism's impact on

residential development and construction industries operating in Parramatta CBD. In effect, this operates as a concession to developers to account for the volatility inherent in the property market.

Apply the PUVS mechanism on residential development in excess of existing planning controls only, as per Council's intent of creating a commercial core through improving the appeal of commercial developments.

This set of recommendations balances two objectives. First, it provides a revenue stream for the construction of CBD infrastructure essential to elevate Parramatta's position at the heart of Greater Sydney's Central City. Second, it provides the development community with cost predictability for five years. The developers will quickly internalise this added costs into their pro forma calculations. and the impact will be on the amount they are willing to pay for land going forward. For future major project removing the uncertainty of the amount of value sharing cost burden and the time required for negotiations will allow Council to continue to communicate its "open for business" attitude essential for the rapid creation of Greater Sydney's Central City.

Responding to the Peer Review recommendations

Together with the feedback received during this consultation period, Council will consider the recommendations from the independent Peer

Review in making decisions around infrastructure planning and funding in the CBD. Please refer to section 7.2 for a more detailed discussion of next steps.

6.6 Value sharing's contribution to closing the funding gap

The Peer Review undertaken by Aurecon and Land Econ Group has recommended value sharing rates of \$150 for Phase 1 and \$375 for Phase 2 (i.e. 20% and 50% of the benchmark \$750/sqm). If these rates were to be applied to development in the CBD, the funding gap established in Chapter 4 would be reduced (though not entirely resolved).

Based on two-thirds build-out of the Planning Proposal and the site consolidation scenarios previously discussed in this report, setting a Phase 1 Value Sharing rate at \$150/sqm (20%) would yield an estimated **\$193 - \$235 million**.

Based on two-thirds build-out of the Planning Proposal and the site consolidation scenarios previously discussed in this report, setting a Phase 1 Value Sharing rate at \$375/sqm (50%) would yield an estimated **\$44 - \$133 million**.

Figures 11 and 12 demonstrate the impact of this recommendation on the funding gap established in Chapter 4. The funding gap is reduced from an estimated **\$394 - \$549 million** to **\$26-\$312 million**.


Figure 11: Value sharing's potential impact on the funding gap (low estimated income

Figure 12: Value sharing's potential impact on the funding gap (high estimated income)



6.7 Chapter 6 discussion questions

- 4. What are your views on using value sharing to fund infrastructure?
- 5. Do the proposed value sharing charges strike an appropriate balance between public and private interests?
- Are there other infrastructure funding mechanisms that should be considered by Council instead of (or in addition to) the PUVS?*

* Remember that Council has limited powers to raise revenue. Stamp duty, land taxes and special area levies are the responsibility of the State Government. Refer to Chapter 4 for further discussion.

7 Discussion Questions and Next Steps

7.1 Discussion questions

This discussion paper has posed the following questions:

- Has Council considered the right types of local* infrastructure projects in its *Draft Parramatta CBD Infrastructure Needs Analysis* at Appendix B?
- 2. In your mind, what are the most important local* infrastructure priorities for Parramatta CBD?
- 3. Is there anything you feel is missing from the draft infrastructure needs analysis?
- 4. What are your views on using value sharing to fund infrastructure?
- 5. Do the proposed value sharing charges strike an appropriate balance between public and private interests?
- Are there other infrastructure funding mechanisms that should be considered by Council instead of (or in addition to) the PUVS?**

* Remember that Council is not directly responsible for infrastructure like schools, hospitals and public transport. While Council advocates for the community and partners on projects where appropriate, provision of these types of infrastructure are generally the responsibility of the State Government.

** Remember that Council has limited powers to raise revenue. Stamp duty, land taxes and special area levies are the responsibility of the State Government.

7.2 Next steps

This Discussion Paper (including the Peer Review and Draft Parramatta CBD Infrastructure Needs Analysis included in the appendices) are being exhibited during March 2017.

Council welcomes your feedback on the matters outlined in this Discussion Paper. Please visit Council's 'On Exhibition' webpage for more information about making a submission:

https://www.cityofparramatta.nsw.gov.au/aboutparramatta/news/on-exhibition

In moving forward, Council will consider all submissions and input received during this consultation period, as well as the peer review recommendations.

The immediate next step for this project will be reporting to Council on the outcomes of the consultation period.

Pending the outcomes of Council's decision at that meeting, the intended next step is for preparation of a detailed draft CBD Infrastructure Strategy containing an updated CBD Works Schedule, as well as the appropriate funding implementation mechanisms.

It is intended that the draft CBD Infrastructure Strategy would then be exhibited alongside the CBD Planning Proposal (pending receipt of a Gateway Determination from the Department of Planning and Environment), at which point the community will have further opportunity to comment on issues on planning and funding infrastructure in the Parramatta CBD.

aurecon





DRAFT PARRAMATTA CBD INFRASTRUCTURE NEEDS ANALYSIS

Summary Table

Priority	Cost Estimate
Growth and Transport	\$139,200,000
Green Spaces and Environment	\$211,030,000
Recreation and Leisure	\$101,100,000
Strong economy and City Centre	\$369,414,000
Community Focus	\$79,749,500
Arts and Culture	\$98,200,000
TOTAL ESTIMATE	\$998,693,500

Notes on exhibited information

1. Infrastructure Planning is an iterative process. The projects identified in this needs analysis reflect Council's understanding of CBD Infrastructure Needs at the time of publication, and are subject to change as work progresses on infrastructure planning in the CBD.

2. This is a Needs Analysis which seeks to identify local infrastructure projects for which Council is likely to be responsible for funding all or part, as well as some regional projects for which there may also be Council funding implications; the projects listed reflect Council's understanding of CBD infrastructure needs.

3. Project costs are best estimates at time of publication, and are subject to change as more detailed planning progresses. Costs have been estimated in 2016/17 dollars. More information including CPI adjustment, operational/maintenance costs, and other modelling will be included in the forthcoming draft Parramatta CBD Infrastructure Strategy.

4. The "Council Funding" column indicates whether it is expected that Council will pay for "All" or "Part" of the project (with Council funds coming from various sources including rates, developer contributions, VPAs, etc). Projects marked as "Part" funding indicate potential funding by another level of Government, and are subject to agreement from the relevant agencies.

5. Some projects listed here are already underway. However, these projects are appropriate for inclusion in the needs analysis, as they are associated with growth consistent with the aims of the CBD Planning Framework.

Disclaimer: The costs provided in this needs analysis are high level estimates only. Costs are generally based on experience with similar projects, benchmarking or analysis undertaken in other strategic work. Significant further cost planning will be required on a project by project basis.

Number	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related Project, Policy or Strategy
1		Pedestrian and Street Lighting under Parkes Street rail overbridge	Parkes Street at Railway Bridge	\$100,000	ALL	Short	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
2	Pedestrian Improvement	Closure of Cowper Street at Parkes Street and a pedestrian fence along the south side of Parkes Street between the rail bridge and Wentworth Street		\$1,500,000	PART	Short	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
3		Widen footpath on north side of Hassall St to accommodate pedestrian volume. Hassall St one-way westbound from between the driveway for No.2 and Station Street. Traffic Signal changes.	Hassall Street, immediately east of Station Street	\$400,000	ALL	Short	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	Council-approved Master Plan (14/12/2015)
4	Median Islands	Median Island in Marsden St between Macquarie St and George St to ensure driveways are left in/left out; includes kerb realignment and new tree planting on the west side of the street.	Marsden Street between Macquarie Street and George Street	\$1,000,000	ALL	Long	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	Identified as part of ongoing CBD traffic management
5	Wedian Islands	Median Island in Parkes Street between Wigram Street and Harris Street to ensure driveways are all left in/left out	Parkes Street between Wigram Street and Harris Street	\$200,000	ALL	Long	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
6		Intersection upgrades	10 locations throughout CBD	\$7,000,000	PART	Short	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	Integrated Transport Plan
7	Intersection Upgrades	Intersection upgrade and road widening at Harris and Parkes St.	Intersection of Harris Street and Parkes Street	\$5,000,000	ALL	Medium	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
8		5 5 , 5	Intersection of Parkes Street and Wigram Street	\$3,000,000	ALL	Medium	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
9	CBD Road Widenings	Road widening in Church Street to provide additional northbound right turn lane into Parkes Street. Minor widening on the north east corner is also required.	Intersection of Church St, Great Western Highway and Parkes St	\$6,000,000	PART	Medium	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	City Ring Road
10	CBD Road Widenings		Throughout North Parramatta and CBD	\$50,000,000	ALL	Medium-Long	C1.2 - Implementing programs focused on safer local roads, managing traffic congestion and the regular turnover of traffic in commercial centres	CBD Planning Proposal project
11		Duplicate Gasworks (Macarthur St) Bridge including a pedestrian and bike path on the east side; to be progressed as Light Rail plans are finalised	Gasworks Bridge	\$25,000,000	PART	Medium	C1.5 - Managing and maintaining civil infrastructure to ensure it is safe, effective and accessible, including roads, footpaths, drainage systems, bridges and street furniture	City Ring Road
12	Bridge Improvement	Morton Street Pedestrian Bridge	Morton Street over the Parramatta River	\$15,000,000	PART	Short	C1.5 - Managing and maintaining civil infrastructure to ensure it is safe, effective and accessible, including roads, footpaths, drainage systems, bridges and street furniture	Elizabeth Street Precinct Planning
13		Barry Wilde Bridge biking and pedestrian upgrade	Barry Wilde Bridge	\$20,000,000	ALL	Short-Medium	C1.5 - Managing and maintaining civil infrastructure to ensure it	River Strategy
14	Smart Parking	Smart wayfinding, ticketless parking, onstreet smart metering, and parking data capture	Throughout CBD	\$5,000,000	ALL	Medium		Smart Cities Program

Number	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related Project, Policy or Strategy
1	CBD Street Trees	Street Tree Planting Project to continue the City of Trees Project	All Streets within City Centre	\$750,000	ALL	Short	D4.9 - Protecting, enhancing and proactively maintaining trees in the city streetscape.	Design Parramatta Public Domain Framework, City Trees project
2		River Square	South Bank between Lennox Bridge and Barry Wilde Bridge	\$54,000,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
3		Sorrell Street foreshore parcel - public domain upgrade including terraced landscape and access way (parcel opposite River Square)	North Bank between Lennox Bridge and Barry Wilde Bridge	\$14,000,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
4	River Square and surrounds	New footbridge across the Parramatta River	Next to Barry Wilde Bridge	\$10,000,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
5		Relocate sewer pipe across the Parramatta River	Next to Barry Wilde Bridge	\$2,000,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
6	Parramatta Quay upgrades	Charles Street Weir - redesigned weir and river crossing to include improved active transport movement and hydraulic flows	Charles Street Weir	\$10,000,000	ALL	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
7	supporting works	Charles Street Square and ferry terminus surrounds	Surrounding Ferry terminus	\$15,000,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
8		Ferry terminus to Gas Works Bridge South bank parcel upgrades	East of Ferry Terminus to Gas Works Bridge, South side of the River	\$2,000,000	ALL	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
9		Northern Terrace parcel foreshore upgrades	Queens Ave Steps to Elizabeth St	\$10,000,000	ALL	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
10		Playground parcel foreshore upgrade	North bank, east of Barry Wilde Bridge	\$5,000,000	ALL	Medium	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
11	Parramatta River Foreshore Parcel upgrades	Southern foreshore parcel upgrade	Wilde Ave to Charles St Weir	\$15,000,000	ALL	Medium	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
12		Justice Precinct parcel foreshore upgrade	Foreshore between Marsden St and O'Connell St	\$3,000,000	ALL	Long	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
13		Kings School parcel Foreshore upgrade	Foreshore between new school and river	\$4,000,000	ALL	Medium	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
14		Marsden St. Weir Upgrades	Marsden St. Weir	\$5,000,000	ALL	Long	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
15		Harrisford Link	East of Ferry Terminus, South bank (near existing State Heritage Item Harrisford House)	\$1,000,000	ALL	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
16	Links to Parramatta River CBD	Stewart Street Link	North Bank, East of Charles St Weir	\$1,980,000	ALL	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
17	Foreshore	O'Connell Street underpass links	Links between between foreshore and Parramatta Park/Old Kings Oval foreshore	\$10,000,000	PART	Medium	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
18		Escarpment Boardwalk	North Bank, East of Charles St Weir, connected to foreshore by Stewart Street Link	\$11,300,000	PART	Short	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
19	Enhanced swimming and recreational opportunity	Enhanced swimming and recreational opportunity associated with the Parramatta River	Location to be confirmed	\$15,000,000	ALL	Medium	A5.6 - Activating lanes, retail precincts and riverbank	River Strategy Program
20	Prince Alfred Square	Improving facilities for events, protecting heritage assets including significant trees, and improving overall amenity and passive recreational use	Prince Alfred Square	\$4,500,000	ALL	Short	D1.3 - Ensure a range of active recreation, leisure and sporting opportunities is available for all ages, genders, ethnicities, ability levels and socio-economic groups	Prince Alfred Square Masterplan
21	Brickfields Creek Naturalisation	Brickfields Creek naturalisation project	North bank, east of Barry Wilde Bridge	\$1,500,000	ALL	Short		River Strategy Program
22	Parramatta Ways	Parramatta Ways links in the CBD - local delivery of Sydney "Green Grid" project	Throughout CBD	\$15,000,000	ALL	Short	C1.1 Developer and implement River City network of pathways (Parramatta Ways) to improve connectivity	Sydney Green Grid Project
23	Lake Parramatta Upgrade	Lake Parramatta new overflow parking facilities and access paths	Lake Parramatta	\$1,000,000	ALL	Short	D1.3 - Ensure a range of active recreation, leisure and sporting opportunities is available for all ages, genders, ethnicities, ability levels and socio-economic groups	Lake Parramatta Reserve Plan of Management

DDOUDDIG		DECODE ACTION AND	TETATIOE
PROVIDING	OPPORTUNITIES FOR	KECKEATION AND	LEISUKE

lumber	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related Project, Policy or Strategy
		New modern aquatics and leisure centre	Location to be confirmed pending				5 7 1 5	CBD Pool relocation and expansion project
1	Aquatic and Leisure Centre		outcomes of Council's site investigations	\$60,000,000	PART	Medium	opportunities is available for all ages, genders, ethnicities, ability levels and socio-economic groups	
			Location(s) to be confirmed				D1.3 - Ensure a range of active recreation, leisure and sporting	Preliminary findings - Open Space and Recreation aud
2	New CBD Sportsgrounds	are needed; includes construction, civil works, amenities, flood lighting, parking and related infrastructure.		\$10,000,000	ALL	Long	opportunities is available for all ages, genders, ethnicities, ability levels and socio-economic groups	
		Improve capacity of existing sportsground playing surfaces	Throughout CBD					Preliminary findings - Open Space and Recreation aud
3				\$2,000,000	ALL	Short	high quality open space including parks, reserves, playground and sporting grounds.	
		Amenities and facilities associated with sportsground use	Throughout CBD					Preliminary findings - Open Space and Recreation audi
4				\$4,500,000	ALL	Short	high quality open space including parks, reserves, playground	
	Existing CBD Sportsgrounds			, ,			and sporting grounds.	
	Upgrades	Sportsground floodlighting	Throughout CBD				D1.2 - Providing strategic planning and asset management for	Preliminary findings - Open Space and Recreation audi
5				\$1,000,000	ALL	Short	high quality open space including parks, reserves, playground	
							and sporting grounds.	
		All weather recreation synthetic playing surface at existing sites	Throughout CBD				D1.3 - Ensure a range of active recreation, leisure and sporting	Preliminary findings - Open Space and Recreation audi
6				\$10,000,000	ALL	Medium	opportunities is available for all ages, genders, ethnicities, ability	
							levels and socio-economic groups	
		Upgrade the quality of parks within the city to cater for increased demand	Throughout CBD				D1.2 - Providing strategic planning and asset management for	Preliminary findings - Open Space and Recreation audi
7		and bring up to a world-class city standard		\$6,000,000	ALL	Medium	high quality open space including parks, reserves, playground	
							and sporting grounds.	
		Multi-purpose outdoor spaces for active recreation (e.g. half-court	Throughout CBD					Preliminary findings - Open Space and Recreation audi
8	CBD Park Improvements	basketball sites)		\$600,000	ALL	Medium	high quality open space including parks, reserves, playground	
							and sporting grounds.	
		Pavilion and Open Space improvements at Robin Thomas Reserve	Robin Thomas Reserve					Robin Thomas Masterplan
9				\$3,500,000	ALL	Short	opportunities is available for all ages, genders, ethnicities, ability	
							levels and socio-economic groups	
		New playground at Belmore Park	Belmore Park	6700.000		ci i	D1.3 - Ensure a range of active recreation, leisure and sporting	Preliminary findings - Open Space and Recreation audit
10				\$700,000	ALL	Short	opportunities is available for all ages, genders, ethnicities, ability	
		Other new CBD Play Spaces	3x locations to be confirmed				levels and socio-economic groups D1.3 - Ensure a range of active recreation, leisure and sporting	Preliminary findings - Open Space and Recreation audit
11	CBD Playgrounds	Other new CBD Play spaces	sx locations to be committed	\$2,100,000	ALL	Medium-Long	opportunities is available for all ages, genders, ethnicities, ability	Preniminary minuings - Open space and Recreation audi
11	CDD Playgrounus			\$2,100,000	ALL	Weulum-Long	levels and socio-economic groups	
		Existing playground upgrade	Location to be confirmed					Preliminary findings - Open Space and Recreation audit
12			Location to be commed	\$700,000	ALL	Short	opportunities is available for all ages, genders, ethnicities, ability	
**				<i>,</i> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	ALL	5001	levels and socio-economic groups	

Number	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related
1	Parramatta Square	Parramatta Square Public Domain	Parramatta Square	\$36,500,000	ALL	Short	Major Priority: Parramatta Square	Parramatta Squar
2	Civic Link	Civic Link Public Domain	2 blocks from Macquarie St to Phillip St	\$40,000,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Civic Link Project
3		George Street Public Domain (including paving, tree cells and multi- function poles)	George Street (btw O'Connell & Harris)	\$23,500,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramati
4		Macquarie Street Public Domain (including paving, tree cells and multi- function poles)	Pitt to Church St (non Light Rail)	\$10,000,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
5		Macquarie Street Public Domain - Council-led works on light rail affected blocks (including paving, tree cells and multi-function poles)	Harris St to Church St (Light Rail)	\$18,000,000	ALL	Short	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramati
6		Church Street Public Domain (including paving, tree cells and multi- function poles)	Macquarie St to Auto Alley (non Light Rail)	\$18,000,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramati
7		Church Street Public Domain - Council-led works on light rail affected blocks (including paving, tree cells and multi-function poles)	Macquarie St to North Parramatta (Light Rail)	\$28,000,000	ALL	Short	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
8	world class city standard	Phillip Street Public Domain (including paving, tree cells and multi- function poles)	Phillip Street (btw Marsden and Charles Square)	\$14,000,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
9	(including paving, tree cells and multi-function poles)	Smith and Station Streets Public Domain (including paving, tree cells and multi-function poles)	Macquarie St to Hassall St	\$14,000,000	ALL	Long	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
10		Wentworth/Valentine Streets Public Domain (including paving, tree cells and multi-function poles)	Wentworth and Valentine Streets	\$10,500,000	ALL	Long	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
11		Hassall Street and Station Street (including paving, tree cells and multi- function poles)	Hassall St from Harris to Station St Station St from Hassall to Parkes St	\$11,000,000	ALL	Long	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
12		O'Connell Street - in conjunction with Western Sydney Stadium redevelopment (including paving, tree cells and multi-function poles)	Ross St to Macquarie St	\$16,000,000	PART	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
13		Charles Street (including paving, tree cells and multi-function poles)	Macquarie St to Phillip St	\$6,500,000	ALL	Medium	A5.10 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Design Parramatt
14		Freemason Arms and Justice Lanes (including paving, tree cells and multi- function poles)	Freemason Arms and Justice Lanes	\$4,000,000	ALL	Medium	A5.10 - Activating lanes, retail precincts and riverbank	Lanes Strategy
15	Develop and Improve CBD Laneway	Erby Place & Lane 13 (including paving, tree cells and multi-function poles)	Erby Place & Lane 13	\$4,000,000	ALL	Medium	A5.10 - Activating lanes, retail precincts and riverbank	Lanes Strategy
16	Network to bring (including paving, tree cells and multi-function poles)	Wentworth Car Park Lanes (including paving, tree cells and multi- function poles)	Lanes around and through Wentworth Car Park	\$5,000,000	ALL	Long	A5.10 - Activating lanes, retail precincts and riverbank	Lanes Strategy
17		Macquarie Street Lanes (including paving, tree cells and multi-function poles)	Lanes North of Macquarie Street, between Smith and Harris Streets	\$5,000,000	ALL	Medium	A5.10 - Activating lanes, retail precincts and riverbank	Lanes Strategy
18		Integrated program of works to address localised flooding and water quality issues	Throughout CBD	\$40,000,000	ALL	Short	B1.1 - Setting policy direction to manage current and future environmental issues	Council's Floodpla
19	Major program of works responding to CBD flooding issues	z Large-scale flood mitigation program of works to address flooding from the Parramatta River	Throughout CBD	\$40,000,000	ALL	Medium	B1.1 - Setting policy direction to manage current and future environmental issues	Council's Floodpla
20		Flood Warning System - installation of Phase 1	Priority area in CBD	\$114,000	ALL	Short	B1.1 - Setting policy direction to manage current and future environmental issues	Council's Floodpla
21	Flood Warning System	Flood Warning System - 3 major expansions and upgrades as CBD grows	Throughout CBD	\$600,000	ALL	Medium-Long	B1.1 - Setting policy direction to manage current and future environmental issues	Council's Floodpla
22		CCTV and associated works	Throughout CBD	\$2,200,000	ALL	Short	D4.10 Developing and maintaining clean and attractive streets and public spaces where people feel safe	Smart Cities Prog
23	Smart Cities	Rationalisation of utilities (e.g. undergrounding power in Auto Alley and North Parramatta)	Throughout CBD	\$20,000,000	ALL	Short-Medium	B1.1 - Setting policy direction to manage current and future environmental issues	Smart Cities Prog
24		Multi-function poles for non-major streets	Throughout CBD	\$2,500,000	ALL	Short	A5.14 - Maintaining the spaces and public domain to the standard that supports the economic growth of the City	Smart Cities Prog

Related Project, Policy or Strategy
amatta Square Urban Design Guidelines

Design Parramatta Public Domain Framework

Council's Floodplain Risk Management Activities

Smart Cities Program

Smart Cities Program

Smart Cities Program

lumber	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related Project, Policy or Strategy
1	Civic Centre at 5 Parramatta Square	Civic Centre including facilities new CBD library, community meeting space, and experience centre	5 Parramatta Square	\$67,000,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study and 5 Parrama Square project
2	New Childcare Centres	2 new CBD childcare centres	Locations to be confirmed	\$4,000,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
3	Flexible Community Spaces	Community rooms of various sizes, including kitchen facilities & accessibility requirements	Throughout CBD	\$3,000,000	ALL	Medium-Long	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
4	New Community Facilities in Harris	Centre - daycare, OOSH, playgroup, meeting space, activity rooms, and	Jubilee Lane	\$3,000,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
5	Park	District level community facility for Harris Park, including childcare centre	Marion Street	\$1,500,000	ALL	Short-Medium	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
6		New multi-purpose community centre sufficient in design and capacity to accommodate multiple user groups	North Parramatta	\$450,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
7	CBD Contribution to New Community Facilities in North Parramatta	Incubator for multiple Community Organisations in the gifted "Admin Building"	North Parramatta	\$75,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
8		Community facility in the gifted "Hall" - fit-out	North Parramatta	\$124,500	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Community Facilities Needs Study
9		Infrastructure for food provision to disadvantaged members of the community	Prince Alfred Square	\$500,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Homelessness Policy
10	Homelessness Support Projects	Upgrade amenities for homeless (laundry, showers, etc)	Within existing non-profit facilities in CBD	\$100,000	ALL	Short	D3.1 - Providing high quality services and engaging with the local community	Homelessness Policy

Number	Type of works	Project Description	Location	Cost Estimate	Council Funding	Timeline	Corporate Plan Link	Related Project, Policy or Strategy
1	Anchor Arts Production and	Expanded Parramatta Artists Studio: Cross-arts professional production facility with art studios, workshop facilities, wet/dry areas, collaboration between old and new technologies	Civic Link	\$5,000,000	ALL	Medium	E2.2 - Grow Creative Industries and provide opportunities for creative practice and cultural production.	Cultural Discussion Paper and current project "Plannin Parramatta's Cultural Infrastructure"
2	Presentation Facilities	Art Exhibition and Gallery Space: exhibition space (including some double- height) and back of house facilities.	Civic Link	\$7,600,000	PART	Medium	E2.2 - Grow Creative Industries and provide opportunities for creative practice and cultural production.	Cultural Discussion Paper and current project "Plannin Parramatta's Cultural Infrastructure"
3	Anchor Performing Arts Facility	Modernised and expanded performing arts anchor facility, including presentation and production incorporating the new Riverside Theatres	River foreshore	\$77,000,000	PART	Long	E4.1 - Undertaking strategic planning for Riverside Theatres as a regional centre for performing arts in Western Sydney and as the key anchor arts and cultural venue in Parramatta CBD	State Infrastructure Strategy (Chapter 9) Create in NSW Parramatta Strategic Framework Council's Cultural Planning Program
4		Aboriginal Cultural Infrastructure	River Foreshore or North Parramatta	\$2,000,000	ALL	Medium	E1.6 - Interpreting the stories of Parramatta.	Cultural Discussion Paper and current project "Plannir Parramatta's Cultural Infrastructure"
5	Aboriginal Cultural Projects	Aboriginal Cityscape Cultural Walk: An interpretative walk of Parramatta including new public artwork, sites of importance and digital tour to showcase local Darug peoples' sites of significance, history and contemporary connection to Parramatta	Throughout CBD	\$1,000,000	ALL	Short	E1.6 - Interpreting the stories of Parramatta.	Cultural Discussion Paper and current project "Plannir Parramatta's Cultural Infrastructure"
6	Creative Industries Cluster	Creative Industries Incubator: incorporates media/digital center, cultural organisations such as Western Sydney Centre for Writing and CuriousWorks, hot desk office space and production offices for film and screen.	Civic Link	\$4,600,000	ALL	Medium	E2.2 - Grow Creative Industries and provide opportunities for creative practice and cultural production.	Cultural Discussion Paper and current project "Plannir Parramatta's Cultural Infrastructure"
7	Cultural Collections Storage	Publicly accessible storage for cultural collections and archaeological items	Location to be confirmed	\$1,000,000	ALL	Short	E1.6 - Interpreting the stories of Parramatta.	Cultural Discussion Paper and current project "Planniı Parramatta's Cultural Infrastructure"





Parramatta CBD Infrastructure Funding **Peer Review** City of Parramatta

28 February 2017 Revision: 3.6 Reference: 253738

Bringing ideas to life



Document control record

Document prepared by:

Aurecon Australasia Pty Ltd

ABN 54 005 139 873 Level 5, 116 Military Road Neutral Bay NSW 2089 PO Box 538 Neutral Bay NSW 2089 Australia

- T +61 2 9465 5599
- **F** +61 2 9465 5598
- E sydney@aurecongroup.com
- W aurecongroup.com

A person using Aurecon documents or data accepts the risk of:

- a) Using the documents or data in electronic form without requesting and checking them for accuracy against the original hard copy version.
- b) Using the documents or data for any purpose not agreed to in writing by Aurecon.

Docι	Document control aurecon								
Repo	rt title	Peer Review							
Docu	ment ID		Project nur	nber	253738				
File p	bath	P:\EAD\Projects\xxx_City of Pa Review v3.6.docx	P:\EAD\Projects\xxx_City of Parramatta\253738_Infrastructure Funding\Peer Review\Peer Review v3.6.docx						
Clien	t	City of Parramatta							
Clien	t contact	Roy Laria	Client reference			Mike Luger			
Rev	Date	Revision details/status	Author	Reviewer	Verifier (if required)	Approver			
0	21 November 2016	Preliminary Draft	BL	ML					
1	05 December 2016	Second Draft	BL	ML		ML			
2 19 December 2016		Draft for Council	BL/RW	ML		ML			
3	28 February 2017	Final draft for Council with incorporated comments BL / RW DA SR				ML			
Current revision		3.6							

Approval			
Author signature	billit. To	Approver signature	J.
Name	Bill Lee	Name	Mike Luger
Title	Land Economist	Title	Project Director

Executive Summary

Parramatta central business district (CBD) is forecasted to grow significantly over the coming decades as it transitions into a world class city. This will mean new residential and commercial developments as well as additional community infrastructure will be required to provide a good environment for people to work, play and live. Community infrastructure is typically funded by local council rates revenue, development contributions, and other funds from State Government Agencies. However, the City of Parramatta Council's (Council) potential income towards infrastructure from these sources is estimated to be between \$449 - \$605 million, which will be insufficient to meet the additional community infrastructure cost of approximately \$1 billion.

As a result, Council has invested considerable time and resources in planning for the future and exploring ways they can reduce the community infrastructure funding gap. This included developing several frameworks and papers on changing existing planning controls for the CBD, and undertaking a study (by GLN Planning) to explore funding options. The GLN Planning study suggested implementation of a planning uplift value share (PUVS) mechanism. The PUVS would apply to residential apartment developments in the CBD in excess of the existing planning controls under a two-phase approach, which essentially involves a reallocation of benefits from the developer to the community in the form of community infrastructure. Our review of these documents indicate that the policy steps were carefully considered, well researched and consistent in approach, and that a PUVS mechanism (implemented via voluntary planning agreements) is an important funding source for additional community infrastructure that will support the overall CBD strategy.

Following the amalgamation of local government areas in May 2016, the NSW State Government appointed an Administrator of the new City of Parramatta Council, who has called for an independent review of the proposed PUVS mechanism for fairness, equity and transparency.

This document provides the requested independent review of the work undertaken to date and in particular the recommendations based on the funding study by GLN Planning. Our review found two issues with some of the model runs, in the form of arithmetic errors and understatement of total development costs, which had a minor impact on project viability outcomes. Other than these minor issues, our opinion is that the study was comprehensive, well researched, thoughtfully modelled, and provided some words of caution and accomplished its primary objective of comparing the likely revenue generation potential of different value sharing mechanisms based on a development pro forma model of hypothetical projects. Aurecon also gleaned some insight into the residential property market by undertaking one-on-one interviews with developers and real estate agents.

Our findings indicate there is continued local market optimism, however, there is a need for caution due to macroeconomic considerations centred mainly on the stage in the business cycle that a PUVS mechanism would be implemented. This is mainly for two reasons: (i) in a strong upswing in the business cycle there tends to be more resilience of the business sector and individuals to absorb government imposts; and (ii) the real estate and property market is inherently risky lending itself to a



level of unpredictability where revenues are concerned. Accordingly, our recommendations for implementing the PUVS mechanism are as follows:

- Implement the PUVS mechanism as promptly as possible to provide Council with an additional source of funding for community infrastructure during this up-market cycle and the development community with cost predictability as the market moves toward less certain times.
- Provide the developer community with cost predictability through smoothing the implementation of PUVS over five years,
 - Set the Phase 1 (Incentive) contribution to a maximum of \$150¹ per square metre for new residential developments that seek to develop beyond the existing planning controls up to the incentive controls.
 - Set the Phase 2 (Opportunity) contribution to a maximum of \$375 per square metre for new residential developments that seek to develop beyond the incentive controls up to the opportunity site controls, applicable to certain areas in Parramatta CBD.

This will allow developers to internalise this contribution into their pro forma calculations, and the impact will be on the amount they are willing to pay for land going forward. For future major projects, removing the uncertainty of the amount of PUVS contribution and the time required for negotiations will allow Council to continue to communicate its "open for business" attitude essential for the continued rapid transition into a world class city.

- Re-evaluate the PUVS process after five years of implementation to make sure the contributions reflect market conditions, and are on track to assist in meeting funding requirements for community infrastructure.
- Build flexibility into the PUVS mechanism that provides Council with the option to either suspend or reduce the dollar per square metre contributions for a finite one to two year period should the selected residential market index decline in five of six successive quarters. In the event of a severe real estate recession (see Section 2.3 for more detail), this provides Council with an efficient tool to temporarily lower development cost and therefore reduce the mechanism's impact on residential development and construction industries operating in Parramatta CBD. In effect, this operates as a concession to developers to account for the volatility inherent in the property market.
- Apply the PUVS mechanism on residential development in excess of existing planning controls only, as per Council's intent of creating a commercial core through improving the appeal of commercial developments.

At the time of writing, there was limited published information regarding the Special Infrastructure Contributions (SIC) levy proposed by the NSW State Government to assist in the funding of regional projects including construction of Parramatta Light Rail. However, the State Government has indicated publically that the levy may require a contribution of \$200 per square metre for areas that benefit from the light rail project (which includes the Parramatta CBD). In the event that the SIC levy materialises, it is expected that this would be *in addition* to the PUVS mechanism. This means that if a developer wants to develop beyond existing planning controls and in an area that benefits from the light rail project, then they will need to make essentially *two* contributions, one to Council under the PUVS mechanism (for community infrastructure), and the second to State Government through the SIC levy (for regional infrastructure). It is important for both transparency and statutory reasons that any contributions under the PUVS mechanism and the SIC levy remain completely separate.

On assessing the impacts the PUVS mechanism will have on the residential market, this review understands that real estate development is a cyclical business. During strong market conditions,

^{1,} Contributions will be indexed against a residential market index sourced from the Australian Bureau of Statistics, rather than a Consumer Price Index, as detailed in this review



where higher demands (in excess of supply) causes prices to increase, developers have a greater interest for developing beyond existing planning controls as it provides them with greater return. Therefore a PUVS mechanism, in most instances, won't significantly impact excess developer profit margins if set above base planning controls. In this regard, the PUVS mechanism is an effective alternative funding strategy for Council to source funds towards community infrastructure in a relatively fair and transparent manner. However, the converse is true under weak market conditions, and is likely to generate little to no funding. Because real estate market fluctuations make the income flow from PUVS funding unpredictable, our recommendation is that Council views this source as an important supplemental rather than primary source of funding for the construction of community infrastructure.

Contents

Parı	ramat	ta CBD Infrastructure Funding	1
City	of Pa	Irramatta	1
1	Intro	duction	1
2	Revie	ew of documents	2
	2.1	Review of Policy and Planning Documents	2
	2.2	Review of Core Documents	6
	2.3	Residential market index	12
	2.4	Review of Developer and Community Responses	13
3	Conc	lusions	15

Figures

Figure 1 Two minor issues with the developer pro forma model developed by GLN Planning (May			
2016). Example shown here is based on Scenario D.	8		
Figure 2 Residential property price index for attached dwellings: eight capital cities – Sydney	12		

Tables

Table 1 Infrastructure funding potential under a Phase 1 and Phase 2 value sharing mechanism 1	10
--	----

1 Introduction

This document provides an independent review of a series of frameworks and papers that acknowledge Parramatta CBD will undergo significant growth over the next 20 years, bringing about new developments, and the need for additional community infrastructure. The City of Parramatta Council's (Council) potential income towards infrastructure from these sources is estimated to be between \$449 - \$605 million, which will be insufficient to meet the additional community infrastructure cost of approximately \$1 billion.

While most of the documents provide policy intentions and background for the reviewer, the 'Infrastructure Funding Model Study and Appendices' by GLN Planning and the recommendations from the Parramatta CBD Infrastructure Funding Review Committee warranted focused attention because the planning uplift value share (PUVS) mechanism, when implemented, will have financial consequences on:

- Future residential developments within Parramatta CBD
- Council's ability to fund community infrastructure and amenities important for the creation of a world class city

The findings from this review will provide Council with a way forward on implementing a PUVS mechanism that is fair, equitable and transparent.

2 Review of documents

As part of this review, Council provided a series of documents, and for convenience, we divided these into three groups:

- Policy and planning documents prepared by Council and consultants, and provided primarily to communicate policy development background.
- Core documents analyse the PUVS mechanism and recommendations from Council's Parramatta CBD Infrastructure Funding Review Committee's meeting minutes.
- **Feedback** from developers and community about the proposed value sharing mechanism and strategy.

This review focuses on the core documents, which inform our final recommendations.

2.1 Review of Policy and Planning Documents

We reviewed the following documents to gain an understanding of the Council's motivations for changing the existing planning controls and amending policies to allow these to changes to materialise. The work to date by Council and its consultants were well prepared and of high professional quality, and are summarised below.

Auto Alley Precinct – Economic Advice

Prepared for Parramatta City Council by SGS Economics & Planning, October 2013

This study provides the market economics perspective for protecting the Auto Alley Precinct intended to accommodate future CBD office expansion, since residential development is now able to support higher land values and displace future office development.

"Therefore, taking a longer-term view, it is important that Council 'holds the line' on commercial development in Auto Alley to 'future proof' the CBD from future commercial floor space shortages. Taking a medium to long-term perspective development controls should allow for at least 400,000 square metres (GFA) of employment only floor space in Auto Alley (in addition to any residential development potential allowed). Any employment related floor space in the mixed use zone would be in addition to this allowance."

This study further cites that Auto Alley has the advantage of providing larger floor plate office sites and the potential to accommodate a substantial cluster of office only buildings where the critical mass provides agglomeration economies of scale and enhances market appeal.



Prepared for Parramatta City Council by SGS Planning and Economics, August 2014

This economic analysis is a companion document to the '*Planning Framework Study*' by Architectus (September, 2014), and it examines three interrelated questions:

- How can Parramatta compete with other centres to attract employment?
- How can Parramatta protect its capacity to accommodate employment and housing in the future?
- What are the dynamic market implications for Parramatta's planning framework?

Parramatta, in order to become a world class city must compete for office development with newly emerging suburban centres. The study indicates:

"In the short term, Parramatta is unlikely to attract many large commercial office buildings (e.g. greater than 30,000 square metres) due to the limited scale of growth in the office market, and need for high proportions of pre-commitment before office developments get off the ground, and competition from other centres noted above. This is not to say additional large office developments will not locate in Parramatta, but that the process of attracting such development needs to be viewed as a long-term undertaking.

Parramatta's planning framework has been built on a clear and consistent strategy to maintain a commercial core (employment only) as a focus for future employment growth. In the face of pressure for residential development it is important that Council takes a clear position to either 'hold the line' on this policy, or, to change policy and accept a mix of uses.

If the development potential in the Parramatta CBD is increased as a result of changes to planning controls (rezoning and/or increases in permissible densities or heights) Council would be justified in seeking to sharing part of the value uplift created by these changes, and to use the proceeds of this value sharing for broader public benefit (for example: upgrades to the public domain, public transport improvements, affordable housing, open space provision, public art, and so on)."

This study suggests that for Council to preserve a commercial core within the CBD, the use of a PUVS mechanism to fund public improvements in support of that objective is justified on economic grounds. The selective or weighted imposition of a value sharing mechanism on residential development due to uplift in floor space ratio (FSR) would skew the development economics away from residential and improve the appeal of commercial developments. Without policy protection, the current robust residential market demand, in part powered by foreign investment, has the potential to erode the CBD's future capacity to become a true employment centre.



Prepared for Parramatta City Council by Architectus (and SGS Economics and Planning) and adopted by Council for exhibition on 8 September 2014

This study relates to the development of Parramatta CBD, in summary it:

- Reviewed the current planning framework that controls the development of Parramatta CBD
- Identified the opportunities, constraints and market conditions that are impacting on development in the Parramatta CBD
- Developed and recommended a planning framework that Council can implement to firmly establish Parramatta CBD and a real alternative to Sydney CBD as an employment and mixeduse centre.

Value sharing recommendations from this study:

- The proposed FSR controls to become the base, and additional higher FSR controls can only be achieved by sharing the value of the uplifts. That is any additional new FSR is to be purchased by landowners based on 50% of the nominated dollar value per sqm of GFA. The dollar value is to be scheduled to provide certainty and reviewed annually.
- This is to operate for residential uses only, not employment uses.
- This system will operate in addition to the existing Section 94A contributions."

On 8 September 2014, Council adopted for public exhibition the recommendations and planning controls from this study (with some amendments), they include adjustments to the city centre boundary, outlining primary built form controls, changed land use mix, introducing a sharing of value uplift resulting from higher FSR controls, maximum floor plates to insure tower slenderness, and incentives for design excellence.

The Draft Auto Alley Planning Framework

Prepared by Parramatta City Council and adopted for exhibition by Council at the meeting on 22 October 2014

This is a planning policy document designed to guide the transformation of the former Auto Alley along Church Street (South of Great Western Highway) into an important and integral part of the future CBD through rezoning and uplift in FSR. However, it explicitly states:

"Existing controls will be maintained until such time that a land owner:

- Contributes to and/or provides public infrastructure/benefits as per Draft Public Domain Plan
- Partakes in a Design Excellence Competition for their site."

This framework supports the economic advice provided by SGS Economics & Planning in October 2013, that Auto Alley will become an important part of growing Parramatta CBD, and that a PUVS mechanism is integral to Council's expansion plans.



Prepared by Parramatta City Council and adopted by Council at the meeting on 27 April 2015

The objectives of the CBD Planning Strategy are as follows:

- To set the vision for the growth of the Parramatta CBD as Australia's next great city.
- To establish principles and actions to guide a new planning framework for the Parramatta CBD.
- To provide a clear implementation plan for delivery of the new planning framework for the Parramatta CBD.

This strategy document confirms the Council's objective to share in the value uplift due to higher FSR controls. It states:

"That additional higher FSR controls than those proposed in this Strategy can only be achieved by sharing the value of the uplift. That is any additional new FSR is to be purchased by landowners based on 50% of the nominated dollar value per sqm of GFA. The dollar value is to be scheduled to provide certainty and reviewed annually. Such a system would apply for residential uses only, not employment uses. Further, the system would operate in addition to any section 94A contributions payable."

Parramatta City Council Section 94A Development Contribution Plan (Amendment No. 4)

Adopted by City of Parramatta Council on 27 April 2015 and commenced 20 May 2015

This is the legal and administrative document under Section 94A of the Environmental Planning and Assessment Act 1979 (EP&A Act) that lays out the development contributions framework for areas outside the Parramatta CBD and within the former Parramatta City Council boundaries. Note there are also other contributions plans, which apply outside the CBD to those areas amalgamated under the new City of Parramatta boundaries. The primary purposes of this plan are:

- To authorise the imposition of a condition on certain development consents and complying development certificates requiring the payment of a development contribution pursuant to Section 94A of the Act.
- To assist Council in providing the appropriate public facilities that are required to maintain and enhance amenity and service delivery within the area covered by this Plan.
- To publicly identify the purposes for which the development contributions are required.

This document provides detailed instructions and forms to establish the cost of the development, which would be used as a basis for a Section 94A contribution. For works to be performed by Council, it provides a schedule of projects and cost estimates plus a set of maps indicating where the works will be performed.



Prepared by Regional Cities Taskforce and initially adopted by Parramatta City Council on 11 July 2007; most recent Amendment (No. 4) adopted on 27 July 2015

This plan provides a description, both in text and graphic form, of the civic infrastructure needed and desired to support the growth and development of the Parramatta CBD and outlines the framework for funding contribution. The plan indicates:

"The public domain projects, special city centre projects, and Parramatta Square projects for the city centre are to be funded in part by monetary contributions levied from new development in the city centre under Section 94A of the Environmental Planning and Assessment Act 1979 (EP&A Act), and by other sources of funds sought by Council."

The public domain projects totalled \$169 million in cost, and the Parramatta Square Project amounted to an additional \$42 million. Note the CIP corresponds to an old strategic planning context for Parramatta CBD, it does not reflect current plans which envision a major step change in the scale of growth projected for in the CBD.

Summary of Review of Policy and Planning Documents

Over the past two and a half years, Council has undertaken a series of policy steps to facilitate the ability of the Parramatta CBD to evolve into a world class city. Aurecon are of the view that the above mentioned documents indicate that the policy steps were carefully considered, well researched and consistent in approach. In short, a value sharing mechanism based on planning uplift (i.e. PUVS) implemented through voluntary planning agreements is an important funding source for additional community infrastructure that will support the overall CBD strategy.

2.2 Review of Core Documents

This review focusses on two core documents that, together, provide an evaluation of the PUVS mechanism.

Infrastructure Funding Model Study and Appendices

Prepared for Parramatta City Council by GLN Planning, May 2016

The recommendations for this review are based on the analysis undertaken in this study, whose key objectives were to:

- Investigate alternative and innovative infrastructure funding mechanisms and models, including specifically schemes that share some of the uplift in value of development sites as a result of the additional FSR.
- Compare both the viability impacts on development projects and the revenue generation likelihood of alternative value sharing mechanisms.
- Recommend a fair, appropriate and workable development contributions system to apply to development in the Parramatta CBD area.

At the core of this study is the application of GLN's real estate development pro-forma model to test the impact of value sharing mechanisms on the financial viability of hypothetical developments planned for different heights and FSR. This modelling and its results are detailed in the Appendices and summarised in the main report. In our review, we uncovered two issues with some of the model runs that had a minor impact on project viability conclusions – these were found in <u>Appendix D: Value</u>

Sharing Development Feasibility Analysis and Assumptions:

- 1. In a number of the summary sheets for the different scenarios, the gross revenue per unit was shown to be less than net revenue per unit. This is shown in Figure 1. The arithmetic error appears to be related to the allocation of GST between the residential units and the commercial portion of the mixed-use project. Since most of the numbers shown in the summary sheet were not used as drivers in the model, this misstatement does not appear to have materially affected the model outcome.
- 2. In the summary sheets of scenarios B, C and D, again housed in Appendix D, the model omitted one key row in summing total development cost. It was the all-important row identifying "value sharing contribution" cost. In Scenario D, the total development cost was shown as \$105.3 million instead of the \$110.1 million if the \$4.83 million in value sharing contribution was included (refer to Figure 1 below). Fortunately, the amount of project surplus brought forward was the correct amount linked to a back sheet. However, because the total development cost was understated by \$4.83 million, the percentage of project surplus (net revenue less total development cost) was over stated by a half percent, 12.6% instead of 12.1%. The following additional impacts also arise from the above issue:
 - a. In Scenarios B and C, the value sharing costs were lower so the impacts on percentage of project surplus were also lower.
 - b. In Scenario B, the correct project surplus percentage should have been 16.2% instead of the 16.4% indicated.
 - c. In Scenario C, it should have been 15.1% instead of the 15.4% indicated. These minor errors did not appear to have a material impact on the conclusions of the analysis.

Scenario D - S94A Levy 3% PLUS V Summary Sheet SITE DETAILS Address Representative Floor Space Ratio 6.9 Land Area 3,000 Site Value 66,110 Number of Dwellings 172	alue Snaring @ 50% o	gln planning	
REVENUE	Average/ Unit (\$) Ex Retail	Development (\$) inc Retail	\$696,802 - 66,805 - 16,769
GROSS REVENUE	\$696,802	\$137,766,667	\$613,228 (not \$717,394)
GST	66,805	11,490,515	
Less Selling Costs	16,769	2,884,333	
NET REVENUE COSTS	\$717,394	\$123,391,818	
Land (including acquisition costs)	66,110	11,371,000	
Acquisition costs	4,610	792,841	
Construction	428,343	73,675,042	
Consultants	17,134	2,947,002	
Section 94A - Commercial	3,125	537,500	
Section 94A - Residential Value Sharing Contribution	9,699 28,081	1,668,251	T.
Statutory Fees & Contributions	6,953	1,195,883	Value sharing contribution not included in summation of total
On Costs	12,850	2,210,251	
Marketing	16,019	2,755,333	development costs
Cost before Interest	564,844	97,153,104	
Finance (incl Loan Est Fees)	47,202	8,118,822	
TOTAL DEVELOPMENT COSTS TOTAL PROJECT SURPLUS PROJECT IRR BEFORE INTEREST	612,046 #2 12.6%	\$105,271,926 \$13,289,893	Project surplus overstated as it does not include value sharing contribution – once included this

Figure 1 Two minor issues with the developer pro forma model developed by GLN Planning (May 2016). Example shown here is based on Scenario D.

Other than these minor issues, our opinion is that the study was comprehensive, well researched, thoughtfully modelled, provided some words of caution and accomplished its primary objective of comparing the likely revenue generation potential of different value sharing mechanisms based on a development pro forma model of hypothetical projects.

Minutes of the City Council Meeting of 27 June 2016 responding to the Parramatta CBD Infrastructure Funding Review Committee Report

The outcomes of the Parramatta CBD Infrastructure Funding Review Committee meeting held on 1 March 2016 were reported to Council on 27 June 2016. The report detailed the GLN Study, the findings of the Committee and also subsequent analysis undertaken by Council officers. The key recommendations of that report were as follows:

- That Council adopt the following recommendations of the Parramatta CBD Infrastructure Funding Review Committee meeting held on 1 March 2016, as follows:
 - That the Committee notes the legal advice received from Council's Legal Services Manager that confirms Council's ability, with the concurrence of the State Government, to amend the Parramatta LEP as proposed to introduce the value sharing scheme and that preparation of necessary clauses and maps include consultation with the NSW Department of Planning and Environment and also the NSW Parliamentary Counsel's Office.
 - That the value sharing rates be included in a Draft Development Guideline which is to be exhibited alongside the Parramatta CBD Planning Proposal and that the Guideline include a mechanism to enable a regular review of the rates from time to time.
 - Further, that the Parramatta CBD Infrastructure Funding Review Committee be dissolved given that it has completed its core function.
- That Council, with respect to infrastructure funding for the Parramatta CBD:
 - endorse a 3% Section 94A levy to apply to all applicable developments, plus 46.7% value sharing for Phase 1 uplift (being \$350/m2) and 50% value sharing for Phase 2 uplift (being \$375/m2); and
 - that the Phase 1 uplift incorporate any SIC levy as determined by the State Government, but only if a minimum 20% (or higher) value sharing rate (being \$150/m2 or higher) is maintained for local infrastructure.
- That Council notes that it requires approximately \$835 million for local infrastructure to make Parramatta Australia's next great city, and that there is a funding gap of between \$512 and \$628 million if Council relies solely on the current section 94A levy of 3% to fund this infrastructure.

This Council report builds on the study by GLN Planning and applies various value sharing strategies that only apply to the residential component of developments;

- Phase 1 ("Incentive") PUVS where development seeks to increase existing planning controls to incentive planning controls. The value share contribution is only applicable to the incremental portion of the uplift.
- Phase 2 ("Opportunity sites") PUVS builds on conditions set under Phase 1, and allows an additional 3:1 FSR in areas that meet certain criteria such as minimum land area and site width. Again, only the incremental portion of uplift is subject to the value sharing mechanism.
- Other FSR schemes include the Design Excellence bonus (additional 1.5:1 FSR), and High Performing Buildings bonus (additional 0.5:1 FSR) – both of which are unaffected by the value sharing mechanism.

These strategies were applied to the entire Parramatta CBD to estimate a low to high funding range to fund community infrastructure and amenities that were estimated at that point in time to cost a total of \$835 million (estimate based on preliminary information available at the time).

Council provided infrastructure funding estimates based on assumed development yields contained in CBD Planning Proposal (which has also been used to estimate future CBD transportation demand).



	Potential Income				
Rate	Low range (\$ million)	High range (\$ million)			
50%	483	589			
40%	387	471			
30%	290	353			
20%	193	235			
10%	97	118			
Phase 2					
50%	44	133			
	50% 40% 30% 20% 10%	Rate Low range (\$ million) 50% 483 40% 387 30% 290 20% 193 10% 97			

Table 1 Infrastructure funding potential under a Phase 1 and Phase 2 value sharing mechanism

Source: City of Parramatta Council Meeting 27 June 2016

The **low funding** estimate of each range assumed that each parcel would develop independently with no property consolidation to facilitate larger projects. The residential yield, which would be subject to the Phase 1 value sharing mechanism, reflects the site-by-site difference between the base FSR and the incentive FSR. Smaller parcels would be subject to a sliding scale FSR constraint to limit maximum yield attainable. Since on an area basis the actual development yield (across the city) almost never reaches planning capacity, an additional assumption was made that at build-out the development yield would only reach two-third of planning capacity. The resulting yield was approximately 1.29 million square metres of floor space. This amount of floor space was multiplied by the different value sharing rates to provide the low end funding estimate for each rate.

The **high funding** estimate of each range assumed that developers and/or land owners are able to consolidate property to avoid the sliding scale penalty imposed upon smaller parcels. Again, the residential yield, which would be subject to the Phase 1 value sharing mechanism, reflects the site-by-site difference between the base FSR and the incentive FSR, and the development yield at build-out is assumed to be two-thirds of planning capacity. The resulting yield was approximately 1.57 million square metres of floor space. This amount of floor space was multiplied by the different value sharing rates to provide the high end funding estimate for each rate.

From Council's perspective, the CBD has a finite amount of land and the 2016 CBD Planning Proposal defines the FSR capacity of this land. Given the high expectations of Parramatta CBD, Council expects essentially all developments will require some reasonable percentage of the permitted FSR. Should a development be delayed or lost, another will take its place at a later date perhaps in the next development cycle. In development economics terms, Council has very low discount rate (essentially zero) for the time value of money. From this perspective, applying a PUVS mechanism against a sample of achievable development capacity is a reasonable approach to estimate the amount of funding for community infrastructure.

However, when considering these potential ranges of alternative income, we advise that consideration needs to be given to the importance of the discount rate or the time value of money. The discount rate accounts for the time value of money since a dollar today is worth more than a dollar next year because it has the capacity to earn interest or an alternative form of return on investment. If Parramatta had \$100 million for community infrastructure today rather than 20 years in the future, it could achieve its objectives much sooner. For this reason, developments lost in a business cycle results in loss of revenue in present value terms.

From our reading of GLN's analysis and experience with real estate market cycles, we want to highlight that, in present value terms, a higher value sharing contribution will not necessarily lead to a proportionately higher level of funding for community infrastructure. The IFRC assumes there is a constant relationship between value sharing rate and range of revenue generated, however, this is a simplified assumption that may not hold true. Higher value sharing contributions could cause more projects to be either postponed or reduced in scale resulting in lower revenue for the Council on a present value basis.

For the same parcel of land, the same construction cost and the same selling prices, the PUVS contribution to Council is effectively viewed by the developer as a higher development cost. As this amount climbs, in order to maintain the level of profit/project rate of return required by either the developer or an external financier, several outcomes may result:

- Land value adjustments may result in delayed funding for Council. If the developer has not already acquired the land, they will reduce the land price offer. The seller, upon seeing recent transaction amounts, may not accept the lower offer. The result is the project comes to a halt at least until the landowner's expectations adjust to the new reality, which is likely to take several years. The result could be cancelation of the current project, possibly replaced by a new project some years later, resulting in delayed funding for Council and lost funding on a present value basis.
- Developers decide to build smaller to protect profit margins and result in reduced funding for Council. In designing a high-rise development, the developer will test the return at alternative heights on a pro forma basis. The test would likely include the additional cost of construction, PUVS contributions, additional sales price, interest cost should the sales be slow and other factors. A rigorous developer may test dozens of alternative heights and unit mix configurations in order to optimise return. The additional (value sharing) contributions required on additional heights may cause the developer to build a somewhat smaller building, especially if the overall market softens and presales decline. The result is a lower level of value sharing funding than estimated.
- Compromise in quality of materials used for developments may mean potential buyers look elsewhere. Another possible strategy to offset the higher cost, particularly if the developer already owns the land, is to reduce the cost of the building, such as less common area or less expensive features within the limitations established by the State Government's Apartment Design Guidelines. However, there is a likely consumer market response to a less expensive building in terms of lower selling price and/or slower sales.

In summary, the relationship between PUVS mechanism and community infrastructure funding to Council is not constant on a present value basis. Because the added cost has an impact on individual development decisions at the margin, as the rate climbs some projects may fall away, be postponed or require less FSR resulting in lower infrastructure funding generated for Council. Since as a percentage of development cost the value sharing contribution is likely in the one to five percent range according to GLN Planning, the impact is marginal. The primary influencing factor on funding flow over the next 20 or 30 years will be the number and depth of recessions in the Parramatta residential market.

This PUVS mechanism has been conceived during a period of robust residential real estate market condition with brisk sales and climbing prices. The mechanism is effective if the market stays strong. However, based upon over four decades of analysing real estate markets mostly in North America and from observing the quarterly changes in the residential market index (see Section 2.3) in the Sydney region, our view is that real estate markets are cyclical. Each up cycle is more or less offset by a down cycle. Even Silicon Valley, arguably the strongest economic region in the world over the last 40 years, has had severe down cycles for residential real estate. Because PUVS elevates the viability threshold of development projects, the impact of the down cycles will be amplified. As sales prices fall and absorption slows, projects will be halted more quickly. During the rebound phase, they will

rekindle more slowly. While value sharing contributions recommended by the Committee, when computed as a percentage of total project development cost, are not high, they will have some impact at the margin in the Parramatta CBD residential development market.

2.3 Residential market index

Council may wish to build in to the PUVS mechanism a concession that, in the event of a severe and sustained real estate recession², either: (i) suspends or (ii) reduces the PUVS contribution for a finite one to two year period to lower development cost, and therefore mitigate the recession's impact on the real estate development and construction industries in Parramatta.

To implement this approach an independent benchmark reference to identify when a sustained downturn is happening in the property market is required. We recommend using a *residential market index*, which has the following attributes:

- It is provided by a trusted source with no vested interests other than to provide objective data.
- It is a source that is easily accessible and ideally free or of nominal cost.
- It covers not only Parramatta but also the areas that compete with Parramatta for residential sales in the Sydney region.
- It provides quarterly data, so as to not overly burden Council with unnecessary detail.
- It measures not only the median price of residential sales but also median price per square metre as not to be influenced by the changing size of property being constructed. (Note, although with a sufficiently large sample, the median size of apartments is not likely to change very much.)

The Australian Bureau of Statistics compiles and publishes the Residential Property Price Index for Eight Capital Cities including Sydney covering all dwelling types including established houses and attached dwellings (flats, units and apartments and semi-detached, row and terrace houses). As the PUVS mechanism will be applicable to high density residential developments (i.e. apartments), we recommend Council use the Attached Dwellings Index (see Figure 2) to closely monitor the market and determine if there is a need to suspend or reduce the PUVS contribution. As an observation, we note the Sydney residential market has not had five quarters of decline within six successive quarters since the index was first compiled in 2003.



Figure 2 Residential property price index for attached dwellings: eight capital cities - Sydney

Source: Australian Bureau of Statistics table 6416.03 based on the period from September 2003 to September 2016 and indexed quarterly.

² For the purposes of the PUVS policy, the definition of a severe and sustained recession might be negative growth in five of six successive quarters.



2.4 Review of Developer and Community Responses

Public Forum Minutes from the Meeting of City of Parramatta Council held on 27 June 2016

We received one document that covered the development community responses in the public forum discussion on the proposed value sharing strategy. Some of the concerns raised primarily by developers and Aurecon's comments are listed below:

- The proposed infrastructure plan is ambitious.
 - Agreed, but great cities are built by leaders who have a clear vision for growth and investment in community infrastructure
- Value sharing mechanism is a form of tax that will have a negative impact on the viability of projects.
 - PUVS is not a tax because it is based upon voluntary agreements with property owners who elect to gain land value due to planning uplift.
- The strategy will cool the development climate in Parramatta as some developers perceive a value sharing mechanism will significantly reduce profit margins, which may shift interests elsewhere.
 - The impact is largely on the landowners who will still gain much value from the planning uplift. Since the impact is likely one to five percent of development cost, providing cost certainty going forward is the key to maintaining a favourable development climate.
- A standard value sharing contribution will impose an uneven and therefore unfair burden on development properties because of their variation in size, location, purchase date, etc.
 - Having individually negotiated contributions, which is the most likely alternative process, would be costly and time consuming for both sides. In addition, individual negotiations lack transparency and can create suspicions of favouritism or corruption.
- Previous time and efforts spent in negotiating Voluntary Planning Agreements (VPAs) become wasted sunk cost.
 - Existing VPAs that already receive some level of Council endorsement will not be renegotiated under a newly introduced PUVS mechanism. Furthermore, VPAs in the CBD are generally currently being negotiated in line with Council's work to date on value sharing.
- There is no direct link between cost of development and the benefits received from the contribution, made through the value sharing mechanism, to the construction of community infrastructure therefore unfair.
 - The details of the individual infrastructure and public amenity projects and their costs and benefits have yet to be presented. These matters are explored in the Discussion Paper.
- More time is required for response.
 - Given that real estate markets fluctuate and the Sydney region is currently in a "hot" market, it
 may not be in the interest of the development community to delay this policy implementation
 and their projects' required approvals.

The objections and criticisms are to be expected, and one of the key purposes of the Administrator's request for a Peer Review and compilation of a Discussion Paper was to encourage informed debate on both the community infrastructure and funding options.



Results of interviews with selected real estate agents and developers on 1 and 2 December 2016

Our interviews with three real estate agents and two developers active in the Parramatta and greater Sydney market indicate some slowing of the Western Sydney apartment market but with no expectation of serious oversupply resulting in a dramatic downturn. The reasons cited include:

- Population growth pushing out from the more developed portions of the Sydney region.
- Government policy support and planned infrastructure investment in and around Parramatta.
- Central location of Parramatta CBD
- High housing costs in the Sydney region means developers are looking elsewhere and Parramatta is seen as a good alternative
- Major development projects being constructed that will elevate the market perception of the Parramatta CBD when completed.

However, there are early signs of a slowing residential market due to some foreign governments beginning to slow capital outflow, which has been a factor in the Parramatta residential market, local banks being more restrictive on financing for apartment investment, and expectation of higher global interest rates due to USA election results.

The developers we interviewed saw the impact of PUVS as falling primarily on the property owners. A developer will run his/her project pro forma to compute how much he/she can pay for the land parcel. To the extent that value sharing adds to project development cost, the developer will simply offer less for the site. A developer will also typically purchase an option on the site and only complete the transaction once all approvals have been secured and major cost items are defined. For these reasons, a skilled developer is not likely to directly pay for the value sharing contribution but will pass that on to the landowner and seller.

The landowner stands to gain a substantial amount of land value from the planning uplift, and the proposed value sharing policy provides him/her with the voluntary option to partake in that value gain while sharing a portion of that gain with Council. The eventual home buyer, if he or she has done the homework, will realise that his/her investment in the Parramatta CBD has substantial long-term value appreciation potential because of the planned investments in transport infrastructure, community infrastructure and public amenities.

3 Conclusions

Over the past two and a half years, Council has undertaken a series of policy steps to facilitate the ability of the Parramatta CBD to evolve into a world class city. The reviewed documents indicate that the policy steps were carefully considered, well researched and consistent in approach. In short, a value sharing mechanism based on planning uplift implemented through voluntary planning agreements is an important funding source (for community infrastructure) that will support the overall CBD strategy.

Balancing the continued local market optimism with the need for caution due to macroeconomic considerations, our recommendations based upon this review are as follows:

- Implement the PUVS mechanism as promptly as possible to provide Council with an additional source of funding for community infrastructure during this up-market cycle and the development community with cost predictability as the market moves toward less certain times.
- Provide the developer community with cost predictability through smoothing the implementation of PUVS over five years,
 - Set the Phase 1 (Incentive) contribution to a maximum of \$150³ per square metre for new residential developments that seek to develop beyond the existing planning controls up to the incentive controls.
 - Set the Phase 2 (Opportunity) contribution to a maximum of \$375 per square metre for new residential developments that seek to develop beyond the incentive controls up to the opportunity site controls, applicable to certain areas in Parramatta CBD.

This will allow developers to internalise this contribution into their pro forma calculations, and the impact will be on the amount they are willing to pay for land going forward. For future major projects, removing the uncertainty of the amount of PUVS contribution and the time required for negotiations will allow Council to continue to communicate its "open for business" attitude essential for the continued rapid transition into a world class city.

- Re-evaluate the PUVS process after five years of implementation to make sure the contributions reflect market conditions, and are on track to assist in meeting funding requirements for community infrastructure.
- Build flexibility into the PUVS mechanism that provides Council with the option to either suspend or reduce the dollar per square metre contributions for a finite one to two year period should the selected residential market index decline in five of six successive quarters. In the event of a severe real estate recession (see Section 2.3 for more detail), this provides Council with an efficient tool to temporarily lower development cost and therefore reduce the mechanism's impact on residential development and construction industries operating in

³. Contributions will be indexed against a residential market index sourced from the Australian Bureau of Statistics, rather than a Consumer Price Index, as detailed in this review

Parramatta CBD. In effect, this operates as a concession to developers to account for the volatility inherent in the property market.

Apply the PUVS mechanism on residential development in excess of existing planning controls only, as per Council's intent of creating a commercial core through improving the appeal of commercial developments.

At the time of writing, there was limited published information regarding the Special Infrastructure Contributions (SIC) levy proposed by the NSW State Government to assist in the funding of transport projects including construction of Parramatta Light Rail. However, the Transport for NSW website has advises that the levy may require a contribution of \$200 per square metre for areas that benefit from the light rail project (which includes the Parramatta CBD). In the event that the SIC levy materialises, it is expected that this would be *in addition* to the PUVS mechanism. This means that if a developer wants to develop beyond existing planning controls and in an area that benefits from the light rail project, then they will need to make essentially *two* contributions, one to Council under the PUVS mechanism (for community infrastructure), and the second to State Government through the SIC levy (for regional transport infrastructure). It is important for both transparency and statutory reasons that any contributions under the PUVS mechanism and the SIC levy remain completely separate.

On assessing the impacts the PUVS mechanism will have on the residential market, this review understands that real estate development is a cyclical business. During strong market conditions, where higher demands (in excess of supply) causes prices to increase, developers' have a greater interest for developing beyond existing planning controls, as it provides them with greater return. Therefore a PUVS mechanism, in most instances, won't significantly impact developer profit margins and is an effective strategy for raising funds towards community infrastructure in a relatively fair and transparent manner. However, the converse is true under weak market conditions, and is likely to generate little to no funding. Because real estate market fluctuations makes the income flow from PUVS funding unpredictable, our recommendation is that Council views this source as an important supplemental rather than primary source of funding for the construction of community infrastructure.

aurecon

Aurecon Australasia Pty Ltd

ABN 54 005 139 873 Level 5, 116 Military Road Neutral Bay NSW 2089 PO Box 538 Neutral Bay NSW 2089 Australia

T +61 2 9465 5599

F +61 2 9465 5598

E sydney@aurecongroup.com

 \boldsymbol{W} aurecongroup.com

Aurecon offices are located in:

Angola, Australia, Botswana, China, Ghana, Hong Kong, Indonesia, Kenya, Lesotho, Macau, Mozambique, Namibia, New Zealand, Nigeria, Philippines, Qatar, Singapore, South Africa, Swaziland, Tanzania, Thailand, Uganda, United Arab Emirates, Vietnam.

planning consulting strategy

5 October 2019

Our Ref: 11122_Ltr_CIFS_LEPclause.docx

Mr Roy Laria Land Use Planning Manager City of Parramatta Council

By email: <u>rlaria@cityofparramatta.nsw.gov.au</u>

Dear Roy,

PROPOSED PARRAMATTA CBD LEP COMMUNITY INFRASTRUCTURE PROVISIONS

1 Background

I refer to Council's request for advice on the 'community infrastructure' provisions in the Parramatta CBD Planning Proposal, specifically in relation to the Gateway Determination conditions issued by the Department of Planning, Industry and Environment (DPIE).

The Gateway Determination issued for the Planning Proposal included a number of conditions, including the following:

Condition No. 1(m):

1. (m) in relation to infrastructure funding:

..

iii. amend the explanation of provisions to clarify that community infrastructure is only able to be provided on the development site; and

iv. consider a funding mechanism to support the provision of community infrastructure, such as the preparation of a new section 7.11 contributions plan or a potential increase to the levy under the current section 7.12 contributions plan.

This letter addresses condition No. 1.(m)(iii).

Please note that in relation to condition No. 1.(m)(iv), our work on the different funding mechanisms is continuing.

2 Effect of implementing condition 1.(m)(iii)

The primary effect of amending the draft LEP provisions so that community infrastructure is only able to be provided on the development site would restrict the type and scale of infrastructure that could be provided by developers on the development site.

The list of infrastructure in Council's draft *Parramatta CBD Infrastructure Needs Analysis* was reviewed to determine which items could possibly be delivered as part of a development on a development site.

A summary of on-site community infrastructure opportunities is shown in **Attachment 1**.

Attachment 1 shows that a minimum of \$150 million worth of community infrastructure could be delivered through adjusting the draft LEP's community infrastructure provisions to reflect the Gateway condition.

However, this amount could be significantly increased if Council was to redefine one of the items in the infrastructure plan.

Item 4 in the 'Providing Opportunities for Recreation and Leisure' category is for the upgrade of parks in the CBD and surrounds. The item, with a value of \$218m, is described as:

Upgrade the quality of parks within the city to cater for increased demand and bring up to a world-class city standard

Council could instead redefine the item and adjust the item description to the following (or similar):

Embellishment of open spaces and public squares provided as part of developments

This redefinition would enable developers to provide new city parks and public squares (at ground level) or accessible podium public spaces (above ground level) on development sites.

Even if 50% of the funding was divided between upgrading existing parks and providing new spaces on development sites, this action would potentially enable \$259 million of community infrastructure to be provided on development sites.

3 Alternative wording that would meet the Gateway condition

GLN Planning is aware of LEPs applying to other Sydney suburban centres that contain provisions for extra floor area linked to the provision of community infrastructure, and that have been in place for several years, and that meet the requirements of the Gateway condition.

A summary of these provisions is included in Table 1.

LEP, relevant clause and area of application	Relevant wording
Burwood LEP 2012	4.4A Exceptions to floor space ratio
Clause 4.4A	(1)-(4)
Applies to the 2 'key areas'	(5) Despite clause 4.4, the floor space ratio for a building on land in Area 1 or Area 2 may exceed the floor space ratio shown for the land on the Floor Space Ratio Map if:
comprising the core and frame of	(a) the floor space ratio for the building does not exceed:
the Burwood CBD	(i) 6.6:1—if the building is in Area 1, or
	(ii) 4.95:1—if the building is in Area 2, and
	(b) the gross floor area of the part of the building used for the purpose of residential accommodation does not exceed the following percentage of the gross floor area of the building:
	(i) 40 percent—if the building is in Area 1, or
	(ii) 70 percent—if the building is in Area 2.
	(6) Subclause (5) applies in relation to proposed development only if:
	(a) the proposed development on the land includes development resulting in community infrastructure or the use of land as community infrastructure, and
	(b) the consent authority is satisfied that the community infrastructure is appropriate for the Burwood Town Centre, taking into account the nature of the community infrastructure and its value to the community working or residing in the Burwood Town Centre.
	(7)-(8)
	(9) In this clause, community infrastructure means any of the following:
	(a) a recreation area,
	(b) a community facility,
	(c) an information and education facility.
Penrith LEP 2010	8.7 Community infrastructure on certain key sites
Clause 8.7	(1) The objectives of this clause are:
Applies to 11 'key sites' in the Penrith	(a) to allow higher density development on certain land in the City Centre where the development includes community infrastructure, and
CBD	(b) to ensure that the greater densities reflect the desired character of the localities in which they are allowed and minimise adverse impacts on those localities.
	(2) This clause applies to land identified as a key site on the Key Sites Map.
	(3) Despite clauses 4.3, 4.4 and 8.4 (5), the consent authority may consent to development on land to which this clause applies (including the erection of a new building or external alteration to an existing building) that exceeds the maximum height shown for the land on the Height of Buildings Map or the floor space ratio for the land shown on the Floor Space Ratio Map, or both, if the proposed development includes community infrastructure.
	(4)
	(5) In deciding whether to grant development consent under this clause, the consent

Table 1 Community infrastructure provisions in other LEPs

.

• •

•

.

.

.

. .

•

.

٠

• •

.

• •

• • • •

•

• •

•

.

•

• •

• •

LEP, relevant clause and area of application	Relevant wording
	authority must have regard to the following:
	(a) the objectives of this clause,
	(b) whether the development exhibits design excellence,
(c) the nature and value of the community infrastructure to the City C	
	(6) In this clause, community infrastructure means development for the purposes of recreation areas, recreation facilities (indoor), recreation facilities (outdoor), recreation facilities (major), public car parks or public roads.

DPIE through its Gateway Determination report has implicitly endorsed the Penrith approach:

A funding mechanism to provide community infrastructure funding as part of development has been implemented in other local government areas. For example, the Penrith LEP contains a provision where community infrastructure is delivered on identified key sites. However, the difference between the Parramatta model and the Penrith model is that within the Parramatta model, there is a broad collection of funds to contribute towards infrastructure projects, whereas in the Penrith model community infrastructure is provided on the development site only.

Department of Planning and Environment, Parramatta CBD Planning Proposal Gateway Determination Report, pp29-30

As described in section 2 above, there are numerous opportunities for infrastructure types listed in the CBD infrastructure plan to be provided on the site where the development is proposed.

Our preliminary assessment shows the opportunity for \$259 million worth of community infrastructure could be provided this way.

While community infrastructure provisions (clause 8.7) under the Penrith LEP applies to particular 'key sites', similar provision wording could be included in the Parramatta CBD LEP. This is because, as demonstrated in section 2, there are many opportunities for community infrastructure to be provided on numerous development sites in the LEP area.

Accordingly, in order to meet Gateway Determination condition 1(m)(iii), we recommend that Council update the relevant clauses in the draft Parramatta CBD Planning Proposal / draft LEP to generally reflect the approach used in clause 8.7 of Penrith Local Environmental Plan 2010.

Should you require clarification in relation to this quote, please do not hesitate to contact me (02) 9249 4100.

4

. • • • • • • • • • • • • • • • • • • • . • • • • • • • • • • • •

Yours faithfully

GLN PLANNING PTY LTD

GREG NEW DIRECTOR

ATTACHMENTS

Attachment 1

5

ATTACHMENT 1

Community infrastructure that potentially could be accommodated on a development site

No.	Type of works	Details	Total value of potential opportunities
Prov	iding Opportunities for Recreation	and Leisure	
5	Playspaces	District and local playspaces in and in close proximity to CBD	\$3,950,000
6	Active recreation facilities	Recreation courts, exercise equipment, youth facilities and dog parks $^{\rm b}$	\$830,000
	Sub total		\$ <i>4,780,000</i> ª
Crea	ting A Strong Economy with a Stro	ng City Centre	
26	Major program of works responding to CBD flooding issues	Integrated program of works to address localised flooding and water quality issues $^{\rm c}$	\$20,000,000
30	Smart Cities	CCTV and associated works	\$2,200,000
	Sub total		\$22,200,000
Havi	ng a Community Focus		
2	New Childcare Centres	4 new Early Learning Centres of 80 places each	\$16,000,000
4	Flexible Community Space	6,800 sqm of flexible community space	\$46,920,000
5	Youth Space	2,000sqm of youth space	\$13,800,000
6	Low-cost leasable office space	3,000sqm low-cost leasable office space (to allow non-Council community services providers and social enterprises to locate in Parramatta CBD)	\$7,035,000
7	Homelessness Support Projects	Infrastructure for food provision to disadvantaged members of the community (Prince Alfred Square)	\$500,000
8		Upgrade amenities for homeless (laundry, showers, etc) within existing non-profit facilities in CBD	\$100,000
	Sub total		\$84,355,000
Supp	oorting Arts and Culture, Celebration	ons and Destinations	
2	Anchor Arts Production and Presentation Facilities	Expanded Parramatta Artists Studio: Cross-arts professional production facility with art studios, workshop facilities, wet/dry areas, collaboration between old and new technologies	\$5,000,000
3		Art Exhibition and Gallery Space: exhibition space (including some double-height) and back of house facilities.	\$7,600,000
4	Aboriginal Cultural Projects	Aboriginal Cultural Infrastructure which	\$2,000,000

ATTACHMENT 1

Community infrastructure that potentially could be accommodated on a development site

No.	Type of works	Details	Total value of potential opportunities
		acknowledges Parramatta's role as a meeting place within the Sydney basin, as well as its history as a place of contact and conflict, contributes to future growth and positive outcomes for Aboriginal people, and helps all people experience Aboriginal culture in Parramatta	
6	Creative Industries Cluster	Creative Industries Incubator: incorporates cultural organisations such as Western Sydney Centre for Writing and CuriousWorks, and office space (for example hot desking for creative industries).	\$4,600,000
7	Screen Culture / Digital Media Centre	Incorporating a public screening presentation facility for screen culture and digital arts; digital media production facility; digital media education centre (potential tertiary institution partnership opportunity); suite of leasable spaces to house local screen/digital arts creative organisations and enterprises.	\$20,000,000
	Sub total		\$39,200,000
	Total		\$150,535,000

Source: Parramatta CBD Infrastructure Needs Analysis

Notes:

a There is the potential to provide a much greater value of on-site items in this category. See discussion below.

b Small allocation. This could be significantly increased to allow recreation facilities (e.g. rooftop outdoor courts) to be provided on development sites.

c Potential opportunity to provide basement detention tanks / on site stormwater harvesting to reduce runoff rates in trunk drainage network. Assume 50% of cost can be allocated to these development site opportunities. Requires testing with Council engineers